### UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

### FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): May 1, 2023



### **VSE CORPORATION**

(Exact name of registrant as specified in its charter) 000-03676

(Commission File Number)

54-0649263 (IRS Employer

Identification Number)

Delaware

(State or Other Jurisdiction

of Incorporation)

6348 Walker Lane 22310 Alexandria, Virginia (Address of Principal Executive Offices) (Zip Code) (703) 960-4600 (Registrant's Telephone Number, Including Area Code) Not Applicable (Former Name or Former Address, if Changed Since Last Report) Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions: Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425) Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12) Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b)) Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c)) Securities registered pursuant to Section 12(b) of the Act: Title of each class Trading Symbol(s) Name of each exchange on which registered Common Stock, par value \$.05 per share VSEC The NASDAQ Global Select Market Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§ 240.12b-2 of this chapter). Emerging growth company □ If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

#### VSE CORPORATION

#### Item 1.01 Entry into a Material Definitive Agreement.

Sale of FDS Business

On May 1, 2023, VSE Corporation (the "Company") announced a definitive agreement to sell its Federal and Defense business to Bernhard Capital Partners ("Purchaser"). Under the terms of the membership interest purchase agreement, dated as of May 1, 2023 (the "Purchase Agreement"), between the Company, one of its wholly-owned subsidiaries, and ASG Operations, LLC, an affiliate of the Purchaser, the FDS Business is anticipated to be sold for a total cash consideration of up to \$100 million (the "FDS Transaction"). This consideration assumes a \$50 million cash payment at the closing of the FDS Transaction and an earn-out payment of up to \$50 million, subject to the achievement of certain milestones. The FDS Transaction is expected to close in late 2023 or early 2024 and is subject to customary closing conditions and approvals. The Company currently intends to use the net proceeds from the FDS Transaction to reduce existing borrowings and execute strategic acquisitions.

The foregoing summary of the Purchase Agreement is subject to, and qualified in its entirety by, the text of the Purchase Agreement, a copy of which is attached hereto as Exhibit 2.1 and is incorporated herein by reference.

Desser Aerospace Acquisition

On May 4, 2023, the Company announced that it had entered into a definitive agreement to acquire Desser-Graham Partnership, L.P. ("Desser Aerospace"). Under the terms of the Agreement and Plan of Merger, dated as of May 3, 2023 (the "Merger Agreement"), by and among VSE Aviation, Inc., a wholly-owned subsidiary of the Company ("VSE Aviation"), one of VSE Aviation's wholly-owned subsidiaries ("Merger Sub"), Desser Aerospace, and Desser Holdings Partnership GP, LLC ("Representative"), Merger Sub will merge with and into Desser Aerospace resulting in Desser Aerospace being a wholly-owned subsidiary of VSE Aviation (the "Merger Transaction"). VSE Aviation will pay total cash consideration in connection with the Merger Transaction of approximately \$124 million, subject to certain customary adjustments. The Merger Transaction is expected to close within 90 days, subject to customary closing conditions and approvals, including certain regulatory approvals in the United Kingdom.

Concurrent with the closing of the Merger Transaction, the Company anticipates, in a separate transaction, the sale of Desser Aerospace's propriety solutions businesses to Loar Group Inc. ("Loar") for cash consideration of approximately \$30 million (the "Loar Sale").

After the execution and closing of the Loar Sale and the concurrent closing of the Merger Transaction, the Company's net total cash consideration exchanged will be approximately \$94 million.

The foregoing description of the Merger Agreement is subject to, and qualified in its entirety by, the text of the Merger Agreement, a copy of which is filed as Exhibit 2.2 hereto and is incorporated herein by reference.

#### Item 9.01 Financial Statements and Exhibits

(d) Exhibits

Exhibit	Description
2.1*†	Membership Interest Purchase Agreement, dated as of May 1, 2023, by and among VSE Corporation, a wholly-owned subsidiary of VSE Corporation, and ASG Operations, LLC.
2.2*†	Agreement and Plan of Merger by and among VSE Aviation, Inc., Diamond Merger Sub 1, LLC, Desser-Graham Partnership, L.P. and Desser Holdings Partnership GP, LLC, as the Representative, dated as of May 3, 2023.
104	Cover Page Interactive Data File

- \* Certain schedules and exhibits have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Company will furnish supplementally a copy of any omitted schedules or exhibits to the Securities and Exchange Commission upon request.
- † Certain information has been omitted pursuant to Item 601(b)(2) of Regulation S-K. The Company will furnish supplementally a copy of any omitted information to the Securities and Exchange Commission upon request.

### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

VSE CORPORATION

(Registrant)

Date: May 5, 2023 By: /s/ Farinaz S. Tehrani

Farinaz S. Tehrani

Chief Legal Officer and Corporate Secretary

CERTAIN INFORMATION CONTAINED IN THIS MEMBERSHIP INTEREST PURCHASE AGREEMENT AND PLAN OF MERGER HAS, PURSUANT TO ITEM 601(B)(2) OF REGULATION S-K, BEEN OMITTED BY MEANS OF REDACTING A PORTION OF THE TEXT AND REPLACING IT WITH [\*\*\*] BECAUSE SUCH INFORMATION IS BOTH NOT MATERIAL AND IS THE TYPE OF INFORMATION THAT THE REGISTRANT TREATS AS PRIVATE OR CONFIDENTIAL.

STRICTLY CONFIDENTIAL

### MEMBERSHIP INTEREST PURCHASE AGREEMENT

BY AND AMONG
VSE CORPORATION,
HURRICANE FDS COMPANY, LLC

**AND** 

### ASG OPERATIONS, LLC

Dated as of May 1, 2023

# TABLE OF CONTENTS

# Page

Article I DEFINITIONS	1
Section 1.1 Definitions	1
Section 1.2 Other Defined Terms	15
Article II PURCHASE AND SALE; CLOSING	20
Section 2.1 Purchase and Sale	20
Section 2.2 Purchase Price	20
Section 2.3 Closing Date	20
Section 2.4 Pre-Closing Reorganization Transactions	21
Section 2.5 Payment; Closing Deliveries	21
Section 2.6 Adjustments to Base Purchase Price	23
Section 2.7 Purchase Price Allocation	25
Section 2.8 Income Tax Treatment	26
Section 2.9 Earn-Out	26
Section 2.10 Withholding	27
Article III REPRESENTATIONS AND WARRANTIES OF SELLER	27
Section 3.1 Organization, Standing and Power	27
Section 3.2 Transferred Entities	27
Section 3.3 Authority; Execution and Delivery; Enforceability	28
Section 3.4 No Conflicts; Consents; Governmental Authorization	28
Section 3.5 Financial Statements	29
Section 3.6 Absence of Undisclosed Liabilities	30
Section 3.7 Absence of Changes or Events	30
Section 3.8 Intellectual Property	30
Section 3.9 Information Technology; Data Protection	32
Section 3.10 Real Property	34
Section 3.11 Contracts	35
Section 3.12 Compliance with Applicable Laws; Permits	38
Section 3.13 Government Contracts	39
Section 3.14 Environmental Matters	42
Section 3.15 Proceedings	43
Section 3.16 Taxes	43
Section 3.17 Labor Relations; Employees and Employee Benefit Plans	45
Section 3.18 Intercompany Agreements	49
Section 3.19 Brokers	50
Section 3.20 Key Customers and Suppliers	50
Section 3.21 Bank Accounts; Powers of Attorney	50
Section 3.22 Insurance	51
Section 3.23 Title; Sufficiency of Assets	
Section 3.24 Security Clearances	51
Article IV REPRESENTATIONS AND WARRANTIES OF PURCHASER	51
Section 4.1 Organization, Standing and Power	52
Section 4.1 Organization, Standing and Power Section 4.2 Authority; Execution and Delivery; Enforceability	52
	52
Section 4.3 No Conflicts; Consents; Governmental Authorizations	52

### TABLE OF CONTENTS

# (continued)

# Page

Section 4.4 Financial Ability to Perform	53
Section 4.5 Proceedings	53
Section 4.6 Brokers	53
Section 4.7 Investigation	53
Section 4.8 Securities Act	54
Section 4.9 Solvency	54
Section 4.10 Acknowledgement of Pre-Closing Services	54
Section 4.11 Foreign Ownership, Control or Influence	54
Section 4.12 Acknowledgement of No Other Representations or Warranties	55
Article V COVENANTS	55
Section 5.1 Approvals; Efforts	55
Section 5.2 Covenants Relating to Conduct of Business	57
Section 5.3 Confidentiality	60
Section 5.4 Access to Information	61
Section 5.5 Publicity	62
Section 5.6 Employee Matters	62
Section 5.7 [Intentionally Omitted]	65
Section 5.8 Names Following Closing	65
Section 5.9 Insurance	66
Section 5.10 Post-Closing Support	66
Section 5.11 Export Control Authorizations	67
Section 5.12 Payments from Third Parties	67
Section 5.13 Novation of Government Contracts; Subcontracting Agreements; Purchaser Approvals	67
Section 5.14 Shared Contracts	68
Section 5.15 Title Commitments; Surveys	70
Section 5.16 R&W Insurance Policy	70
Section 5.17 Termination of Intercompany Balances and Intercompany Agreements	71
Section 5.18 Directors' and Officers' Indemnification	71
Section 5.19 Misallocated Assets	72
Section 5.20 Pre-Closing Reorganization Transaction	73
Section 5.21 Transition Services	73
Section 5.22 Financing Obligations	73
Section 5.23 Financing Cooperation	73
Section 5.24 Non-Competition; Non-Solicitation; Non-Disparagement	74
Section 5.25 Specified Matters	76 -
Article VI CERTAIN TAX MATTERS	76
Section 6.1 Certain Tax Returns	76
Section 6.2 Certain Income Tax Treatment and Apportionment of Taxes	76
Section 6.3 Cooperation and Exchange of Information	77

### TABLE OF CONTENTS

# (continued)

# Page

Section 6.4 Transfer Taxes	78
Section 6.5 Tax Sharing Agreements	78
Section 6.6 Certain Tax Proceedings	78
Section 6.7 Purchaser Tax Acts	78
Section 6.8 Tax Refunds	79
Section 6.9 Seller Consolidated Taxes; Seller Consolidated Tax Returns	79
Section 6.10 Pre-Closing Tax Indemnity	79
Article VII CONDITIONS PRECEDENT	80
Section 7.1 Conditions to Each Party's Obligations to Close	80
Section 7.2 Conditions to Obligations of Purchaser to Close	80
Section 7.3 Conditions to Obligations of Seller to Close	81
Section 7.4 Frustration of Closing Conditions	81
Article VIII TERMINATION; EFFECT OF TERMINATION	82
Section 8.1 Termination	82
Section 8.2 Effect of Termination	83
Section 8.3 Notice of Termination	83
Article IX INDEMNIFICATION	83
Section 9.1 Survival of Representations, Warranties, Covenants and Agreements.	83
Section 9.2 Exclusive Remedy	83
Section 9.3 Indemnification by Seller	84
Section 9.4 Indemnification by Purchaser	84
Section 9.5 Indemnification Procedures.	84
Section 9.6 Limitation of Liability	85
Section 9.7 R&W Insurance Policy	87
Section 9.8 Escrow Release	87
Article X GENERAL PROVISIONS	87
Section 10.1 Entire Agreement	87
Section 10.2 Assignment	87
Section 10.3 Amendments and Waivers	88
Section 10.4 No Third-Party Beneficiaries	88
Section 10.5 Notices	88
Section 10.6 Specific Performance	89
Section 10.7 Governing Law and Jurisdiction	89
Section 10.8 Waiver of Jury Trial	90
Section 10.9 Severability	90
Section 10.10 Counterparts	90
Section 10.11 Expenses	91
Section 10.12 Interpretation; Absence of Presumption	91
Section 10.13 Waiver of Conflicts Regarding Representation: Nonassertion of Attorney-Client Privilege	92

TABLE OF CONTENTS	
-------------------	--

# (continued)

Page

Section 10.14	Disclosure Schedules	93
Section 10.15	Non-Recourse	93

iv

### (continued)

**Page** 

### EXHIBITS1

Exhibit A	Form of Escrow A	Agreement
-----------	------------------	-----------

Form of Subcontract Agreement Pending Novation Allocation Schedule Exhibit B

Exhibit C

Exhibit D Sample Working Capital Statement Transaction Accounting Principles
Pre-Closing Reorganization Documents Exhibit E Exhibit F Exhibit G Assignment of Membership Interests

Exhibit H

Company Certificate
Specified Reports and MRO Site Matters Exhibit I

DCMA Audit Exhibit J

Exhibit K Purchaser Knowledge Exhibit L Entry into Contracts

### **SCHEDULES**

Schedule 1.1 Pre-Closing Reorganization Transaction Steps Schedule

Schedule 2 Earn-Out Mechanics

Schedules and exhibits have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Company will furnish supplementally a copy of any omitted schedules or exhibits to the Securities and Exchange Commission upon request.

### MEMBERSHIP INTEREST PURCHASE AGREEMENT

This Membership Interest Purchase Agreement, dated as of May 1, 2023 (this "<u>Agreement</u>"), is by and among VSE Corporation, a Delaware corporation ("<u>Seller</u>"), Hurricane FDS Company, LLC, a Delaware limited liability company (the "<u>Company</u>"), and ASG Operations, LLC, a Delaware limited liability company ("<u>Purchaser</u>").

- WHEREAS, Seller and certain Subsidiaries of Seller are, directly or indirectly, engaged in, among other things, the Business;
- WHEREAS, Seller owns all of the issued and outstanding membership interests of the Company (the "Company Equity Interests");
- **WHEREAS**, as promptly as practicable after the date hereof, Seller shall take the actions set forth on the Pre-Closing Reorganization Transaction Steps Schedule to transfer and assign the Business to the Company (the transactions set forth on the Pre-Closing Reorganization Transaction Steps Schedule, the "**Pre-Closing Reorganization Transactions**"), in each case, upon the terms and conditions set forth herein and in the Pre-Closing Reorganization Transaction Steps Schedule;
- WHEREAS, following the consummation of the Pre-Closing Reorganization Transactions, the Company shall collectively own and operate the Business;
- **WHEREAS**, simultaneously with the execution of this Agreement, each of the Key Employees is entering into an employment agreement or offer letter (collectively, the "**Employment Agreements**") with the Company; and
- **WHEREAS**, following the consummation of the Pre-Closing Reorganization Transactions, Seller desires to sell, assign, transfer and convey to Purchaser, and Purchaser desires to purchase and acquire from Seller, the Company Equity Interests, in each case, upon the terms and conditions contained in this Agreement (the "**Transaction**").
- **NOW, THEREFORE**, in consideration of the representations, warranties, covenants and agreements contained in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, on the terms and subject to the conditions of this Agreement, the parties hereto hereby agree as follows:

### ARTICLE I DEFINITIONS

Section 1.1 <u>Definitions</u>. As used herein, the following terms have the meanings set forth below:

"Affiliate" means, with respect to any Person, any other Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by Contract or otherwise. For purposes of this Agreement, Seller and its Affiliates

shall be deemed not to be Affiliates of Purchaser or, from and after the Closing, of the Business or the Transferred Entities.

"Affiliated Group" means, an affiliated group as defined in Section 1504 of the Code (or any analogous combined, consolidated, unitary or similar group under state, local or non-U.S. Law).

"Approvals" means any approvals, authorizations, notices and consents by or to a third-party person (including Governmental Entities).

"Benefit Plan" means each compensation or benefits plan, program or arrangement (including those subject to ERISA, employment agreements, cash, or equity-based compensation, stock ownership, stock purchase, option, stock appreciation right, performance unit, restricted stock unit, restricted share, phantom stock, bonus or incentive arrangements, deferred compensation, thrift, savings, stock bonus, tax gross-up, change of control, retention, severance arrangements, separation, termination, cafeteria, paid time off, perquisite, fringe benefit, vacation policies, disability, death benefit, hospitalization, medical or other health and welfare plans or other employee benefit plan, program, policy, arrangement or understanding, whether oral or written, formal or informal, funded or unfunded), in each case (a) sponsored, maintained, or contributed to by Seller or any of its ERISA Affiliates for the benefit of any Business Employee, or (b) any such plan, program or arrangement sponsored, maintained, or contributed to, or required to be sponsored, maintained, administered or contributed to by a Transferred Entity or with respect to which a Transferred Entity has any liability; provided that in no event will a "Benefit Plan" include any plan, program or arrangement sponsored or maintained by a Governmental Entity.

"Business" means collectively, the following businesses and their related assets, Intellectual Property and past performance used, or held for use, primarily or exclusively in the operation or conduct of such businesses as conducted in the twelve months prior to the date hereof and as proposed to be conducted immediately following the Closing by the Seller's Federal and Defense Services business segment, including its Contracts with the state- and local-level Governmental Entities set forth on Schedule 1.1(a),: (a) supply chain management solutions to the U.S. government, federal agencies and international defense contractors; (b) data management solutions to the U.S. government, federal agencies and international defense contractors; (c) maintenance, repair and overhaul services to the U.S. government, federal agencies and international defense contractors; (e) learning and training support services and similar services to the U.S. government, federal agencies and international defense contractors; (e) learning and training support services to the U.S. government, federal agencies and international defense contractors; and (f) energy and management consulting, as currently conducted by Energetics. For the avoidance of doubt, the Business shall expressly exclude the business and operations of VSE Aviation, Inc. and/or Wheeler Fleet Solutions, Co and their respective Subsidiaries.

"Business Day" means any day, other than a Saturday, Sunday or day on which commercial banks are required or authorized to be closed in New York, New York or Baton Rouge, Louisiana.

"Business Employee" means each employee of Seller or its Affiliates whose services are Relating to the Business.

"Business Intellectual Property" means all Intellectual Property owned or purported to be owned (in whole or in part), licensed, used or held for use by the Transferred Entities.

"Business Material Adverse Effect" means any event, change, effect, circumstance, state of facts, occurrence or development (each an "Effect") that, individually in the aggregate with all other Effects, (x) has a material adverse effect on the business, financial condition or results of operations of the Business, assets or Liabilities of the Business, taken as a whole or (y) would prevent or materially impair or delay Seller's ability to consummate the Transaction as contemplated by this Agreement and the other Transaction Documents; provided, that no such Effect resulting or arising from or in connection with any of the following matters shall be deemed, either alone or in combination, to constitute or contribute to a Business Material Adverse Effect: (a) the general conditions in the industries in which the Business operates, including competition in any of the geographic areas in which the Business operates; (b) any national, global or regional, political, economic, business, monetary, financial, banking, inflationary, securities, or capital or credit market conditions or trends (including any changes in interest rates, the price of commodities or raw materials, or the decline in price of any security or any market index), including with respect to government spending, budgets and related matters; (c) geopolitical conditions, trade wars, tariffs or sanctions, any act of civil unrest, war, sabotage or terrorism (including by cyberattack or otherwise), including an outbreak or escalation of hostilities involving the United States or any other country or the declaration by the United States or any other country or jurisdiction of a national emergency or war; (d) any natural or manmade disasters or weather developments, including earthquakes, hurricanes, tsunamis, typhoons, lightning, hail storms, blizzards, tornadoes, droughts, floods, cyclones, arctic frosts, mudslides and wildfires, acts of God, or any virus, pandemic (including COVID-19), epidemic or disease or similar force majeure events, including any material worsening of such conditions; (e) the failure of the financial or operating performance of Seller or the Business to meet internal, Purchaser or analyst projections, forecasts, milestones, estimates, guidance or budgets or financial or operating predictions of revenue, earnings, cash flow or cash position for any period (provided that the underlying facts or basis for any failure to meet such projections, forecasts, milestones, estimates, guidance, budgets or predictions which are not otherwise excluded may be taken into account in determining whether there has been or would reasonably be expected to be a Business Material Adverse Effect); (f) the execution, announcement, pendency, performance or consummation of this Agreement, the Transaction, or the other transactions contemplated hereby or thereby, or the identity of Purchaser; (g) changes in any Law (including any governmental or quasigovernmental action) or changes in GAAP or (h) [\*\*\*]; provided, further, that any adverse events, changes or effects resulting from the matters described in the foregoing clauses (a), (b), (c) or (g) may be taken into account in determining whether there has been a Business Material Adverse Effect to the extent, and only to the extent that they have a materially disproportionate effect on the Business in the aggregate relative to similarly situated businesses in the industries in which the Business operates (in which case only such incremental materially disproportionate impact may be taken into account in determining whether there has been a Business Material Adverse Effect).

"Business IT Assets" means the computers, servers, workstations, routers, hubs, switches, data communications lines and other information technology equipment, networks, or systems owned or controlled by the Transferred Entities.

"Business Technology" means any Business IT Assets, websites or other online properties or services, and any software or hardware products or services made available, provided, sold, licensed, or leased to any customer by or on behalf of the Seller or any

Transferred Entities, including any microchips, firmware, on-premise software, mobile applications or browser extensions.

- "CARES Act" means the Coronavirus Aid, Relief, and Economic Security Act, H.R. 748, 116th Cong., 2d Sess. (signed into law on March 27, 2020) and any similar or successor federal, state, local and foreign Law, including any applicable guidance (including IRS Notice 2020-65, and IRS Notice 2021-11) issued thereunder or relating thereto.
- "Cash Amounts" means, of any Person and as of any time, all cash and cash equivalents, bank and other depositary accounts and safe deposit boxes, deposits in transit, uncashed and uncleared checks received or deposited for the account of any of the Transferred Entities, demand accounts, certificates of deposit, time deposits, negotiable instruments, marketable securities and securities and brokerage accounts, net of outstanding checks, wire transfers, or other pending withdrawals, in each case of such Person as of such time, such amounts calculated in a manner consistent with the Transaction Accounting Principles.
- "Closing Cash Amounts" means an amount equal to the sum of the Cash Amounts of the Business as of 12:01 a.m. (Eastern Daylight Time) on the Closing Date.
- "Closing Indebtedness" means an amount equal to the sum of the Indebtedness of the Business as of 12:01 a.m. (Eastern Daylight Time) on the Closing Date.
- "Closing Purchase Price" means (a) the Base Purchase Price, plus (b) the Estimated Closing Cash Amounts, minus (c) the Estimated Unpaid Company Transaction Expenses, minus (c) the Estimated Closing Indebtedness, plus (d) the Estimated Working Capital Amount (which, for the avoidance of doubt, may be a negative number).
- "Closing Unpaid Company Transaction Expenses" means an amount equal to the sum of the Unpaid Company Transaction Expenses of the Business as of 12:01 a.m. (Eastern Daylight Time) on the Closing Date.
  - "Closing Working Capital" means the Working Capital as of 12:01 a.m. (Eastern Daylight Time) on the Closing Date.
  - "Code" means the U.S. Internal Revenue Code of 1986, as amended.
- "Collective Bargaining Agreement" means each collective bargaining, works council or other Contract, agreement, or arrangement with a union, works council, labor organization, or other representative of employees.
- "Contract" means any written or oral contract, agreement, note, bond, guarantee, mortgage, deed of trust, lease, sublease, license, sublicense, instrument, purchase order, commitment, loan or credit agreement, indenture or agreement, or other legally binding right, commitment, obligation, arrangement or understanding, other than a Permit or Export Control Authorization.
- "COTS Software" means generally available, commercial off-the-shelf Software licensed from a third party on standard commercial terms for which license fees are less than \$25,000 per year and that is used solely in connection with the internal operations of the Business.

- "Covered Loss" means any and all losses, Liabilities, claims, fines, deficiencies, damages, payments (including those arising out of any settlement or Judgment relating to any Proceeding), Taxes, penalties and reasonable attorneys' and accountants' fees and disbursements.
- "COVID-19" means SARS-CoV-2 or COVID-19, and any evolutions or mutations thereof or related epidemics, pandemics or disease outbreaks.
- "Current Government Contract" means any Government Contract the period of performance of which has not yet expired or been terminated, or which remains open to government audit as of the Closing Date.
- "<u>Data Protection Law</u>" means all applicable Laws relating to the protection or processing of Personal Data, data privacy, cybersecurity or the privacy of electronic communications in any relevant jurisdiction, including (a) the California Consumer Privacy Act, as amended by the California Privacy Rights Act, (b) the General Data Protection Regulation ((EU) 2016/679), and any national implementing Laws relating thereto and, (c) in respect of the United Kingdom, the Data Protection Act 2018.
- "<u>Data Protection Obligations</u>" means (i) all applicable Data Protection Laws; (ii) commitments made under any privacy policy or notice published or made available by Seller or any Transferred Entity; (iii) all public commitments or promises made by the Seller or any Transferred Entity relating to the Processing or security of Protected Information; and (iv) obligations related to Processing of Protected Information pursuant to Contracts to which the Seller or a Transferred Entity is a party.
  - "Deductible Escrow Amount" means \$243,750.
  - "DCSA" means the Defense Counterintelligence and Security Agency within the U.S. Department of Defense.
  - "Energetics" means Energetics Inc., a Maryland corporation.
- "Environmental Laws" means, collectively, any and all Laws and Judgments relating to pollution or the protection or regulation of human health or safety or the environment, natural resources or the use, generation, transportation, treatment, disposal, recycling, labeling, processing, production, manufacture, remediation, exposure to, storage or handling of Hazardous Materials.
- "Environmental Permits" means Permits issued pursuant to Environmental Laws required for the operation of the Business as presently conducted.
  - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "ERISA Affiliate" means, with respect to any entity, trade or business, any other entity, trade or business that is, or was at the relevant time, a member of a group described in Section 414(b), (c), (m) or (o) of the Code or Section 4001(b)(1) of ERISA that includes or included the first entity, trade or business, or that is, or was at the relevant time, a member of the same "controlled group" as the first entity, trade or business pursuant to Section 4001(a)(14) of ERISA.

- "Escrow Agent" means JPMorgan Chase Bank, N.A.
- "Escrow Release Date" means twelve (12) months following the Closing Date.
- "Estimated Working Capital Amount" means the amount calculated in accordance with Schedule 3 to this Agreement.
- "Export Control Authorization" means any and all Approvals, including Approvals under the International Traffic in Arms Regulations (ITAR) (22 C.F.R. Parts 120-130) or the Export Administration Regulations (EAR) (15 C.F.R. Parts 730-774), required for the lawful conduct of the Business following the Closing Date in substantially the same manner as conducted as of the date of this Agreement pursuant to the Export Control Laws as administered by the relevant Governmental Entities, including the Department of Commerce, Bureau of Industry and Security; the United States Department of State, Directorate of Defense Trade Controls; and the Department of the Treasury, Office of Foreign Assets Control; and any other export controls and economic sanctions laws or regulations in any other foreign (non-U.S.) jurisdiction. This shall also mean any import permits under 27 C.F.R. Part 447, as administered by the Department of Justice, Bureau of Alcohol, Tobacco, Firearms and Explosives.
- "Export Control Laws" means (a) the Arms Export Control Act (22 U.S.C. § 2778 et seq.), as amended, and the ITAR; (b) the Export Administration Act (50 U.S.C. App. §§ 2401 et seq.), as amended and continued in force by presidential order; (c) the Export Control Reform Act of 2018 (Pub. L. 115-232), and the EAR; (d) any U.S. sanctions programs, regulations and executive orders, including those promulgated under the Countering America's Adversaries Through Sanctions Act (22 U.S.C. § 9525), the International Emergency Economic Powers Act (50 U.S.C. §§ 1701-1706), the National Emergencies Act (50 U.S.C. §§ 1601-1651), the Trading with the Enemy Act (50 U.S.C. App. §§ 5, 16); (e) European Union controls on exports of dual-use items and technology implemented pursuant to Council Regulation (EC) No 428/2009; (f) additional international sanctions programs administered by the Department of the Treasury, Office of Foreign Assets Control, or Department of State; (g) any other regulations or Laws promulgated under the aforementioned acts or issued under the aforementioned regimes; and (h) any applicable non-U.S. export control or sanctions Laws.
- "Fraud" means actual and intentional common law fraud under Delaware law, which, for the avoidance of doubt, does not include constructive or equitable fraud, with respect to the making of the representations and warranties set forth in <a href="Article III">Article III</a> or <a href="Article IV">Article IV</a>, or any Transaction Document.

"GAAP" means United States generally accepted accounting principles, as consistently applied by Seller.

[\*\*\*]

"Government Contract" means any prime contract, subcontract, grant, subaward, purchase order, task order, delivery order, basic ordering agreement, pricing agreement, teaming agreement, letter contract or other similar written arrangement of any kind between Seller or a Transferred Entity, on the one hand, and (a) any Governmental Entity or (b) any higher-tier contractor of a Governmental Entity in its capacity as a higher-tier contractor; provided, that a task, purchase, work or delivery order under a Government Contract shall not constitute a

separate Government Contract, for purposes of this definition, but shall be part of the Government Contract to which it relates.

- "Government Official" means any person qualifying as a public official or public employee under any legal requirement of any relevant jurisdiction, and also includes:
  - (a) a person who holds a legislative, administrative or judicial position in or with a Governmental Entity;
- (b) a person holding an official position, such as an employee, officer or director, in or with any Governmental Entity or state-owned or controlled commercial or other enterprise;
- (c) an individual "acting in an official capacity," such as pursuant to a delegation of authority, from a Governmental Entity to carry out official responsibilities;
- (d) an official of a public international organization such as the United Nations, the World Bank, the International Monetary Fund or a regional development bank;
  - (e) an official of a political party or a candidate for political office;
- (f) an immediate family member, such as a Purchaser, spouse, sibling or child of a person referred to in any of clauses "(a)" through "(e)" above; and
  - (g) an agent or intermediary of any person referred to in any of clauses "(a)" through "(f)" above.
- "Governmental Entity" means any national, state, local, supranational or foreign government or any court of competent jurisdiction, administrative agency or commission or other national, state, local, supranational or foreign governmental authority or instrumentality.
- "<u>Hazardous Material</u>" means any substance, pollutant, contaminant, material or waste that is listed, defined, regulated by or classified in any applicable Environmental Law, and including asbestos, asbestos-containing materials, polychlorinated biphenyls, petroleum or petroleum products, radioactive materials, and radon gas.
- "Indebtedness" means, of any Person and as of any time, the aggregate amount of the following obligations of such Person as of such time, without duplication, in each case such amount calculated in a manner consistent with the Transaction Accounting Principles: (a) the outstanding principal amount of any indebtedness for borrowed money (other than trade payables arising in the ordinary course of business), including all accrued but unpaid interest thereon; (b) the outstanding principal amount of and accrued interest under all other obligations evidenced by bonds, debentures, notes or similar instruments of indebtedness, including all accrued but unpaid interest thereon; (c) lease obligations to the extent any such lease is classified as a finance lease in the Business Financial Statements or should be recorded as a finance lease in accordance with GAAP; (d) any accrued but unpaid income Taxes of the Transferred Entities attributable to the Pre-Closing Tax Period; (e) all obligations of the Company arising under letters of credit, bank guarantees or similar facilities, in each case solely to the extent drawn; (f) all outstanding obligations arising from installment purchases of property or representing the deferred purchase price of securities, assets, property or services for which and of the Transferred Entities is liable

(in each case, other than current trade payables incurred in the ordinary course of business), whether contingent or otherwise, and amounts under any earn-out or similar performance payment or any seller notes or post-closing true-up obligations; (g) all unpaid severance payable by the Transferred Entities arising with respect to terminations of employment which notice to employees occurred prior to the Closing, as well as the employer portion of any payroll, social security, unemployment or similar Tax imposed on such amounts (including, for the avoidance of doubt, any payroll or similar Taxes deferred pursuant to the CARES Act); (h) all obligations of the Transferred Entities owed to any Person under any Transferred Entity Benefit Plan that is a retirement or deferred compensation plan or arrangement, (i) all accounts payable or accrued liabilities in connection with any discontinued projects; (j) forward loss reserves on Contracts as determined in accordance with GAAP; (k) any premium, break fee, penalty or other cost payable in connection with terminating, unwinding, or prepaying any obligations described in preceding clauses (a) – (j); and (l) all obligations of the type referred to in the preceding clauses (a) – (k) that are secured by a Lien on the assets of the Transferred Entities or for which any Transferred Entity is obligated pursuant to a guaranty; provided, that in no event shall Indebtedness include (i) any Liabilities to be repaid or extinguished pursuant to this Agreement in connection with the Closing, (ii) any Liabilities included in the calculation of Working Capital, and (iii) any leases in respect of leased real property.

"Intellectual Property" or "Intellectual Property Rights" means any and all common law or statutory, proprietary or intellectual property rights existing anywhere in the world, whether registered or unregistered, including the following: (a) patents, patent applications (including provisionals, continuations, continuations-in-part, and divisionals), statutory invention registrations, industrial and utility models, registered designs; extensions, reissues, re-examinations or renewals of any of the foregoing; and similar or equivalent rights in inventions and designs, and all rights therein provided by international treaties and conventions ("Patents"); (b) trademarks, service marks, trade dress, trade names, logos and other designations of origin; registrations and applications to register the foregoing, and any renewals thereof; and all goodwill associated therewith ("Trademarks"); (c) domain names, uniform resource locators, Internet Protocol addresses, social media handles and other names, identifiers and locators associated with Internet addresses, sites and services and registrations and applications for registration for any of the foregoing ("Internet Properties"); (d) copyrights (whether in published or unpublished works); original works of authorship fixed in any tangible medium of expression; data, databases, data collections, mask works, and website content; registrations of manufacturing processes; compilations, collective works and derivative works of any of the foregoing; registrations and applications for registration for any of the foregoing and any renewals or extensions thereof ("Copyrights"); (e) trade secrets, industrial secrets, know-how, and other confidential and proprietary information, ideas, discoveries, improvements, inventions and invention disclosures (whether or not patentable), including customer, business or technical information and reports, processes, formulae, methods, techniques, schematics and technology ("Trade Secrets"); (f) Software; and (g) claims or causes of action related to, and rights to recover damages for, any past, present and future infringement, misappropriation, dilution, or other violation of any of the foregoing.

"Intercompany Agreements" means all Contracts in effect as of the date of this Agreement that are between or among Seller and/or its Subsidiaries and/or Affiliates (other than the Transferred Entities) on the one hand, and any of the Transferred Entities, on the other hand, other than the organizational documents of the Transferred Entities.

- "Judgment" means any judgment, injunction, order, ruling, writ, injunction, stipulation or decree issued or promulgated by any Governmental Entity or alternative dispute resolution body.
  - "Key Employee" means the individuals set forth on Schedule 4.
- "Knowledge" means the actual knowledge after reasonable due inquiry, with respect to Seller, of any Person listed in Section 1.1(c) of the Seller Disclosure Schedules, and the actual knowledge after reasonable due inquiry, with respect to Purchaser, of any Person listed in Exhibit K.
- "Law" means any national, state, local, supranational or foreign law, statute, law, common law ruling, code, order, ordinance, rule, regulation, treaty (including any Tax treaty) or Judgment or other binding directive, in each case, issued, promulgated or enforced promulgated by a Governmental Entity.
- "<u>Liabilities</u>" means all debts, liabilities, guarantees, assurances, commitments and obligations, whether fixed, contingent or absolute, asserted or unasserted, matured or unmatured, liquidated or unliquidated, accrued or not accrued, known or unknown, due or to become due, whenever or however arising (including whether arising out of any Contract or tort based on negligence or strict liability).
- "Lien" means any mortgage, lien, license, deed of trust, claim, mortgage, pledge, security interest, charge, easement, license, sublicense, hypothecation, declaration, covenant, encroachment, title defect, right of first refusal or first offer or similar encumbrance of any kind.
- "Losses" means actual losses, damages, Liabilities, awards, claims, deficiencies, fines, costs, fees, expenses, payments (including those arising out of any settlement or Judgment relating to any Proceeding), Taxes, including reasonable fees and disbursements of attorneys, auditors, consultants, experts and other agents associated therewith.
- "NISPOM" means the National Industrial Security Program Operating Manual, codified in 32 CFR Part 117, as required by Executive Order 12829 and issued under the authority of the Department of Defense Instruction 5220.22, National Industrial Security Program ("NISP"), or subsequent Executive Orders, directives or regulations for the protection of classified information released or disclosed to industry in connection with classified contracts under the NISP.
- "Open Source Software" means any Software that is distributed or made available under any "open source" or "free software" terms, including (a) any license approved by the Open Source Initiative (www.opensource.org), or (b) any license that meets the Open Source Definition as defined by the Open Source Initiative or the Free Software Definition as defined by the GNU Project (www.gnu.org) (any license referred to in (a) and/or (b), an "Open Source Software License").
- "PATRIOT Act" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act (USA Patriot Act) of 2001 (Title III of Pub. L. 107-56 (signed into law October 26, 2001)).

- "Permits" means permits, approvals, authorizations, consents, licenses, identification numbers, exemption, exceptions or certificates issued by any Governmental Entity.
- "Permitted Liens" means the following Liens: (a) Liens for Taxes that are not yet due or payable or that are being contested by appropriate Proceedings, that may thereafter be paid without penalty or for which an adequate reserve has been established and reflected in the Business Financial Statements; (b) statutory Liens of landlords that are so called rights of distraint, Liens of carriers, warehousemen, mechanics, materialmen, workmen, repairmen and other Liens imposed by Law in the ordinary course of business for amounts which are not due and payable or which are being contested in good faith by appropriate proceedings; (c) with respect to real property, (i) easements, declarations, covenants, rights-of-way, restrictions and other charges, instruments or encumbrances affecting title to real estate or other documents or writings recorded in the public records, and (ii) Liens not created by Seller or any of its Subsidiaries that affect the underlying fee interest of any leased real property, including master leases or ground leases; provided, that with respect to this clause (c), any such item does not materially interfere with the ordinary conduct of the Business and is not material to such real property; (d) Liens which will be released at or prior to the Closing; and (e) non-exclusive licenses granted with respect to Intellectual Property in the ordinary course of business.
- "Person" means any individual, firm, corporation, partnership, limited liability company, trust, joint venture, Governmental Entity or other entity.
- "Personal Data" means any information about an identifiable natural person that alone or in combination with other information identifies, or could be used to identify, a natural person, and includes information that constitutes "personal data," "personally identifiable information," "individually identifiable health information," "protected health information" or "personal information" (or any analogous term) under any applicable Data Protection Law.
- "Pre-Closing Reorganization Closing Condition Steps" means the following steps set forth on the Pre-Closing Reorganization Transaction Steps Schedule: 1, 2, 3, 4, 5, 6, 8, 9(a), 10, 13, 14, 15, 16, 17 and 18.
  - "Pre-Closing Reorganization Transaction Steps Schedule" means the Transaction Steps Schedule set forth in Schedule 1.1.
- "Pre-Closing Reorganization Transactions Transfer Taxes" means any and all Transfer Taxes arising from or attributable to the Pre-Closing Reorganization Transactions.
- "Pre-Closing Tax Period" means any taxable period ending on or before the Closing Date and the portion of any Straddle Period ending on the Closing Date.
- "Proceeding" means any judicial, administrative or arbitral action, charge, suit, countersuit, litigation, investigation, audit, arbitration, mediation or proceeding by or before any Governmental Entity or arbitral or mediation body.
- "Process" (and inflections thereof, when used in connection with Protected Information) means any operation or set of operations performed on Protected Information, whether or not by automatic means, such as access, receipt, collection, monitoring, maintenance, creation, recording, organization, structuring, storage, adaptation or alteration, retrieval, consultation, use.

processing, analysis, transfer, transmission, disclosure, dissemination or otherwise making available, alignment or combination, blocking, erasure, destruction, or security.

"Protected Information" means any information (i) that is Personal Data; (ii) that Seller or any Transferred Entity received from or on behalf of customers of Seller or any Transferred Entity on a confidential basis; or (iii) that is subject to a confidentiality obligation, that is proprietary information of Seller or any Transferred Entity, or in which Seller or a Transferred Entity has Intellectual Property Rights.

"Purchaser Fundamental Representations" means the representations and warranties set forth in Section 4.2.

- "Regulatory Approvals" means all Approvals from Governmental Entities that are required under applicable Law to permit the consummation of the Transaction and the other transactions contemplated by this Agreement.
- "Relating to the Business" or "Related to the Business" used, or held for use, primarily or exclusively in the operation or conduct of the Business as conducted in the twelve months prior to the date hereof and as proposed to be conducted immediately following the Closing.
- "Release" means any presence, emission, spill, seepage, leak, escape, leaching, discharge, injection, pumping, pouring, emptying, dumping, disposal, migration, or release of Hazardous Materials from any source into or upon the indoor or outdoor environment.
- "Representatives" of a Person means any officer, director or employee of such Person or any investment banker, attorney, accountant or other advisor, agent or representative of such Person.
- "Restricted Party" means any Person that appears on, or is owned 50% or more or controlled by a Person that appears on, restricted party lists maintained by any U.S., EU or other Governmental Entity (including the U.S. Department of the Treasury's Specially Designated Nationals List, the Foreign Sanctions Evaders List, the Sectoral Sanctions Identifications List, the Non-SDN Menu-Based Sanctions List and the U.S. Department of Commerce's Entity List, Unverified List, or Denied Persons List) (collectively, "Restricted Party").
- "Retained Businesses" means the businesses and operations of Seller and its Subsidiaries other than the Business, and which are not primarily or exclusively used in the operation or conduct of the Business.

"Sample Working Capital Statement" means the calculation statement set forth on Exhibit D.

"Section 721" means Section 721 of the Defense Production Act of 1950, as amended, including amendments made by the Foreign Investment and National Security Act of 2007 (codified at 50 U.S.C. § 4565), and the regulations promulgated by the U.S. Committee on Foreign Investment (CFIUS) thereunder, codified at 31 C.F.R. Part 800, et seq., or any successor statute and/or regulations thereto.

"Securities Act" means the U.S. Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

- "Seller Benefit Plan" means each Benefit Plan that is not a Transferred Entity Benefit Plan.
- "Seller Consolidated Taxes" means any U.S. federal, state or local or non-U.S. income Taxes for any period that are imposed on an Affiliated Group that includes a Transferred Entity, on the one hand, and Seller or any Affiliate of Seller, on the other hand, for such period.
- "Seller Consolidated Tax Return" means any Tax Return for any period of an Affiliated Group that includes a Transferred Entity, on the one hand, and Seller or any Affiliate of Seller, on the other hand, for such period.
- "Seller Disclosure Schedules" means those certain Seller Disclosure Schedules, dated as of the date of this Agreement, provided by Seller to Purchaser.
- "Seller Fundamental Representations" means the representations and warranties set forth in Section 3.1, Section 3.2, Section 3.3, Section 3.4(a), Section 3.18, Section 3.19, and, for purposes of Section 9.2 only, Section 3.16.
- "Software" means any and all computer software and programs in any form or format (including object code, source code, firmware and embedded versions thereof), however fixed, and all related documentation.
  - "Specified Matters" has the meaning set forth in Section 2 of Exhibit I.
  - "Specified Reports" means the reports listed on Section 1 of Exhibit I.
  - "Straddle Tax Period" means any taxable period beginning on or before and ending after the Closing Date.
- "Subsidiary" means, with respect to any Person, any corporation, partnership, limited liability company or other entity whether incorporated or unincorporated, of which (a) such first Person directly or indirectly owns or controls at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions or (b) such first Person is a general partner or managing member. For purposes of this agreement, from and after the Closing, no Transferred Entity shall be deemed to be a Subsidiary of Seller or an Affiliate of Seller.
  - "Target Working Capital" means \$16,750,823.
- "Tax" means (i) any federal, state, provincial, local or foreign tax imposed by a Taxing Authority, including any net income, gross receipts, sales, use, value-added, goods and services, profits, license, withholding, payroll, employment, unemployment, social security, employer health, excise, premium, registration, real property, personal property, escheat, unclaimed property, capital gains, transfer, stamp, environmental, alternative or add-on minimum, occupation, and franchise tax, windfall profits, customs, and any other duty, assessment or governmental charge; and (ii) any and all interest, penalties and additions imposed with respect to amounts described in (i) whether disputed or not.
  - "Tax Proceeding" means any audit, examination, contest, litigation or other Proceeding with or against any Taxing Authority.

- "<u>Tax Return</u>" means any return, declaration, report, claim for refund or information return or statement required to be filed with any Taxing Authority relating to Taxes, including any amendment thereof or attachment thereto.
  - "Taxing Authority" means any Governmental Entity responsible for the administration or the imposition of any Tax.
  - "Transaction Accounting Principles" means the accounting principles set forth on Exhibit E.
- "Transaction Documents" means this Agreement, the Escrow Agreement, the Subcontract Agreement Pending Novation, the Pre-Closing Reorganization Documents, and any other agreement or certificate to be delivered by a party hereto pursuant to the terms of this Agreement.
- "Transferred Entity Equity Interests" means the Company Equity Interests, the Energetics Equity Interests and the VS2 Equity Interests.
- "Transfer Taxes" means any federal, state, local, foreign and other sales, use, transfer (including real property transfer), registration, documentary, stamp, duty, value added, goods and services, recording, conveyance or similar Taxes and related fees and costs imposed on or payable in connection with the transactions contemplated by this Agreement or the recording of any sale, transfer, conveyance, lease, sublease or assignment of property (or any interest therein) effected pursuant to or contemplated by this Agreement.
- "Transferred Entity" means each of the Company, Energetics and VS2. For the avoidance of doubt, Seller's contribution of the VS2 Equity Interests only represents 51% of the issued and outstanding membership interests of VS2.
- "Transferred Entity Benefit Plan" means each Benefit Plan that is sponsored, maintained or contributed to solely by one or more Transferred Entities, including the Benefit Plans the sponsorship of which are transferred to a Transferred Entity in the Pre-Closing Reorganization Transactions.
- "Unpaid Company Transaction Expenses" means, without duplication, the sum of (i) the aggregate unpaid fees and expenses of any lawyers, accountants, investment bankers, financial advisors, brokers, consultants, agents, accountants, data room administrators and other third parties, to the extent such fees are incurred or payable by any Transferred Entity and remain unpaid and an obligation of a Transferred Entity as of the Closing, relating to the negotiation, documentation and consummation of the transactions contemplated by this Agreement and/or the Transaction Documents, but excluding any such fees and expenses expressly specified to be paid by Purchaser pursuant to Section 10.11, (ii) any transaction, retention, change-of-control or similar payments and/or benefits payable by the Transferred Entities pursuant to arrangements entered into prior to Closing (including the employer portion of any payroll, social security, unemployment or similar Tax imposed on such amounts and including, for the avoidance of doubt, any payroll or similar Taxes deferred pursuant to the CARES Act), (iii) any amounts payable to Seller or its Affiliates by the Transferred Entities, including any Liabilities, if any, in connection with the termination of the Intercompany Agreements contemplated by his Agreement, (iv) fifty percent (50%) of the costs of the R&W Insurance Policy in accordance with Section 10.11, (v) Seller's share of any Transfer Taxes in accordance with Section 6.4 and

(vi) all other unpaid Liabilities, costs, fees and expenses relating to or arising from the Pre-Closing Reorganization Transactions; provided, however, that, for the avoidance of any doubt, any amounts included in Closing Date Net Working Capital or Closing Date Indebtedness shall be excluded from Unpaid Company Transaction Expenses.

"VS2" means VS2 LLC, a Delaware limited liability company.

"WARN" means the Worker Adjustment Retraining and Notification Act, 29 U.S.C. Section 2101 et seq., as amended or state or local law equivalent.

"Working Capital" means, as of any time, the net working capital of the Business calculated by subtracting (a) the sum of current liabilities for the Business, from (b) the sum of current assets for the Business, in each case calculated in accordance with the Transaction Accounting Principles; provided, that in no event shall Working Capital include (i) any amount included within the definition of Cash Amounts or Indebtedness, (ii) any Liabilities repaid or extinguished pursuant to this Agreement in connection with the Closing, or (iii) any amounts with respect to income Tax assets or income Tax liabilities. A sample calculation of the Working Capital for illustrative purposes is set forth in Exhibit D.

Section 1.2 Other Defined Terms. In addition, the following terms shall have the meanings ascribed to them in the corresponding Section of this Agreement:

### Term Section

[\*\*\*] Section 2.9 Affiliate Section 1.1 Affiliated Group Section 1.1 Aggregate Purchase Price Section 2.2 Agreement Preamble Allocation Section 2.7(a) Allocation Schedule Section 2.7(a) Approval Section 1.1 Assigned Contracts Section 5.13(a) Award Period Schedule 2(c) Balance Sheet Date Section 3.5 Base Purchase Price Section 2.2 Benefit Plan Section 1.1 Business Section 1.1 Business Balance Sheet Section 3.5 Business Day Section 1.1 Business Employee Section 1.1 Business Financial Statements Section 3.5 Business Insurance Policies Section 3.22 Business Intellectual Property Section 1.1 Business IT Assets Section 1.1 Business Material Adverse Effect Section 1.1 Business Permits Section 3.12(b) Business Technology Section 1.1 CARES Act Section 1.1 Cash Amounts Section 1.1

Closing Section

Closing Cash Amounts Section 1.1

Closing Date Section 2.3

Closing Fringe Reserve Section 5.6(1) Closing Indebtedness Section 1.1

Closing Purchase Price Section 1.1

Closing Statement Section 2.6(a)

Closing Unpaid Company Transaction Expenses Section 1.1

Closing Working Capital Section 1.1

Code Section 1.1

Collective Bargaining Agreement Section 1.1

Commitment Section 5.15(a)

Company Preamble

Company Business Intellectual Property Section 3.8(a)(i)

Company Equity Interests Recitals

Company Inbound IPR Agreements Section 3.11(a)(v)

Company Outbound IPR Agreements Section 3.11(a)(v)

Company Software Section 3.8(a)(i)

Competing Business Section 5.24(a)

Confidential Business Information Section 5.3(b)

Confidentiality Agreement Section 5.3(a)

Continuation Period Section 5.6(b)

Continuing Employee Section 5.6(b)

Contract Section 1.1

Copyrights Section 1.1

COTS Software Section 1.1

Covered Employee Section 5.24(b) Covered Loss Section 1.1

COVID-19 Section 1.1

Current Government Contract Section 1.1

Current Representation Section 10.13(a)

D&O Indemnifiable Claim Section 5.18(a)

D&O Indemnified Party Section 5.18(a)

Data Protection Law Section 1.1

Data Protection Obligations Section 1.1

DCSA Section 1.1

Debt Financing Section 5.23(a)

Deductible Escrow Amount Section 1.1

Deferred Compensation Plan Section 5.6(k)

Designated Person Section 10.13(a)

DFARS Section 3.13(b)

Dispute Notice Section 2.6(c)

Dispute Resolution Period Section 2.6(c)

Disqualified Individual Section 3.17(h)

Earn-Out Section 2.2

[\*\*\*] Section 2.9 Effect Section 1.1

Employment Agreements Recitals

Energetics Section 1.1

Energetics Equity Interests Section 3.2

Environmental Laws Section 1.1 Environmental Permits Section 1.1 Equity Commitment Letter Section 4.4(a) Equity Financing Section 4.4(a) Equity Investors Section 4.4(a)

ERISA Section 1.1

ERISA Affiliate Section 1.1

Escrow Agent Section 1.1

Escrow Agreement Section 2.5(b)(ii) Escrow Release Date Section 1.1

Estimated Closing Cash Amounts Section 2.6(a) Estimated Closing Indebtedness Section 2.6(a)

Estimated Closing Unpaid Company Transaction Expenses Section 2.6(a)

Estimated Closing Working Capital Amount Section 2.6(a)

Estimated Working Capital Amount Section 1.1, Schedule 3(1) Estimated Working Capital Excess Schedule 3(1)(a)

Estimated Working Capital Underage Schedule 3(1)(b)

Export Control Authorization Section 1.1

Export Control Laws Section 1.1

Facility Clearance Section 5.1(a)

FAR Section 3.13(b)

Final Purchase Price Section 2.6(e) Final Transition Time Section 5.13(b)

Final Working Capital Amount Schedule 2(2) Final Working Capital Excess Schedule 3(2)(a)

Final Working Capital Underage Schedule 3(2)(b)

FLSA Section 3.17(i)

Fraud Section 1.1

GAAP Section 1.1 [\*\*\*] Section 1.1

Government Contract Section 1.1
Government Official Section 1.1

Governmental 8

Hazardous Material Section 1.1

Inactive Employees Section 5.6(b)

Indebtedness Section 1.1

Indemnified Party Section 9.5(a) Indemnifying Party Section 9.5(a)

Independent Accounting Firm Section 2.6(c)

Institutions Section 3.8(a)(i)

Intellectual Property Section 1.1

Intercompany Agreements Section 1.1

Internet Properties Section 1.1

Judgment Section 1.1

Key Customers Section 3.20(a)
Key Employee Section 1.1
Key Suppliers Section 3.20(b)

Knowledge Section 1.1

Law Section 1.1

Liabilities Section 1.1

Lien Section 1.1

Losses Section 1.1

Material Contracts Section 3.11(a)

Material Government Contract Section 3.13(a)

New VEBA Section 5.6(1)

NISP Section 1.1

NISPOM Section 1.1

[\*\*\*] Schedule 2(d)
OCI Section 3.13(g)
Open Source Software Section 1.1
Open Source Software License Section 1.1

Outside Date Section 8.1(d)

Patents Section 1.1

PATRIOT Act Section 1.1

Permits Section 1.1 Permitted Liens Section 1.1

Person Section 1.1

Personal Data Section 1.1

Post-Closing Representation Section 10.13(a)

Post-Closing Statement Section 2.6(b)

Pre-Closing Reorganization Documents Section 2.4

Pre-Closing Reorganization Transaction Steps Schedule Section 1.1

Pre-Closing Reorganization Transactions Recitals

Pre-Closing Reorganization Transactions Transfer Taxes Section 1.1

Pre-Closing Tax Period Section 1.1

Privileged Communications Section 10.13(b)

Proceeding Section 1.1 Process Section 1.1

Property Rights Section 1.1

Protected Information Section 1.1

Purchase Price Section 2.2

Purchased Assets Section 3.23(a)

Purchaser Preamble

Purchaser 401(k) Plan Section 5.6(j)

Purchaser Fundamental Representations Section 1.1

Purchaser Indemnified Person Section 9.3

Purchaser Material Adverse Effect Section 4.1

Purchaser's Allocation Notice Section 2.7(a)

R&W Insurance Policy Section 5.16 Real Property Section 3.10(c)

Real Property Lease Section 3.10(a)

Registered Intellectual Property Section 3.8(a)(i)

Regulatory Approvals Section 1.1

Relating to the Business Section 1.1

Release Section 1.1

Replacement Contract Section 5.14(a)

Representatives Section 1.1

Restricted Party Section 1.1

Retained Businesses Section 1.1

Section 721 Section 1.1

Securities Act Section 1.1 Seller Preamble Seller 401(k) Plans Section 5.6(j) Seller Benefit Plan Section 1.1 Seller Consolidated Tax Return Section 1.1 Seller Disclosure Schedules Section 1.1 Seller Fundamental Representations Section 1.1 Seller Indemnified Person Section 9.4 Seller Related Party Section 3.18 Seller's Allocation Section 2.7(a) Shared Contracts Section 5.14(a) Software Section 1.1 [\*\*\*] Section 2.9 Solvent Section 4.9 Specified Contract Section 5.13(a)
Specified Contract Assignments Section 5.13(b) Specified Matters Section 1.1 Specified Reports Section 1.1 Straddle Tax Period Section 1.1 Subcontract Agreement Pending Novation Section 2.5(b)(iii) Subsidiary Section 1.1 Surveys Section 5.15(b) Target Working Capital Section 1.1 Tax Section 1.1 Tax Proceeding Section 1.1 Tax Return Section 1.1 Taxing Authority Section 1.1 Third-Party Claim Section 9.5(a) Title Insurance Policies Section 5.15(a) Trade Secrets Section 1.1 Trademarks Section 1.1 Transaction Recitals Transaction Documents Section 1.1 Transfer Taxes Section 1.1 Transferred Entity Section 1.1 Transferred Entity Benefit Plan Section 1.1 Transferred Entity Equity Interests Section 1.1
Transferred Leased Properties Section 3.10(a)
Transferred Owned Property Section 3.10(b) Unpaid Company Transaction Expenses Section 1.1 VS2 Section 1.1 VS2 Equity Interests Section 3.2 WARN Section 1.1 Working Capital Section 1.1

### ARTICLE II PURCHASE AND SALE; CLOSING

Section 2.1 <u>Purchase and Sale</u>. Subject to the terms and conditions of this Agreement, at the Closing, Seller hereby agrees to sell, assign, transfer and convey to Purchaser, and

Purchaser agrees to purchase and acquire from Seller, the Company Equity Interests free and clear of all Liens other than transfer restrictions imposed by applicable federal and state securities Laws, for good and valuable consideration, including the payment by Purchaser of the Aggregate Purchase Price.

- Section 2.2 <u>Purchase Price</u>. The aggregate purchase price payable by Purchaser to Seller in consideration for the purchase of the Company Equity Interests shall be an amount equal to (a) \$50,000,000 (the "<u>Base Purchase Price</u>"), plus (i) Closing Cash Amounts, minus (ii) Unpaid Company Transaction Expenses minus (iii) Closing Indebtedness, plus (iv) the Estimated Working Capital Amount (which, for the avoidance of doubt, may be a negative number), in each case in cash (as adjusted pursuant to <u>Section 2.6</u> and <u>Schedule 3</u>, the "<u>Purchase Price</u>") plus (b) \$50,000,000 in cash to the extent payable subject to and in accordance with <u>Section 2.9</u> and <u>Schedule 2</u> (the "<u>Earn-Out</u>"). The Purchase Price together with the Earn-Out shall be collectively referred to as the "<u>Aggregate Purchase Price</u>".
- Section 2.3 <u>Closing Date</u>. The closing of the Transaction (the "<u>Closing</u>") shall take place at 10:00 a.m. Eastern Time, at the offices of Jones Day, 600 Brickell Avenue, Miami, FL 33131 (or, at the request of either party to this Agreement, by means of a virtual closing through electronic exchange of documents and signatures), on the third Business Day following the date on which the last of the conditions set forth in <u>Article VII</u> (other than those conditions that are to be satisfied by action taken at the Closing, but subject to the satisfaction or waiver of such conditions at the Closing) have been satisfied (or, to the extent permitted, waived by the parties entitled to the benefits thereof) or at such other place, time and date as may be agreed between Seller and Purchaser. The date on which the Closing occurs is referred to in this Agreement as the "<u>Closing Date</u>."
- Section 2.4 <u>Pre-Closing Reorganization Transactions.</u> Subject to the terms and conditions of this Agreement, including <u>Section 5.13</u>, prior to the Closing, each of Seller and the Company shall implement the Pre-Closing Reorganization Transactions in accordance with the Pre-Closing Reorganization Transaction Steps Schedule and shall enter into or amend certain Contracts to effect the Pre-Closing Reorganization Transactions (such Contracts, the "<u>Pre-Closing Reorganization Documents</u>"). If Seller or the Company make any amendments to the Pre-Closing Reorganization Documents that contain material terms or material conditions which are not expressly set forth on the Pre-Closing Reorganization Transaction Steps Schedule, Seller shall obtain Purchaser's consent (not to be unreasonably withheld, conditioned or delayed). Following the consummation of the Pre-Closing Reorganization Transactions, the Company will own and operate the Business.

### Section 2.5 Payment; Closing Deliveries.

- (a) <u>Closing Date Payments</u>. At the Closing, Purchaser shall make (or cause to be made):
- (i) a payment in an amount equal to the Closing Purchase Price <u>less</u> the Deductible Escrow Amount, by wire transfer(s) of immediately available funds to one or more bank accounts designated in writing by Seller (such designation to be made by Seller at least two (2) Business Days prior to the Closing Date);
- (ii) subject to Section 2.9 and <u>Schedule 2</u>, a payment in an amount equal to the Earn-Out to the Seller, by wire transfer(s) to one or more bank accounts designated in

writing by Seller (such designation to be made by Seller at least two (2) Business Days prior to the Closing Date); and

- (iii) a payment in an amount equal to the Deductible Escrow Amount to the Escrow Agent, by wire transfer(s) of immediately available funds to one or more bank accounts designated in writing by the Escrow Agent as set forth in the Escrow Agreement.
  - (b) <u>Closing Deliveries of Purchaser</u>. At the Closing, Purchaser shall deliver, or cause to be delivered, to Seller, the following:
    - (i) the certificate to be delivered pursuant to <u>Section 7.3(c)</u>;
- (ii) a counterpart of the Escrow Agreement substantially in the form attached hereto as <u>Exhibit A</u> (the "<u>Escrow Agreement</u>"), duly executed by Purchaser; and
- (iii) a counterpart of the Subcontract Agreement Pending Novation in substantially the form attached hereto as <u>Exhibit B</u> (the "<u>Subcontract Agreement Pending Novation</u>"), duly executed by Purchaser.
- (c) <u>Closing Deliveries of Seller</u>. At the Closing, Seller shall deliver, or cause to be delivered by one of its Affiliates, to Purchaser the following:
- (i) an assignment of membership interests substantially in the form attached as <u>Exhibit G</u>, duly endorsed by the Seller to transfer the Company Equity Interests to the Purchaser;
  - (ii) the certificate to be delivered pursuant to <u>Section 7.2(d)</u>;
  - (iii) duly executed copies of the Pre-Closing Reorganization Documents substantially in the form attached as <u>Exhibit</u>
    - (iv) a properly completed and duly executed IRS Form W-9 of Seller;
- (v) customary releases in form and substance reasonably satisfactory to the Purchaser with respect to all guarantee obligations and other Indebtedness of the Transferred Entities under Seller's credit facilities and all UCC-3 termination statements and other releases, if any, relating to the Transferred Entities' assets, properties or rights secured by such Indebtedness;
  - (vi) a counterpart of the Escrow Agreement duly executed by Seller and the Escrow Agent;
  - (vii) a counterpart of the Subcontract Agreement Pending Novation, duly executed by Seller and the Company;
- (viii) a certificate substantially in the form attached as <u>Exhibit H</u>, duly executed by an officer of the Company, dated as of the Closing Date and attaching with respect to each of the Transferred Entities true and complete copies of (i) such Transferred Entity's articles of incorporation or certificate of formation (or equivalent governing document) and all amendments thereto (if any), in effect as of the Closing Date, (ii) such Transferred Entity's

<u>F</u>;

bylaws or operating agreement (or equivalent governing document) and all amendments thereto (if any), in effect as of the Closing Date, and (iii) resolutions of the board of directors or managers (or equivalent governing body) of Seller and the Company as to the authorization of this Agreement and the Transaction;

- (ix) resignation letters, effective as of the Closing Date and in form reasonably acceptable to Purchaser, from the managers and officers of the Transferred Entities from their respective capacities as such, as requested by Purchaser in writing not less than five (5) Business Days prior to the Closing Date;
- (x) a certificate of good standing from each of the Transferred Entity's jurisdiction of formation (for such Transferred Entity) and certificates of good standing (or equivalent documentation, to the extent applicable) from each jurisdiction in which a Transferred Entity is qualified or registered to do business, in each case, that is dated within ten (10) Business Days of the Closing Date;
  - (xi) the Specified Reports;
- (xii) evidence reasonably satisfactory to Purchaser of final invoices (with wire instructions) for the payees of all Unpaid Company Transaction Expenses of the Transferred Entities and Seller as of Closing (except for any retention, change-of-control or similar payments, for which the payees will provide wire instructions no less than two (2) Business Days prior to the Closing); and
- (xiii) evidence of consent in form and substance reasonably satisfactory to Purchaser of the third parties identified on Schedule 2.5(c)(xiii).

### Section 2.6 Adjustments to Base Purchase Price.

- (a) At least five (5) Business Days prior to the Closing Date, Seller shall cause to be prepared and delivered to Purchaser a closing statement (the "Closing Statement") setting forth a good-faith estimate of (i) the Closing Working Capital (such estimate, the "Estimated Closing Working Capital Amount"), (ii) the Closing Cash Amounts (such estimate, the "Estimated Closing Cash Amounts"), (iii) the Estimated Unpaid Company Transaction Expenses (such estimate, the "Estimated Unpaid Company Transaction Expenses"), and (iv) the Closing Indebtedness (such estimate, the "Estimated Closing Indebtedness") and the resulting calculation of the Closing Purchase Price. The Closing Statement shall be prepared in accordance with the Transaction Accounting Principles, including for purposes of calculating the Working Capital, the use of the same line items and line item entries set forth on and used in the preparation of the Sample Working Capital Statement.
- (b) Within ninety (90) days after the Closing Date, Purchaser shall cause to be prepared and delivered to Seller a written statement (the "Post-Closing Statement"), setting forth the calculation of the Closing Working Capital, the Closing Cash Amounts, the Closing Unpaid Company Transaction Expenses and the Closing Indebtedness. The Post-Closing Statement shall be prepared in accordance with the Transaction Accounting Principles, including for purposes of calculating the Working Capital, the use of the same line items and line item entries set forth on and used in the preparation of the Sample Working Capital Statement.

Within forty-five (45) days following receipt by Seller of the Post-Closing Statement, Seller shall deliver written notice to Purchaser of any dispute Seller has with respect to the preparation or content of the Post-Closing Statement (the "Dispute Notice"); provided, that if Seller does not deliver any Dispute Notice to Purchaser within such forty-five (45)-day period, the Post-Closing Statement will be final, conclusive and binding on the parties hereto. The Dispute Notice shall set forth in reasonable detail the basis for any dispute included therein, the amounts involved and Seller's determination of the Closing Working Capital, the Closing Cash Amounts, the Closing Unpaid Company Transaction Expenses and the Closing Indebtedness; provided, that any dispute set forth in the Dispute Notice shall be limited to the determination of the Closing Working Capital, the Closing Cash Amounts, the Closing Unpaid Company Transaction Expenses and the Closing Indebtedness, and Seller may not dispute the accounting principles, practices, methodologies and policies used in preparing the Post-Closing Statement unless they are inconsistent with the Transaction Accounting Principles. Upon receipt by Purchaser of a Dispute Notice, Purchaser and Seller shall negotiate in good faith to resolve any disputed items and amounts set forth therein, and the parties hereto agree that other than to the extent disputed in the Dispute Notice, the Post-Closing Statement will be final, conclusive and binding on the parties hereto. If Purchaser and Seller, such good faith effort notwithstanding, fail to resolve any such dispute within fifteen (15) Business Days following receipt by Purchaser of the Dispute Notice (the "Dispute Resolution Period"), then Purchaser and Seller jointly shall engage, within ten (10) Business Days following the expiration of the Dispute Resolution Period, BDO US LLP or if BDO US LLP is unavailable, then a nationally recognized accounting firm selected jointly by Seller and Purchaser (the "<u>Independent Accounting Firm</u>") to resolve any such dispute (and only such unresolved disputes in the Dispute Notice). If BDO US LLP is unavailable and Seller and Purchaser are unable to agree on the Independent Accounting Firm (if the firm agreed to is unavailable or conflicted), then each of Seller and Purchaser shall select a nationally recognized major accounting firm, and the two (2) firms will mutually select a third nationally recognized major accounting firm to serve as the Independent Accounting Firm. As promptly as practicable, and in any event not more than fifteen (15) days following the engagement of the Independent Accounting Firm, Purchaser and Seller shall each prepare and submit a presentation detailing each party's complete statement of proposed resolution of each issue still in dispute to the Independent Accounting Firm. Purchaser and Seller shall cause the Independent Accounting Firm to, as soon as practicable after the submission of the presentations described in the immediately preceding sentence and in any event not more than thirty (30) days following such presentations, make a final determination, binding on the parties hereto, of the appropriate amount of each of the line items that remain in dispute as indicated in the Dispute Notice. With respect to each disputed line item, such determination, if not in accordance with the position of either Seller or Purchaser, shall not be in excess of the higher, nor less than the lower, of the amounts advocated by Seller or Purchaser, as applicable, in their respective presentations to the Independent Accounting Firm described above. The scope of the disputes to be resolved by the Independent Accounting Firm shall be limited to determining any items still in dispute in the Dispute Notice in accordance with the terms of this Agreement, and the Independent Accounting Firm is not to make any other determination. All fees and expenses relating to the work, if any, to be performed by the Independent Accounting Firm shall be allocated between Purchaser and Seller in inverse proportion to the relative amounts of the aggregate of the disputed amounts determined by the Independent Accounting Firm to be for the account of Purchaser and Seller, respectively (i.e., so that the prevailing party bears a lesser amount of such costs, fees and expenses). All determinations made by the Independent Accounting Firm, and the Post-Closing Statement, as modified by the Independent Accounting Firm, will be final, conclusive and binding on the parties hereto.

- (d) For purposes of complying with the terms set forth in this Section 2.6, from and after the Closing, Seller and Purchaser shall reasonably cooperate with and make available to each other, the Independent Accounting Firm and each of their respective Representatives all information, records, data and working papers, in each case to the extent related to the Business or Transferred Entities, and shall permit reasonable access, upon advance written notice, during normal business hours to the books and records of the Transferred Entities and their personnel (subject to reasonable confidentiality restrictions and to providing such assurances, releases, indemnities or other agreements as accountants may customarily require in such circumstances), as may be reasonably required in connection with the preparation, analysis and review of the Post-Closing Statement and the resolution of any disputes thereunder.
- (e) The "<u>Final Purchase Price</u>" means an amount equal to the Base Purchase Price, *plus* (i) the Closing Cash Amounts as finally determined pursuant to this Section <u>2.6</u>, *minus* (ii) the Closing Unpaid Company Transaction Expenses as finally determined pursuant to this <u>Section 2.6</u>, *minus* (iii) the Closing Indebtedness as finally determined pursuant to this <u>Section 2.6</u>, *plus* (iv) the Final Working Capital Amount as finally determined pursuant to this <u>Section 2.6</u> and <u>Schedule 3</u> (which, for the avoidance of doubt, may be a negative number).
- (f) If the Closing Purchase Price exceeds the Final Purchase Price, then Seller shall pay or cause to be paid an amount in cash equal to such excess to Purchaser by wire transfer of immediately available funds to an account or accounts designated in writing by Purchaser to Seller; or if the Final Purchase Price exceeds the Closing Purchase Price, then Purchaser shall pay or cause to be paid an amount in cash equal to such excess to Seller by wire transfer of immediately available funds to an account or accounts designated in writing by Seller to Purchaser. Any such payment pursuant to this Section 2.6(f) is to be made within five (5) Business Days of the date on which the Closing Working Capital, the Closing Cash Amounts, the Closing Unpaid Company Transaction Expenses and the Closing Indebtedness are finally determined pursuant to this Section 2.6.
- (g) The process set forth in this <u>Section 2.6</u> shall be the sole and exclusive remedy of any of the parties and their respective Affiliates for any disputes related to the Closing Working Capital, the Closing Cash Amounts, the Closing Unpaid Company Transaction Expenses, the Closing Indebtedness, the Closing Purchase Price, the Final Purchase Price and the calculations and amounts on which they are based or set forth in the related statements and notices delivered in connection therewith.

#### Section 2.7 Purchase Price Allocation.

- (a) Seller and Purchaser agree to allocate and, as applicable, to cause their relevant Affiliates to allocate, the Purchase Price (as finally determined pursuant to Section 2.6) and any other items that are treated as additional consideration for Tax purposes, (i) among the Energetics Equity Interests and (ii) any other assets of the Company, pursuant to this Agreement and any other Transaction Document, in accordance with Exhibit C attached hereto (the "Allocation Schedule").
- (b) No later than ninety (90) days after the date on which the Purchase Price is finally determined pursuant to Section 2.6, Seller shall deliver to Purchaser a proposed allocation of the Purchase Price (as finally determined pursuant to Section 2.6) and any other items that are treated as additional consideration for U.S. income Tax purposes between Energetics Equity Interests and any other assets of the Company, determined in a manner consistent with

Section 1060 of the Code and the Treasury Regulations promulgated thereunder, any other relevant provisions of applicable Tax Law, and the Allocation Schedule ("Seller's Allocation"). If Purchaser disagrees with Seller's Allocation, Purchaser may, within thirty (30) days after receipt of Seller's Allocation, deliver a notice ("Purchaser's Allocation Notice") to Seller to such effect, specifying those items as to which Purchaser disagrees and setting forth Purchaser's proposed allocation. If Purchaser's Allocation Notice is duly delivered, Seller and Purchaser shall, during the thirty (30) days immediately following such delivery, use commercially reasonable efforts to reach agreement on the disputed items or amounts in order to determine the allocation of the Purchase Price (as finally determined pursuant to Section 2.6) and any other items that are treated as additional consideration for U.S. income Tax purposes. If Seller and Purchaser are unable to resolve such disputed items or amounts within thirty (30) days after the delivery of Purchaser's Allocation Notice, such disputed items or amounts will be resolved by the Independent Accounting Firm in accordance with Section 2.6(c). The Seller's Allocation, as prepared by Seller if no Purchaser's Allocation Notice has been given, as adjusted pursuant to any agreement between Seller and Purchaser and/or resolved by the Independent Accounting Firm (the "Allocation"), shall be conclusive and binding on the parties hereto and neither Seller nor Purchaser shall (and they shall cause their respective Affiliates not to) take any position for income Tax purposes inconsistent with the Allocation on any Tax Return or in any Tax Proceeding or otherwise, in each case, except to the extent otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code (or any analogous provision of state, local or foreign law). The Allocation shall be adjusted, as necessary, to reflect any subsequent adjustments to the Aggregate Purchase Price and any other

Section 2.8 <u>Income Tax Treatment</u>. Seller, Purchaser, the Transferred Entities, and their respective Affiliates shall treat any and all payments under this <u>Article II</u> as an adjustment to the Purchase Price for Tax purposes, except to the extent otherwise required by Law (including any portion of the Earn-Out amount deemed to be interest for U.S. federal income tax purposes under Treasury Regulations Section 1.483-4 or 1.1275-4(c)).

Section 2.9 <u>Earn-Out</u>. The Seller submitted [\*\*\*] set forth on <u>Schedule 2</u> hereto (the "[\*\*\*]"). If [\*\*\*] on or prior to the three-year anniversary of the Closing Date [\*\*\*], then, at or following the Closing, Purchaser shall pay to Seller the Earn-Out subject to and in accordance with the terms set forth on <u>Schedule 2</u> to this Agreement. In the event that [\*\*\*] is issued prior to the second (2nd) anniversary of the Closing Date but after the first (1st) anniversary of the Closing Date, Seller shall be entitled to eighty percent (80%) of the Earn-Out; in the event that [\*\*\*] issued on or after the second (2nd) anniversary of the Closing Date but prior to the third (3rd) anniversary of the Closing Date, Seller shall be entitled to forty percent (40%) of the Earn-Out. For the avoidance of doubt, in the event that the Business does not receive [\*\*\*] within three (3) years of the Closing Date, no Earn-Out shall be payable under this Agreement. Notwithstanding anything to the contrary, the parties hereto acknowledge and agree that the Seller and/or the Company obtaining or being [\*\*\*] is not a condition to Closing. If the Earn-Out is not paid at the Closing, the Purchaser shall not, and shall cause the Transferred Entities not to, directly or indirectly, enter into any Contract that restricts, prevents or otherwise prohibits the ability of Purchaser or the Company including any successor thereto to timely and fully pay the Earn-Out pursuant to the terms of this Agreement. If the Earn-Out is not paid at the Closing, the Purchaser and the Company and each of the Transferred Entities, and each of their successors

and assigns, shall be jointly and severally liable to the Seller for the payment of the Earn-Out pursuant to and in accordance with the terms of this Agreement.

Section 2.10 <u>Withholding</u>. Seller and Purchaser are each permitted to deduct and withhold amounts from any payment made under this Agreement (a) as required under applicable Law or (b) to offset any obligations of the Seller owed (i) to Purchaser pursuant to <u>Section 2.6</u> (as finally determined in accordance with the terms of <u>Section 9.5</u>) or (ii) to any Purchaser Indemnified Persons pursuant to <u>Section 9.3</u> (as finally determined in accordance with the terms of <u>Section 9.5</u>). Seller or Purchaser, as applicable, shall (except with respect to withholding of employment and payroll taxes, or withholding arising from a failure to provide the deliverable in Section 2.5(c)(iv)) exercise commercially reasonable efforts to the extent permitted by applicable Law, to reasonably cooperate with the other party to reduce or eliminate any such withholding. Any amounts so deducted and withheld under the foregoing clause (a) shall be remitted by the withholding party to the appropriate Governmental Entity and shall be treated for all purposes of this Agreement as having been paid to the party in respect of which such deduction and withholding was made.

# ARTICLE III REPRESENTATIONS AND WARRANTIES OF SELLER

Except as set forth in, or qualified by any matter set forth in, the Seller Disclosure Schedules (it being agreed that the disclosure of any matter in any Section in the Seller Disclosure Schedules shall be deemed to have been disclosed in any other Section in the Seller Disclosure Schedules to which the applicability of such disclosure is reasonably apparent on the face of such disclosure), Seller hereby represents and warrants to Purchaser as follows:

Section 3.1 <u>Organization, Standing and Power.</u> Seller and each Transferred Entity is duly organized or incorporated, validly existing and, to the extent such concept is applicable in the relevant jurisdiction, in good standing or duly registered under the Laws of its jurisdiction of organization or incorporation, and Seller and each Transferred Entity has (or, with respect to the Company, following the Pre-Closing Reorganization Transactions will have) all necessary organizational power and authority to carry on the Business as presently conducted, except where the failure to have such power and authority would not, individually or in the aggregate, result in material Liability to Seller or any of the Transferred Entities or reasonably be expected to prevent or materially delay the ability of Seller and each Transferred Entity to consummate the Transaction. Each of the Transferred Entities is qualified or otherwise authorized to act as a foreign corporation (or equivalent entity) and is in good standing under the Laws of every other jurisdiction in which such qualification or authorization is necessary under applicable Law, except in any case as would not reasonably be expected to have, individually or in the aggregate, a Business Material Adverse Effect.

Section 3.2 <u>Transferred Entities</u>. As of the date hereof, all of the Company Equity Interests are held beneficially and of record by Seller, Seller has good and marketable title to the Company Equity Interests, and at the Closing the Company Equity Interests will be held free and clear of all Liens. As of the date hereof, all of the issued and outstanding equity interests of Energetics (the "<u>Energetics Equity Interests</u>")are held beneficially and of record by Seller, Seller has good and marketable title to the Energetics Equity Interests. As of the Closing, pursuant to the Pre-Closing Reorganization Transactions, Energetics has been converted to a Delaware limited liability company, and as of the Closing, the Energetics Equity Interests are held free and clear of all Liens by the Company. As of the date hereof, Seller holds beneficially

and of record, fifty one percent (51%) of the issued and outstanding membership interests of VS2 (the membership interests held by Seller in VS2, the "VS2 Equity Interests"). Seller has good and marketable title to the VS2 Equity Interests, and at the Closing the VS2 Equity Interests will be held free and clear of all Liens by the Company. Following the consummation of the Pre-Closing Reorganization Transactions, the Company will own the Energetics Equity Interests and the VS2 Equity Interests, in each case beneficially and of record, and immediately prior to the Closing, free and clear of all Liens. All of the Transferred Entity Equity Interests have been validly issued, and are fully-paid and nonassessable. Except as contemplated by this Agreement or set forth in Section 3.2 of the Seller Disclosure Schedules, there are no outstanding options, warrants, calls, purchase rights, subscription rights, exchange rights or other similar rights, convertible securities, agreements or commitments of any kind pursuant to which any of the Transferred Entities is or may become obligated to (a) issue, transfer, sell or otherwise dispose of any of its equity interests, or (b) redeem, purchase or otherwise acquire any outstanding Transferred Entity Equity Interests, in each case, other than as may be set forth in the organizational documents of such Transferred Entity (current and complete copies of which have been made available to Purchaser).

Section 3.3 Authority; Execution and Delivery; Enforceability. Each of the Seller and the Company has all necessary corporate power and authority to execute this Agreement and to consummate the Transaction and the other transactions contemplated hereby. The execution and delivery by Seller and the Company of this Agreement and each Transaction Document to which the Seller and/or the Company is a party and the consummation by Seller and the Company of the Transaction and the other transactions contemplated hereby and under the Transaction Documents have been duly authorized by all necessary corporate or other action of Seller and the Company. Each of Seller and the Company has duly executed and delivered this Agreement and, at the Closing, will duly execute and deliver each other Transaction Document to which it is a party, and assuming due authorization, execution and delivery by Purchaser, each of this Agreement and the other Transaction Documents will constitute its valid and binding obligation, enforceable against it in accordance with its terms, subject to the effect of any Laws relating to bankruptcy, reorganization, insolvency, moratorium, fraudulent conveyance or preferential transfers, or similar Laws relating to or affecting creditors' rights generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a Proceeding in equity or at law).

Section 3.4 No Conflicts; Consents; Governmental Authorization. Except as set forth on Section 3.4 of the Seller Disclosure Schedule, the execution and delivery by each of Seller and the Company of this Agreement does not, and the execution by Seller and the Company of the other Transaction Documents will not, and the consummation of the Transaction and the other transactions contemplated hereby and thereby and compliance by Seller and the Company with the terms hereof and thereof will not, conflict with, or result in any violation of, breach or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation under, or result in the creation of any Lien (other than Permitted Liens) under, any provision of (a) the certificate of incorporation, bylaws, articles of association, limited liability company agreement, certificate of formation, or equivalent governing and/or constitutional documents of Seller or any Transferred Entity, (b) any Judgment or Law applicable to the Business or any of the Transferred Entities or (c) any Material Contract or Real Property Lease. Except as set forth on Section 3.4 of the Seller Disclosure Schedules, no Approval of any Governmental Entity is required to be obtained or made by or with respect to Seller or the Company in connection with the execution, delivery and performance of this Agreement and the other Transaction Documents or the consummation of the

Transaction and the other transactions contemplated hereby and thereby, other than those that, if not obtained, made or given, would not reasonably be expected to be material to any of the Transferred Entities or the Business.

Section 3.5 Financial Statements. Section 3.5 of the Seller Disclosure Schedules sets forth true and complete copies of (a) the unaudited balance sheet of the Business (together with the notes and schedules thereto, if any, the "Business Balance Sheet") as of December 31, 2022 (the "Balance Sheet Date"), (b) the unaudited statement of income of the Business for the twelve (12) month period ending on December 31, 2022, (c) the unaudited statement of income and the unaudited balance sheet for the Business for and as of the fiscal year ended December 31, 2021, if any, and (d) the unaudited statement of income and the unaudited balance sheet for the Business for the period commencing January 1, 2023 and ending March 31, 2023 (clauses (a), (b), (c) and (d), collectively, the "Business Financial Statements"). The Business Financial Statements (i) have been prepared from the books and records of the Business, which are maintained in the ordinary course of business and in accordance with good business practices, and which reflect only valid transactions and are reliable, and in all material respects, complete and accurate (except, in each case, (A) as expressly noted therein, (B) for failures to be so prepared that would not result in an unfair presentation of the financial position and the results of operations of the Business, in the aggregate, on the basis of presentation outlined in Section 3.5 of the Seller Disclosure Schedules, and (C) subject to the absence of notes, and, in the case of clause (a) of the definition of Business Financial Statements, subject to normal year-end adjustments), (ii) have been prepared in accordance with GAAP, on a consistent basis during the periods covered thereby, and (iii) fairly present in all material respects the financial condition and results of operations of the Business as of the respective dates thereof and for the respective periods covered; provided, however, that the Business Financial Statements and the foregoing representations and warranties are qualified by the fact that (i) the Business has not operated on a separate stand-alone basis and has historically been reported within Seller's or its Affiliates' combined financial statements and (ii) the Business Financial Statements assume certain allocated charges and credits which do not necessarily reflect amounts that would have resulted from arm's-length transactions or that the Business would incur on a stand-alone basis. The Seller has established and maintains systems of internal accounting controls that are designed to provide reasonable assurances that all transactions are recorded as necessary to permit the preparation of proper and accurate financial statements in accordance with GAAP. The Seller has not, and, to the Knowledge of Seller, its accountants or representatives have not received any unresolved material written complaint, allegation or assertion of a problem or claim regarding the accounting or auditing practices, procedures, methodologies or methods of the Seller or the Seller's accounting controls. To the Knowledge of the Seller, there are no material weaknesses or significant deficiencies in the design or operations of the internal financial controls utilized by Seller. As of the Closing, the books and records of the Transferred Entities and the Business are in the possession of the Company.

Section 3.6 <u>Absence of Undisclosed Liabilities</u>. The Business does not have any Liabilities, other than Liabilities that (a) are reflected or reserved against in the Business Balance Sheet, (b) were incurred since the Balance Sheet Date in the ordinary course of business and not material in amount (but excluding Liabilities arising out of a breach of, or default under, any Contract, breach of warranty, tort or infringement claim or lawsuit), (c) will be reflected, reserved, accrued, recorded or included in the Closing Working Capital, or the Closing Indebtedness, (d) are incurred in connection with the Transaction and the other transactions contemplated by this Agreement or the other Transaction Documents, or (e) are not individually or in the aggregate material to the Business. There are no "off balance sheet

arrangements" (within the meaning of Item 303 of Regulation S-K promulgated by the Securities and Exchange Commission) with respect to the Business.

Section 3.7 <u>Absence of Changes or Events</u>. Except in connection with the Transaction and the other transactions contemplated by this Agreement or the other Transaction Documents, from and after the Balance Sheet Date, (a) except as set forth on <u>Section 3.7(a)</u> of the Seller Disclosure Schedules, Seller has not taken any action that, if taken post-Closing, would be prohibited by the terms of <u>Section 5.2</u> and (b) there has not been any event, change, effect or development that is reasonably expected to have, individually or in the aggregate, a Business Material Adverse Effect.

# Section 3.8 <u>Intellectual Property</u>.

- (a) <u>Section 3.8(i)</u> of the Seller Disclosure Schedules lists all (i) registrations and applications with a Governmental Entity or a domain name registrar for the registration of Business Intellectual Property owned by Seller or its Subsidiaries (collectively, the "<u>Registered Intellectual Property</u>"); and (ii) unregistered Intellectual Property that is material to the conduct of the Business owned by Seller or its Subsidiaries as of the date of this Agreement (the Intellectual Property in subsection (i) and (ii), collectively, the "<u>Company Business Intellectual Property</u>"). For each listed item, <u>Section 3.8(a)</u> of the Seller Disclosure Schedules provides, as applicable, the registered owner of such Intellectual Property, the countries in which such Intellectual Property is patented or registered (or in which applications for patents or registrations have been filed), the patent or registration number, and the filing and expiration dates thereof. No opposition, cancellation, reexamination, invalidation or other Proceeding is pending challenging the validity, enforceability or the Transferred Entities' ownership of any Registered Intellectual Property, nor, to the Knowledge of Seller, is there a reasonable basis for any such challenge. Except as required by the Pre-Closing Reorganization Transactions, neither Seller nor the Transferred Entities have assigned or otherwise transferred ownership of, or agreed to assign or otherwise transfer ownership of, any Company Business Intellectual Property to any other Person.
- (b) (i) As of the date hereof, Seller and its Subsidiaries and (ii) as of the Closing Date, the Transferred Entities, in each case solely and exclusively own all right, title and interest in and to, or otherwise have a right to use pursuant to a binding, written license agreement all Business Intellectual Property, including each item of Registered Intellectual Property, free and clear of all Liens, other than Permitted Liens.
- (c) Except as will be owned by the Company following the consummation of the Pre-Closing Reorganization Transactions, neither Seller nor any of its Subsidiaries (other than the Transferred Entities) hold any right, title or interest in or to any Company Business Intellectual Property.
- (d) (i) Each item of the Registered Intellectual Property is subsisting and valid, and all material filings and fees required todate related to each item of Registered Intellectual Property have been timely filed and paid to the relevant Governmental Entity; (ii) the operation of the Business as currently conducted as of the date of this Agreement and as previously conducted within the five (5) year period prior to the date of this Agreement, does not and has not infringed, misappropriated or otherwise violated the Intellectual Property of any Person; (iii) to the Knowledge of Seller, no third party is infringing, misappropriating or otherwise violating any Company Business Intellectual Property; (iv) neither Seller nor any of its

Subsidiaries has made any written allegation against a third party of any infringement, misappropriation, or other violation of any Company Business Intellectual Property; (v) neither Seller nor any of its Subsidiaries has received any written notice within the five (5) year period prior to the date of this Agreement alleging that the operation of the Business infringes, misappropriates, violates or otherwise conflicts with the Intellectual Property Rights of any other Person; and (vi) within the five (5) year period prior to the date of this Agreement, there have been no (1) claims or Proceedings against the Transferred Entities or Seller relating to the Business in which claims are asserted alleging infringement, misappropriation or other violations of the Intellectual Property of any Person that are pending or (2) claims threatened in writing against the Transferred Entities or Seller relating to the Business alleging infringement, misappropriation or other violations of the Intellectual Property of any Person. No Company Business Intellectual Property or, to the Knowledge of Seller, any Business Intellectual Property exclusively licensed by Seller or any of its Subsidiaries is subject to any outstanding consent, settlement or Judgment restricting the use or ownership thereof.

- (e) All current and former officers, employees, agents, consultants and independent contractors of the Seller or any of its Subsidiaries that were involved in the creation or development of any material Company Business Intellectual Property have executed and delivered to the Seller or the applicable Subsidiary Contracts pursuant to which such Person (i) agrees to maintain the confidentiality of, and not use or disclose, the confidential information of the Seller or the applicable Subsidiary; and (ii) assigns to the Seller or the applicable Subsidiary all Intellectual Property related to the Business created or developed by such Person in the course of such Person's employment or other relationship with the Seller or the applicable Subsidiary. To the Knowledge of Seller, no former or current employees, agents, consultants or independent contractors of the Seller or the applicable Subsidiary have breached or violated any of the Contracts referenced in the immediately preceding sentence. No current or former officer or employee of, or consultant or independent contractor to, the Seller or any of its Subsidiaries is asserting or, to the Knowledge of Seller, has grounds to assert any rights to any of the Company Business Intellectual Property.
- (f) The Seller and its Subsidiaries have taken commercially reasonable measures to protect, enforce and maintain the confidentiality (where applicable) of the Business Intellectual Property (including any Trade Secrets). To the Knowledge of Seller, there have been no unauthorized uses or disclosures of any such confidential information except pursuant to a valid and binding confidentiality agreement that imposes confidentiality obligations on the recipient of such confidential information.
- (g) Neither Seller nor any of its Subsidiaries has used any Open Source Software in any manner that does or would, with respect to any Software included in the Company Business Intellectual Property, products or services (the "Company Software"), (i) impose a requirement or condition that any such Company Software using, incorporating, distributed with or derived from such Open Source Software: (A) be made available, disclosed or distributed in source code form; (B) be licensed for the purpose of making modifications or derivative works; or (C) be redistributable at no charge; or (ii) impose any other material limitation, restriction or condition on the right or ability of the Business to use or distribute any product or service of the Business.
- (h) Except as set forth in <u>Section 3.8(h)</u> of the Seller Disclosure Schedule,] no Governmental Entity, hospital, educational institution or research center (collectively, "<u>Institutions</u>") has provided facilities or funding (in each case, directly or indirectly) for the

development of any Company Business Intellectual Property, and no Institution has any rights in or with respect to any developments of or to any Company Business Intellectual Property. Neither Seller nor its Subsidiaries is or was a member or promoter of, or a contributor to, any industry standards body or similar organization that would require or obligate Seller or its Subsidiaries to grant or offer to any other Person any license or right to any Company Business Intellectual Property.

# Section 3.9 Information Technology; Data Protection.

- (a) The Business IT Assets operate and perform reliably, in material conformance with applicable specifications and documentation for such assets, and substantially as needed by Seller and its Subsidiaries to adequately conduct the Business as currently conducted. The Business IT Assets are free of viruses, malware, and other corruptants, including any computer code intentionally designed to disrupt, disable, or harm in any manner the operation of, or to provide any unauthorized access to, any Business IT Asset, in each case, except as, individually or in the aggregate, that would be material to the Business. During the five (5) year period immediately prior to the date of this Agreement, there have been no failures, breakdowns, or other incidents adversely affecting any such Business IT Assets or any software, data, information or materials contained therein, other than temporary problems arising in the ordinary course of business that did not materially disrupt the operations of Seller or the Transferred Entities. The Company maintains commercially reasonable incident response, backup, disaster recovery and business continuity plans and procedures and has taken commercially reasonable measures to protect the security and integrity of the material Business IT Assets.
- (b) Except as set forth in Section 3.9(b) of the Seller Disclosure Schedules, during the five (5)-year period immediately prior to the date of this Agreement, there have not been any vulnerabilities or defects that would reasonably be expected to result in any security breaches or unauthorized access or other security access incidents (a) affecting the Business IT Assets or (b) resulting in a partial or complete loss of control of any Business assets, in each case, except as, individually or in the aggregate, would not reasonably be expected to have a Business Material Adverse Effect.
- (c) The Seller and each Transferred Entity take commercially reasonable actions to protect the confidentiality, integrity, operation, and security of Protected Information, Business Technology, and Business IT Assets. The Seller and each Transferred Entity have implemented appropriate technical and organizational measures, including standards, policies, procedures and documentation, designed to ensure, the Processing of Protected Information by the Seller and each Transferred Entity has been performed during the five (5)-year period immediately prior to the date of the Agreement in all material respects in accordance with applicable Data Protection Obligations.
- (d) The Seller and each Transferred Entity have taken all commercially reasonable measures to secure all Business Technology and Business IT Assets prior to selling, distributing, deploying or making it available and have made patches and updates to such Business Technology and Business IT Assets in accordance with good industry standards.
- (e) The Seller and each Transferred Entity has taken commercially reasonable measures to ensure that all third parties authorized to Process Protected Information will Process such Protected Information in accordance with applicable Data Protection Obligations.

- (f) Neither the execution, delivery or performance of this Agreement or any other Transaction Document, nor the consummation of any of the Transactions, nor the Seller's or any Transferred Entity's provision to Purchaser or Purchaser's possession of Protected Information or its Processing of such data consistent with historical practice, will result in any material violation of any Data Protection Obligations, so long as Purchaser Processes that data in a manner consistent with the Seller's or Transferred Entity's Processing of such data. The Transferred Entities shall continue to have at least the same rights to Process Protected Information immediately after Closing as each such Transferred Entity had immediately before Closing.
- (g) The Seller and the Transferred Entities are and have been during the three (3)-year period immediately prior to the date of this Agreement in material compliance with all applicable Data Protection Obligations.
- (h) None of the Seller or the Transferred Entities has, during the five (5)-year period immediately prior to the date of this Agreement, received any written notice or, to the Knowledge of Seller, any communication from any applicable Governmental Entity against or relating to any Transferred Entity alleging a violation of any Data Protection Laws by any Transferred Entity, nor has Seller or any Transferred Entity been threatened in writing to be charged with any such violation by any Governmental Entity, and, to the Knowledge of Seller, there are no circumstances as of the date hereof which could reasonably be expected to give rise to any such notices or communications, in each case, except as, individually or in the aggregate, that would not be material to the Business. During the five (5)-year period immediately prior to the date of this Agreement, there has been no unauthorized Processing or disclosure of, Personal Data in their possession or under the control of, or, to the Knowledge of Seller, Processed by or on behalf of, the Seller or any Transferred Entity, in each case, except as, individually or in the aggregate, that would be material to the Business.

## Section 3.10 Real Property.

(a) As of the date of this Agreement, the Seller has valid title to the leasehold estate (as lessee) in the real property set forth in Section 3.10(a) of the Seller Disclosure Schedules (the "Transferred Leased Property"), free and clear of all Liens, other than Permitted Liens. Following the consummation of the Pre-Closing Reorganization Transactions, the Company will have valid title to the leasehold estate (as lessee in the Transferred Leased Property), free and clear of all Liens, other than Permitted Liens. True and complete copies of each Lease (including all amendments and modifications) under which the Transferred Leased Property is held (each, a "Real Property Lease") have been made available to Purchaser. Each Real Property Lease is in full force and effect and is enforceable in accordance with its terms, subject to the effect of any Laws relating to bankruptcy, reorganization, insolvency, moratorium, fraudulent conveyance or preferential transfers, or similar Laws relating to or affecting creditors' rights generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a Proceeding in equity or at law). To the Knowledge of Seller, there is no default under any Real Property Lease by the Seller or, to the Knowledge of Seller, by any other party thereto and no event has occurred or circumstance exists which, with the delivery of notice, the passage of time or both, would constitute a breach or default under such Real Property Lease. Seller has not received written notice that (i) it is in default under any Real Property Lease and such default has not been cured or (ii) a party to a Real Property Lease (other than the Seller) is exercising a termination right with respect to such Real Property Lease. To the Knowledge of Seller, Seller's possession and quiet enjoyment of the

Transferred Leased Property under the applicable Real Property Lease has not been disturbed and, to the Seller's Knowledge, there are no disputes with respect to any Real Property Lease.

- (b) As of the date of this Agreement, the Seller holds valid title to the real property set forth in Section 3.10(b) of the Seller Disclosure Schedules (the "Transferred Owned Property"), free and clear of all Liens, other than Permitted Liens. Following the consummation of the Pre-Closing Reorganization Transactions, the Company will hold good and valid fee simple title to the Transferred Owned Property, free and clear of all Liens, other than Permitted Liens. Except as set forth in Schedule 3.10(b), the Seller has not leased or otherwise granted to any other Person the right to use or occupy any Transferred Owned Property or any portion thereof. There are no outstanding options, rights of first offer or rights of first refusal to purchase any Transferred Owned Property or any portion thereof or interest therein. Neither the Seller nor the Company is a party to any agreement or option to purchase any real property or interest therein.
- (c) The Transferred Leased Property together with the Transferred Owned Property (collectively, the "Real Property") constitute all of the real property necessary to conduct the business of the Company, as currently being conducted, following the consummation of the Pre-Closing Reorganization Transactions.
- (d) Except as would not have a Business Material Adverse Effect, (i) all improvements, equipment, machinery and fixtures located at, on, or within the demised premises constituting the Real Property are in good operating condition and repair, have been maintained in accordance with commercially reasonable standards and are adequate and suitable in all material respects for the present and continued use and operation thereof as used or operated as of the date of this Agreement, ordinary wear and tear excepted; (ii) to Seller's Knowledge, there are no pending or threatened condemnation or eminent domain proceedings, and the Seller has not received any notice of any such condemnation or eminent domain proceedings, affecting the Real Property; and (iii) the Seller has not leased, subleased, licensed or otherwise granted any third party the right to use or occupy any portion of the Real Property.
- (e) Each parcel of the Transferred Owned Property, and to the Knowledge of the Seller, the Transferred Leased Property, has full and free access to and from public ways, and all water, sewer, sanitary and storm drain facilities, community services, and all public utilities which access is sufficient for the purposes of the business of the Company following the consummation of the Pre-Closing Reorganization Transactions, and Seller does not have any knowledge of any fact or condition that would result in the interruption or termination of such access, except for any restrictions to access that would not be material to the operations of the Business. Seller has not received any written notice from any utility company or municipality of any fact or condition which would result in the discontinuation of presently available or otherwise necessary sewer, water, electric, gas, telephone or other utilities or services for the Real Property.
- (f) To the Knowledge of the Seller, (i) there are no violations of Law with respect to any Real Property; (ii) there are no violations of any covenant, restriction or easement affecting the Real Property or any part of it, or with respect to the use or occupancy of the Real Property or any part of it, from any Governmental Entity having jurisdiction over the Real Property or from any other Person entitled to enforce the same; and (iii) there have been no work orders, deficiency notices or other similar notices of non-compliance issued by any

Governmental Entity or otherwise with respect to the Real Property, which in each of the foregoing instances, have not been cured.

#### Section 3.11 Contracts.

- (a) Except as set forth in <u>Section 3.11</u> of the Seller Disclosure Schedules, as of the date hereof neither Seller nor any of the Transferred Entities is a party to or bound by any of the following Contracts Relating to the Business (other than any Benefit Plans) (the "<u>Material Contracts</u>"):
- (i) the ten (10) largest Contracts (measured by dollar value based on the fiscal year ended December 31, 2022) for the supply of services, products or materials for use in the Business;
- (ii) the ten (10) largest Contracts (measured by dollar value based on the fiscal year ended December 31, 2022) with customers of the Business;
  - (iii) any Contract containing any future capital expenditure obligations of the Business in excess of \$1,000,000;
- (iv) any Contract that is a joint venture, partnership or other similar agreement involving co-investment between the Business and a third party;
- (v) (A) any Contract pursuant to which Seller or any of its Subsidiaries licenses or is otherwise permitted by a third party to use any Intellectual Property (other than any license for COTS Software) involving payments in excess of \$80,000 per annum ("Company Inbound IPR Agreements"), or (B) any Contract pursuant to which a third party licenses from Seller or any of its Subsidiaries any Company Business Intellectual Property (other than non-exclusive licenses granted in the ordinary course of business, including in connection with the sale or licensing of any products or services) ("Company Outbound IPR Agreements");
- (vi) (A) any Contract relating to Indebtedness with respect to which a Transferred Entity is an obligor, other than any Indebtedness to be repaid or extinguished pursuant to this Agreement at or prior to the Closing, or (B) any Contract that authorizes or creates a Lien on any assets of the Business or Transferred Entity Equity Interests, other than any Liens that will be extinguished at or prior to the Closing;
- (vii) any Contract relating to the acquisition or disposition of any business (whether by merger, sale of stock, sale of assets or otherwise) under which, after Closing, a Transferred Entity will have an obligation with respect to an "earn out," contingent purchase price or similar contingent payment obligation;
- (viii) Contracts limiting the freedom of the Transferred Entities to engage in any line of business, acquire any entity or compete with any Person in any market or geographical area or during any period of time;
  - (ix) Contracts relating to loans or advances to, guarantees for the benefit of, or investments in, any Persons;

- (x) outstanding Contracts of guaranty, surety or indemnification, direct or indirect, by the Business or any of the Transferred Entities:
- (xi) Contracts evidencing any obligation of any Transferred Entity with respect to the issuance, sale, repurchase or redemption of any equity or other interests of any Transferred Entity;
- (xii) Contracts containing exclusivity obligations or restrictions binding on any Transferred Entity, including (A) "most favored nation" pricing terms, (B) granting of any right or first offer or right of first refusal, (C) covenants not to solicit any employees, contractors or agents of another Person, (D) covenants not to acquire any product or other asset or any services from any other Person, or (E) covenants not to transact business with any other Person;
- (xiii) Contracts containing non-disclosure, confidentiality, non-solicitation, non-competition or other restrictive covenant obligations with current and to the extent obligations exist thereunder, former Business Employees or other employees or individual service providers of the Transferred Entities;
- (xiv) Contracts relating to the settlement of any Proceedings during the last three years from the date of this Agreement or under which any of the parties thereto still have outstanding rights or obligations, in each case involving the Business or any Transferred Entity;
- (xv) any Contract not otherwise listed above which requires payments from the Business to a third party in excess of \$1,000,000 per annum and which is not terminable by Purchaser or a Transferred Entity on notice of ninety (90) calendar days or less without a premium or penalty;
  - (xvi) any Collective Bargaining Agreements with a Transferred Entity or involving any Business Employee;
- (xvii) (A) Contracts with any Business Employee or any employee or other individual service provider or consultant of a Transferred Entity or (B) Contracts with any third party professional employer organization related to the Business Employees or employees or other individual service providers or consultants of a Transferred Entity, in each case other than any "at will" offer letter that may be terminated without penalty or Liability upon thirty (30) calendar days' or less advance notice;
- (xviii) Contracts that contain severance, retention, accelerated payments, change in control, deferred compensation or other similar payments, including those where vesting, benefit enhancement, payment, funding or limitations would be triggered by entry into this Agreement, any of the other Transaction Documents, or the consummation of the Transaction;
- (xix) any Contract not otherwise listed above which provides for payments from a third party to the Business in excess of \$1,000,000 per annum; and
- (xx) any outstanding written or otherwise binding commitment to enter into any Contract of the type described in the immediately preceding subsections (i)-(xix).

- (b) Other than expirations or non-renewals following the date of this Agreement in accordance with the terms of the applicable Material Contract, each Material Contract is in full force and effect and is valid, binding and enforceable against Seller or a Transferred Entity and, to the Knowledge of Seller, against the other parties signatory thereto in accordance with its terms, in each case, subject to the effect of any Laws relating to bankruptcy, reorganization, insolvency, moratorium, fraudulent conveyance or preferential transfers, or similar Laws relating to or affecting creditors' rights generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a Proceeding in equity or at law). As of the date hereof, none of Seller, any Transferred Entity or, to the Knowledge of Seller, any other party to a Material Contract is in breach or violation of, or default under, any Material Contract and no event has occurred that with or without notice or lapse of time or both would constitute a breach or default (whether by lapse of time or notice or both). Prior to the execution of this Agreement, Seller has made available to Purchaser true and correct copies of each Material Contract (including all amendments and modifications thereto). Except as set forth on Section 3.11(b) of the Seller Disclosure Schedules, there are no outstanding or pending claims, requests for equitable adjustment or contract disputes in excess of \$1,000,000 arising under or relating to a Material Contract.
- Section 3.12 <u>Compliance with Applicable Laws; Permits.</u> (a) Neither Seller nor any of the Transferred Entities nor any of Seller's Affiliates is, or has been at any time since April 1, 2018, in violation in any material respect of any Law applicable to the Business.
- (b) The Seller and Energetics validly hold all material Permits necessary for the conduct of the Business as currently conducted or for the ownership and use of the assets of the Business (the "Business Permits"). Seller and Energetics are in compliance in all material respects with the terms of the Business Permits. The Seller has not received written notice of any Proceeding related to the revocation, withdrawal, modification or termination of any such Business Permit.
- (c) Export Control Laws. The Seller and Energetics are and for the last five (5) years have been in compliance in all material respects with all Export Control Laws;
- (i) The Seller and Energetics have, for the last five (5) years, obtained and complied in all material respects with the requirements of all applicable Export Control Authorizations;
- (ii) Neither Seller nor Energetics has during the last five (5) years engaged in any transactions or other activities directly or indirectly with or involving: (A) any Person located or ordinarily resident in Cuba, Iran, North Korea, Sudan, Syria, Crimea, Russia or the so-called Donetsk Peoples' Republic or Luhansk Peoples' Republic, (B) the Government of Venezuela or any Entity owned or controlled by the Government of Venezuela, or (C) any other Restricted Party; and
- (iii) Neither Seller nor Energetics has during the last five (5) years: (A) been cited or fined for failure to comply with applicable Export Control Laws, and no proceeding, investigation, complaint or inquiry, with respect to any alleged non-compliance with applicable Export Control Laws by Seller or Energetics, is pending or threatened; or (B) made any disclosures (voluntary or directed) to any relevant Governmental Entity with respect to any

potential violation or liability of Seller or Energetics arising under or relating to any Export Control Laws.

- (d) Anti-Corruption and Anti-Bribery. Neither Seller nor any of the Transferred Entities nor any of their respect directors, officers, employees, nor any other Person acting for or on behalf of either Seller or any of the Transferred Entities has during the last five (5) years, directly or indirectly, in connection with any activity of Seller or the Transferred Entities:
- (i) solicited, made, offered or agreed or promised to make or offer any payment, loan or transfer of anything of value, including any reward, advantage or benefit of any kind, to or for the benefit of any Government Official, candidate for public office, political party or political campaign, or any official of such party or campaign, for the purpose of (A) influencing any act or decision of such Government Official, candidate, party or campaign or any official of such party or campaign, (B) inducing such Government Official, candidate, party or campaign, or any official of such party or omit to do any act, (C) obtaining or retaining business for or with any Person, (D) expediting or securing the performance of official acts of a routine nature or (E) otherwise securing any improper advantage;
- (ii) solicited, paid, offered or agreed or promised to pay or offer any bribe, payoff, influence payment, kickback, unlawful rebate or other unlawful payment of any nature;
- (iii) solicited, made, offered or agreed or promised to make or offer any unlawful contributions, gifts, entertainment or other unlawful expenditures;
  - (iv) established or maintained any unlawful fund of corporate monies or other properties;
  - (v) created or caused the creation of any false or inaccurate books and records related to any of the foregoing;
- (vi) established or maintained accounts which do not appear in any of the books and records that they are required to keep in accordance with applicable accounting and auditing standards; or
- (vii) otherwise violated (A) any provision of the Foreign Corrupt Practices Act of 1977, 15 U.S.C. §§ 78dd-1, et seq., or (B) any other applicable Law of any Governmental Entity or any jurisdiction relating to anti-corruption and/or anti-bribery matters.

## Section 3.13 Government Contracts.

(a) <u>Section 3.13</u> of the Seller Disclosure Schedule sets forth a true and complete list as of the date hereof of all of the Current Government Contracts that involve aggregate payments to the Transferred Entities per annum that are reasonably expected to be in excess of one million Dollars (\$1,000,000) (each, a "<u>Material Government Contract</u>"). To the Knowledge of Seller, each Material Government Contract was legally awarded, and is valid, binding, and in full force and effect and enforceable against the Business in accordance with its terms, except as such enforceability may be limited by (i) bankruptcy, insolvency,

reorganization, moratorium or other similar Laws affecting or relating to creditors' rights generally and (ii) the availability of injunctive relief and other equitable remedies.

- (b) (i) The Business is not in material breach of or default under any Current Government Contract, and, to the Knowledge of Seller, no event has occurred which, with the giving of notice or the lapse of time or both, would constitute such a material breach or default by the Business; (ii) the Business is in compliance in all material respects with all applicable Laws, including the Federal Acquisition Regulation ("FAR"); Defense FAR Supplement ("DFARS"); Cost Accounting Standards; Service Contract Act of 1963, as amended (including requirements for paying applicable Service Contract Act wage rate and fringe benefit rates); the Truth in Negotiations Act; and the Anti-Kickback Act, where and as applicable to each Current Government Contract or Government Bid; (iii) since April 1, 2018, each representation and certification made by the Business in connection with a Current Government Contract or Government Bid was current, accurate and complete in all material respects as of its effective date; (iv) there are no outstanding or pending claims, requests for equitable adjustment or contract disputes in excess of one million Dollars (\$1,000,000) arising under or relating to a Current Government Contract or Government Bid; and (v) no Current Government Contract or Government Bid is currently the subject of any bid protest before any Governmental Entity.
- (c) Except as set forth Schedule 3.13, since April 1, 2018, (i) none of the Transferred Entities or any Principals (as defined in FAR 52.209-5) have been debarred, suspended or excluded from participation in, or the award of, Government Contracts or doing business with any Governmental Entity, no suspension, debarment, or exclusion action has been commenced or, to the Knowledge of Seller, threatened against any of the Transferred Entities or any of their respective officers or employees, and, to the Knowledge of Seller, there exist no circumstances that require any Transferred Entity to answer any of the questions in FAR 52.209-5 in the affirmative; (ii) no Governmental Entity has notified the Business in writing of any breach or violation of any applicable Law or of any certification, representation, clause, provision or requirement of any such Current Government Contract; (iii) the Business has not received any notice of termination for default, cure notice or show cause notice pertaining to any Current Government Contract; (iv) the Business has not received any notice of an unresolved significant weakness or deficiency with respect to the cost accounting system of the Transferred Entities; (v) the Business has not received any notice of any audits or investigations by any Governmental Entity with respect to a Current Government Contract or Government Bid (other than in the Ordinary Course of Business, including routine audits by the DCAA and United States Office of Federal Contract Compliance Programs); (vi) the Business has not received written notice from any Governmental Entity or other counterparty to a Material Government Contract that the counterparty to such Material Government Contract (A) has ceased or will cease to be a customer of the Business, (B) intends to terminate or materially modify (including by materially decreasing the rate or amount of services obtained from or provided to the Business) any Material Government Contract, (C) intends to change the type of contracting vehicle for the services provided pursuant to such Material Government Contract in a manner that may preclude the Business from continuing to provide such services or (D) seeks to convert any Material Government Contract that establishes an exclusive or single source purchasing arrangement or relationship between such counterparty and the Business into a non-exclusive or multi-source arrangement or relationship; and (vii) the Business has not made any disclosures to any Governmental Entity with respect to any irregularity, misstatement, significant overpayment or violation of applicable Law arising under or relating to any Current Government Contract or Government Bid, nor, to the Knowledge of Seller, has any violation occurred for which the Business is required to make any such disclosure to a Governmental Entity.

- (d) Since April 1, 2018, the Business has complied in all material respects with the requirements of Government Contracts issued under the General Services Administration Federal Supply Schedule program, including applicable requirements related to the Price Reductions Clause (48 CFR 552.238-81) and the Industrial Funding Fee and Sale Reporting clause (48 CFR 552.238-80).
- (e) Except as set forth in Section 3.13(e) of the Seller Disclosure Schedules, since April 1, 2018, (i) the Business has complied in all material respects with applicable Cost Accounting Standards and Cost Principles; (ii) the cost accounting systems and business systems (as defined in DFARS 242.7001 & 252.242-7005) used by the Transferred Entities and the associated entries reflected in the financial and business records of the Transferred Entities with respect to Government Contracts and Government Bids are (and have been) in compliance in all material respects with applicable Law; (iii) the Transferred Entities' business systems have been approved, where applicable, by the Defense Contract Management Agency as adequate for accumulating and billing costs under and otherwise for complying with Government Contracts, to the extent evaluated; and (iv) to the Knowledge of Seller, such cost accounting systems are adequate to meet the standards promulgated by the Cost Accounting Standards Board required for complying with the terms and conditions of the Current Government Contracts and applicable Law.
- (f) None of the Intellectual Property owned by the Transferred Entities was (i) with respect to patents, conceived or first actually reduced to practice in performance of a Government Contract or (ii) with respect to "technical data" and "computer software," as those terms are defined in FAR Parts 27 and 52 and/or DFARS Parts 227 and 252, funded partially or exclusively at a Governmental Entity's expense, and any Governmental Entity of the United States has only "Limited" and "Restricted" rights in such "technical data" and "computer software," respectively. The Business has complied with all data rights marking requirements under FAR Parts 27 and 52 and/or DFARS Parts 227 and 252.
- (g) Except as set forth on <u>Schedule 3.13(g)</u>, (i) none of the Current Government Contracts restrict any Transferred Entity's ability to bid on or perform work on future Government Contracts due to an actual or potential organizational conflict of interest (the "<u>OCI</u>") and (ii) the Business is not operating under or subject to any current OCI mitigation plan.
- (h) <u>Schedule 3.13(h)</u> sets forth, for each Government Contract having backlog as of December 31, 2022, the dollar amounts of Funded Backlog and the name of the customer. All of the Government Contracts constituting Funded Backlog and Unfunded Backlog of the Transferred Entities (i) were entered into in the Ordinary Course of Business and (ii) management of the Transferred Entities believes in good faith that such Government Contracts are capable of performance in accordance with the terms and conditions of such Government Contract by the applicable Transferred Entity.
- (i) The Business has complied in all material respects with the prohibitions on the sale and use of covered telecommunications equipment and services as required by its Government Contracts, including FAR clause 52.204-24, FAR clause 52.204-25, and FAR clause 52.204-26.
- (j) Since April 1, 2018, (i) the Business has complied with and is in compliance in all material respects with all applicable information system security requirements

regarding the safeguarding of information related to Government Contracts, including FAR clause 52.204-21, DFARS clause 252.204-7008, DFARS clause 252.204-7012, DFARS clause 252.204-7019, and DFARS clause 252.204-7020; and (ii) the Business has not experienced any cyber incident that would require reporting to the U.S. Department of Defense under DFARS clause 252.204-7012.

## Section 3.14 Environmental Matters.

- (a) Except as set forth in Section 3.14(a) of the Seller Disclosure Schedule, each of Seller and its Affiliates (with respect to the Business) (i) are, and since April 1, 2018, have been, in compliance in all material respects with all applicable Environmental Laws, (ii) have no material Liability under Environmental Laws or with respect to Hazardous Materials, and, (iii) except in the usual course of business in connection with obtaining or providing services, have not agreed to indemnify or assume the liability of any third party pursuant to Environmental Law or with respect to Hazardous Materials.
- (b) (i) Except as set forth in Section 3.14(b)(i) of the Seller Disclosure Schedule, each of Seller and its Affiliates (with respect to the Business) have timely applied for, obtained, maintained, and been since April 1, 2018 in material compliance with all Environmental Permits, and (ii) Section 3.14(b)(ii) of the Seller Disclosure Schedule sets forth a true and complete list of all material Environmental Permits required for the operation of the Business, all of which are valid and in full force and effect.
- (c) Except as set forth in Section 3.14(c) of the Seller Disclosure Schedule, with respect to the Business, (i) as of the date of this Agreement, there are no Proceedings pending, nor, to the Knowledge of Seller, threatened in writing against Seller or any of its Affiliates, and (ii) since April 1, 2019, neither Seller nor any Affiliate has received or been threatened in writing with any notice or action, in either case of (i) or (ii) alleging a material violation of or material Liability under Environmental Laws or with respect to Hazardous Materials or threatening to revoke, suspend or materially modify any material Environmental Permit.
- (d) Except as set forth in Section 3.14(d) of the Seller Disclosure Schedule or as would not reasonably be expected to result in material Liability to Seller, there has been no Release of Hazardous Materials by Seller or any of its Affiliates or by any other person at, on, under or from (i) any real property currently owned, operated or leased by Seller or any of its Affiliates in connection with the Business, or (ii) any real property formerly owned, operated or leased by Seller or any of its Affiliates in connection with the Business during Seller's or any of its Affiliates ownership, operation or lease of such real property, or (iii) any other real property to which Seller or any or any of its Affiliates has arranged, by contract, agreement or otherwise, for the transportation, treatment or disposal of Hazardous Materials in connection with the Business.
- (e) Except as set forth in <u>Section 3.14(e)</u> of the Seller Disclosure Schedule none of the following are present at any Transferred Owned Property, or to the Knowledge of Seller, any of the Transferred Leased Property: (i) "underground storage tanks" as defined under Environmental Law, or (ii) any dump or landfill.
- (f) Seller has made available to Purchaser true and complete copies of all material reports, assessments, audits, Environmental Permits and all other material documents in

Seller's or any Affiliate's possession or reasonable control related to Seller's or any Affiliate's (with respect to the Business) compliance with Environmental Laws or the environmental condition of any real property currently or formerly owned, operated, leased or used by Seller or any Affiliate in connection with the Business.

- Section 3.15 <u>Proceedings.</u> Except as set forth on <u>Section 3.15</u> of the Seller Disclosure Schedules, since April 1, 2018, there are no and there have not been any Proceedings pending or, to the Knowledge of Seller, threatened in writing (or, to the Knowledge of Seller, orally), against the Seller or any of its Affiliates with respect to the Business. Since April 1, 2018, there is no, and has not been, any Judgment outstanding against the Seller or any of its Affiliates involving or relating to the Business.
- Section 3.16 <u>Taxes</u>. For purposes of this <u>Section 3.16</u>, all references to a Transferred Entity shall be deemed to include the Seller solely with respect to the Business.
- (a) All income and other material Tax Returns required to be filed by or with respect to the Transferred Entities have been timely filed and all Taxes required to be paid by or with respect to the Transferred Entities have been timely paid, regardless of whether shown on such Tax Returns as due and payable. All such Tax Returns are true, correct and complete in all material respects. There are no Liens for any Taxes on any of the properties or assets of the Transferred Entities other than Permitted Liens. Each Transferred Entity has established on its books and records, including the Business Financial Statements, in accordance with GAAP (and consistently applied in accordance with such Transferred Entity's historical practices insofar as such practices are consistent with GAAP), reserves that are adequate for the payment of any Taxes not yet due and payable.
- (b) Each Transferred Entity has: (i) complied in all material respects with all applicable Laws relating to the payment, reporting and withholding (including any amount not withheld because of exemption or similar circumstance) of material Taxes; (ii) within the time and in the manner prescribed by applicable Law, paid over to the proper Governmental Entity (or is properly holding for such timely payment) all amounts required to be so withheld and paid over in connection with any amounts paid or owing to Seller or any employee, independent contractor, creditor, member, or other third party; and (iii) properly collected and remitted sales, value added, and similar Taxes with respect to sales made to, or purchases made by, its customers or users.
- (c) No Transferred Entity has received any written notice of assessment or proposed assessment, deficiency, claim, adjustment or proposed adjustment in connection with any amount of Taxes, which notice has not been resolved, and to the Knowledge of Seller there are no pending audits, examinations, or administrative or judicial proceedings regarding any amount of Taxes of any Transferred Entity.
- (d) No Transferred Entity has waivers or extensions currently in effect with respect to any statute of limitations applicable to any claim for, or the period for the collection or assessment or reassessment of, any amount of Taxes and no request for any such waiver or extension is currently pending.
- (e) No Transferred Entity has requested or received a ruling from any Taxing Authority or signed any binding agreement with any Taxing Authority that might affect the amount of Tax due from any Transferred Entity after the Closing Date.

- (f) No Transferred Entity is a party to any agreement or contract relating to the sharing, allocation or indemnification of Taxes (excluding any customary commercial agreement or contract arising in the Ordinary Course of Business and not primarily related to Taxes). No Transferred Entity has any liability for Taxes of any other Person by reason of being a member of an affiliated group, within the meaning of Section 1504(a) of the Code, or filing consolidated U.S. federal income Tax Returns under Treasury Regulations Section 1.1502-6 or any similar provision of state, local or foreign Law or as a member of any combined, unitary or other similar group for tax purposes, or as a transferee or successor, by Contract or otherwise.
- (g) No Transferred Entity that is classified as a corporation for U.S. federal income tax purposes has constituted a "distributing corporation" or "controlled corporation" in a distribution of stock intended to qualify for Tax-free treatment under Section 355 of the Code (i) in the two (2) years prior to the date of this Agreement or (ii) in a distribution which could otherwise constitute part of a "plan" or "series of related transactions" (within the meaning of Section 355(e) of the Code) in conjunction with the transactions contemplated by this Agreement.
- (h) No Transferred Entity has been a party to a reportable or listed transaction as defined under Treasury Regulation Section 1.6011-4(b)(2).
- (i) At all times since formation, the Company was disregarded as an entity separate from its owner for U.S. federal income tax purposes. From (i) the date of incorporation until the Pre-Closing Reorganization Transactions, Energetics was a corporation and (ii) the Pre-Closing Reorganization Transactions until the Closing Date, Energetics was disregarded as an entity separate from its owner, in each case for U.S. federal income tax purposes. At all times since formation, VS2 was a partnership for U.S. federal income tax purposes.
- (j) No Transferred Entity will be required to include any material item of income in, or exclude any material item of deduction or loss from, taxable income, or make any adjustment under Section 481(a) of the Code, for any taxable period (or portion thereof) ending after the Closing Date as a result of any: (i) change in or improper use of any method of accounting for any taxable period ending on or prior to the Closing Date, (ii) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or other Law) executed prior to the Closing Date, (iii) installment sale or open transaction disposition made on or prior to the Closing Date, (iv) prepaid amount received or deferred revenue accrued on or prior to the Closing Date, (v) election under Section 108(i) of the Code or (vi) the application of Section 965 of the Code.
- (k) No Transferred Entity is, or has ever been, subject to Tax in any country other than its respective country of incorporation or formation by virtue of having a permanent establishment or other place of business in that country. No claim has ever been made by a Taxing Authority in a jurisdiction where any Transferred Entity does not file Tax Returns that such Transferred Entity is or may be subject to taxation by that jurisdiction.
- (l) Except as set forth on <u>Section 3.16(m)</u> of the Seller Disclosure Schedule, no Transferred Entity has taken advantage of any relief or Tax deferral provisions or any carryback of net operating losses or similar tax items related to COVID-19 for Tax purposes, whether federal, state, local or foreign, including the CARES Act.

- (m) No Transferred Entity is a beneficiary of any Tax exemption, Tax holiday or other Tax incentive that would terminate or be subject to recapture or clawback by the transactions contemplated hereby.
- (n) No Transferred Entity has participated in, or cooperated with, an international boycott within the meaning of Section 999 of the Code.
  - (o) No election under Section 965(h) of the Code is in effect with respect to any Transferred Entity.
- (p) No Transferred Entity is or has ever been a passive foreign investment company within the meaning of Section 1297 of the Code (determined without regard to Section 1297(d) of the Code).
  - (q) None of the Transferred Entities is a United States real property holding company.
  - (r) None of the Transferred Entities has issued or granted any power of attorney with respect to Taxes that remains in effect.

# Section 3.17 <u>Labor Relations; Employees and Employee Benefit Plans.</u>

- (a) Each Collective Bargaining Agreement previously entered into by Seller or its Subsidiaries involving current or former Business Employees as of the date of this Agreement is listed on Section 3.17(a) of the Seller Disclosure Schedules. No Collective Bargaining Agreements (including those disclosed pursuant to Section 3.17(a)) are in effect with respect to the Transferred Entities or any Business Employee. No union or labor organization is currently certified or recognized to represent any Business Employees, no demand for recognition has ever been made by any union or labor organization with respect to the current Business Employees (and except as set forth on Section 3.17(a), no demand for recognition has ever been made by any union or labor organization with respect to former Business Employees), and there are no pending or threatened labor organizing activities, strikes, slowdowns, work stoppages, petitions for representation, pickets, concerted refusals to work, lockouts or walkouts by Business Employees or against the Transferred Entities, nor has there been any such event since April 1, 2018. There is no unfair labor practice, charge or complaint pending or unresolved or threatened before any court, arbitrator or other Governmental Entity against the Transferred Entities or with respect to any of the current or former Business Employees.
- (b) Each Seller Benefit Plan is listed on Section 3.17(b)(i) of the Seller Disclosure Schedules, Seller has made available to Purchaser correct and complete copies of written summaries of each such Seller Benefit Plan (provided that agreements with individual Business Employees may be provided on an anonymized basis and templates may be provided in lieu of copies of individual agreements that conform to such templates). Each Transferred Entity Benefit Plan is listed on Section 3.17(b)(ii) of the Seller Disclosure Schedules, and Seller has made available to Purchaser correct and complete copies of (i) each such Transferred Entity Benefit Plan and in the case of any unwritten Transferred Entity Benefit Plan, a written descriptions thereof, in each case including any amendments thereto, (ii) the three most recent actuarial valuations, if any, annual reports required to be filed, and such similar reports, statements, information returns or material correspondence required to be filed with or delivered to any Governmental Entity, if any, with respect to each Transferred Entity Benefit Plan

(including reports filed on Form 5500 with accompanying schedules and attachments), (iii) the most recent summary plan description (if any), and any summary of material modifications, prepared for each Transferred Entity Benefit Plan, (iv) each trust agreement and group annuity or insurance Contract and other documents relating to the funding or payment of benefits under any Transferred Entity Benefit Plan and (v) the most recent determination, qualification or opinion letter or similar document issued by any Governmental Entity for each Transferred Entity Benefit Plan intended to qualify for favorable tax treatment and any pending application therefor and a complete and accurate list of all amendments to any such Transferred Entity Benefit Plan as to which a favorable determination or qualification letter has not yet been received. All participant data necessary to administer each Transferred Entity Benefit Plan will be in the possession of the Transferred Entities prior to Closing.

- (c) Each Seller Benefit Plan has been administered in material compliance with its terms and the administration thereof are and is in material compliance with all applicable provisions of ERISA, the Code, the Patient Protection and Affordable Care Act of 2010, as amended, and all other applicable Laws, whether Federal, state or local, domestic or foreign, and the terms of all collective bargaining agreements. Each Seller Benefit Plan required to have been approved by any non-U.S. Governmental Entity (or permitted to have been approved to obtain any beneficial tax or other status) has been so approved or timely submitted for approval, no such approval has been revoked (nor, to the Knowledge of the Seller, has revocation been threatened) and no event has occurred since the date of the most recent approval or application therefor that is reasonably likely to affect any such approval.
- (d) Each Transferred Entity Benefit Plan has been administered in material compliance with its terms and the administration thereof are and at all times have been in compliance with all applicable provisions of ERISA, the Code, the Patient Protection and Affordable Care Act of 2010, as amended, and all other applicable Laws, whether Federal, state or local, domestic or foreign, and the terms of all collective bargaining agreements. Each amendment to a Transferred Entity Benefit Plan has been made in material compliance with the terms of such Transferred Entity Benefit Plan, and with all applicable Laws. No contractual commitments, undertakings or representations have been made or given to any participant regarding the continued operation, extension, amendment or replacement of any Transferred Entity Benefit Plan or of the benefits available under any Transferred Entity Benefit Plan. The Seller has not filed, and is not considering filing, an application under the IRS Employee Plans Compliance Resolution System or the Department of Labor's Voluntary Fiduciary Correction Program with respect to any Transferred Entity Benefit Plan. Each Transferred Entity Benefit Plan required to have been approved by any non-U.S. Governmental Entity (or permitted to have been approved to obtain any beneficial tax or other status) has been so approved or timely submitted for approval, no such approval has been revoked (nor, to the Knowledge of the Seller, has revocation been threatened) and no event has occurred since the date of the most recent approval or application therefor that is reasonably likely to affect any such approval or increase the costs relating thereto.
- (e) All contributions, premiums and benefit payments, fees and expenses under or in connection with each Transferred Entity Benefit Plan that are required to have been made by the Seller or the Transferred Entity Benefit Plan, in accordance with the terms of such Transferred Entity Benefit Plan and applicable Laws have been timely made, in all material respects.

- (f) No Transferred Entity has sponsored, maintained, contributed to or been obligated to sponsor, maintain or contribute to, or has any actual or contingent liability under, any Benefit Plan that is a "defined benefit plan" (as defined in Section 3(35) of ERISA) or a "multiemployer plan" (within the meaning of Section 4001(a)(3) of ERISA), that is subject to Section 302 or Title IV of ERISA or Section 412 of the Code or that is a "multiple employer welfare arrangement" (within the meaning of Section 3(40) of ERISA). No event has occurred and, to the Knowledge of the Seller, no condition exists, that has subjected, or would reasonably be expected to subject, any Transferred Entity to any tax, fine, Lien or penalty imposed by ERISA.
- (g) Except as otherwise set forth in <u>Section 3.17(g)</u> of the Seller Disclosure Schedules, no current or former employee or other service provider of Seller or any of its Subsidiaries is a party to any benefit under a Benefit Plan, or employment or other Contract with Seller or any Subsidiary thereof that entitles him or her to compensation or other consideration (or the acceleration of any such compensation or consideration) based on, related to or contingent upon the consummation of the transactions contemplated by this Agreement.
- (h) Except as set forth on Section 3.17(h) of the Seller Disclosure Schedules, no amount, economic benefit or other entitlement that could be received (whether in cash or property or the vesting of property) as a result of any of the transactions contemplated by this Agreement (alone or in combination with any other event) by any Person who is a "disqualified individual" (as defined in Treasury Regulation Section 1.280G-1) with respect to the Seller (each, a "Disqualified Individual") would be characterized as an "excess parachute payment" (as defined in Section 280G(b)(1) of the Code), and no such Disqualified Individual is entitled to receive any additional payment from the Seller or its Subsidiaries in the event that the excise tax required by Section 4999(a) of the Code is imposed on such Disqualified Individual.
- (i) Seller has made available to Purchaser a complete and accurate list of each Business Employee as of a date that is no more than five (5) Business Days prior to the date of this Agreement and including, with respect to each such Business Employee, as applicable (i) their dates of hire; (ii) their position(s); (iii) their current annual base salaries or hourly wages; (iv) their 2022 bonus compensation; (v) their work location; (vi) their status as a full-time or part-time employee; (vii) their classification as exempt or non-exempt under the Fair Labor Standards Act ("FLSA"); (viii) their status as a fixed term (i.e. temporary) or permanent employee; (ix) their status as a regular or leased employee; (x) their status as an active or inactive employee, as well as the date of commencement of their leave and their expected return date (as applicable of each employee); (xi) the value of their accrued vacation time and sick leave or other paid time off; (xii) the name of their employing entity; and (xiii) whether the Business Employee's employment is governed by an employment agreement or an offer letter. Except as set forth on Schedule 3.17(i), all Business Employees are employed on an at-will basis.
- (j) Seller has made available to Purchaser a complete and correct list of all individual independent contractors engaged directly by Seller or its Subsidiaries with respect to the Business who have received more than Fifty Thousand Dollars (\$50,000) in remuneration in the past twelve (12) months or are expected to receive more than Fifty Thousand Dollars (\$50,000) in remuneration in the next twelve (12) months, who are actively providing services or provided any services in the past twelve (12) months that reflects (i) their dates of engagement; (ii) their rate of remuneration; (iii) the amount they were paid in calendar year 2022; (iv) a description of the services they provide; and (v) the name of the entity by which they are engaged. Each such independent contractor is party to an independent contractor agreement, and

all such agreements have been made available to Purchaser. To the Knowledge of Seller, for the purpose of applicable Law, all independent contractors who are currently, or since April 1, 2018, have been, engaged directly by Seller or its Subsidiaries with respect to the Business are bona fide independent contractors under applicable Law and not employees of the Seller or any of its Subsidiaries. Each current independent contractor directly engaged by Seller or its Subsidiaries with respect to the Business is terminable on thirty (30) or fewer days' notice, without any obligation to pay severance or a termination fee or any remuneration other than for services already provided, or services which will be provided prior to the contract's termination date, by such independent contractor.

- (k) Since March 1, 2020, there have been no material Proceedings or to the Knowledge of Seller, threatened Proceedings involving Business Employees before any Governmental Entity against the Seller or its Subsidiaries.
- (1) The Transferred Entities are, and as it relates to the Business Employees the Seller and its Subsidiaries are, in compliance in all material respects with the rules and regulations governing the conduct of federal contractors, with respect to employees and all applicable Laws pertaining to employment, employment practices and the employment of labor, including all such Laws relating to labor relations, equal employment opportunities, fair employment practices, prohibited discrimination, wages, hours, meal and rest breaks, employee leave (including paid sick leave), overtime, worker classification, privacy, immigration, unemployment insurance, affirmative action, occupational safety and health, workers' compensation, plant closings (including hiring and termination of employees). The Seller has with respect to the Business Employees and the Transferred Entities timely paid all salaries, wages and fees, which are payable to any employees or independent contractors. No Transferred Entity is a party to a conciliation agreement, consent decree or other Contract or order with any Governmental Entity with respect to employment practices, nor are Seller or any of its Subsidiaries party to such an agreement, consent decree, or other Contract or order with respect to any Business Employee.
- (m) To the Knowledge of Seller, the Business Employees, and all employees or independent contractors of the Transferred Entities, are legally authorized to work in the locations where they perform work for the Company, and have all work permits, immigration permits, visas or other authorizations, each as required by applicable Law for such employee or independent contractor given the duties, location and nature of such individual's employment or engagement, except where the continuing failure to have all such work permits, immigration permits, visas or other authorizations would not be material. The entity employing or engaging such individual has completed and retained employment verification paperwork required under appliable Law, including the Immigration Reform and Control Act of 1986, as required by Law.

## (n) Except as set forth in Schedule 3.17(n):

- (i) Since March 1, 2020, neither Seller nor any of its Subsidiaries has received a written notice, citation, complaint or charge asserting any violation or Liability under the federal Occupational Safety and Health Act of 1970, or any similar applicable Law regulating employee health and safety Relating to the Business or involving Business Employees.
- (ii) Since April 1, 2018, neither Seller nor any of its Subsidiaries, with respect to the Business Employees, nor the Transferred Entities, effectuated any "plant closing" or "mass layoff" of employees as those terms are defined by WARN;

- (iii) To the Knowledge of Seller, no Key Employee, and no Business Employee with base annual compensation at or above \$150,000, intends to terminate their employment prior to the one (1) year anniversary of the Closing Date (other than as a result of a loss of a government contract or subcontract termination);
- (iv) To the Knowledge of Seller, no Business Employee, or independent contractor Relating to the Business is in violation of any nondisclosure agreement, noncompetition agreement, or restrictive covenant obligation owed to any third party with respect to such person's right to be employed or engaged by the Company or with respect to the Business.
- (o) Since April 1, 2018, no allegation, complaint, charge or claim (formal or informal) of sexual harassment, sexual assault, sexual misconduct, or similar behavior (a "Sexual Misconduct Allegation") has been made against any Person who is or was an officer, director, manager or supervisory-level Business Employee or other officer, director, manager or supervisory-level employee of one or more of the Transferred Entities in such Person's capacity as such. Since April 1, 2018, the Transferred Entities have not, and the Seller and its Subsidiaries have not with respect to any Business Employee, entered into any settlement agreement, tolling agreement, non-disparagement agreement, confidentiality agreement or non-disclosure agreement, or any Contract or provision similar to any of the foregoing, relating directly or indirectly to any Sexual Misconduct Allegation against the Seller or its Subsidiaries (including or more of the Transferred Entities), or any Person who is or was an officer, director, manager or supervisory-level employee of the Seller or its Subsidiaries (including or more of the Transferred Entities).
- Section 3.18 Intercompany Agreements. Other than (a) the Transaction Documents and the Contracts expressly contemplated thereby and (b) any Contracts that will be terminated at or prior to the Closing or will be terminated pursuant to this Agreement, Section 3.18 of the Seller Disclosure Schedules lists all Intercompany Agreements. Except as set forth on Section 3.18 of the Seller Disclosure Schedules, neither Seller nor any of its Affiliates nor any of their officers, directors, members, stockholders (each, a "Seller Related Party") is currently a party to any transaction or Contract with the Transferred Entities or has any material interest in any property used by the Transferred Entities, other than employment or consulting agreements entered into with individuals in the ordinary course of business or the Transaction Documents.
- Section 3.19 <u>Brokers</u>. Other than Jefferies Group LLC, no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the Transaction and the other transactions contemplated by this Agreement based upon arrangements made by or on behalf of Seller and its Subsidiaries.

# Section 3.20 Key Customers and Suppliers.

(a) <u>Section 3.20(a)</u> of the Seller Disclosure Schedules sets forth a true and complete list of the ten (10) most significant customers of the Business (measured by dollar volume) for each of the twelve (12)-month periods ended December 31, 2021 and December 31, 2022 (the "<u>Key Customers</u>"). None of the Key Customers have by providing written (or, to the Knowledge of Seller, oral) notice (i) cancelled, terminated or materially and adversely modified (including by materially decreasing the rate or amount of services obtained from the Business) its relationship with the Business or (ii) sought to convert any exclusive or single-source purchasing

arrangement or relationship between such Key Customer and the Business into a non-exclusive or multi-source arrangement or relationship. None of Seller or its Affiliates have received any written notice that any Key Customer has ceased or will cease to be a customer of the Business or that such Key Customer intends to terminate or materially modify existing Contracts with the Business.

(b) <u>Section 3.20(b)</u> of the Seller Disclosure Schedules sets forth a true and complete list of the ten (10) most significant suppliers (including subcontractors or vendors) of the Business (measured by dollar volume) for each of the twelve (12)-month periods ended December 31, 2021 and December 31, 2022 (the "<u>Key Suppliers</u>"). None of the Key Suppliers have by providing written (or, to the Knowledge of Seller, oral) notice (i) cancelled, terminated or materially and adversely modified its relationship with the Business or (ii) sought to convert any exclusive or single-source purchasing arrangement or relationship between such Key Supplier and the Business into a non-exclusive or multi-source arrangement or relationship. None of the Seller or its Affiliates have received any written notice that any Key Supplier has ceased or will cease to be a supplier of the Business or that such Key Supplier intends to terminate or materially modify existing Contracts with the Business.

# Section 3.21 Bank Accounts; Powers of Attorney.

- (a) Section 3.21(a) of the Seller Disclosure Schedules sets forth a true and complete list of all banks, trust companies, savings and loan associations, mutual fund or stock brokerage firms and other financial institutions at which any Transferred Entity maintains accounts or safe deposit boxes of any nature, the account numbers of all such accounts and the names of all Persons authorized to draw thereon or make withdrawals therefrom.
- (b) Except as set forth on Section 3.21(b) of the Seller Disclosure Schedules, no Transferred Entity has any obligation to act under, or has granted, any outstanding power of attorney or Liability, as guarantor, surety, consignor, endorser (other than for purposes of collection in the ordinary course of business), co-maker or indemnitor in respect of the obligation of any Person.
- Section 3.22 <u>Insurance</u>. <u>Section 3.22</u> of the Seller Disclosure Schedule sets forth a true and complete list, as of the date hereof, of all material policies of insurance currently owned, held or maintained by, at the expense of or for the benefit of the Business (the "<u>Business Insurance Policies</u>"), and all premiums due and payable on the Business Insurance Policies have been paid in full and all premiums with respect thereto concerning all periods up to and including the Closing will have been paid. The Seller has, with respect to any claim that relates to any damage, impairment or loss that is potentially covered under the Business Insurance Policies, duly provided all required notices under the Business Insurance Policies and have duly made all available claims under such Business Insurance Policies covering or related to any such claim in accordance with the terms and conditions of the Business Insurance Policies. There are no pending claims against any Business Insurance Policy as to which the insurers have denied Liability. No owner of a Business Insurance Policy has received any written notice of cancellation, amendment or dispute or rejection as to coverage with respect to any Business Insurance Policy.

Section 3.23 Title; Sufficiency of Assets.

- (a) Except as set forth in <u>Section 3.23(a)</u> of the Seller Disclosure Schedules, as of the Closing, one of the Transferred Entities will have (i) good and valid title to and is the sole and exclusive owner of, all of the owned assets of the Business, including the Energetics Equity Interests and the VS2 Equity Interests, and (ii) valid leasehold interest to use all the leased and subleased assets of the Business (collectively, (i) and (ii) the "<u>Purchased Assets</u>"), in each case, free and clear of any Liens other than Permitted Liens. All of the material, tangible Purchased Assets (whether owned or leased) are in good working order, operating condition and state of repair (ordinary wear and tear excepted). As of immediately following the Closing, one of the Transferred Entities will own and have good and valid title to, or a valid leasehold interest in (as applicable), the Purchased Assets, free and clear of all Liens, except for Permitted Liens.
- (b) Except as set forth on Section 3.23(b) of the Seller Disclosure Schedules, as of the Closing Date, (i) the Purchased Assets and (ii) the assets, services, products, Intellectual Property and real property to be provided, acquired, leased or licensed under the Transaction Documents or Company Inbound IPR Agreements constitute all of the assets (other than Open Source Software, COTS Software, Benefit Plans and policies of insurance) sufficient and necessary to conduct the Business in substantially the same manner, in all material respects, as conducted by Seller during the twelve (12)-month period prior to the Closing.
- Section 3.24 <u>Security Clearances</u>. <u>Schedule 3.24</u> sets forth all facility security clearances held by the Business that the Seller is permitted by Law to disclose. The Business is in compliance in all material respects with applicable national security requirements, including the NISPOM and all applicable requirements under each Current Government Contract. To the Knowledge of Seller, no facts exist which are reasonably expected to give rise to the revocation, invalidation or suspension of any facility security clearance held by the Business or any personnel security clearance held by any employee of the Seller. Since April 1, 2020, the Business has not received a rating less than "Satisfactory" in connection with any DCSA review or audit, and, to the Knowledge of Seller, there has been no unauthorized disclosure of classified information by employees of the Seller.

# ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to Seller as follows:

- Section 4.1 <u>Organization, Standing and Power.</u> Purchaser is duly organized, validly existing and in good standing under the Laws of the jurisdiction in which it is organized and has all necessary organizational power and authority to carry on its business as presently conducted, except in any case as would not reasonably be expected to impair or materially delay the ability of Purchaser to (a) perform its obligations under this Agreement, including payment of the Purchase Price, or (b) consummate the Transaction, the Transaction Documents and the other transactions contemplated hereby any thereby (each of clause (a) and clause (b), a "<u>Purchaser Material Adverse Effect</u>").
- Section 4.2 <u>Authority; Execution and Delivery; Enforceability</u>. Purchaser has all necessary limited liability company power and authority to execute this Agreement, the Transaction Documents and to consummate the Transaction and the other transactions contemplated hereby and thereby. The execution and delivery by Purchaser of this Agreement, the Transaction Documents and the consummation by Purchaser of the Transaction and the other transactions contemplated hereby and thereby have been duly authorized by all necessary

company or other action of Purchaser. Purchaser has duly executed and delivered this Agreement, and assuming due authorization, execution and delivery by Seller and the Company, this Agreement and each of the Transaction Documents will constitute its valid and binding obligation, enforceable against it in accordance with its terms, subject to the effect of any Laws relating to bankruptcy, reorganization, insolvency, moratorium, fraudulent conveyance or preferential transfers, or similar Laws relating to or affecting creditors' rights generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a Proceeding in equity or at law).

Section 4.3 No Conflicts; Consents; Governmental Authorizations. The execution and delivery by Purchaser of this Agreement does not, and the execution by Purchaser of the other Transaction Documents will not, and the consummation of the Transaction and the other transactions contemplated hereby and thereby and compliance by Purchaser with the terms hereof and thereof will not, conflict with, or result in any violation of, breach or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation under, or result in the creation of any Lien upon any of the properties or assets of Purchaser (including, after the Closing, the Business) or any of its Subsidiaries (including, after the Closing, the Transferred Entities) under, any provision of (a) the certificate of incorporation, bylaws, articles of association, limited liability company agreement, certificate of formation or equivalent governing and/or constitutional documents of Purchaser, (b) any Judgment or Law applicable to Purchaser or its Subsidiaries, or the properties or assets of Purchaser or its Subsidiaries, or (c) any material Contract of Purchaser or any of its Subsidiaries, except, in the case of clauses (a), (b) and (c), for any such items that would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect. No Approval of any Governmental Entity is required to be obtained or made by or with respect to Purchaser or its Subsidiaries in connection with the execution, delivery and performance of this Agreement or the consummation of the Transaction and the other transactions contemplated hereby.

# Section 4.4 Financial Ability to Perform.

(a) Purchaser is a party to and has accepted a fully executed commitment letter dated May 1, 2023 (the "Equity Commitment Letter" from certain Persons (collectively, the "Equity Investors") pursuant to which the Equity Investors have agreed and committed, subject to the terms and conditions thereof, to invest in Purchaser the Aggregate Purchase Price. The Equity Commitment Letter provides that Seller is a third-party beneficiary thereof and is entitled to enforce each such agreement. The equity financing committed pursuant to the Equity Commitment Letter is referred to in this Agreement as the "Equity Financing." Purchaser has delivered to Seller a true, complete and correct copy of the Equity Commitment Letter. The Equity Commitment Letter constitutes the legal, valid, binding and enforceable obligations of Purchaser and, to the Knowledge of Purchaser, all other parties thereto and is in full force and effect. To the Knowledge of Purchaser, no event has occurred that, with or without notice, lapse of time, or both, could reasonably be expected to constitute a default, breach or a failure to satisfy a condition precedent on the part of Purchaser under the terms and conditions of the Equity Commitment Letter. The Equity Commitment Letter has not been modified, amended or altered. Assuming the Equity Financing is funded in accordance with the terms of the Equity Commitment Letter and assuming the satisfaction or, if permissible, waiver of the conditions set forth in Section 7.1 and Section 7.2 (other than those conditions to be satisfied at the Closing, but subject to the satisfaction or, if permissible, written waiver of such conditions), Purchaser will

have at the Closing sufficient available cash on hand necessary to consummate the transactions contemplated by this Agreement on the terms and subject to the conditions set forth herein.

- (b) In no event shall the receipt or availability of any funds or financing by Purchaser or any Affiliate be a condition to any of Purchaser's obligations under this Agreement.
- Section 4.5 <u>Proceedings</u>. There are no Proceedings or claims pending or, to the Knowledge of Purchaser, threatened against Purchaser or any of its Affiliates that would reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect.
- Section 4.6 <u>Brokers</u>. Other than William Blair, no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the Transaction and the other transactions contemplated by this Agreement based upon arrangements made by or on behalf of Purchaser.
- Section 4.7 <u>Investigation</u>. Purchaser has such knowledge and experience in financial and business matters, and is capable of evaluating the merits and risks of the Transaction and the other transactions contemplated by this Agreement and the Transaction Documents. Purchaser confirms that Seller has made available to Purchaser and its Affiliates and Representatives the opportunity to ask questions of the officers and management of Seller and the Business and to acquire additional information about the business and financial condition of the Business, and Purchaser confirms that it has made an investigation, analysis and evaluation of the Business.
- Section 4.8 Securities Act. Purchaser is acquiring the Company Equity Interests solely for the purpose of investment and not with a view to, or for sale in connection with, any distribution thereof in violation of the Securities Act. Purchaser acknowledges that the Company Equity Interests are not registered under the Securities Act, any applicable state securities Laws or any applicable foreign securities Laws, and that such Company Equity Interests may not be transferred or sold except pursuant to the registration provisions of the Securities Act and applicable state and foreign securities Laws or pursuant to an applicable exemption therefrom. Purchaser has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of its investment in the Company Equity Interests and is capable of bearing the economic risks of such investment.
- Section 4.9 <u>Solvency</u>. No transfer of property is being made and no obligation is being incurred in connection with the Transaction with the intent to hinder, delay or defraud either present or future creditors of Purchaser and its Subsidiaries (including, following the Closing, the Transferred Entities). Immediately after giving effect to the consummation of the Transaction, Purchaser and its Subsidiaries, taken as a whole, will be Solvent. For purposes of this <u>Section 4.9</u>, "<u>Solvent</u>" means, with respect to Purchaser and its Subsidiaries, taken as a whole, that:
- (i) the fair saleable value (determined on a going concern basis) of the assets of Purchaser and its Subsidiaries, taken as a whole on a consolidated basis, shall be greater than the total amount of the liabilities (including all liabilities, whether or not reflected in a balance sheet prepared in accordance with GAAP, and whether direct or indirect, fixed or contingent, secured or unsecured, disputed or undisputed) of Purchaser and its Subsidiaries, taken as a whole on a consolidated basis; and

- (ii) Purchaser and its Subsidiaries, taken as a whole on a consolidated basis, shall be able to pay their debts and obligations in the ordinary course of business as they become due.
- Section 4.10 <u>Acknowledgement of Pre-Closing Services</u>. Purchaser acknowledges that Seller and its Subsidiaries provide various services, rights and support to both the Business and the Retained Business (including with respect to the following matters: tax, legal, compliance and governmental affairs, information technology support, audit, accounting, treasury, insurance, business development, capital raising and intercompany financing and access to shared facilities), in each case, that will not continue after the Closing except to the extent expressly provided in this Agreement, or other Transaction Documents.
- Section 4.11 <u>Foreign Ownership, Control or Influence</u>. Purchaser is not (a) a "foreign person" within the meaning of 31 C.F.R. § 800.224 or (b) a "foreign interest" within the meaning of 32 C.F.R. § 117.3. Purchaser is a "U.S. Person" within the meaning of 22 C.F.R. § 120.16. To Purchaser's Knowledge, there are no facts pertaining to Purchaser or its Affiliates that would reasonably be expected to require a 60-day advance notice to the Directorate of Defense Trade Controls pursuant to 22 CFR 122.4(b) in connection with the Transactions.

# Section 4.12 Acknowledgement of No Other Representations or Warranties.

- (a) Purchaser acknowledges and agrees that, except for the representations and warranties contained in <u>Article III</u> and the other Transaction Documents, neither Seller, nor any of its Affiliates, Representatives or any other Person makes any express or implied representation or warranty with respect to Seller, the Transferred Entities or their respective Affiliates, the Business or with respect to any other information provided, or made available, to Purchaser or any of its Affiliates or Representatives in connection with the transactions contemplated hereby and thereby. Purchaser acknowledges and agrees that it is not relying on any representation or warranty of Seller or any of its Affiliates or Representatives or any other Person, other than those representations and warranties specifically and expressly set forth in <u>Article III</u> and the other Transaction Documents. Purchaser acknowledges and agrees that Seller and its respective Affiliates disclaims any and all representations and warranties, whether express or implied, except for the representations and warranties contained in <u>Article III</u> and the other Transaction Documents. Purchaser acknowledges and agrees that neither Seller nor any of its Affiliates makes any express or implied representation or warranty with respect to the Retained Businesses.
- (b) Purchaser acknowledges that neither Seller nor any of its respective Affiliates or Representatives has made any warranty, express or implied, as to the prospects of the Business or its profitability for Purchaser, or with respect to any forecasts, projections or business plans (including any Confidential Information (as defined in the Confidentiality Agreement)) delivered to Purchaser or any of its Affiliates or Representatives in connection with Purchaser's review of the Business and the negotiation and execution of this Agreement, including as to the accuracy or completeness thereof or the reasonableness of any assumptions underlying any such forecasts, projections or business plans.

ARTICLE V COVENANTS

Section 5.1 Approvals; Efforts.

- (a) Seller shall as reasonably as practicable cause the Company to apply for a facility security clearance from the DCSA that is, to the extent permitted by DCSA consistent with the facility security clearance held by Seller as of the date hereof (a "Facility Clearance").
- From and after the date hereof, each of the parties hereto shall use commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or advisable under any applicable Law to, assist the Seller and the Company in obtaining the Facility Clearance, obtain any required Approval to consummate and make effective in the most expeditious manner possible the Transaction and the other transactions contemplated by this Agreement and to cause the conditions to the Transaction set forth in Article VII to be satisfied as promptly as reasonably practicable. In furtherance and not in limitation of the foregoing, each of the parties hereto shall use commercially reasonable efforts to accomplish the following as promptly as reasonably practicable: (i) the obtaining of all actions or non-actions, including any consents, approvals, registrations, waivers, permits, authorizations, orders, expirations or terminations of waiting periods and other confirmations from any Governmental Entity or other person that are or may become necessary in connection with the consummation of the Transaction and the other transactions contemplated by this Agreement, (ii) the preparation and filing of all forms, registrations filings, petitions, statements, notices, submissions of information, applications and other documents (including filings with Governmental Entities) that are required to be filed to consummate the Transaction and the other transactions contemplated by this Agreement, (iii) the taking of all actions necessary, proper or advisable to obtain (and cooperating with each other in obtaining) any Approval of, or to avoid an action by, any third party, including any Governmental Entity (which actions shall include furnishing all information required in connection with any Regulatory Approvals) in connection with the Transaction, and the other transactions contemplated by this Agreement, (iv) the defending of any lawsuits or other Proceedings, whether judicial or administrative, challenging this Agreement or that would otherwise prevent or delay the consummation of Transaction, or the transactions contemplated by this Agreement, performed or consummated by each party in accordance with the terms of this Agreement, including seeking to have any stay, temporary restraining order or injunction entered by any court or other Governmental Entity vacated or reversed, and (v) the execution and delivery of any additional instruments that are or may reasonably become necessary, proper or advisable to consummate the Transaction and the other transactions contemplated by this Agreement and to fully carry out the purposes of this Agreement.
- (c) Prior to the Closing, subject to the Confidentiality Agreement and Section 5.3 of this Agreement, each party shall (i) promptly consult with the other party to this Agreement to provide any necessary information with respect to (and, in the case of correspondence, provide the other party (or their counsel) copies of) any governmental filings, submissions or other documents made by such party with any Governmental Entity in connection with this Agreement, the Transaction and the other transactions contemplated by this Agreement, (ii) promptly inform the other party, and if in writing, furnish the other party with copies of (or, in the case of oral communications, advise the other party orally of) any communication with or from any Governmental Entity or any official, representative or staff thereof regarding the Transaction and the other transactions contemplated by this Agreement, and permit the other party to review and discuss in advance, and consider in good faith the views, and secure the participation, of the other party in connection with, any such governmental filing, submission or other document or communication with any such Governmental Entity and (iii) cooperate in responding as promptly as reasonably practicable to any investigation or other inquiry from a Governmental Entity in connection with this Agreement, the Transaction and the other

transactions contemplated by this Agreement. If either party or any Representative of such party receives a request for additional information or documentary material from any Governmental Entity with respect to the Transaction or the other transactions contemplated by this Agreement, then such party will use commercially reasonable efforts to make, or cause to be made, promptly and after consultation with the other party, an appropriate response in compliance with such request. For the avoidance of doubt, the Seller's obligations under this Section 5.1(c) shall not apply to the Retained Business.

- (d) Whether or not the Transaction is consummated, each party agrees to be responsible for its own fees and payments to any third party or any Governmental Entity in order to obtain any Approvals pursuant to this Section <u>5.1</u>, including the fees of and payments to such party's legal and professional advisors.
- (e) Any other provision in this Section 5.1 or otherwise in this Agreement notwithstanding, neither Seller nor any of its Affiliates shall under any circumstance (i) be required to pay or commit to pay any amount or incur any obligation in favor of or offer or grant any accommodation (financial or otherwise, regardless of any provision to the contrary in the underlying Contract, including any requirements for the securing or posting of any bonds, letters of credit or similar instruments, or the furnishing of any guarantees) to any Person to obtain any Approval or (ii) have any Liability whatsoever to Purchaser arising out of or relating to the failure to obtain any Approvals that may be required in connection with the Transaction and the other transactions contemplated by this Agreement, or because of the termination of any Contract as a result of any of the foregoing. Purchaser acknowledges that no representation, warranty or covenant of Seller contained herein shall be breached or deemed breached solely as a result of (A) the failure to obtain any Approval, (B) any such termination of a Contract or (C) any Proceeding commenced or threatened by or on behalf of any Person arising out of or relating to the failure to obtain any such Approval or any such termination.
- Section 5.2 <u>Covenants Relating to Conduct of Business.</u> (a) Except (i) as set forth in <u>Section 5.2</u> of the Seller Disclosure Schedules, (ii) as required by applicable Law, (iii) as otherwise required or contemplated by the terms of this Agreement, including <u>Section 5.13</u>, or (iv) for actions required or contemplated by the Pre-Closing Reorganization Transaction Steps Schedule or the Pre-Closing Reorganization Transactions, from the date of this Agreement to the Closing, and except as Purchaser may otherwise consent to in writing (such consent not to be unreasonably withheld, conditioned or delayed), Seller shall (and shall cause the Transferred Entities to) use commercially reasonable efforts to (w) conduct the Business in all material respects in the ordinary course of business and in compliance with applicable Laws, (x) preserve intact the Business and the Purchased Assets (including the reputation and goodwill associated therewith) and preserve the relationships of the Business with customers, suppliers, licensors, licensees, distributors, agents and contractors; (y) maintain accurate books and records of the Business consistent with past practice; and (z) make all filings related to the Business Permits required to be filed in connection with the operation of the Business.
- (b) Except (i) as set forth in <u>Section 5.2</u> of the Seller Disclosure Schedules, (ii) as required by applicable Law, (iii) as otherwise required or contemplated by the terms of this Agreement, including <u>Section 5.13</u>, or (iv) actions required or expressly contemplated by the Pre-Closing Reorganization Transaction Steps Schedule or the Pre-Closing Reorganization Transactions, from the date of this Agreement to the Closing, and solely with respect to the Business, Seller shall not, and shall cause its Subsidiaries not to, do any of the following without

the prior written consent of Purchaser (such consent not to be unreasonably withheld, conditioned or delayed):

- (i) authorize or effect any amendment to, or change, the organizational documents of any Transferred Entity;
- (ii) issue, sell, pledge or transfer or propose to issue, sell, pledge or transfer any equity interests of any of the Transferred Entities, or securities convertible into, or exchangeable or exercisable for, or options with respect to, or warrants to purchase, or rights to subscribe for, equity interests of any of the Transferred Entities:
- (iii) except (A) in the ordinary course of business, (B) as may be required under any Benefit Plan or contemplated pursuant to Section 5.6, (C) as otherwise reflected in the Business Financial Statements, or (D) for any grant for which Seller or its Affiliates shall be solely obligated to pay, grant to any Business Employee who is a member of the senior leadership team of the Business (which persons are set forth on Schedule 5.2(b)(iii)(D)) any material increase in compensation or benefits;
- (iv) (A) make any material acquisition of any assets or businesses in excess of \$1,000,000 other than acquisitions of businesses or assets in the ordinary course of business, already contracted by Seller, any Transferred Entity or their respective Affiliates or (B) sell, pledge, dispose of or encumber any material assets or businesses other than in the ordinary course of business, and sales or dispositions of businesses or assets already contracted by Seller, any Transferred Entity or their respective Affiliates, inventory or obsolete assets, or as may be required by applicable Law;
- (v) (A) enter into, renew or extend the term of any Real Property Lease or (B) terminate, deliver a notice of termination, waive or accelerate any material claim or right, or breach or default under any Real Property Lease;
  - (vi) acquire any real property;
- (vii) enter into any new project or Contract with regards to a new project (A) which is outside of the ordinary course of business consistent with past practice or (B) other than a Government Bid for which Seller or the Transferred Entities are not contractually obligated to accept any resulting Contract award which involves a Collective Bargaining Agreement;
- (viii) settle any Proceeding, except for any Proceeding in the ordinary course of business and for an amount not in excess of \$250,000:
- (ix) make any material change in any method of financial accounting or financial accounting practice or policy applicable to the Business, other than (A) such changes as are required by GAAP or applicable Law and (B) any changes as requested or mandated by any Governmental Entity, including the Defense Contract Audit Agency or the Defense Contract Management Agency;
- (x) (A) other than non-exclusive licenses or non-exclusive sublicenses granted in the ordinary course of business, sell, assign, transfer or grant any license or sublicense

to any Company Business Intellectual Property or (B) allow any Registered Intellectual Property to lapse or go abandoned;

- (xi) permit any Transferred Entity to incur, create or assume, (A) any indebtedness for borrowed money (including debt evidenced by loans, notes, bonds, debentures or other similar instruments) or (B) any Lien, other than Permitted Liens, with respect to any material asset of the Business;
- (xii) (A) make, revoke or change any material Tax election, (B) adopt or change any material accounting method with respect to Taxes other than as required by applicable Law, (C) surrender any right to claim a refund of a material amount of Taxes, (D) consent to any extension or waiver of the limitations period applicable to any Tax claim or assessment, (E) amend, modify or otherwise change any filed Tax Return, (F) settle any material Tax audits or examinations, or (G) enter into any closing agreement in respect of Taxes;
- (xiii) enter (or commit to enter) into, amend, terminate or extend any (A) Transferred Entity Benefit Plan or (B) other than as required to bid on, or fulfill, a specific project, Collective Bargaining Agreement (or enter into negotiations to do any of the foregoing), with respect to the Transferred Entities or any Business Employee;
- (xiv) (A) plan, announce, implement or effect any reduction in force, layoff, early retirement program, severance program or effort concerning the termination of Business Employees or other employees or individual service providers of the Transferred Entities (other than reductions as a result of a project termination or other loss of work of a specific project that ends between the date of this Agreement and the Closing ("Covered Project Termination")), (B) terminate the employment or engagement of any Business Employee or other employee or individual service provider of the Transferred Entities (other than employment terminations of Business Employees for cause, as reasonably determined by the Company, death or disability, or for Business Employees who are not Key Employees, as a result of a Covered Project Termination) or (C) except for Business Employees or other employees or individual service providers of the Transferred Entities directly billed to customers, hire or engage any Business Employee or other individual service provider Relating to the Business of the Company, other than hiring or engaging any such individual in order to fill a vacancy in the Ordinary Course of Business or upon a termination for cause or due to death or disability, provided that the annual base salary of such replacement employee or annual compensation of such other individual service provider is less than \$150,000 per year; or
- (xv) terminate, establish, adopt, amend, modify or agree to terminate, establish, adopt, amend or modify (or announce an intention to terminate, establish adopt, amend or modify) any, Transferred Entity Benefit Plans or the Employment Agreements;
- (xvi) take an action (or fail to take any action) which would constitute "good reason", in each case as defined under the respective Employment Agreement for each such Key Employee; or
  - (xvii) authorize any of, or commit or agree to take, whether in writing or otherwise, or do any of, the foregoing actions.
- (c) Seller shall not, and shall cause its Subsidiaries not to, enter into any Contract (including any Contract awarded as a result of a pending Government Bid) which is set

forth on Exhibit L without the prior written consent of Purchaser (which consent shall not be unreasonably withheld, conditioned or delayed).

- (d) Anything to the contrary in this Agreement notwithstanding, the parties acknowledge and agree that nothing in this Section 5.2 shall be deemed to limit the transfer, use operation, acquisition or disposition of the Retained Business prior to, at or after the Closing. Anything to the contrary in this Agreement notwithstanding, nothing in this Section 5.2 shall prohibit or otherwise restrict in any way the operation of the business of Seller, the Transferred Entities, or their respective Affiliates, except solely with respect to the conduct of the Business by the Seller, the Transferred Entities and their respective Affiliates.
- (e) Nothing contained in this Agreement shall be construed to give to Purchaser, directly or indirectly, rights to control or direct the Business's operations prior to the Closing. Prior to the Closing, Seller and its Subsidiaries shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision of the operations of the Business.

# Section 5.3 Confidentiality.

- (a) Purchaser acknowledges that the information being provided to it in connection with the Transaction and the other transactions contemplated hereby is subject to the terms of that certain non-disclosure agreement between Purchaser and Seller, dated as of December 20, 2022 (the "Confidentiality Agreement"), the terms of which are incorporated herein by reference in their entirety. Effective upon, and only upon, the Closing, the Confidentiality Agreement shall terminate and no longer be of any force and effect.
- (b) Prior to the Closing, the Business was supplied copies of proprietary and confidential information relating to strategic, technical, and/or marketing plans of Seller and its Affiliates and their various operations unrelated to the Business. Although Seller has attempted to recover such information from the Business, some may still be present within the Business. Purchaser therefore agrees that it shall not use or disclose such information for any purpose whatsoever, and shall destroy any remaining copies in its possession after the Closing.
- (c) For a period of twenty-four (24) months from the Closing Date, Seller shall, and shall cause its Affiliates and Representatives to, hold in confidence any nonpublic information that is proprietary or competitively sensitive ("Confidential Business Information") to the extent relating to the Business or Purchaser and its Affiliates; provided that the foregoing restriction shall not apply to information (i) that is in the public domain or enters into the public domain through no fault of Seller or any of its Affiliates, (ii) to the extent used by Seller or any of its Affiliates (A) as may be reasonably required by Seller or its Subsidiaries in connection with any governmental investigation or similar Proceeding involving Seller or its Subsidiaries (but solely to the extent required to be disclosed in connection with such Proceeding), (B) in connection with any financial reporting required pursuant to applicable Law, including any applicable rules of any stock exchange or quotation system or (C) to comply with the terms of this Agreement or any of the Transaction Documents or any other Contract between Seller or any of its Subsidiaries, on the one hand, and Purchaser or any of its Affiliates, on the other hand, or (iii) that Seller or any of its Subsidiaries is required by Law or required pursuant to legal or regulatory process to disclose; provided, that, in the event of any disclosure in accordance with the foregoing clauses (ii) or (iii), Seller and its Affiliates shall use commercially reasonable efforts to obtain confidential treatment of any such Confidential Business Information

disclosed, and shall promptly notify Purchaser of any such disclosure to the extent permitted under applicable Law. For the avoidance of doubt, the obligations set forth on this Section 5.3(c) shall in no way apply to the Retained Business.

- Section 5.4 <u>Access to Information</u>. (a) Seller shall afford to Purchaser reasonable access, upon reasonable notice during normal business hours, consistent with applicable Law (including, for the avoidance of doubt, all applicable Data Protection Laws), in furtherance of the consummation of the Transaction and the other transactions contemplated hereby and in accordance with the procedures established by Seller, during the period prior to the Closing, to the facilities, properties, books, Contracts, records and personnel of the Seller and Transferred Entities; <u>provided</u>, that prior to the Closing Date, Purchaser shall not conduct any Phase II Environmental Site Assessment or conduct any invasive testing or any sampling of soil, sediment, surface water, ground water or building material at, on, under or within any facility on the Transferred Leased Property or any other property of Seller, the Transferred Entities or any of their respective Affiliates, without prior written permission of Seller, which shall not be unreasonably withheld, conditioned or delayed.
- (b) Purchaser agrees that any investigation undertaken pursuant to the access granted under <u>Section 5.4(a)</u> shall be conducted in such a manner as not to unreasonably interfere with the operation of the Business (or any other business of Seller or any of its Affiliates) and in a manner that complies with all applicable Laws (including, for the avoidance of doubt, all applicable Data Protection Laws), and none of Purchaser or any of its Affiliates or Representatives shall communicate with any of the employees of the Business (except for the employees set forth on Schedule 5.4(b)) without the prior written consent of Seller, which consent shall not be unreasonably withheld, conditioned or delayed. Notwithstanding anything to the contrary in this Agreement, neither Seller nor any of its Affiliates shall be required to provide access to or disclose information where, in the reasonable judgment of Seller, such access or disclosure would jeopardize attorney-client or other applicable privilege or protection, or contravene any Laws. Purchaser agrees that it will only process Personal Data provided to it by Seller or any of its Affiliates in accordance with all applicable Data Protection Laws (provided, however, that Seller and its Affiliates are solely responsible for informing Purchaser of the Data Protection Obligations that apply to such Personal Data, and implementing any contractual language required by Data Protection Laws, each prior to any disclosure of Personal Data under this Section 5.4(b)) and implement and maintain at all times appropriate technical and organizational measures designed to protect such Personal Data against unauthorized or unlawful processing and accidental loss, destruction, damage, alteration and disclosure. In addition, Purchaser agrees to provide commercially reasonable assistance to Seller in respect of any of Seller's or its Affiliates' obligations under applicable Data Protection Laws affecting the disclosure of such Personal Data to Purchaser and will not knowingly process such Personal Data in such a way to cause Seller or any of its Affiliates to violate any of its or their obligations under any applicable Data Protection Laws.
- (c) At and after the Closing (except to the extent relating to Tax matters (access, cooperation and procedures with respect to which are governed exclusively by <u>Article VI</u>)), Purchaser shall, and shall cause its Affiliates to, afford Seller, its Affiliates and their respective Representatives, during normal business hours, upon reasonable written notice, consistent with applicable Law (including, for the avoidance of doubt, all applicable Data Protection Laws), in furtherance of the consummation of the Transaction and the other transactions contemplated hereby and in accordance with the procedures established by Purchaser, access to the properties, books, Contracts, records and employees of the Business and

the Transferred Entities to the extent that such access may be reasonably requested by Seller, including in connection with financial statements, reporting obligations, defense of claims and compliance with applicable Laws.

- (d) Except for Tax Returns and other documents governed by Article VI, Purchaser agrees to hold all the books and records of the Business existing on the Closing Date and not to destroy or dispose of any thereof for a period of seven (7) years from the Closing Date or such longer time as may be required by Law, and thereafter, if it desires to destroy or dispose of such books and records, to offer first in writing at least sixty (60) days prior to such destruction or disposition to surrender them to Seller.
- (e) At and after the Closing (except to the extent relating to Tax matters (access, cooperation and procedures with respect to which are governed exclusively by Article VI), solely with respect to the Business and to the extent such information was not transferred to a Transferred Entity pursuant to the Pre-Closing Reorganization Transactions, Seller shall, and shall cause its Affiliates to, afford Purchaser, its Affiliates and their respective Representatives, during normal business hours, upon reasonable written notice, consistent with applicable Law (including, for the avoidance of doubt, all applicable Data Protection Laws), in furtherance of the consummation of the Transaction and the other transactions contemplated hereby and in accordance with the procedures established by Seller, access to the properties, books, Contracts, records and employees of Seller to the extent that such access may be reasonably requested by Purchaser, including in connection with financial statements, reporting obligations, defense of claims and compliance with applicable Laws.
- Section 5.5 <u>Publicity.</u> No party to this Agreement nor any Affiliate or Representative of such party shall issue or cause the publication of any press release or public announcement in respect of this Agreement or the transactions contemplated by this Agreement without the prior written consent of the other party (which consent shall not be unreasonably withheld, conditioned or delayed), except as may be required by Law or stock exchange rules, in which case the party required to publish such press release or public announcement shall use reasonable efforts to provide the other party a reasonable opportunity to comment on such press release or public announcement in advance of such publication; provided, that, without prior consent of the other party, each party hereto or any of its Affiliates or Representatives may disseminate information substantially consistent with information included in a press release or other document previously approved for external distribution pursuant to this Section 5.5.

# Section 5.6 Employee Matters.

- (a) <u>Compensation and Benefit Continuation</u>. The Seller and its Affiliates will take such action as is necessary such that, as of the Closing, the Business Employees other than the Inactive Employees (as defined below in <u>Section 5.6(b)</u>) and Transferred Entities cease participation in each Benefit Plan that is not a Transferred Entity Benefit Plan. The Seller and its Affiliates will retain all Liabilities under each Seller Benefit Plan, whenever incurred, and all Liabilities under each Transferred Entity Benefit Plan incurred on or prior to the Closing Date.
- (b) <u>Inactive Employees</u>. Notwithstanding anything to the contrary contained herein, all Business Employees who are, as of immediately prior to the Closing not actively employed due to an approved leave of absence with an expected return to work date that is more than six (6) months after the start of such leave of absence ("<u>Inactive Employees</u>") shall not be employed by any of the Transferred Entities as of the date of Closing, provided, however, that

Purchaser or its Subsidiaries shall, at least ten (10) business days prior to the Closing, make offers of employment (consistent with the terms of Section 5.6(d) hereof for active Business Employees) to be employed by one of the Transferred Entities (or another Subsidiary or Affiliate of Purchaser) to all such Inactive Employees contingent upon the Closing and such Inactive Employees being able to return to full-time active status within six (6) months following the Closing. Notwithstanding any other provisions of this Agreement, Purchaser shall have no obligation with respect to any Inactive Employee who does not accept such an offer of employment or begin work pursuant to such offer letter within six (6) months following the Closing. Additionally, notwithstanding any other provision of this Agreement, Purchaser and its Subsidiaries shall have no obligations whatsoever regarding any Inactive Employee prior to the date such Inactive Employee begins work pursuant to such offer letter following the Closing.

# (c) [Reserved].

- (d) Offers to Certain Business Employees. No later than ten (10) business days prior to the Closing Date, Purchaser shall or shall cause one of its Subsidiaries to, make a written offer of employment (i) consistent with the requirements of this Section 5.6 and applicable Law, and (ii) in a position that is substantially comparable to such individual's position immediately prior to the time of offer to each Business Employee listed on Section 5.6(d) of the Seller Disclosure Schedules (each an "Offer Employee").
- (e) With respect to (i) each Business Employee who is employed by one of the Transferred Entities (or another Subsidiary or Affiliate of Purchaser) following the Closing and (ii) each Offer Employee who accepts an offer of employment with Purchaser or its Subsidiaries (in the case of (i) or (ii), a "Continuing Employee"), Purchaser shall maintain, for the longer of (A) the twelve (12) month period following the Closing, and (B) the applicable period required by local Law (such period, the "Continuation Period"), (1) the same wage rate or salary in effect for such employee immediately prior to the Closing, (2) target cash bonus opportunities (excluding equity-based compensation, retention or transaction bonus plans) that are substantially similar in the aggregate to those provided by the Seller prior to the Closing (3) and other employee benefits not expressly covered in the foregoing (1) and (2), excluding any equity-based compensation, retention or transaction bonus plans, that are substantially similar, in the aggregate, to those provided by the Seller immediately prior to the Closing but excluding any grandfathered or closed plans. Notwithstanding the foregoing, nothing herein shall be interpreted as, nor shall it, require Purchaser to employ any Business Employee for any period of time after Closing.
- (f) <u>Service Credit</u>. As of and after the Closing, Purchaser shall provide to each Business Employee credit for the length of service with Seller and its Affiliates (and their predecessors) and credit under each employee benefit plan, policy or arrangement and each seniority-based entitlement (including vacation sick time, paid time off leave and severance policies), sponsored by the Purchaser or any of its Affiliates in each case, in which Business Employees participate (excluding any such employee benefit plans that are equity-based, frozen or grandfathered as of the Closing) as of the Closing for purposes of any waiting period, vesting, eligibility and benefit entitlement; provided that such service shall not be credited for purposes of benefit accrual under any defined benefit pension plan or other plan providing for post-employment benefits, or for purposes of eligibility for retirement vesting under the equity incentive plans of the Purchaser and its Affiliates; provided, further, that such service shall not be credited to the extent such credit would result in any duplication of compensation or benefits.

- (g) Health and Welfare Coverages. Purchaser shall use commercially reasonable efforts to cause each Continuing Employee (and his or her "eligible dependents," as defined in the Seller Benefit Plans) to be covered as soon as reasonably practicable following the Closing by a group health plan or plans and dental, vision, and life insurance plans that (i) comply with the provisions of Section 5.6(a), (ii) do not limit or exclude coverage on the basis of any pre-existing condition of such Continuing Employee or dependent (other than any limitation already in effect under the applicable group health Seller Benefit Plan) or on the basis of any other exclusion or waiting period not in effect under the applicable group health Seller Benefit Plan, and (iii) provide each Continuing Employee full credit, for the year in which the Closing Date occurs, for any deductible or co-payment already incurred by the Continuing Employee under the applicable group health Seller Benefit Plan and for any other out-of-pocket expenses that count against any maximum out-of-pocket expense provision of the applicable group health Seller Benefit Plan or Purchaser's group health plans subject in each case to the Seller providing adequate data to the Purchaser, in a format reasonably acceptable and agreed to by Seller and the Purchaser.
- (h) <u>Severance</u>. With respect to each Business Employee whose employment is terminated without cause (as defined in <u>Section 5.6(h)</u> of the Seller Disclosure Schedules) during the Continuation Period, Purchaser shall provide the severance benefits described in <u>Section 5.6(h)</u> of the Seller Disclosure Schedules, and the amount of such severance benefits shall be calculated taking into account the Business Employee's period of employment with Seller and its Affiliates (and their predecessors) prior to the Closing and with Purchaser or its Affiliates on and after the Closing; <u>provided</u>, however that this <u>Section 5.6(h)</u> shall not apply with respect to any Business Employee for whom 100% of compensation and benefits are billed directly to a customer, and further provided that, for the avoidance of doubt, Purchaser shall have no obligation to provide severance benefits with respect to Inactive Employees who do not perform work for one of the Transferred Entities (or another Subsidiary or Affiliate of Purchaser) following the Closing.
- (i) <u>WARN and Corresponding State Laws</u>. Purchaser shall be solely responsible for the obligation to provide appropriate notices under WARN, with respect to any Business Employee who suffers an "employment loss" under WARN after the Closing Date solely as a result of the actions of Purchaser or any of its Affiliates. Seller and/or its Affiliates shall retain this Liability for any such actions occurring on or before the Closing Date. Seller shall reasonably cooperate and assist (or cause its Affiliates to reasonably cooperate) Purchaser in providing WARN notices, to the extent providing such notices are required by Law.
- (j) 401(k) Plan. As soon as reasonably practicable after the Closing, Purchaser shall cause each Continuing Employee who both (i) as of immediately prior to the Closing, was eligible to participate in a Seller Benefit Plan that is a U.S. tax-qualified defined contribution plan (collectively, the "Seller 401(k) Plans") and (ii) as of immediately prior to the Closing is not a participant in a Transferred Entity Benefit Plan that is a U.S. tax-qualified defined contribution plan, to be eligible to participate in Purchaser's U.S. tax-qualified defined contribution plan (the "Purchaser 401(k) Plan"). Purchaser shall cause the Purchaser 401(k) Plan to accept "eligible rollover distributions" (as such term is defined under Section 402 of the Code) of Continuing Employees from the Seller 401(k) Plans (including the loans and related notes).
- (k) <u>Deferred Compensation</u>. Seller shall be solely responsible for the payment of the account balances held by each Offer Employee in the VSE Corporation Deferred

Supplemental Compensation Plan (the "<u>Deferred Compensation Plan</u>") (including the employer portion of any payroll, social security, unemployment or similar Tax imposed on such amounts). Seller shall retain the Liability for all such amounts paid to the Offer Employees from the Deferred Compensation Plan.

- (1) New Transferred Entity Benefit Plans. To the extent requested by Purchaser within sufficient time prior to Closing, Seller shall, prior to Closing, use commercially reasonable efforts to cause the Company to establish and sponsor a qualified defined contribution plan and health and welfare benefits (which may include establishing a voluntary employees' beneficiary association (VEBA) trust, within the meaning of Section 501(c)(9) of the Code (the "New VEBA")), to provide benefit plan coverage to Continuing Employees, effective as of the Closing Date. To the extent that, as of the Closing, any fringe reserve attributable to Continuing Employees who are Service Contract Act employees exists and is an obligation of the Company (the "Closing Fringe Reserve"), (i) an amount equal to the Closing Fringe Reserve will be reflected as a current liability in the calculation of the Closing Working Capital, and (ii) Seller will cause a cash amount equal to the Closing Fringe Reserve to be reflected as a current asset in the calculation of the Closing Working Capital and either (x) transferred to the general assets of the Company or (y) contributed in cash to the New VEBA, if established.
- (m) No Third-Party Rights. Without limiting Section 10.4, this Section 5.6 will not create any third-party beneficiary rights, express or implied, and will not be enforceable by any current or former employee, union representative, or any spouse, dependent or beneficiary of any current or former employee, or any other Person who is not a party to this Agreement. This Section 5.6 will not be deemed to be an establishment of or amendment to any employee benefit plan. No term of this Agreement will be deemed to create any Contract with any employee or to give any employee the right to be retained in the employment of Seller or any of its Affiliates, or, after the Closing, Purchaser or any of its Affiliates, to terminate the employment of any employee at any time or for any reason.

# Section 5.7 [Intentionally Omitted]

•

## Section 5.8 Names Following Closing.

- (a) Neither Purchaser nor any of its Affiliates (including the Transferred Entities) shall acquire any rights in, or use, or have the right to use, the "VSE," or "VSE Corporation" names or any variations or derivatives thereof or any trademarks or logos of Seller or any of its Affiliates containing "VSE," "VSE Corporation" or "Wheeler" (the "Names"), or any name that, in the reasonable judgment of Seller, is similar to the Names, except as provided in Section 5.8(b).
- (b) The Company may continue to temporarily use the Names following the Closing, to the extent and in the same manner as used immediately prior to the Closing, so long as Purchaser shall, and shall cause its Affiliates (including the Business), to (i) immediately after the Closing cease to hold itself out as having any affiliation with Seller or any of its Affiliates and (ii) use commercially reasonable efforts to minimize and eliminate use of the Names by the Business from and after the Closing; provided, that as soon as practicable after the Closing Date, and in any event by the later of (x) December 31, 2023 or (y) ninety (90) days from the Closing

Date, Purchaser shall, and shall cause each of its Affiliates (including the Transferred Entities), to (A) cease and discontinue use of all Names and (B) complete the removal of the Names from all products, signage, vehicles, properties, technical information and promotional or other marketing materials and other assets; provided, further, that (a) products of the Business in finished goods inventory (to the extent the same bear any of the Names or any corporate symbol or logo related thereto on the Closing Date) may be disposed of without remarking in the ordinary course of business for a period of six (6) months following the Closing or until the supply is exhausted, whichever is the first to occur, and (b) notwithstanding anything to the contrary, nothing in this Section 5.8(b) shall preclude Purchaser or its Affiliates from (X) describing the historical relationship of the Transferred Entities with the Seller or its Subsidiaries or (Y) complying with its obligations under Section 2.9 or Schedule 2.

Section 5.9 <u>Insurance</u>. From and after the Closing Date, the Business, the Transferred Entities, and the operations and assets and Liabilities in respect thereof, shall cease to be insured by Seller's or its Affiliates' insurance policies or by any of their self-insured programs. Seller or its Affiliates may amend any insurance policies in the manner it deems appropriate to give effect to this <u>Section 5.9</u>; <u>provided, that, Seller shall not amend its policies in a manner that would be adverse to any existing claims of the Business as of the Closing Date without the prior written consent of Purchaser (not to be unreasonably withheld, conditioned or delayed). From and after the Closing, Purchaser shall be responsible for securing all insurance it considers appropriate for the Business, the Transferred Entities, and the operations and assets and Liabilities in respect thereof. At the Closing, Purchaser agrees to take over and assume all the known and incurred but not reported claims in respect of the Business, the Transferred Entities, and the operations and assets and Liabilities in respect thereof, solely to the extent arising from or after the Closing. Seller agrees to use commercially reasonable efforts to assist Purchaser and the Transferred Entities in exercising any rights or claims of, or in respect of, the Business, the Transferred Entities, and the operations and assets and Liabilities in respect thereof, under or in respect of any past or current insurance policy under which any of the foregoing is a named insured.</u>

Section 5.10 Post-Closing Support. In the event that and for so long as Seller or any of its Affiliates is prosecuting, contesting or defending any Proceeding by or against a third party in connection with or relating to any fact, situation, circumstance, status, condition, activity, practice, plan, occurrence, event, incident, action, failure to act, or transaction relating to, in connection with or arising from the Business or the Transferred Entities, Purchaser shall, and shall cause its Affiliates (and its and their officers and employees and Representatives) to, reasonably cooperate with Seller and its counsel (at Seller's sole cost and expense) in such prosecution, contest or defense, including making available its personnel, and providing such testimony and access to its books and records and other information as shall be reasonably necessary in connection with such prosecution, contest or defense; provided, that, nothing herein shall require Purchaser or its Affiliates to disclose any privileged information. For the avoidance of doubt, and notwithstanding anything in this Agreement to the contrary, Seller shall retain full control of prosecuting, contesting, defending, compromising, settling or taking any other action related to or in connection with any Proceeding by or against a third party related to any Retained Businesses, whether arising before or after the Closing, and neither Purchaser nor its Affiliates shall have any rights in connection therewith; and further, Seller shall retain full responsibility, including any damages, attorneys' fees, costs, sanctions, penalties, or other liabilities, obligations, or responsibilities arising from or relating to any such Proceeding by or against a third party related to any Retained Businesses, whether arising before or after the Closing, and neither Purchaser nor its Affiliates shall have any liabilities, obligations, or responsibilities arising before or after the Closing, and neither Purchaser nor its Affiliates shall have any liabilities, obligations,

Section 5.11 <u>Export Control Authorizations</u>. The parties hereto acknowledge that certain Export Control Authorizations may be required by the Company to lawfully operate the Business after the Closing. The Company shall be solely responsible for obtaining, amending, or providing notices with respect to such Export Control Authorizations as may be necessary after the Closing. Without limitation to the foregoing, Purchaser shall cause the Company to submit a "Material Change Notification" to the Directorate of Defense Trade Controls within five (5) days of the Closing to the extent required pursuant to International Traffic in Arms Regulations § 122.4(a).

#### Section 5.12 Payments from Third Parties.

- (a) Seller shall, or shall cause its applicable Affiliate to, (i) promptly pay or deliver to Purchaser (or Purchaser's designated Affiliate) any monies or checks that are in respect of the Business that have been delivered to Seller or any of its Subsidiaries following the Closing, including any monies or checks sent by customers, suppliers or other contracting parties in respect of the Business and (ii) promptly reimburse Purchaser (or its designated Affiliates) for any amounts paid by Purchaser (or its designated Affiliates) to the extent such payments are or are in respect of the Retained Business.
- (b) Purchaser shall, or shall cause its applicable Affiliate to, (i) promptly pay or deliver to Seller (or its designated Affiliates) any monies or checks that have been sent to Purchaser or any of its Affiliates after the Closing to the extent such monies or checks are in respect of the Retained Business and (ii) promptly reimburse Seller (or its designated Affiliates) for any amounts paid by Seller to the extent such payments are or are in respect of the Business.
- Section 5.13 Novation of Government Contracts; Subcontracting Agreements; Purchaser Approvals . Notwithstanding anything to the contrary in this Agreement:
- (a) Promptly, and as soon as reasonably practicable following the date hereof (with Seller using commercially reasonable efforts to complete within 20 Business Days), Seller shall transfer and assign to the Company each of the Contracts set forth on Section 5.13(a) of the Seller Disclosure Schedules (the "Assigned Contracts"). The parties hereto acknowledge that the Closing may occur prior to completion of the process required to novate all of the Assigned Contracts under 48 CFR 42.1204. This Agreement shall not constitute an agreement to transfer any Assigned Contract if an attempted transfer thereof, without the Approval of a Governmental Entity, would constitute a breach or other contravention thereof or in any way adversely affect the rights of Seller or any of its Affiliates thereunder, including each Assigned Contract listed on Section 5.13(a) of the Seller Disclosure Schedules (each such listed Assigned Contract, a "Specified Contract"). Any such Specified Contract shall be transferred to the Purchaser in accordance with the following provisions of this Section 5.13 and notwithstanding any failure to transfer the Specified Contracts at the Closing, the Closing shall nonetheless take place subject only to the satisfaction or waiver of the conditions set forth in Article VII.
- (b) From and after the Closing and until the receipt of all Approvals required to effect the transfer of each Specified Contract (the "Specified Contract Assignments") in accordance with this Section 5.13(b) (the "Final Transition Time"): (i) in each case pursuant to and in accordance with the terms of the Subcontract Agreement Pending Novation, (A) the Company (or one of its Subsidiaries) shall, as a subcontractor of the Seller, perform Seller's obligations under the Specified Contracts to the maximum extent permitted under the terms of such Specified Contracts, and (B) to the extent the Company (or one of its Subsidiaries) cannot

perform the Seller's obligations as a subcontractor under the specified Contracts as provided in clause (A), the Company (or one of its Subsidiaries) shall second necessary employees to the Seller; and (ii) the parties hereto shall, subject to applicable Law, use commercially reasonable efforts to obtain the required Approvals and complete novation of the Specified Contracts from Seller to the Company in accordance with 48 CFR 42.1204. The parties hereto shall, as promptly as reasonably practicable, effect the Specified Contract Assignments. Subject to and without limiting Section 5.1 in any respect, from and after the date hereof, the parties hereto shall, subject to applicable Law, use their respective commercially reasonable efforts to obtain the Approval of any Governmental Entity or, subject to Section 5.1(e), any other third party, as applicable, in each case, required in respect of the subcontracting arrangements contemplated by the Subcontract Agreement Pending Novation as promptly as practicable.

#### Section 5.14 Shared Contracts.

Seller and Purchaser acknowledge that Seller and/or its Subsidiaries are parties to certain vendor or supply Contracts (collectively, the "Shared Contracts") that relate in part to both (i) the operations or conduct of the Business and (ii) the operations or conduct of the Retained Businesses. Such Shared Contracts are set forth on Schedule 5.14. Subject to applicable Law, unless Seller or its applicable Subsidiary and Purchaser otherwise agree or the benefits of any Shared Contract described in this Section 5.14(a) are otherwise expressly conveyed to the applicable party pursuant to this Agreement or any other Transaction Document, Seller or its applicable Subsidiary and Purchaser shall cooperate with each other and use their respective commercially reasonable efforts prior to the Closing to cause each Shared Contract (other than any Shared Contract that is immaterial to the Business) to be apportioned (including by using their respective commercially reasonable efforts to obtain the Approval of such counterparty to enter into a new contract or amendment, or splitting or assigning in relevant part such Shared Contract), effective as of the Closing, between Seller or its applicable Subsidiary and the Company, pursuant to which Seller or its applicable Subsidiary will assume all of the rights and obligations under such Shared Contract to the extent relating to the Retained Businesses, on the one hand, and the Company will assume all of the rights and obligations under such Shared Contract to the extent relating to the Business, on the other hand and use their commercially reasonable efforts to, as soon as practicable, terminate the Purchaser or Company portion of such Shared Contract and to enter into a new Contract with the Company in respect of such portion, such new Contract (a "Replacement Contract") being effective as of the Closing Date and limited to the Business but otherwise on substantially similar terms to those that existed under the applicable Shared Contract or on terms that are at least no less favorable than current market terms offered by such counterparty to the Shared Contract. To the extent a Replacement Contract has not been concluded for a Shared Contract prior to the Closing Date, the Seller and the Purchaser shall, in good faith, enter into a subcontract or sublicensing arrangement for the Business to continue to receive such services until the earlier of (x) the natural expiration of such Contract but not to exceed six (6) months from the Closing Date or (y) the Company's entry into a Replacement Contract. From and after the Closing, (1) the Purchaser or the Company shall reimburse, indemnify and hold harmless Seller and its Subsidiaries against all Covered Losses arising from or relating to the portion of any Shared Contract apportioned to the Business, (2) the Seller shall reimburse, indemnify and hold harmless Purchaser and its Subsidiaries (including the Company) against all Covered Losses arising from or relating to the portion of any Shared Contract apportioned to the Retained Business and (3) Seller and its Affiliates shall not amend the terms of any Shared Contract in a manner that would materially and adversely affect the Business without the prior written consent of Purchaser (not to be unreasonably withheld, conditioned or delayed).

- (b) Notwithstanding anything in this Agreement to the contrary, any apportionment to Purchaser of any Shared Contract that shall require Approval of a third party shall be made subject to such Approval being obtained, and Seller and its Subsidiaries shall not be required to agree to any arrangement or take any action in connection with the matters contemplated by Section 5.14(a) that would (i) constitute a breach or other contravention in respect of any Shared Contract, (ii) be ineffective, void or voidable, or (iii) adversely affect the contractual rights of Seller or its Subsidiaries or any of their respective officers, directors, agents or Affiliates. If any such third-party Approval referred to in this Section 5.14 is not obtained prior to Closing, the Closing shall nonetheless take place subject only to the satisfaction or waiver of the conditions set forth in Article VII.
- (c) The Seller shall, and shall cause its Affiliates to following the Closing Date, provide access to Purchaser to all relevant books, records and other documents and information in relation to the Purchaser or Company portion of any Shared Contract as the Purchaser may reasonably require.
- (d) Solely to the extent related to the Business, Seller shall promptly forward to Purchaser a copy of all correspondence, notice or any other document or items received, or sent, with respect to the Shared Contract; provided, that, for the avoidance of doubt, Purchaser and its Affiliates may not contact any counterparty to a Shared Contract or any of such counterparty's Affiliates without the prior written consent of the Seller (including review of any documentation or information proposed to be sent to such counterparty).
- (e) Upon the entry into a Replacement Contract for any Shared Contract, any contractual arrangements between the parties hereto in accordance with Section 5.14(a), if any, shall automatically terminate or be discontinued, as the case may be and (i) Seller and its Affiliates shall owe no further obligations to Purchaser or its Affiliates with respect thereto and (ii) Purchaser and its Affiliates shall owe no further obligations to Seller or its Affiliates with respect thereto.

### Section 5.15 <u>Title Commitments; Surveys.</u>

- (a) Purchaser shall have the right to order, at its sole cost and expense, a title commitment for the Transferred Owned Real Property ("Commitment") from a title insurance company selected by Purchaser sufficient in form to allow Purchaser to obtain, at Purchaser's sole cost and expense, an owner's title insurance policy ("Title Insurance Policies") insuring the fee simple interest in the Transferred Owned Real Property. The Seller shall, and shall cause the Transferred Entities to, deliver any information as may be reasonably and customarily required by the title company under the requirements section of each Commitment or otherwise in connection with the issuance of the final Title Insurance Policies. The Transferred Entities agree to provide a commercially reasonable affidavit of title and/or such other reasonable information as the title company may reasonably require in order for the title company to delete the mechanics' lien and parties in possession exceptions and to insure over the gap (i.e., the period of time between the effective date of the title company's last bringdown of title to such Transferred Owned Real Property and the Closing Date) and to use commercially reasonable efforts to cause the title company to delete all standard exceptions from the final Title Insurance Policies.
- (b) In addition, Purchaser shall have the right to order a survey for each Transferred Owned Real Property ("Surveys"), at its sole cost and expense. Seller shall, and

shall cause the Transferred Entities to, agree to deliver to Purchaser any information reasonably and customarily required to obtain the Surveys.

Section 5.16 <u>R&W Insurance Policy</u>. Prior to or concurrently with the execution of this Agreement, Purchaser shall cause to be bound and incepted a representations and warranties insurance policy (a copy of which Purchaser shall have provided Seller prior to the execution thereof) (the "<u>R&W Insurance Policy</u>"). Purchaser will use commercially reasonable efforts to comply with all terms of such binder agreement to cause the proper and timely issuance of the R&W Insurance Policy. Purchaser, on the one hand, and Seller on the other hand, shall each timely pay, or cause to be paid, 50% of the premium and other costs required for issuance of the R&W Insurance Policy. Purchaser will use commercially reasonable efforts to, and to cause each of its applicable insurance brokers and insurers to, take reasonable actions necessary for the R&W Insurance Policy to be issued to Purchaser as soon as practicable after Closing. Following issuance of the R&W Insurance Policy, Purchaser agrees to use commercially reasonable efforts to keep the R&W Insurance Policy in full force and effect for the policy period set forth therein. Upon its final issuance, Purchaser shall deliver a copy of the R&W Insurance Policy to Seller. The parties hereto acknowledge that Purchaser obtaining the R&W Insurance Policy is a material inducement to Seller entering into the transactions contemplated by this Agreement, and Seller is relying on Purchaser's covenants and obligations set forth in this <u>Section 5.16</u>. The R&W Insurance Policy shall include a provision whereby the insurer expressly waives, and irrevocably agrees not to pursue, directly or indirectly, any subrogation rights against Seller or any of its Affiliates, or any former shareholders, managers, members, directors, officers and employees of any of the foregoing with respect to any claim made by any insured thereunder. Purchaser shall not waive, amend, modify or otherwise alter or revise any provision in the R&W Insurance Policy or consent to the waiver, amendment, modification, alter

Section 5.17 Termination of Intercompany Balances and Intercompany Agreements. Immediately prior to the Closing (or prior thereto, if so determined by Seller), all intercompany balances and accounts (other than intercompany balances and accounts set forth in Section 5.17 of the Seller Disclosure Schedules) between the Seller and any of its Affiliates (other than the Transferred Entities), on the one hand, and the Transferred Entities, on the other hand, shall be settled or otherwise eliminated in such a manner as Seller shall determine in its sole discretion (including, if so determined by Seller or any of its Affiliates removing from any Transferred Entity any or all Cash Amounts or funds from cash pools by means of dividends, distributions, contribution, the creation or repayment or refinancing of intercompany debt, increasing or decreasing of cash pool balances or otherwise); provided, that, no such settlement or elimination will result in a Liability to any Transferred Entity following the Closing. Intercompany balances and accounts solely among any of the Transferred Entities shall not be affected by the above provisions of this Section 5.17. Immediately prior to the Closing (or prior thereto, if so determined by Seller), except for the Transaction Documents to be entered into in connection with this Agreement and any arrangements, understandings or Contracts set forth in Section 5.17 of the Seller Disclosure Schedules, all Intercompany Agreements shall automatically be terminated without further payment or performance and cease to have any further force and effect, such that no party thereto shall have any further obligations or Liabilities therefor or thereunder.

# Section 5.18 <u>Directors' and Officers' Indemnification</u>.

- (a) For a period of six (6) years from and after the Closing, Purchaser shall cause each of the Transferred Entities to indemnify, defend and hold harmless, and advance expenses as incurred, in each case to the fullest extent permitted under applicable Law (and to the extent the organizational documents of the Transferred Entities as of Closing provide therefor or under any applicable Contracts between a Transferred Entity and such D&O Indemnified Party which are set forth on <u>Schedule 5.18</u>), each present and former manager, director and officer of the Transferred Entities (each, a "<u>D&O Indemnified Party</u>") against any Covered Losses incurred in connection with any pending or completed Proceeding, whether civil, criminal, administrative or investigative, based on, arising out of or related to such D&O Indemnified Parties' service as a manager, director or officer of the Transferred Entities at or prior to the Closing, whether asserted or claimed prior to, at or after the Closing (a "<u>D&O Indemnifiable Claim</u>").
- (b) The obligations of Purchaser under this <u>Section 5.18</u> shall not be terminated or modified in such a manner as to adversely affect in any material respect any D&O Indemnified Party without the express written consent of such affected D&O Indemnified Party. The rights of the D&O Indemnified Parties under this <u>Section 5.18</u> shall be in addition to any rights such D&O Indemnified Parties may have under the organizational documents of the Transferred Entities, or under any applicable Contracts or Laws. All rights to indemnification and exculpation from liabilities for acts or omissions occurring at or prior to the Closing and rights to advancement of expenses relating thereto now existing in favor of any D&O Indemnified Party as provided in the organizational documents of the Transferred Entities or any indemnification agreement between such D&O Indemnified Party which has been disclosed on <u>Schedule 5.18</u> and the Transferred Entities shall survive the Closing and shall not be terminated, amended, repealed or otherwise modified in any manner that would adversely affect in any material respect any right thereunder of any such D&O Indemnified Party without the consent of such D&O Indemnified Party and each Person entitled to insurance coverage under this <u>Section 5.18</u>, respectively, and his or her heirs and legal representatives.
- (c) At the Closing, Seller shall, or shall cause the Transferred Entities to, at Seller's expense, obtain and fully pay the premium for irrevocable "tail" insurance policies naming the D&O Indemnified Persons as direct beneficiaries with a claims period of six (6) years from the Closing Date from an insurance carrier with the same or better financial-strength rating as the Seller's current insurance carrier with respect to directors' and officers' liability insurance on terms, conditions, retentions and limits of liability that are at least as favorable in the aggregate to the D&O Indemnified Parties as the Seller's existing policies with respect to matters existing or occurring at or prior to the Closing Date.

#### Section 5.19 Misallocated Assets.

(a) If, at any time within twelve (12) months following the Closing, any right, property or asset not forming part of the Business, is found to have been transferred to the Company or its Affiliates (including any Transferred Entities) in error, either directly or indirectly, Purchaser shall transfer, or shall cause its Affiliates (including the Transferred Entities) to transfer, at Purchaser's sole cost and expense (including payment of all Transfer Taxes) and for no additional consideration, such right, property or asset as soon as practicable to the Seller.

- (b) If, at any time within twelve (12) months following the Closing, any right, property or asset forming part of the Business, is found to have been retained by Seller in error, either directly or indirectly, Seller shall transfer, or shall cause the Seller to transfer, at Seller's sole cost and expense (including payment of all Transfer Taxes) and for no additional consideration, such right, property or asset as soon as practicable to a Transferred Entity, indicated by Purchaser in writing.
- Section 5.20 <u>Pre-Closing Reorganization Transaction</u>. From and after the date hereof, Seller shall, and shall cause its Affiliates to, use commercially reasonable efforts to consummate the Pre-Closing Reorganization Transactions as promptly as is reasonably practicable.
- Section 5.21 <u>Transition Services</u>. Prior to the Closing, Purchaser and Seller shall, if reasonably necessary for the operations of the Seller or the Business following the Closing, negotiate in good faith and enter into a transition services agreement effective as of the Closing. Such transition services agreement shall include the services that are reasonably required to operate in the ordinary course of business consistent with past practice following the Closing (a) the Business and/or (b) the Seller's Retained Businesses, in each case that have been historically provided to each respective business from the other during the twelve (12)-month period prior to the Closing Date, in the ordinary course of business consistent with past practice, and such services shall be provided on an at-cost basis and on terms and conditions consistent with the level of such services provided and costs charged by each respective business to the other during the twelve (12)-month period prior to the Closing Date.
- Section 5.22 <u>Financing Obligations</u>. Purchaser shall not, without the prior written consent of Seller, (a) permit any amendment or modification to, or any waiver of any provision or remedy under, the Equity Commitment Letter, if such amendment, modification, waiver or remedy could reasonably be expected to (i) add new (or adversely modify any existing) conditions to the consummation of all or any portion the Equity Financing, (ii) reduce the aggregate amount of the Equity Financing, (iii) adversely affect the ability of Purchaser to enforce its rights against other parties to the Equity Commitment Letter or (iv) prevent or materially delay the consummation of the transactions contemplated by this Agreement; or (b) cause the Equity Commitment Letter to lapse, or otherwise terminate the Equity Commitment Letter, prior to the earlier to occur of the Closing and the expiration of such Equity Commitment Letter in accordance with its terms.

# Section 5.23 <u>Financing Cooperation</u>.

(a) From the date of this Agreement until the Closing Date, Seller shall, and shall cause the Company and its and their respective and appropriate directors, officers, employees, and advisers, to provide to the Purchaser such assistance and cooperation as is reasonably requested by the Purchaser and as is customary in relation to the financing arrangements to be entered into by the Purchaser in connection with the transactions contemplated under this Agreement (the "**Debt Financing**"). Such assistance and cooperation may include (A) making appropriate directors, officers and employees of the Company available for participation in a reasonable number of meetings, conference calls, due diligence sessions, and rating agency presentations with providers of the Debt Financing; (B) reasonably cooperating with the Purchaser in its preparation of marketing materials in connection with such Debt Financing; (C) assisting in the preparation of a customary bank information memorandum, offering memorandum, lender presentation, investor presentation, marketing materials and similar marketing and syndication documents, customary materials for rating agency

presentations, and customary authorization letters (and execute such authorization letters); (D) preparing and furnishing Purchaser with reasonable and customary historical financial and other pertinent business information regarding the Company as may be reasonably requested in writing by Purchaser, for use in connection with the Debt Financing and providing reasonable assistance to the Purchaser's preparation of pro forma financial information and projections; provided, that, for the avoidance of doubt, the information described in this clause (D) shall not include, and Purchaser shall be solely responsible for, the actual preparation of pro forma financial information; (E) furnishing Purchaser with all documentation and other information regarding the Company required by Governmental Entities with respect to the Debt Financing under applicable "know your customer" and anti-money laundering rules and regulations, including the PATRIOT Act and the Beneficial Ownership Regulation; (F) facilitating the pledging of collateral for the Debt Financing (including delivering original stock certificates and original stock powers), including executing and delivering any customary pledge and security documents, credit agreements, ancillary loan documents, original stock certificates and customary closing certificates and documents (in each case, subject to and only effective upon the occurrence of the Closing Date), including assisting in preparing any schedules thereto as may be reasonably requested by Purchaser; and (G) reasonably cooperating with Purchaser's legal counsel in connection with customary legal opinions required of Purchaser in connection with the Debt Financing (it being understood and agreed that the Company (and its legal counsel) shall not be required to deliver any legal opinions in connection with the Debt Financing). Seller consents to the customary and reasonable use of the logos in respect of the Company solely in connection with any such Debt Financing; provided that such logos are used solely in a manner that is not intended, or reasonably likely, to harm or disparage Seller or the reputation or goodwill of Seller. Purchaser acknowledges and agrees that neither the Seller, the Company nor any of their respective Affiliates or representatives shall incur any liability to any Person prior to the Closing under or in connection with the Debt Financing that is not simultaneously reimbursed by Purchaser at or prior to the Closing.

(b) Notwithstanding anything to the contrary in this Agreement, (1) nothing in Section 5.23 shall require cooperation to the extent it would (i) subject the Seller or any of its Affiliates or Representatives to any actual or potential personal liability (provided that this clause shall not limit cooperation with respect to any customary authorization letters described in clause (a) above), (ii) unreasonably interfere with the business or operations of the Seller or the Company, (iii) reasonably be expected to conflict with, or violate, the Seller's certificate of formation, operating agreement or any organizational documents or any applicable Law, or result in the contravention of, or violation or breach of, or default under, any material Contract to which the Seller or any of its Affiliates is a party, in each case, as in effect on the date hereof, (iv) cause any representation, warranty, covenant or other obligation in this Agreement or any Transaction Document to be breached or any closing condition to fail to be satisfied, or (v) require the Seller to agree to (A) pay or pay any fees or reimburse any expenses or make any other payment in connection with any financing which are not reimbursed or indemnified hereunder, or (B) give any indemnities or incur any liabilities in connection with any financing or any information utilized in connection therewith. Neither the Seller, the Company or any of their respective Affiliates and Representatives shall have any liability to the Purchaser in respect of any financial information or data or other information provided pursuant to this Section 5.23.

# Section 5.24 Non-Competition; Non-Solicitation; Non-Disparagement.

(a) For a period of three (3) years from the Closing Date, each of Seller and its Affiliates agrees that, without the prior written consent of Purchaser, it will not, and will cause

its Affiliates not to, directly or indirectly (whether by itself, through an Affiliate in partnership or conjunction with or as a manager, member, owner, consultant or agent of, any other Person or otherwise), engage in any business in competition with the Business as conducted on the Closing Date (a "Competing Business"); provided, however, that nothing in this Section 5.24(a) shall be deemed to limit in any way or preclude Seller or its Affiliates from owning securities of any entity engaged in any Competing Business which has outstanding publicly traded securities, so long as Person's direct holdings in any such entity shall not in the aggregate constitute more than five percent (5.0%) of the voting power of such entity. The Purchaser acknowledges and agrees that (i) the business and operations of VSE Aviation, Inc. and/or Wheeler Fleet Solutions, Co and their respective Subsidiaries, as currently operated as of the date of this Agreement, including the distribution of aircraft parts and maintenance repair and overhaul, are expressly excluded from the Business and not considered a Competing Business and (ii) that nothing in this Section 5.24(a) shall be deemed to limit in any way or preclude Seller or any of its Affiliates from (A) performing component and accessory maintenance repair and overhaul to support military aircraft customers that is similar in nature and scope to existing operations of Seller (which may include performing maintenance, repair and overhaul for products, parts or components for military aircraft that Seller or any of its Affiliates currently does not service as long as such maintenance, repair and overhaul is similar in nature to existing operations of Seller) and (B) distributing products, parts or components for military aircraft to the extent Seller currently engages in such distribution as part of its Retained Businesses (which may include distributing products, parts or components for military aircraft that Seller or any of its Affiliates does not currently distribute), and in each case, are expressly excluded from the Business and not considered a Competing Business; provided, that, notwithstanding the foregoing (i) and (ii), the parties hereto agree that platform level maintenance, repair and operations are specifically excluded from the foregoing (i) and (ii).

- (b) For a period of three (3) years from the Closing Date, Seller shall not, and shall cause its Affiliates not to, without the prior written consent of Purchaser, directly or indirectly (whether through an Affiliate in partnership or conjunction with or as a manager, member, owner, consultant or agent of, any other Person or otherwise), solicit or hire or employ or seek to entice away from Purchaser or the Transferred Entities for employment, any employee of Purchaser or its Transferred Entities who was an employee as of the Closing ("Covered Employee"); provided that (i) neither Seller nor any of its Affiliates will be deemed to have solicited any such Covered Employee who responds to any general media advertisement or job posting placed by or on behalf of Seller or any of its Affiliates prior to any direct or indirect solicitation by Seller or any of its Affiliates, and (ii) neither Seller nor any of its Affiliates may solicit any such Covered Employee whose employment has been terminated by Purchaser or its Subsidiaries not less than one (1) year prior to any direct or indirect solicitation by Seller or any of its Affiliates.
- (c) Each party hereto agrees that it will not, and will direct each of its Affiliates not to, make, publish, or communicate any disparaging or defamatory comments regarding any other party hereto or their respective Affiliates, or their respective current or former directors, officers, employees or owners. This Section shall not be violated by truthful statements in connection with litigation or other legal process, required governmental testimony or filings or administrative or arbitral proceedings (including oral questions or requests for information or documents in any proceeding, interrogatory, subpoena, civil investigative demand or similar process) or in connection with any stock exchange rules or financing reporting required pursuant to applicable Law.

- (d) In the event that any of the covenants contained in this Section 5.24 shall be determined by any court of competent jurisdiction to be unenforceable by reason of its extending for too long a period of time or over too large a geographical area or by reason of being too extensive in any other respect, the covenants contained in this Section 5.24 shall be interpreted to extend only over the longest period of time for which they may be enforceable, and/or over the largest geographical area as to which they may be enforceable and/or to the maximum extent in all other aspects as to which they may be enforceable, all as determined by such court in such action.
- Section 5.25 <u>Specified Matters</u>. Seller shall, and shall cause the Transferred Entities to, take the actions set forth in Section 3 of <u>Exhibit I</u>, subject to the terms and conditions set forth therein.

#### ARTICLE VI CERTAIN TAX MATTERS

## Section 6.1 <u>Certain Tax Returns</u>.

- (a) Subject to Section 6.9, Seller will prepare and timely file (or cause to be prepared and timely filed) all Tax Returns of a Transferred Entity for any taxable period ending on or prior to the Closing Date, regardless of when such Tax Returns are due. Any such Tax Return prepared pursuant to this Section 6.1(a) shall be prepared in a manner consistent with past practice (unless otherwise required by Law or otherwise pursuant to this Agreement). Such Tax Returns shall be submitted by Seller (together with schedules, statements and supporting documentation) to Purchaser at least thirty (30) days prior to the due date thereof (taking into account properly obtained extensions) of such Tax Returns (or such shorter period as reasonable under the circumstances) for Purchaser's review and comment, such reasonable comments to be incorporated into such Tax Returns as filed.
- (b) Subject to Section 6.9, Purchaser will prepare and timely file (or cause to be prepared and timely filed) all Tax Returns of a Transferred Entity for any Straddle Period. Any such Tax Returns prepared pursuant to this Section 6.1(b) shall be prepared in a manner consistent with past practice (unless otherwise required by Law or otherwise pursuant to this Agreement); provided, however, that each such Tax Return of VS2 shall, to the extent a Section 754 Election is not already in effect, reflect a Section 754 Election. Any such Tax Returns prepared pursuant to this Section 6.1(b), shall be submitted by Purchaser (together with schedules, statements and supporting documentation) to Seller at least thirty (30) days prior to the due date thereof (taking into account properly obtained extensions) of such Tax Returns (or such shorter period as reasonable under the circumstances) for Seller's review and comment, such reasonable comments to be incorporated into such Tax Returns as filed.
- Section 6.2 <u>Certain Income Tax Treatment and Apportionment of Taxes</u> (a) . In the case of Taxes (other than Transfer Taxes) that are payable with respect to any Straddle Tax Period, the portion of any such Tax that is attributable to the portion of the Straddle Tax Period ending on the Closing Date shall be: (i) in the case of income Taxes or any other Taxes resulting from, or imposed on, sales, receipts, uses, transfers or assignments of property or other assets, payments or accruals to another Person (including, without limitation, wages), or any other similar transaction or transactions, the amount that would be payable for the portion of the Straddle Tax Period ending on the Closing Date if the applicable Transferred Entity filed a separate Tax Return with respect to such Taxes solely for the portion of the Straddle Tax Period

ending on the Closing Date, assuming an interim closing of the books as of the close of business on the Closing Date (and for the avoidance of doubt, with respect to VS2, any such Taxes shall be determined based on a "closing of the books" method (as described in Treasury Regulations Section 1.706-4(a)(3)(iii)(on the basis of the calendar day conventions as described in Treasury Regulations Section 1.706-4(c)(1)(i))); and (ii) in the case of all other Taxes, an amount equal to the amount of Taxes for the entire Straddle Tax Period multiplied by a fraction the numerator of which is the number of calendar days in the portion of the period ending on the Closing Date and the denominator of which is the number of calendar days in the entire Straddle Tax Period. For purposes of clause (i), any item determined on an annual or periodic basis (including amortization and depreciation deductions and the effects of graduated rates) shall be allocated to the portion of the Straddle Tax Period ending on the Closing Date based on the relative number of days in such portion of the Straddle Tax Period as compared to the number of days in the entire Straddle Tax Period.

# Section 6.3 Cooperation and Exchange of Information.

- (a) From and after the Closing, each party to this Agreement shall, and shall cause its Affiliates to, provide to the other party to this Agreement such cooperation, documentation and information as either of them may reasonably request in (i) filing any Tax Return, amended Tax Return or claim for refund, (ii) determining a Liability for Taxes or a right to refund of Taxes, or (iii) conducting any Tax Proceeding. Such cooperation and information shall include providing copies of all relevant portions of relevant Tax Returns, together with all relevant portions of relevant accompanying schedules and relevant work papers, relevant documents relating to rulings or other determinations by Taxing Authorities and relevant records concerning the ownership and Tax basis of property and other information, which any such party hereto may possess. Notwithstanding anything to the contrary in this Agreement, such cooperation shall not include disclosure of, and neither Purchaser nor any of its Affiliates shall be entitled to any Tax Return (or copy thereof) of Seller or any of its respective Affiliates (other than relating to a Transferred Entity or the Business) or any Seller Consolidated Tax Return. Each party hereto shall make its employees reasonably available upon reasonable prior notice and on a mutually convenient basis at its cost to provide an explanation of any documents or information so provided.
- (b) Each party to this Agreement shall retain all Tax Returns, schedules and work papers, and all material records and other documents relating to Tax matters, of the relevant entities for their respective Tax periods ending on or prior to the Closing Date until the later of (i) the expiration of the statute of limitations for the Tax periods to which the Tax Returns and other documents relate or (ii) seven (7) years following the due date (without extension) for such Tax Returns. Thereafter, the party hereto holding such Tax Returns or other documents may dispose of them after offering the other party hereto reasonable notice and opportunity to take possession of such Tax Returns and other documents at such other party's own expense.

Section 6.4 <u>Transfer Taxes</u>. All Transfer Taxes arising from the Transaction shall be borne fifty percent (50%) by Seller and fifty percent (50%) by Purchaser. The party responsible under applicable Law for filing the Tax Returns with respect to any such Transfer Taxes shall prepare and timely file such Tax Returns and promptly provide a copy of such Tax Return to the other party. Seller and Purchaser shall, and shall cause their respective Affiliates to, cooperate to timely prepare and file any Tax Returns or other filings relating to such Transfer Taxes, including any claim for exemption or exclusion from the application or imposition of any Transfer Taxes. Notwithstanding the foregoing, Seller shall bear all Pre-Closing Reorganization

Transactions Transfer Taxes and shall timely prepare and file any Tax Returns or other documentation required with respect to such Pre-Closing Reorganization Transactions Transfer Taxes.

Section 6.5 <u>Tax Sharing Agreements</u>. On or before the Closing Date, the rights and obligations of the Transferred Entities pursuant to all Tax sharing agreements or arrangements (other than this Agreement), if any, to which any of the Transferred Entities, on the one hand, and Seller or any of its Affiliates, on the other hand, are parties, shall terminate, and neither Seller nor any of its Affiliates, on the one hand, nor any of the Transferred Entities, on the other hand, shall have any rights or obligations to each other after the Closing in respect of such agreements or arrangements.

#### Section 6.6 <u>Certain Tax Proceedings</u>.

- (a) Purchaser shall notify Seller within ten (10) Business Days of receiving notice of a Tax Proceeding relating to a Pre-Closing Tax Period with respect to the Transferred Entities; provided that a failure by Purchaser to give such notice shall not impair or diminish Purchaser's rights and remedies hereunder except to the extent that such failure has actually and materially prejudiced Seller's rights with respect to such Tax Proceeding. With respect to any such Tax Proceeding relating solely to a Pre-Closing Tax Period, Seller may choose in its sole discretion (at its expense) to control any such proceeding, subject to Purchaser's right to participate (at its expense and by employing counsel of its choosing) in such proceeding and Seller shall not compromise or settle any such proceeding without first obtaining the prior written consent of Purchaser, which consent shall not be unreasonably withheld, conditioned or delayed.
- (b) Except for any Tax Proceedings, Seller chooses to control pursuant to <u>Section 6.6(a)</u>, Purchaser shall control all Tax Proceedings with respect to the Transferred Entities.
- Section 6.7 <u>Purchaser Tax Acts.</u> Without the prior written consent of Seller (not to be unreasonably withheld, conditioned or delayed), Purchaser shall not (a) make, and shall cause its Affiliates (including the Transferred Entities) not to make, any election with respect to any Transferred Entity (including any entity classification election pursuant to Treasury Regulations Section 301.7701-3 and any election under section 338 of the Code), or change any method of Tax accounting or any Tax accounting period of any Transferred Entity, which election or change would be effective for any Pre-Closing Tax Period, (b) except as otherwise provided in <u>Section 6.1(b)</u>, voluntarily approach a Tax authority in respect of any Pre-Closing Tax Period, or (c) except as otherwise provided in <u>Section 6.1(b)</u>, file or amend any Tax Return of any Transferred Entity for any Pre-Closing Tax Period.

Section 6.8 <u>Tax Refunds</u>. Seller will be entitled to payment for the amount of any Tax refunds that are received from a Governmental Entity in cash (or tax credits that are actually utilized) by Purchaser or the Transferred Entities after the Closing Date in respect of Taxes paid by any Transferred Entity on or prior to the Closing Date, or paid on or prior to the Closing Date in respect of any assets of a Transferred Entity; provided that (a) the amount to be paid to Seller will be reduced by any net Taxes payable with respect to such refund (or credit) and by any third-party costs or expenses associated with obtaining such refund (or credit) and (b) no amount will be required to be paid to Seller to the extent it arises from a carryback of a Tax attribute generated in any period (or portion thereof) beginning after the Closing Date. Subject to the

foregoing, Purchaser will (or will cause the Transferred Entity to) forward the amount of any such refunds or credits by wire transfer of immediately available funds to Seller within ten (10) days of receipt or realization thereof. In the event that Purchaser or any Transferred Entity after the Closing is required to pay to a Governmental Entity all or any portion of a refund that was received and paid over to Seller pursuant to this Section 6.8, Seller shall pay back to Purchaser or the applicable Transferred Entity such amount (plus any penalties, interest or other charges imposed by the relevant Governmental Entity) promptly upon receiving notice from Purchaser.

Section 6.9 <u>Seller Consolidated Taxes; Seller Consolidated Tax Returns.</u> Notwithstanding anything in this Agreement to the contrary, (a) in no event will Purchaser or any Affiliate of Purchaser (including, after the Closing, a Transferred Entity) have any rights or access to any Seller Consolidated Tax Return, (b) Seller and its Affiliates will have the exclusive right to control the preparation and filing of such Seller Consolidated Tax Returns and (c) Seller and its Affiliates will have the exclusive right of control over any Tax Proceeding relating to any Seller Consolidated Taxes or Seller Consolidated Tax Return and make all decisions taken in connection therewith in its sole discretion.

Section 6.10 Pre-Closing Tax Indemnity. From and after the Closing, Seller shall, indemnify each Purchaser Indemnified Person from and against any and all Losses incurred or sustained by, or imposed upon, any Purchaser Indemnified Person based upon, arising out of, or with respect to (i) any Taxes imposed on or with respect to the Transferred Entities or the Business for any and all Pre-Closing Tax Periods; (ii) any Taxes imposed on the Transferred Entities as a result of any of the Transferred Entities being a member of a combined, consolidated or unitary group prior to the Closing or Taxes imposed on the Transferred Entities as a transferee, successor, by Contract or pursuant to any Law which Taxes relate to an event or transaction occurring prior to the Closing; (iii) the Transfer Taxes (if any) for which Seller is liable pursuant to Section 6.4; (iv) any payments required to be made after the Closing Date under any Tax sharing, Tax indemnity, Tax allocation or similar Contracts to which any Transferred Entity was obligated, or was a party, in each case prior to the Closing; (v) any Taxes of the Seller arising as a result of the Pre-Closing Reorganization Transactions; (vi) any Taxes of Seller arising as a result of the sale of the Transferred Entity Interests or the sale of the Transferred Equity Interests; and (vii) any Taxes attributable to income inclusion by Purchaser or any of its Affiliates (including any Transferred Entity after the Closing Date) from a partnership or other pass-through entity for the taxable year of the Transferred Entity that includes the Closing Date to the extent such Taxes are attributable to income of such partnership or other pass-through entity for any period or portion of a period occurring on or before the Closing Date. The obligations of Seller pursuant to this Section 6.10 shall survive until the earlier of (x) sixty (60) days following the expiration of the applicable statute of limitations or (y) six (6) years after the Closing Date (and, if a claim is made hereunder wi

#### ARTICLE VII CONDITIONS PRECEDENT

Section 7.1 <u>Conditions to Each Party's Obligations to Close</u>. The respective obligations of Seller and Purchaser to effect the Closing are subject to the satisfaction or (to the extent permitted by Law) waiver by Seller and Purchaser at or prior to the Closing of the following conditions:

- (a) Facility Clearance. DCSA shall have issued a Facility Clearance, or an interim Facility Clearance, to the Company;
- (b) No Injunctions, Proceedings or Restraints. No Judgment issued by any Governmental Entity of competent jurisdiction shall have been entered and remain in effect which prevents the consummation of the Closing and no Proceedings by any Governmental Entity seeking to restrain or enjoin the consummation of the Transaction shall be ongoing.
- (c) Pre-Closing Reorganization Closing Condition Steps. The Pre-Closing Reorganization Closing Condition Steps shall have occurred except that notwithstanding anything to the contrary, the recognition of the transfer and assignment by novation of the Assigned Contracts shall not be a condition to Closing.
- Section 7.2 <u>Conditions to Obligations of Purchaser to Close.</u> The obligation of Purchaser to effect the Closing is subject to the satisfaction (or waiver by Purchaser) at or prior to the Closing of the following additional conditions:
- (a) Representations and Warranties. The representations and warranties of Seller contained in Article III of this Agreement (other than those contained in the Seller Fundamental Representations) shall be true and correct as of the Closing Date as if made on and as of the Closing Date, except (i) that representations and warranties that are made as of a specific date shall be true and correct only on and as of such date and (ii) where the failure of such representations and warranties to be true and correct (without giving regard to any materiality or "Business Material Adverse Effect" qualifications set forth therein) would not reasonably be expected to have, individually or in the aggregate, a Business Material Adverse Effect. The Seller Fundamental Representations shall be true and correct in all respects as of the Closing Date as if made on and as of the Closing Date (or, in the case of representations and warranties that are made as of a specific date, as of such date);
- (b) Performance of Obligations of Seller. The covenants and agreements of Seller to be performed on or before the Closing Date in accordance with this Agreement shall have been performed in all material respects;
- (c) No Business Material Adverse Effect. Since the date of the Agreement, there shall not have occurred a Business Material Adverse Effect;
- (d) Officer's Certificate. Purchaser shall have received a certificate, dated as of the Closing Date and signed on behalf of Seller by an executive officer of Seller, stating that the conditions specified in Section 7.2(a), Section 7.2(b) and Section 7.2(c) have been satisfied; and
- (e) Closing Deliverables. Purchaser shall have received each of the deliverables required to be delivered by Seller or the Company pursuant to Section 2.5(c).
- Section 7.3 <u>Conditions to Obligations of Seller to Close</u>. The obligation of Seller to effect the Closing is subject to the satisfaction (or waiver by Seller) at or prior to the Closing of the following additional conditions:
- (a) Representations and Warranties. The representations and warranties of Purchaser contained in <u>Article IV</u> of this Agreement (other than the Purchaser Fundamental

Representations) shall be true and correct as of the Closing Date as if made on and as of the Closing Date, except (i) in that representations and warranties that are made as of a specific date shall be true and correct only on and as of such date and (ii) where the failure of such representations and warranties to be true and correct (without giving regard to any materiality or "Purchaser Material Adverse Effect" qualifications set forth therein) would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect. The Purchaser Fundamental Representations shall be true and correct in all respects as of the Closing Date as if made on and as of the Closing Date (or, in the case of representations and warranties that are made as of a specific date, as of such date);

- (b) Performance of Obligations of Purchaser. The covenants and agreements of Purchaser to be performed on or before the Closing Date in accordance with this Agreement shall have been performed in all material respects;
- (c) Officer's Certificate. Seller shall have received a certificate, dated as of the Closing Date and signed on behalf of Purchaser by an executive officer of Purchaser, stating that the conditions specified in <u>Section 7.3(a)</u> and <u>Section 7.3(b)</u> have been satisfied; and
- (d) Closing Deliverables. Seller shall have received each of the deliverables required to be delivered by Purchaser pursuant to Section 2.5(b).
- Section 7.4 <u>Frustration of Closing Conditions</u>. Neither Purchaser nor Seller may rely as a basis for terminating this Agreement on the failure of any condition set forth in this Article <u>VII</u> to be satisfied if such failure was caused by such party's failure to act in good faith or to use the efforts to cause the Closing to occur as required by this Agreement, including <u>Section 5.1</u>.

### ARTICLE VIII TERMINATION; EFFECT OF TERMINATION

- Section 8.1 <u>Termination</u>. Notwithstanding anything to the contrary in this Agreement, this Agreement may be terminated and the Transaction and the other transactions contemplated by this Agreement abandoned at any time prior to the Closing:
  - (a) by mutual written consent of Seller and Purchaser;
- (b) by Seller upon written notice to Purchaser, if Purchaser is in breach of any of its representations and warranties contained in <u>Article IV</u> of this Agreement or Purchaser has breached or failed to perform in any material respect any of its covenants or other agreements contained in this Agreement, and such failure or breach would give rise to the failure of a condition set forth in <u>Section 7.3(a)</u> or Section 7.3(b) and cannot be or has not been cured by the earlier of (i) the date that is thirty (30) days after the date that Seller has notified Purchaser (or Purchaser has notified Seller) in writing of such failure or breach and (ii) the Outside Date; provided, that Seller shall not be entitled to terminate this Agreement pursuant to this Section 8.1(b) if Seller or the Company (as applicable) is also in breach of any of its representations, warranties, covenants or agreements contained in this Agreement, which has prevented or would prevent the satisfaction of the conditions set forth in <u>Section 7.2(a)</u> or <u>Section 7.2(b)</u>;
- (c) by Purchaser upon written notice to Seller, if Seller is in breach of any of its representations and warranties contained in Article III of this Agreement or Seller has breached or failed to perform in any material respect any of its covenants or other agreements

contained in this Agreement, and such failure or breach would give rise to the failure of a condition set forth in Section 7.2(a) or Section 7.2(b) and cannot be or has not been cured by the earlier of (i) the date that is thirty (30) days after the date that Purchaser has notified Seller (or Seller has notified Purchaser) in writing of such failure or breach and (ii) the Outside Date; provided, that Purchaser shall not be entitled to terminate this Agreement pursuant to this Section 8.1(c) if Purchaser is also in breach of any of its representations, warranties, covenants or agreements contained in this Agreement, which has prevented or would prevent the satisfaction of the conditions set forth in Section 7.3(a) or Section 7.3(b);

- (d) by Seller or by Purchaser upon written notice to the other, if the Closing shall not have occurred on or prior to May 1, 2024; provided, that if any of the conditions set forth in Section 7.1 has not been satisfied by such date, either Seller or Purchaser may extend such date to August 1, 2024 (such date, as may be so extended, the "Outside Date"); provided, further, that the right to terminate this Agreement under this Section 8.1(d) shall not be available to (i) any party whose failure to perform any covenant or obligation under this Agreement has been the primary cause of, or primarily resulted in, the failure of the Closing to occur on or before such date or (ii) any party during the pendency of any Proceeding brought by the other party for specific performance of this Agreement; or
- (e) by Seller or by Purchaser upon written notice to the other, if a permanent injunction or other permanent Judgment issued by a court of competent jurisdiction in the United States preventing the consummation of the Transaction shall have become final and nonappealable; provided, that the party seeking to terminate this Agreement pursuant to this <u>Section 8.1(e)</u> shall have used commercially reasonable efforts to prevent the entry of such permanent injunction or other permanent Judgment, or to prevent such notification, as applicable, in each case, to the extent required by Section 5.1.; provided, further, that the foregoing notwithstanding, nothing in this Agreement shall require any party to this Agreement to institute a Proceeding against any Governmental Entity or any other Person.
- Section 8.2 <u>Effect of Termination</u>. If this Agreement is terminated and the Transaction is abandoned as described in <u>Section 8.1</u>, this Agreement shall become null and void and of no further force and effect, except that the provisions of <u>Section 5.3</u>, <u>Section 8.1</u>, this <u>Section 8.2</u>, <u>Section 8.3</u>, <u>Article IX</u> and <u>Article X</u> will survive the termination hereof. Nothing in this <u>Section 8.2</u> shall be deemed to release any party from any liability for Fraud or willful breach by such party of the terms and provisions of this Agreement.
- Section 8.3 <u>Notice of Termination</u>. In the event of termination by Seller or Purchaser pursuant to <u>Section 8.1</u>, written notice of such termination shall be given by the terminating party to the other party to this Agreement.

# ARTICLE IX INDEMNIFICATION

# Section 9.1 Survival of Representations, Warranties, Covenants and Agreements.

(a) The representations and warranties of Seller (other than the Seller Fundamental Representations) contained in this Agreement will survive until the date that is twelve (12) months after the Closing Date; provided, that, the representation in Section 3.16 shall survive the Closing until the earlier of (i) sixty (60) days following the expiration of the applicable statute of limitations or (ii) six (6) years after the Closing Date. The Seller

Fundamental Representations will survive until the date that is six (6) years after the Closing Date.

- (b) The representations and warranties of Purchaser contained in this Agreement will survive until the date that is twelve (12) months after the Closing Date. The Purchaser Fundamental Representations will survive until the date that is six (6) years after the Closing Date.
- (c) All covenants and agreements contained herein shall survive the Closing until sixty (60) calendar days following the expiration of the period in which such covenants and agreements are to be performed in accordance with their terms.
- Section 9.2 <u>Exclusive Remedy</u>. The parties hereby agree that, following the Closing, absent Fraud, the sole and exclusive remedy of a party for any breach or inaccuracy of any representation, warranty, covenant or agreement contained in this Agreement shall be the applicable rights set forth in this <u>Article IX</u>. In furtherance of the foregoing, each party hereby waives, from and after Closing, to the fullest extent permitted under Law, any and all rights, claims and causes of action for any breach of any representation, warranty, covenant or agreement set forth herein or otherwise relating to the subject matter of this Agreement it may have against the other party hereto and its Affiliates and each of their respective Representatives arising under or based upon any Law, except pursuant to the indemnification provisions set forth in this <u>Section 9.2</u>. Nothing in this <u>Section 9.2</u> shall limit any Person's right to seek and obtain any equitable relief to which any Person shall be entitled pursuant to Section <u>10.6</u> or to seek any remedy on account of Fraud or limit Purchaser's ability to pursue recovery under the R&W Insurance Policy. Furthermore, for the avoidance of doubt, nothing in this <u>Section 9.2</u> shall limit the rights and remedies of any Purchaser Indemnified Person pursuant to <u>Section 6.10</u>.
- Section 9.3 <u>Indemnification by Seller.</u> Subject to the other terms and conditions of this Section 9.3, from and after Closing, Seller shall indemnify and hold harmless the Purchaser and its Affiliates (including, after the Closing, the Transferred Entities) and its and their respective directors, officers, employees, agents and Representatives (each, a "<u>Purchaser Indemnified Person</u>") from and against any and all Losses incurred or sustained by, or imposed upon, any Purchaser Indemnified Person based upon, arising out of, or with respect to (a) any inaccuracy or breach of any of the representations and warranties set forth in <u>Article III</u>, (b) any inaccuracy in or breach of any of the Seller Fundamental Representations, (c) any breach or non-fulfillment of any covenant, agreement or obligation to be performed by Seller after the Closing pursuant to this Agreement, (d) the Specified Matters set forth on Section 2 of <u>Exhibit I</u>, and (e) the matters set forth on Section 1 of <u>Exhibit J</u>. For purposes of determining (i) whether any inaccuracy or breach in any representation or warranty of Seller has occurred, and (ii) the amount of Losses resulting from any such inaccuracy or breach, the determination shall, in each case, be made without references to the terms "materially," "Business Material Adverse Effect," "material adverse effect" or similar qualifications as to materiality in any such representation or warranty.
- Section 9.4 <u>Indemnification by Purchaser</u>. Subject to the other terms and conditions of this Section 9.4, from and after Closing, Purchaser shall indemnify and hold harmless the Seller and its Affiliates and its and their respective directors, officers, employees, agents and Representatives (each, a "<u>Seller Indemnified Person</u>") from and against any and all Losses incurred or sustained by, or imposed upon, any Seller Indemnified Person based upon, arising out of, or with respect to (a) any inaccuracy or breach of any of the representations and warranties

set forth in Article IV, (b)any inaccuracy in or breach of any of the Purchaser Fundamental Representations or (c) any breach or non-fulfillment of any covenant, agreement or obligation to be performed by Purchaser after the Closing pursuant to this Agreement.

# Section 9.5 <u>Indemnification Procedures</u>.

- (a) A Person that may be entitled to be indemnified under this Agreement (the "Indemnified Party") shall promptly notify the party or parties hereto liable for such indemnification (the "Indemnifying Party") in writing of any pending or threatened claim or demand that the Indemnified Party has determined gives or would reasonably be expected to give rise to a right of indemnification under this Agreement (including a pending or threatened claim or demand asserted by a third party against the Indemnified Party, such claim being a "Third-Party Claim"), describing in reasonable detail the facts and circumstances with respect to the subject matter of such claim and or demand and indicating the amount (estimated, if necessary) and, if applicable, will send a copy of such written notice of a Third-Party Claim to the Escrow Agent during the period for which the Escrow Agreement remains in effect; provided, however, that the delay or failure to provide such notice shall not release the Indemnifying Party from any of its obligations under this Section 9.5(a) except to the extent that the Indemnifying Party is actually and materially prejudiced by such delay or failure, it being agreed that notices for claims in respect of a breach of a representation, warranty, covenant or agreement must be delivered prior to the expiration of any applicable survival period specified in Section 9.1 for such representation, warranty, covenant or agreement.
- Upon receipt of a notice of a Third-Party Claim for indemnity from an Indemnified Party pursuant to Section 9.3 or Section 9.4 the Indemnifying Party will be entitled, by notice to the Indemnified Party delivered within fifteen (15) Business Days of the receipt of notice of such Third-Party Claim (or sooner if the notice of the Third-Party Claim so requires), to assume the defense and control of such Third-Party Claim (at the expense of such Indemnifying Party); provided that the Indemnifying Party shall allow the Indemnified Party a reasonable opportunity to participate in the defense of such Third-Party Claim with its own counsel and at its own expense. If the Indemnifying Party does not assume the defense and control of any Third-Party Claim pursuant to this Section 9.5(b), the Indemnified Party shall be entitled to assume and control such defense, but the Indemnifying Party may nonetheless participate in the defense of such Third-Party Claim with its own counsel and at its own expense. If the Indemnifying Party assumes the defense and control of a Third-Party Claim, then the Indemnifying Party shall select counsel, contractors and consultants of recognized standing and competence and shall use commercially reasonable efforts in the defense or settlement of such Third-Party Claim. Purchaser and Seller, as the case may be, shall, and shall cause each of their Affiliates and Representatives to, reasonably cooperate with the Indemnifying Party in the defense of any Third-Party Claim, including by furnishing books and records, personnel and witnesses, as reasonably required for any defense of such Third-Party Claim. If the Indemnifying Party has assumed the defense and control of a Third-Party Claim, it shall be authorized to consent to a settlement of, or the entry of any judgment arising from, any Third-Party Claim, in its sole discretion and without the consent of any Indemnified Party; provided that such settlement or judgment does not (A) impose any equitable or other non-monetary remedies or obligations on the Indemnified Party but involves solely the payment of money damages for which the Indemnified Party will be indemnified in full hereunder, (B) involve a finding or admission of wrongdoing or misconduct or any violation of Law or any violation of the rights of any Person by the Indemnified Party, and the Indemnifying Party shall (I) pay or cause to be paid all amounts in such settlement or judgment and (II) obtain, as a condition of any

settlement or other resolution, a complete and unconditional release of the Indemnified Parties potentially affected by such Third-Party Claim. No Indemnified Party will consent to the entry of any judgment or enter into any settlement or compromise with respect to a Third-Party Claim without the prior written consent of the Indemnifying Party, with such consent not to be unreasonably withheld, conditioned or delayed; provided that, notwithstanding the foregoing, the Indemnified Party shall have the right to pay or settle any such claim without the Indemnifying Party's consent if it irrevocably waives in a writing delivered to the Indemnifying Party any right to indemnity therefor under this Agreement.

- Section 9.6 <u>Limitation of Liability</u>. The indemnification provided for in <u>Section 9.3</u> and <u>Section 9.4</u> shall be subject to the following limitations:
- (a) The Seller Indemnifying Person shall be liable to the Purchaser Indemnified Person for indemnification under <u>Section 9.3(a)</u> for an aggregate amount of Losses up to the Deductible Escrow Amount, in which event the Purchaser Indemnified Person shall assume liability for any such Losses in connection with a claim under <u>Section 9.3(a)</u> in excess of the Deductible Escrow Amount (subject to Purchaser's right to recover under the R&W Insurance Policy); <u>provided</u>, that no Loss may be claimed by a Purchaser Indemnified Party under <u>Section 9.3(a)</u>, or shall be reimbursable by Seller, or shall be included in calculating the aggregate Losses, until such Losses resulting from a claim or claims arising out of related facts exceed \$10,000. Notwithstanding anything to the contrary, the limitations set forth in this <u>Section 9.6(a)</u> shall not apply to (i) any Losses based upon, arising out of, or otherwise in respect of any breach of any Seller Fundamental Representations or (ii) instances of Fraud.
- (b) The Seller Indemnifying Person shall be liable to the Purchaser Indemnified Person for indemnification for Losses under Section 9.3(d) subject to the limitations set forth in Section 4 of Exhibit I.
- (c) Except in the case of Fraud, the Sellers will not have any Liability for any Losses in excess of: (i) in the case of <u>Section 9.3(a)</u>, the Deductible Escrow Amount, (ii) in the case of <u>Section 9.3(b)</u> and <u>Section 6.10</u> (Pre-Closing Tax Indemnity), the amount of the Purchase Price, (iii) in the case of <u>Section 9.3(c)</u>, the amount of the Purchase Price and (iv) in the case of <u>Section 9.3(d)</u>, the amount of the cap set forth in Section 4 of <u>Exhibit I</u>.
- (d) Payments by an Indemnifying Party pursuant to Section 9.3 or <u>Section 9.4</u> in respect of any Loss shall be limited to the amount of any liability or damage that remains after deducting therefrom any insurance proceeds and any indemnity, contribution or other similar payment received or reasonably expected to be received by the Indemnified Party in respect of any such claim. The Indemnified Party shall use its reasonable efforts to recover under insurance policies or indemnity, contribution or other similar agreements for any Losses prior to seeking indemnification under this Agreement.
- (e) In no event shall any Indemnifying Party be liable to any Indemnified Party for any punitive damages, except to the extent awarded to a third party.
- (f) No Indemnified Party shall be entitled to recover from an Indemnifying Party more than once in respect of Losses resulting from the same individual claim or series of related claims.

- (g) To the extent required by applicable Law, each Indemnified Party shall take, and cause its Affiliates to take, all reasonable steps to mitigate any Loss upon becoming aware of any event or circumstance that would be reasonably expected to, or does, give rise thereto, including incurring costs only to the minimum extent necessary to remedy the breach that gives rise to such Loss.
- (h) All indemnification payments made under this Agreement shall be treated by the parties as an adjustment to the Purchase Price for Tax purposes, unless otherwise required by Law.
- Section 9.7 <u>R&W Insurance Policy</u>. Notwithstanding the foregoing, to the extent coverage for a claim under <u>Section 9.3</u> is available under the R&W Insurance Policy (and not subject to a retention or exclusion and the policy limits under such R&W Insurance Policy have not been exhausted), before proceeding against Seller pursuant to <u>Section 9.3</u>, Purchaser shall use commercially reasonable efforts (which shall not include an obligation to litigate) to obtain payment or recovery of Losses for any such claim from the insurer under the R&W Insurance Policy; <u>provided, however</u>, that Purchaser may provide a notice of such claim to Seller at any time prior to, concurrently with or after such efforts to recover under the R&W Insurance Policy. Within five (5) Business Days of the Closing Date, Seller will deliver to Purchaser (or to recipients identified by Purchaser) two (2) or more USB electronic storage devices containing a complete and accurate (as of the date of delivery) electronic copy of the electronic data room hosted by Datasite with regards to the Transaction.
- Section 9.8 <u>Escrow Release</u>. No later than three (3) Business Days following the Escrow Release Date, the Purchaser and Seller shall execute and deliver a joint written instruction to the Escrow Agent to release the Deductible Escrow Amount, except for the aggregate amount of all unsatisfied claims for indemnification that a Purchaser Indemnified Person has made on or before such date and which are subject to satisfaction (in whole or in part) from the Deductible Escrow Amount. Any remaining funds in the escrow account holding the Deductible Escrow Amount following the Escrow Release Date in respect of any such pending claim will be released and paid, upon receipt of joint written instruction from the Purchaser and the Seller to the Escrow Agent, as soon as practicable following the full and final resolution of such pending claim. Each of the Purchaser and the Seller shall from time to time submit joint written instructions to the Escrow Agent instructing the Escrow Agent to distribute such portion of the Deductible Escrow Amount in accordance with this Section 9.8 and the Escrow Agreement.

# ARTICLE X GENERAL PROVISIONS

Section 10.1 Entire Agreement. This Agreement and the other Transaction Documents, and the Schedules and Exhibits hereto and thereto, and the Confidentiality Agreement, along with the Seller Disclosure Schedules, constitute the entire agreement and understanding among the parties hereto with respect to the subject matter hereof and thereof and supersede all prior agreements and understandings, whether written or oral, relating to such subject matter. Neither party shall be liable or bound to the other party in any manner by any representations, warranties or covenants relating to such subject matter, except as specifically set forth herein and therein. In the event of a conflict between the terms of this Agreement and the terms of any Transaction Document, the terms of this Agreement shall control.

Section 10.2 <u>Assignment</u>. Neither this Agreement nor any of the rights and obligations hereunder may be assigned or transferred by either party (whether by operation of Law or otherwise) without the prior written consent of the other party hereto. Any attempted assignment in violation of this <u>Section 10.2</u> shall be void. Subject to the two (2) preceding sentences, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and assigns.

Section 10.3 <u>Amendments and Waivers</u>. This Agreement may not be amended, except by an instrument in writing signed on behalf of each of the parties hereto. By an instrument in writing, Purchaser, on the one hand, or Seller, on the other hand, may waive compliance by the other with any term or provision of this Agreement that the other party was or is obligated to comply with or perform. Such waiver or failure to insist on strict compliance with such term or provision shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure of compliance.

Section 10.4 No Third-Party Beneficiaries. Except for the present and former directors and officers of the Transferred Entities solely with respect to Section 5.18, this Agreement, together with the other Transaction Documents and the Exhibits and Schedules hereto and thereto are not intended to confer in or on behalf of any Person not a party to this Agreement (and their successors and assigns) any rights, benefits, causes of action or remedies with respect to the subject matter or any provision hereof.

Section 10.5 <u>Notices</u>. All notices and other communications to be given to any party hereunder shall be sufficiently given for all purposes hereunder if in writing and delivered by hand, courier, or delivered by electronic mail (which is confirmed) or sent by overnight courier (providing proof of delivery), and shall be directed to the addresses set forth below (or at such other address as such party shall designate by like notice):

# (i) if to Purchaser,

400 Convention Street, Suite 1010
Baton Rouge, Louisiana 70802
Attention: Thomas Henley
Jeffrey Koonce
Lucie R. Kantrow
Email: Thomas@bernhardcapital.com
Koonce@bernhardcapital.com
Lucie@bernhardcapital.com

with a copy (which shall not constitute notice) to:

Hogan Lovells US LLP 555 Thirteenth Street NW Washington, DC 20004 Attention: William J. Curtin, III Email: william.curtin@hoganlovells.com Fax: (202) 637-5910

and

and

Hogan Lovells US LLP 8350 Broad Street, 17th Floor Tysons, VA 22102 Attention: Ashlee Sawyer Gilson Email: ashlee.gilson@hoganlovells.com

Fax: (703) 610-6200

#### (ii) if to Seller,

VSE Corporation 6348 Walker Lane Alexandria, VA 22310 Attention: Chief Executive Officer and SVP/Chief Legal Officer Email: JACuomo@VSECorp.com FTehrani@VSECorp.com

with a copy (which shall not constitute notice) to:

Jones Day 600 Brickell Avenue Miami, FL 33131

Attention: Lorne S. Cantor; Luis A. Velez

Email: lcantor@jonesday.com;lvelez@jonesday.com

Section 10.6 Specific Performance. The parties hereto agree that irreparable damage, for which monetary damages (even if available) would not be an adequate remedy, would occur in the event that the parties hereto do not perform any provision of this Agreement in accordance with its specified terms or otherwise breach such provisions. Accordingly, the parties acknowledge and agree that the parties shall be entitled to seek an injunction, specific performance and other equitable relief to prevent breaches or threatened breaches of this Agreement and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which such party is entitled in law or in equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that the other party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity. Any party seeking an injunction or injunctions to prevent breaches or threatened breaches of this Agreement or to enforce specifically the terms and provisions of this Agreement shall not be required to provide any bond or other security in connection with such remedy. The foregoing is in addition to any other remedy to which any party is entitled at law, in equity or otherwise.

Section 10.7 Governing Law and Jurisdiction. This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. In addition, each of the parties hereto (a) submits to the personal jurisdiction of the Delaware Court of Chancery, or in the event (but only in the event) that such Delaware, or in the event (but only in the event) that such United States District Court also does not have jurisdiction over such dispute, any Delaware State court, in the event any dispute (whether in contract, tort or otherwise) arises out of this Agreement or the

Transaction or the other transactions contemplated hereby, (b) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court and (c) agrees that it will not bring any Proceeding relating to this Agreement or the Transaction or the other transactions contemplated hereby in any court other than the Delaware Court of Chancery, or in the event (but only in the event) that such Delaware Court of Chancery does not have subject matter jurisdiction over such action, the United States District Court for the District of Delaware, or in the event (but only in the event) that such United States District Court also does not have jurisdiction over such Proceeding, any Delaware State court. Each party hereto agrees that service of process upon such party in any such Proceeding shall be effective if notice is given in accordance with Section 10.5.

Section 10.8 <u>Waiver of Jury Trial</u>. EACH PARTY TO THIS AGREEMENT WAIVES TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY ANY OF THEM AGAINST THE OTHER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS AGREEMENT, OR ANY OTHER AGREEMENTS EXECUTED IN CONNECTION HEREWITH OR THE ADMINISTRATION THEREOF OR THE TRANSACTION OR ANY OF THE OTHER TRANSACTIONS CONTEMPLATED HEREIN OR THEREIN. NO PARTY TO THIS AGREEMENT SHALL SEEK A JURY TRIAL IN ANY LAWSUIT, PROCEEDING, COUNTERCLAIM OR ANY OTHER LITIGATION PROCEDURE BASED UPON, OR ARISING OUT OF, THIS AGREEMENT OR ANY RELATED INSTRUMENTS. NO PARTY HERETO WILL SEEK TO CONSOLIDATE ANY SUCH ACTION IN WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. EACH PARTY TO THIS AGREEMENT CERTIFIES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT OR INSTRUMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS SET FORTH ABOVE IN THIS SECTION 10.8. NO PARTY HAS IN ANY WAY AGREED WITH OR REPRESENTED TO ANY OTHER PARTY THAT THE PROVISIONS OF THIS <u>SECTION 10.8</u> WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.

Section 10.9 <u>Severability</u>. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other competent authority to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party hereto. Upon such a determination, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the Transaction and the other transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

Section 10.10 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, all of which shall be considered an original, with the same effect as if the signatures thereto and hereto were upon the same instrument, and shall become effective when one or more such counterparts have been signed by each party and delivered (by facsimile, email or otherwise) to the other party. Signatures to this Agreement transmitted by facsimile transmission, by electronic mail in "portable document format" from, or by any other electronic means intended to preserve the original graphic and pictorial appearance of a document, will have the same effect as physical delivery of the paper document bearing the original signatures.

This Agreement has been executed in the English language. If this Agreement is translated into another language, the English language text shall in any event prevail.

Section 10.11 <u>Expenses</u>. Whether or not the Closing takes place, and except as set forth otherwise in this Agreement, all costs and expenses incurred in connection with this Agreement, the Transaction and the other transactions contemplated hereby shall be paid by the party incurring such expense; provided, that, the parties agree that each of Seller, on the one hand, and Purchaser, on the other hand shall each bear (a) fifty percent (50%) of the costs of the R&W Insurance Policy in accordance with <u>Section 5.16</u> and (b) fifty percent (50%) of the costs of obtaining the Specified Reports.

Section 10.12 Interpretation; Absence of Presumption. It is understood and agreed that the specification of any dollar amount in the representations and warranties or covenants and agreements contained in this Agreement or the inclusion of any specific item in the Seller Disclosure Schedules is not intended to imply that such amounts or higher or lower amounts, or the items so included or other items, are or are not material, and no party shall use the fact of the setting of such amounts or the fact of the inclusion of any such item in the Seller Disclosure Schedules in any dispute or controversy between the parties as to whether any obligation, item or matter not described in this Agreement or included or not included in the Seller Disclosure Schedules is or is not material for purposes of this Agreement. Nothing herein (for the avoidance of doubt, including the Seller Disclosure Schedules) shall be deemed an admission by either party or any of its Affiliates, in any Proceeding, that such party or any such Affiliate, or any third party, is or is not in breach or violation of, or in default in, the performance or observance of any term or provisions of any Contract or any Law. For the purposes of this Agreement, (a) words in the singular shall be held to include the plural and vice versa, and words of one gender shall be held to include the other gender as the context requires; (b) references to the terms Article, Section, paragraph, Exhibit and Schedule are references to the Articles, Sections, paragraphs, Exhibits and Schedules to this Agreement unless otherwise specified; (c) the terms "hereof," "herein," "hereby," "hereto," and derivative or similar words refer to this entire Agreement, including the Schedules and Exhibits hereto and the words "date hereof" refer to the date of this Agreement; (d) references to "Dollars" or "\$" shall mean U.S. dollars; (e) the word "including" and words of similar import when used in this Agreement and the Transaction Documents shall mean "including, without limitation," unless otherwise specified; (f) the word "or" shall not be exclusive; (g) references to "written" or "in writing" include in electronic form; (h) the headings contained in this Agreement and the other Transaction Documents are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement and the other Transaction Documents; (i) Seller and Purchaser have each participated in the negotiation and drafting of this Agreement and the other Transaction Documents and if an ambiguity or question of interpretation should arise, this Agreement and the other Transaction Documents shall be construed as if drafted jointly by the parties thereto and no presumption or burden of proof shall arise favoring or burdening either party by virtue of the authorship of any of the provisions in this Agreement or the other Transaction Documents; (j) a reference to any Person includes such Person's successors and permitted assigns; (k) any reference to "days" means calendar days unless Business Days are expressly specified; (1) when calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded and if the last day of such period is not a Business Day, the period shall end on the next succeeding Business Day; (m) "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if"; (n) any reference to a Law means such Law as amended from time to time and includes any successor

legislation thereto and any rules and regulations promulgated thereunder; (o) any reference to a Contract means such Contract as amended, supplemented and modified from time to time to the extent permitted by the provisions thereof; and (p) "ordinary course of business" means, with respect to any Person or business, the ordinary course of business consistent with the applicable Person's or business's past custom and practice. The phrase "delivered" or "made available" shall mean that the information referred to has been physically or electronically delivered to the relevant parties hereto or their representatives, including, in the case of "made available" to Purchaser, material that has been posted to the electronic data room hosted by Datasite, in each case, prior to 10 a.m. Eastern Daylight Time on April 30.

#### Section 10.13 Waiver of Conflicts Regarding Representation; Nonassertion of Attorney-Client Privilege.

- (a) Purchaser waives and will not assert, and agrees to cause its Affiliates, including, following the Closing, the Transferred Entities, to waive and not assert, any conflict of interest arising out of or relating to the representation, after the Closing (the "Post-Closing Representation"), of Seller or any of its Affiliates, or any shareholder, officer, employee or director of Seller or any of its Affiliates (any such Person, a "Designated Person"), in any matter involving this Agreement, the other Transaction Documents or any other agreements or transactions contemplated hereby or thereby, by Jones Day (such representation, the "Current Representation").
- (b) Purchaser waives and will not assert, and agrees to cause its Affiliates, including, following the Closing, the Transferred Entities, to waive and not assert, any attorney-client or other applicable legal privilege or protection with respect to any communication between Jones Day and any Designated Person (including with respect to any communication occurring at or prior to the Closing, the Business) occurring during the Current Representation (the "Privileged Communications") or in connection with any Post-Closing Representation, including in connection with a dispute with Purchaser or its Affiliates (including, following the Closing, any Transferred Entity or any of its Subsidiaries), it being the intention of the parties hereto that all such rights to such attorney-client and other applicable legal privilege or protection and to control such attorney-client and other applicable legal privilege or protection shall be retained by Seller and its Affiliates and that Seller, and not Purchaser or its Affiliates or the Transferred Entities, shall have the sole right to decide whether or not to waive any attorney-client or other applicable legal privilege or protection. Accordingly, from and after Closing, none of Purchaser or its Affiliates, including the Transferred Entities, shall have any access to any such communications or to the files of the Current Representation, all of which shall be and remain the property of Seller and not of Purchaser or its Affiliates, including the Transferred Entities, or to internal counsel relating to such engagement, and none of Purchaser or its Affiliates, including, following the Closing, the Transferred Entities, or any Person acting or purporting to act on their behalf shall seek to obtain the same by any process on the grounds that the privilege and protection attaching to such communications and files belongs to Purchaser or its Affiliates, including, following the Closing, the Transferred Entities, or does not belong to Seller. Notwithstanding the foregoing, after the Closing, (i) none of Seller nor any of its Affiliates shall waive such privilege other than in connection with the enforcement or defense of its rights or obligations under this Agreement and the other Transaction Documents and (ii) in the event of a dispute between Purchaser, the Company or any of their respective Affiliates, on the one hand, and a Person other than the Seller or its Affiliates, on the other hand, the Company may assert the attorney-client privilege to prevent the disclosure of such communications and files by Jones Day, the Seller or their Affiliates to such Person.

Section 10.14 <u>Disclosure Schedules</u>. The Seller Disclosure Schedules and all schedules attached thereto, and all Exhibits attached to this Agreement shall be construed with and as an integral part of this Agreement to the same extent as if the same had been set forth verbatim herein. Any capitalized terms used in any Exhibit or in the Seller Disclosure Schedules but not otherwise defined therein shall be defined as set forth in this Agreement. Any information, item or other disclosure set forth in any Section of the Seller Disclosure Schedules, as the case may be, shall be deemed to be disclosed with respect to any other Section of this Agreement (or to have been set forth in any other Section of the Seller Disclosure Schedules, as the case may be), if the relevance of such disclosure to such other Section is reasonably apparent on the face of such disclosure notwithstanding the omission of a reference or a cross-reference with respect thereto and notwithstanding any reference to a Section of the Seller Disclosure Schedules, as applicable, in such Section of this Agreement.

Section 10.15 <u>Non-Recourse</u>. Notwithstanding anything herein to the contrary, this Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement or the transactions contemplated by this Agreement, may only be brought against, the entities that are expressly named as parties hereto and then only with respect to the specific obligations set forth herein with respect to such party. Except to the extent a party hereto (and then only to the extent of the specific obligations undertaken by such named party in this Agreement), no Purchaser Indemnified Person or any of their respective other past, present or future director, officer, employee, incorporator, member, partner, stockholder, agent, attorney, advisor or Representative or Affiliate shall have any liability (whether in contract, tort, equity or otherwise) for any one or more of the representations, warranties, covenants, agreements or other obligations or liabilities of Seller or Purchaser under this Agreement or of or for any claim based on, arising out of, or related to this Agreement or the transactions contemplated hereby.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, Seller, Company and Purchaser have duly executed this Agreement as of the date first written above.

# VSE CORPORATION

By: /s/ John A. Cuomo Name: John A. Cuomo Title: President and CEO

HURRICANE FDS COMPANY, LLC

By: /s/ John A. Cuomo Name: John A. Cuomo Title: President and CEO

ASG OPERATIONS, LLC

By: /s/ Thomas Henley
Name: Thomas Henley

Title: Authorized Representative

[Signature Page to Membership Interest Purchase Agreement]

CERTAIN INFORMATION CONTAINED IN THIS AGREEMENT AND PLAN OF MERGER HAS, PURSUANT TO ITEM 601(B)(2) OF REGULATION S-K, BEEN OMITTED BY MEANS OF REDACTING A PORTION OF THE TEXT AND REPLACING IT WITH [\*\*\*] BECAUSE SUCH INFORMATION IS BOTH NOT MATERIAL AND IS THE TYPE OF INFORMATION THAT THE REGISTRANT TREATS AS PRIVATE.

**EXECUTION VERSION** 

# AGREEMENT AND PLAN OF MERGER

BY AND AMONG

VSE AVIATION, INC.,

DIAMOND MERGER SUB 1, LLC,

DESSER-GRAHAM PARTNERSHIP, L.P.

AND

DESSER HOLDINGS PARTNERSHIP GP, LLC, AS THE REPRESENTATIVE

DATED AS OF

May 3, 2023

# TABLE OF CONTENTS

Page

ART	TICLE 1. THE MERGER; EFFECTS OF THE MERGER	2
1.1	The Merger	2
1.2	Effective Time	2
1.3	Effects of the Merger.	2
1.4	Treatment of Options.	3
1.5	Merger Consideration; Calculation and Payment of Estimated Closing Merger Consideration.	3
1.6	Representative.	6
1.7	Withholding	8
ART	TICLE 2. CLOSING	8
2.1	Time and Place of Closing	8
2.2	Deliveries by the Company and the Representative	9
2.3	Deliveries by Parent	10
ART	TICLE 3. REPRESENTATIONS AND WARRANTIES REGARDING THE ACQUIRED COMPANIES	10
3.1	Organization and Good Standing; Holding Company.	11
3.2	No Conflicts.	11
3.3	Capitalization; Investments and Subsidiaries.	12
3.4	Power and Authorization.	13
3.5	Financial Statements	13
3.6	Real Property.	14
3.7	Personal Property	15
3.8	Taxes.	15
3.9	Litigation	17
3.10	Labor Matters.	18
3.11	Intellectual Property Rights and Data Protection.	19
3.12	Contracts and Commitments	21
3.13	Recent Events	23
3.14	Employee Benefit Plans.	24
3.15	Compliance with Laws.	27
3.16	Environmental Compliance	27
3.17	Customers and Suppliers.	28
3.18	Transactions with Affiliates	29
3.19	Insurance	29
3.20	Brokers	30
3.21	Condition and Sufficiency of Assets	30
3.22		30
3.23	Bank Accounts	30
3.24	Government Contracts	30
3.25	Inventory; Accounts Receivable	31
3.26		31
3.27		32
3.28		32
3.29	Funded Indebtedness; Seller Expenses	33

# TABLE OF CONTENTS (continued)

Page

64 64

3.30 CARES Act and COVID Relief Programs.	33
3.31 Disclaimer of Warranties	33
ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	33
4.1 Organization and Good Standing.	33
4.2 Power and Authorization	34
4.3 No Conflicts	34
4.4 Litigation	34
4.5 Solvency	34
4.6 Brokers	35
4.7 Investigation and Evaluation; Disclaimer Regarding Projections.	35
4.8 Sufficiency of Funds	35
4.9 NO ADDITIONAL REPRESENTATIONS.	36
ARTICLE 5. CERTAIN OBLIGATIONS OF THE PARTIES	36
5.1 Public Announcements	36
5.2 Indemnification and Insurance.	37
5.3 Certain Tax Matters.	38
5.4 R&W Insurance Policy	42
5.5 Employee Matters.	42
5.6 Data Room.	42
5.7 Release of Claims	43
5.8 Regulatory Matters.	43
5.9 Interim Operations of the Business	44
5.10 Reasonable Access; Information Technology	45
5.11 Exclusivity	45
5.12 Parachute Payments	45
5.13 Obligations Related to Employee Plan Matters	46
5.14 Notice of Certain Events.	46
ARTICLE 6. Conditions Precedent	46
6.1 Conditions to Obligations of the Company and Representative	46
6.2 Conditions to Obligations of Parent and Merger Sub	47
6.3 Conditions to Obligations of All Parties	48
ARTICLE 7. TERMINATION	48
7.1 Termination	48
7.2 Effect of Termination.	49
ARTICLE 8. NO SURVIVAL; EXCLUSIVE REMEDY; NONRECOURSE	49
8.1 No Survival	49
3.2 EXCLUSIVE REMEDY; NONRECOURSE.	50
3.3 Materiality Scrape	50
ARTICLE 9. DEFINITIONS	50
ARTICLE 10. MISCELLANEOUS	63
10.1 Construction.	63

10.2 Exhibits and Schedules

10.3 Further Assurances

# TABLE OF CONTENTS (continued)

Page

10.4	Costs and Expenses	64
10.5	Notices	64
10.6	Assignment and Benefit.	65
10.7	Governing Law; Jurisdiction	65
10.8	Amendment, Modification and Waiver	66
10.9	Section Headings and Defined Terms	66
10.10	Severability	66
10.11	Counterparts	66
10.12	Entire Agreement	66
10.13	Retention of Counsel	67
10.14	Specific Performance	67

# EXHIBITS1

E 1 1 2 A	A : 10 :
Exhibit A:	Acquired Companies
Exhibit B:	Form of Certificate of Merger
Exhibit C:	Form of Estimated Statement
Exhibit D:	Form of Company FIRPTA Certificate
Exhibit E:	Form of Aero-Cee Bailey's Holding Company FIRPTA Certificate
Exhibit F:	Form of Amendments to Employee Plans
Exhibit G:	Forms of Assignment
Exhibit H:	Forms of Terminations
Exhibit I:	Accounting Principles and Net Working Capital Example
Exhibit J:	Form of Earn Out Agreement
Exhibit K:	Form of Escrow Agreement

Appendix I: Allocation of Merger Consideration

Schedules and exhibits have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Company will furnish supplementally a copy of any omitted schedules or exhibits to the Securities and Exchange Commission upon request.

#### AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (this "<u>Agreement</u>"), dated as of May 3, 2023, is entered into by and among VSE Aviation, Inc., a Florida corporation ("<u>Parent</u>"), Diamond Merger Sub 1, LLC, a Delaware limited liability company and a wholly-owned subsidiary of Parent ("<u>Merger Sub</u>"), Desser-Graham Partnership, L.P., a Delaware limited partnership (the "<u>Company</u>"), and Desser Holdings Partnership GP, LLC, a Delaware limited liability company, in its capacity as representative of the Former Equityholders (the "<u>Representative</u>"). Parent, Merger Sub, the Company and the Representative are sometimes referred to in this Agreement individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>."

#### BACKGROUND

WHEREAS, the Company owns beneficially and of record all of the issued and outstanding Equity Interests of Desser Holding Company, LLC, a Delaware limited liability company ("<u>Desser</u>"), and Desser owns, directly or indirectly, all of the issued and outstanding Equity Interests of the Persons identified on <u>Exhibit A</u> hereto (collectively with Desser and the Company, the "<u>Acquired Companies</u>" and each, an "<u>Acquired Company</u>");

WHEREAS, the Parties intend that at the Effective Time Merger Sub be merged with and into the Company (the "Merger"), with the Company surviving the Merger and continuing as a wholly-owned subsidiary of Parent, all on the terms and subject to the conditions set forth herein;

WHEREAS, pursuant to the Agreement of Limited Partnership of the Company (as amended, the "<u>LP Agreement</u>"), Desser Holdings Partnership GP, LLC, a Delaware limited liability company and the general partner of the Company (the "<u>General Partner</u>"), has (a) determined that it is in the best interests of the Company and the limited partners of the Company (collectively, the "<u>Limited Partners</u>"), and declared it advisable, to enter into this Agreement and (b) authorized and approved the execution, delivery and performance by the Company of this Agreement and the consummation of the transactions contemplated hereby, including the Merger;

WHEREAS, the General Partner and its Affiliates collectively hold a majority of each of the Original Partner Interests and Preferred Interests;

WHEREAS, the Majority Preferred Holders (as such term is defined in the LP Agreement) have approved the Company's entry into this Agreement and the Merger;

WHEREAS, the applicable governing bodies of each of Merger Sub and Parent have each unanimously (a) determined it is advisable and in the best interests of Merger Sub and Parent, respectively, to enter into this Agreement and (b) authorized and approved the execution, delivery and performance by Merger Sub and Parent of this Agreement and the transactions contemplated hereby, including the Merger;

WHEREAS, each of the Limited Partners and the General Partner has executed and delivered a letter of transmittal on or prior to the date hereof agreeing to certain obligations in connection with the transactions contemplated herein;

WHEREAS, the Company, the Representative and Loar Group, Inc., ("Loar") will enter into the Earn Out Agreement simultaneously with the effectiveness of this Agreement; and

WHEREAS, the Parties desire to make certain representations, warranties, covenants and agreements in connection with the Merger and the transactions contemplated by this Agreement.

NOW, THEREFORE, in consideration of the premises and the representations, warranties, covenants and agreements herein contained, the Parties, all intending to be legally bound, hereby agree as follows:

#### ARTICLE 1. THE MERGER; EFFECTS OF THE MERGER

- 1.1 <u>The Merger.</u> On the terms and subject to the conditions of this Agreement, and in accordance with the Delaware Revised Uniform Limited Partnership Act (as amended from time to time, the "<u>DLPA</u>"), at the Effective Time (a) Merger Sub shall be merged with and into the Company and (b) the separate legal existence of Merger Sub shall cease and the Company shall continue its existence under the DLPA as the surviving entity (sometimes referred to herein as the "Surviving Company").
- 1.2 <u>Effective Time</u>. Subject to the provisions of this Agreement, at the Closing, the Parties will cause a certificate of merger, in the form attached hereto as <u>Exhibit B</u> (the "<u>Certificate of Merger</u>"), to be executed, acknowledged and filed with the Secretary of State of the State of Delaware in accordance with the applicable provisions of the DLPA. The Merger will become effective at the time and date of the filing of the Certificate of Merger or at such later time and date as may be agreed by the Parties in writing and specified in the Certificate of Merger in accordance with the applicable provisions of the DLPA (the effective time of the Merger is hereinafter referred to as the "<u>Effective Time</u>").

## 1.3 Effects of the Merger.

- (a) The Merger shall have the effects set forth herein and in the applicable provisions of the DLPA. Without limiting the generality of the foregoing, and subject thereto, from and after the Effective Time, all property, rights, privileges, immunities, powers, franchises, licenses and authority of the Company and Merger Sub shall vest in the Surviving Company, and all debts, liabilities, obligations, restrictions and duties of the Company and Merger Sub shall become the debts, liabilities, obligations, restrictions and duties of the Surviving Company.
- (b) At the Effective Time, as a result of the Merger and without any action on the part of Parent, Merger Sub, the Company or the holders of (i) the issued and outstanding capital stock of Merger Sub ("Merger Sub Capital Stock") or (ii) the issued and outstanding Company Equity Interests, said securities shall be converted, or cancelled and retired, as follows:
- (i) <u>Merger Sub Capital Stock</u>. Each share of Merger Sub Capital Stock that is issued and outstanding immediately prior to the Effective Time shall be converted on a one-for-one basis into Equity Interests of the Surviving Company.
- (ii) <u>Company Equity Interests</u>. As of the Effective Time, by virtue of the Merger and without any action on the part of any holder of Company Equity Interests or any Party, each issued and outstanding Company Equity Interest immediately prior to the Effective Time shall be converted into the right of the holders thereof to receive (x) such Person's Closing Proceeds, payable in cash, without interest thereon, and (y) such Person's Ownership Percentage of any Distribution, payable in cash, without interest thereon, in accordance with the terms and conditions of this Agreement.
- (c) At and as of the Effective Time, the Company Certificate shall be the certificate of limited partnership of the Surviving Company until amended in accordance with its terms and applicable Law.
- (d) At and as of the Effective Time, the LP Agreement as in effect immediately before the Effective Time shall be terminated and of no further force and effect and replaced by the Amended and Restated LP Agreement.
- (e) At and as of the Effective Time, an Affiliate of Parent shall be admitted as the general partner of the Surviving Company and shall succeed to the interests, rights and obligations of the General Partner, including the General Partner Interest.
- (f) From and after the Effective Time, until their successors are duly elected or appointed and qualified in accordance with the limited partnership agreement of the Surviving Company and applicable Law, the officers of Merger Sub immediately before the Effective Time shall comprise all of the officers of the Surviving Company.

- 1.4 <u>Treatment of Options.</u> As of the Effective Time and without any further action on the part of any Party or any holder of Unit Options, the Unit Option Plan and each Unit Option outstanding immediately prior to the Effective Time shall be cancelled and terminated and, in consideration for the termination and cancellation of such Unit Options, the Option Holder thereof shall be entitled to receive from the Surviving Company, less applicable withholding taxes (including any UK employee National Insurance Contributions and other applicable social security contributions), (x) such Person's Closing Proceeds (which shall be calculated net of the aggregate exercise price of the Unit Options held by such Option Holder) and (y) with respect to any Distribution, an amount equal to such Person's Ownership Percentage of such Distribution (the amounts in clause (x) and (y), the "Option Payments"). The Company shall take all requisite actions to ensure the actions provided for in this Section 1.4 are completed.
  - 1.5 Merger Consideration; Calculation and Payment of Estimated Closing Merger Consideration.
- (a) <u>Merger Consideration</u>. "<u>Merger Consideration</u>" shall mean a dollar amount equal to (i) \$124,000,000 (the "<u>Base Merger Consideration</u>"), <u>plus</u> (ii) the Net Working Capital Adjustment (which may be a negative number), <u>minus</u> (iii) the amount of Closing Date Funded Indebtedness, <u>minus</u> (iv) the amount of any Seller Expenses unpaid as of the Closing, <u>plus</u> (v) the aggregate amount of Closing Date Cash and Cash Equivalents, and <u>plus</u> (vi) the Aggregate Option Exercise Price (clauses (i) (vi) collectively, the "<u>Closing Merger Consideration</u>"), and <u>plus</u> (vii) the amounts, if any, payable pursuant to the Earn Out Agreement (such amounts, the "<u>Earn Out Payments</u>").
- (b) <u>Estimated Closing Merger Consideration</u>. No later than two (2) Business Days prior to the Closing Date, the Company and the Representative shall deliver to Parent a calculation of the Estimated Closing Merger Consideration in the form set forth on <u>Exhibit C</u>. The "<u>Estimated Closing Merger Consideration</u>" shall be a good faith estimate of the Closing Merger Consideration and its components, as determined by the Company and the Representative (subject to adjustment as provided below). In connection with determining the Estimated Closing Merger Consideration, the Company and the Representative shall estimate in good faith (i) the Net Working Capital Adjustment, (ii) the amount of Closing Date Funded Indebtedness (such amount the "<u>Estimated Closing Date Funded Indebtedness</u>"), (iii) the amount of Closing Date Cash and Cash Equivalents, (iv) the amount of Seller Expenses unpaid as of the Closing, (v) a schedule of inventory of the Acquired Companies as of the Measurement Time and (vi) the Aggregate Option Exercise Price. The Estimated Closing Merger Consideration and the components thereof shall be calculated in accordance with the requirements of this <u>Section 1.5</u> and, where applicable, the Accounting Principles.
- (c) <u>Payment of Estimated Closing Merger Consideration; Closing Date Payments</u>. On the Closing Date, contemporaneously with the Effective Time, Parent shall pay, or shall cause Merger Sub or the Surviving Company to pay, the following amounts:
- (i) (A) an amount of cash equal to \$2,000,000 (the "Merger Consideration Adjustment Escrow Funds") shall be deposited into an escrow account (the "Merger Consideration Adjustment Escrow Account"), (B) an amount of cash equal to \$350,000 shall be deposited into an escrow account (the "UK Exchange Tax Escrow Account"), (C) an amount of cash equal to \$1,000,000 shall be deposited into an escrow account (the "UK Withholding Tax Escrow Account", and together with the UK Exchange Tax Escrow Account, the "UK Tax Escrow Account"), each of which shall be established pursuant to the Escrow Agreement;
- (ii) an amount of cash equal to the Expense Funds shall be paid to the Representative by wire transfer of immediately available funds to an account as directed by the Representative in writing to Parent no later than two (2) Business Days prior to the Closing Date;
- (iii) an amount of cash equal to the Estimated Closing Date Funded Indebtedness in respect of the Funded Indebtedness set forth on Schedule 1.5(c)(iii) shall be paid to the holders of such Estimated Closing Date Funded Indebtedness on behalf of the applicable Acquired

Company by wire transfer of immediately available funds to such account or accounts as directed in the applicable Payoff Letter;

- (iv) any Seller Expenses unpaid as of the Closing set forth on <u>Schedule 1.5(c)(iv)</u> shall be paid to each of the payees thereof on behalf of the Company or the Former Equityholders (the Estimated Closing Merger Consideration minus the amounts set forth in clauses (i)-(iv), the "<u>Aggregate Closing Proceeds</u>"); and
  - (v) to each holder of Company Equity Interests and each Option Holder, their Closing Proceeds.
  - (d) Preparation of the Final Statement of Closing Merger Consideration.
- (i) As soon as practicable, but no later than ninety (90) days after the Closing Date, Parent shall prepare and deliver to the Representative (A) a proposed calculation of the Net Working Capital as of the Measurement Time (the "Proposed Closing Date Net Working Capital"), (B) a proposed calculation of the Closing Date Funded Indebtedness (the "Proposed Closing Date Indebtedness"), (C) a proposed calculation of the Closing Date Cash and Cash Equivalents (the "Proposed Closing Date Cash and Cash Equivalents"), (D) a proposed calculation of the Seller Expenses unpaid as of the Closing (the "Proposed Seller Expenses"), (E) a proposed calculation of the Aggregate Option Exercise Price (the "Proposed Aggregate Option Exercise Price"), (F) a schedule of inventory of the Acquired Companies as of the Measurement Time (the "Proposed Inventory") and (G) a proposed calculation of the resulting Closing Merger Consideration (the "Proposed Merger Consideration Calculation") based on the Proposed Closing Date Net Working Capital, the Proposed Closing Date Indebtedness, the Proposed Closing Date Cash and Cash Equivalents, the Proposed Seller Expenses, the Proposed Closing Date Cash and Cash Equivalents, the Proposed Inventory and the Proposed Merger Consideration Calculation shall collectively be referred to herein from time to time as the "Proposed Closing Date Calculations." The Proposed Closing Date Calculations shall be calculated in good faith in accordance with the requirements of this Section 1.5 and, where applicable, the Accounting Principles and shall include supporting schedules and other reasonable supporting documentation.
- (ii) The Representative may dispute the Proposed Closing Date Calculations by delivering a written notice of dispute (a "Merger Consideration Dispute Notice") to Parent within thirty (30) days after receiving the Proposed Closing Date Calculations. During the thirty (30)-day period following the Representative's receipt of the Proposed Closing Date Calculations, the Representative and its accountants (which may be the Company's current auditors) shall, at the Representative's expense, be permitted reasonable access to review the working papers of Parent and Parent's accountant relating to the Proposed Closing Date Calculations; provided, however, that (A) the provision of any information or access pursuant to this Section 1.5(d)(ii) will be subject to appropriate confidentiality undertakings and, if applicable, execution of customary release letters in favor of the accountant as requested by the accountant in connection with the sharing of work papers and (B) nothing in this Section 1.5(d) will require any Party to disclose information that is subject to attorney-client privilege. Parent and the Representative will use commercially reasonable efforts to resolve any such dispute during the thirty (30)-day period commencing on the date Parent receives the Merger Consideration Dispute Notice from the Representative. If the Representative and Parent do not obtain a final resolution within such thirty (30)-day period, then the items in dispute shall be submitted promptly to BDO USA, LLP or, if BDO USA, LLP declines to serve, such other independent accounting firm of national standing mutually agreed to by Parent and the Representative (the "Accounting Firm"). The Accounting Firm shall be instructed to render a determination of the applicable dispute and the resulting final Closing Merger Consideration within thirty (30) days (or such other period of time as may be required by the Accounting Firm) after engagement of the Accounting Firm with respect to the dispute, which determination must be in writing and must set forth, in reasonable detail, the basis therefor. The determination of the Accounting Firm shall be conclusive and binding upon the Representative, Parent and the other parties hereto and judgment may be entered upon the determination of the Accounting Firm in any court having jurisdiction over the party against which such determination is to be enforced.

The Accounting Firm shall determine, based solely on written submissions by Parent and the Representative and their respective representatives, and not by independent review, only those items disputed in the Merger Consideration Dispute Notice that have not been agreed upon in writing by the parties. In resolving any disputed item, (A) the Accounting Firm shall be bound by the Accounting Principles, the applicable definitions set forth in this Agreement and the other requirements of this Section 1.5 and shall not assign a value to any item greater than the greatest value for such item claimed by either party or less than the smallest value for such item claimed by either party, and (B) the Accounting Firm shall not be bound by the amount of Taxes reflected on a Parent Tax Return filed after the Closing Date to the extent, if any, that such amount has been reasonably disputed by the Representative under Section 5.4(d)(ii).

The "Final Statement of Closing Merger Consideration" shall mean (A) if a Merger Consideration Dispute Notice is not delivered within the time period required by this Section 1.5(d)(ii), the amount of the Proposed Merger Consideration Calculation, (B) the amount agreed as the final Closing Merger Consideration at any time in writing by Parent and the Representative, or (C) the final Closing Merger Consideration as set forth in the written determination of the Accounting Firm made in accordance with the provisions of this Section 1.5. If Parent does not deliver the Proposed Closing Date Calculation within the time period required by Section 1.5(d)(i), the Representative may submit in writing to Parent its own Proposed Closing Date Calculation within thirty (30) days of the expiration of the deadline for Parent's delivery of the Proposed Closing Date Calculation (whereupon Parent and the Representative shall follow the requirements of this Section 1.5(d)(ii) with Parent being substituted for the Representative, mutatis mutandis, including with Parent having a thirty (30)-day period following the Representative's delivery of the Proposed Closing Date Calculations to deliver a Merger Consideration Dispute Notice).

(iii) In the event the Representative and Parent submit any unresolved objections to the Accounting Firm for resolution as provided in Section 1.5(d)(ii) above, the fees, costs and expenses of the Accounting Firm (A) shall be paid by Parent in the proportion that the aggregate dollar amount of such disputed items so submitted that are successfully disputed by the Representative (as finally determined by the Accounting Firm) bears to the aggregate dollar amount of such items so submitted and (B) shall be paid, first, from the Merger Consideration Adjustment Escrow Funds, and second (if the Merger Consideration Adjustment Escrow Funds are not sufficient for the payment of such fees, costs and expenses), by the Representative, in the proportion that the aggregate dollar amount of such disputed items so submitted that are unsuccessfully disputed by the Representative (as finally determined by the Accounting Firm) bears to the aggregate dollar amount of such items so submitted.

# (e) Adjustment to Estimated Merger Consideration.

(i) If the Actual Adjustment is a positive amount, then within three (3) Business Days after the date on which the Closing Merger Consideration is finally determined pursuant to Section 1.5(d), then (x) the Surviving Company shall pay, or cause to be paid, to the Former Equityholders (other than the Option Holders) and to the Surviving Company or one of its Subsidiaries on behalf of the Option Holders (in each case pro rata in accordance with their Ownership Percentages) an amount equal to the Actual Adjustment and (y) Parent and the Representative shall promptly deliver joint written instructions to the Escrow Agent instructing the Escrow Agent to deliver to the Former Equityholders or to the Surviving Company on behalf of the former Option Holders (in each case pro rata in accordance with their Ownership Percentages) an amount equal to the amount then in the Merger Consideration Adjustment Escrow Account.

(ii) If the Actual Adjustment is a negative amount, then within three (3) Business Days after the date on which the Closing Merger Consideration is finally determined pursuant to Section 1.5(d) above, then Parent and the Representative shall promptly deliver joint written instructions to the Escrow Agent instructing the Escrow Agent to deliver (x) to the Surviving Company or another Person designated in writing by Parent an amount equal to the Actual Adjustment; provided, however, that if the Actual Adjustment exceeds the amount then in the Merger Consideration Adjustment Escrow Account then the Representative shall pay to the Surviving Company or another Person designated in writing by Parent such excess amount and (y) the remaining amount in the Merger Consideration Adjustment Escrow Account, if any, to the Former Equityholders or to the Surviving

Company on behalf of the former Option Holders (in each case pro rata in accordance with their Ownership Percentages).

## 1.6 Representative.

- (a) Parent shall be entitled to deal exclusively with and rely upon the Representative on all matters relating to this Agreement, the Escrow Agreement, the Earn Out Agreement and the other Transaction Documents and shall be entitled to rely conclusively (without further evidence of any kind whatsoever) on any document executed or purported to be executed by the Representative on behalf of any Former Equityholder (collectively, the "Represented Parties"), and on any other action taken or purported to be taken by the Representative on behalf of any of the Represented Parties by the Representative, as fully binding upon the Represented Parties. Parent and its Affiliates shall be relieved from any liability to any Person for any acts done by them in accordance with any decision, act, consent or instruction of the Representative.
- (b) Parent hereby waives any claims it may have or assert, including those that may arise in the future, against the Representative for any action or inaction taken or not taken by the Representative in connection with such entity's capacity as the Representative, except to the extent that such action or inaction shall have been held by a court of competent jurisdiction to constitute bad faith or willful misconduct.
- (c) The Represented Parties and their respective successors shall be irrevocably bound by any and all actions taken by the Representative under or otherwise relating to this Agreement, the Escrow Agreement and any other Transaction Document, and the transactions contemplated hereby and thereby as if such actions were expressly ratified and confirmed by each of them. The approval of this Agreement and the transactions contemplated hereby by the requisite Represented Parties necessary to approve such matters shall constitute the consent and agreement of each Represented Party to the appointment and authority of the Representative to act on their behalf pursuant to this Agreement, the Escrow Agreement and the other Transaction Documents.
- (d) The Representative shall not be entitled to any fee, commission or other compensation for the performance of its services hereunder, but shall be entitled to the payment of all its out-of-pocket costs and expenses incurred as the Representative. In connection with the foregoing, \$250,000 (the "Expense Funds") of the Estimated Closing Merger Consideration shall be paid at the Closing to an account designated by the Representative pursuant to Section 1.5(c)(ii) to be used by the Representative to pay the costs and expenses incurred by the Representative in its capacity as the Representative. The Parties agree that, for all Tax purposes, the Expense Funds shall be treated as having been received and voluntarily set aside by the Former Equityholders and the Option Holders at the time of Closing. In addition, to the extent that the Expense Funds are at any time insufficient (as determined by the Representative in its sole discretion) to cover all of the costs and expenses incurred by the Representative in its capacity as the Representative or amounts due hereunder, then the Representative may, at its option: (i) retain such portion of the Merger Consideration Adjustment Escrow Funds (when and to the extent that the Merger Consideration Adjustment Escrow Funds are released to the Representative (on behalf of the Former Equityholders) in accordance with this Agreement and the Escrow Agreement) as determined by the Representative in its sole discretion for purposes of reimbursement of such costs and expenses; (ii) retain such amount of the proceeds received by the Former Equityholders after the Closing Date under any term or provision of this Agreement; or (iii) seek reimbursement of such costs and expenses directly from the Former Equityholders. Once the Representative determines, in its sole discretion, that the Representative will not incur any additional expenses in its capacity as the Representative, then the Representative will distribute the remaining unused Expense Funds (if any) pro rata to the Former Equi
- (e) For purposes of exercising Representative's rights or satisfying the Representative's obligations under this Agreement and the Earn Out Agreement, or as necessary for any reasonable business purpose, including financial reporting and accounting matters, the preparation and filing of any Tax Return, the defense of any Tax Claim or assessment, Parent shall retain and provide the

Representative and its representatives with access, at the Representative's sole expense, upon prior reasonable written request specifying the need therefor, during regular business hours, to (i) the officers and employees of any Acquired Company and (ii) the books of account and records of any Acquired Company, but, in each case, only to the extent relating to the assets, liabilities or business of any Acquired Company prior to the Effective Time, and the Representative and its representatives shall have the right to make copies of such books and records, at the sole cost and expense of the Representative; provided, however, the foregoing right of access shall not be exercisable in such a manner as to interfere unreasonably with the normal operations and business of the applicable Acquired Company; and provided, further, as to so much of such information as constitutes trade secrets or confidential business information of any Acquired Company, the Representative and its representatives will enter into a confidentiality agreement reasonably acceptable to Parent and use due care to not disclose such information except (1) with the prior written consent of Parent, (2) where such information becomes available to the public generally, through sources other than the Representative and its representatives or (3) for a disclosure that is required by Law or a securities exchange or in connection with a filing by the Representative under federal or state securities Laws or is reasonably believed to be so required in which case, the Representative shall notify Parent prior to such disclosure and allow Parent the ability to seek a protective order. Nothing in this Section 1.6(e) will require Parent or the Surviving Company to disclose information that is subject to attorney-client privilege. The Surviving Company may nevertheless destroy the financial books and records contemplated by this Section 1.6(e) on or after the five (5)-year anniversary of the Closing Date if Parent sends to the Representative written notice of its intent to destroy such records; provided that any records relevant to the performance or enforcement of obligations under the Earn Out Agreement shall not be destroyed until the completion or expiration of all such obligations; provided further that no such notice shall be required after the six (6)-year anniversary of the Closing Date. Such records may then be destroyed after the 60th day after such notice is given unless the Representative objects to the destruction, in which case the Surviving Company shall deliver such records to the Representative, at the sole cost and expense of the Representative.

Withholding. Parent, the Company, the Surviving Company, the Representative, and each of their respective Affiliates and agents shall be entitled to deduct and withhold (or cause to be deducted and withheld) from any amount payable pursuant to this Agreement and any other Transaction Document such Taxes as are required to be deducted and withheld under applicable Law; provided that, except with respect to any amounts paid to or for the benefit of any Person pursuant to the terms of this Agreement that constitute wages or compensation subject to employment or withholding Tax, so long as a Former Equityholder provides the IRS Form W-9 required under Section 2.2 of this Agreement, no Taxes shall be deducted or withheld from any amounts payable to such Former Equityholder. To the extent that any amounts are ultimately required to be deducted or withheld under this <u>Section 1.7</u> from amounts payable by Parent or any of its Affiliates after the Closing Date (other than amounts required to be deducted or withheld pursuant to the next sentence of this <u>Section 1.7</u> and other than amounts required to be deducted or withheld pursuant to Section 1446(f) of the Code), (a) Parent shall provide the Representative with written notice of any such required deduction or withholding no later than five (5) days prior to the date on which the relevant payment is required to be made hereunder, and (b) Parent shall cooperate with the Representative, as reasonably requested, to reduce or eliminate any withholding that otherwise would be required. Notwithstanding anything to the contrary in this Agreement, any amounts paid to or for the benefit of any Person pursuant to the terms of this Agreement that constitute wages or compensation subject to employment or withholding Tax may be paid to the Acquired Company that employs such Person, which Acquired Company will in turn pay the applicable Person such amounts through its payroll system no later than the end of the next regularly scheduled pay period (less applicable withholding, payroll and similar Taxes, which will be deposited with the appropriate Governmental Entity in accordance with applicable Law). Amounts withheld pursuant to this Section 1.7 and timely paid over to the appropriate Tax authority shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made. Parent shall furnish to the Representative a copy of the receipt issued by such Tax authority, if any, or otherwise such other

documentation reasonably satisfactory to the Representative, evidencing such payment, in each case, as soon as reasonably practicable.

## ARTICLE 2. CLOSING

- 2.1 <u>Time and Place of Closing</u>. The closing of the transactions pursuant to this Agreement (the "<u>Closing</u>") shall take place remotely on the third Business Day following the satisfaction or waiver of each of the conditions set forth in <u>Article 6</u> (other than those conditions that are to be satisfied at the Closing), or on such other date or at such other time and place as Parent and Representative mutually agree in writing, via the exchange of electronic copies of documents and signatures, and the confirmation of other matters necessary to consummate the Closing, by the Parties (the "<u>Closing</u> <u>Date</u>").
- 2.2 <u>Deliveries by the Company and the Representative</u>. At or prior to the Closing, the Company and the Representative delivered or caused to be delivered to Parent the following items:
  - (a) a copy of the Certificate of Merger, duly executed by the Company;
  - (b) the Escrow Agreement, duly executed by the Representative and the Escrow Agent;
- (c) (i) a properly completed and duly executed IRS Form W-9 from each of the Former Equityholders that is a "United States person" as defined in Section 7701(a)(30) of the Code and (ii) a properly completed and duly executed applicable IRS Form W-8 from each of the Former Equityholders that is not a "United States person" as defined in section 7701(a)(30) of the Code, together with a certification that complies with the requirements of Treasury Regulation Section 1.1446(f)-2(c)(2)(ii)(B) or (C);
- (d) a duly executed certificate from the Company, dated as of the Closing Date and in the form attached hereto as <u>Exhibit D</u>, that meets the requirements of Treasury Regulations Section 1.1445-11T(d)(2)(i);
- (e) a duly executed statement of Aero-Cee Bailey's Holding Company and notice to the Internal Revenue Service, dated as of the Closing Date, stating under penalty of perjury that Aero-Cee Bailey's Holding Company is not, and has not been, a "United States real property holding corporation" as defined in Section 897(c)(2) of the Code during the applicable period described in Section 897(c)(1)(A)(ii) of the Code, and in the form attached hereto as Exhibit E;
- (f) the Payoff Letters with respect to the Funded Indebtedness set forth on <u>Schedule 1.5(c)(iii)</u> and any necessary UCC authorizations or other releases as may be reasonably required to evidence the satisfaction (or deemed satisfaction) thereof, in form and substance reasonably acceptable to Parent;
- (g) a certificate or other equivalent evidence as of a recent date as to the good standing of each Acquired Company in its respective jurisdiction of formation;
- (h) a certificate of an officer of the Company, duly executed on behalf of the Company and not in such officer's individual capacity, certifying as to (i) resolutions duly adopted by the General Partner authorizing the execution and delivery of this Agreement by the Company and the performance by the Company of the transactions contemplated hereby, (ii) resolutions duly adopted by the Majority Preferred Holders (as defined in the LP Agreement) authorizing the execution and delivery of this Agreement by the Company and the performance by the Company of the transactions contemplated hereby, (iii) resolutions duly adopted by the Representative authorizing the execution and delivery of this Agreement by the Representative and the performance by the Representative of the transactions contemplated hereby;

- (i) written resignations of the directors and officers of each Acquired Company as set forth on Schedule 2.2(i);
- (j) to the extent not held or maintained at their registered office or at the Properties, the organizational documents, minute books and similar documentation of each of the Acquired Companies;
- (k) amendments to the Employee Plans, in the form attached hereto as Exhibit F, to remove AOG-Seginus Holding Company LLC, a Delaware limited liability company ("AOG-Seginus"), AOG Aviation Spares LLC, a Delaware limited liability company ("AOG Aviation"), Seginus Aerospace LLC, a Delaware limited liability company ("Seginus Aerospace"), and DAC Engineered Products, LLC, a Delaware limited liability company ("DAC"), as participating employers under the Employee Plans, effective as of the Closing;
- (l) evidence reasonably acceptable to Parent that the employment of the individuals listed on <u>Schedule 2.2(l)</u> have transferred (and their applicable employment agreements assigned) to one of AOG-Seginus, AOG Aviation, Seginus Aerospace or DAC;
- (m) a duly executed assignment, in the form attached hereto as Exhibit G, of the agreements set forth on Schedule 2.2(m) from the Acquired Company party thereto to one of AOG-Seginus, AOG Aviation, Seginus Aerospace or DAC, as applicable; and
- (n) duly executed agreements, in the forms attached hereto as Exhibit H, validly terminating the Contracts set forth on Schedule 2.2(n).
- 2.3 <u>Deliveries by Parent</u>. At or prior to the Closing, Parent delivered or caused to be delivered to the Representative (on behalf of the Representative or Acquired Companies, as applicable) or such other Person specified herein the following items:
  - (a) payment of the amounts in accordance with <u>Section 1.5</u> to the Person or Persons entitled thereto;
  - (b) the Escrow Agreement, duly executed by Parent and the Escrow Agent;
- (c) a certificate of the Secretary of State of the State of Delaware as of a recent date as to the good standing of each of Parent and Merger Sub in such jurisdiction;
- (d) a certificate of the Secretary of Parent, duly executed by him on behalf of Parent and not in his individual capacity, certifying as to the resolutions duly adopted by the Board of Directors of Parent authorizing the execution and delivery of this Agreement by Parent and the performance by Parent of the transactions contemplated hereby; and
- (e) a certificate of the Secretary of Merger Sub, duly executed by him on behalf of Merger Sub and not in his individual capacity, certifying as to the resolutions duly adopted by the managing member of Merger Sub authorizing the execution and delivery of this Agreement by Merger Sub and the performance by Merger Sub of the transactions contemplated hereby.

# ARTICLE 3. REPRESENTATIONS AND WARRANTIES REGARDING THE ACQUIRED COMPANIES

The Company represents and warrants to Parent and Merger Sub that, except as set forth on an applicable Disclosure Schedule, as of the date hereof and as of the Closing Date:

- 3.1 Organization and Good Standing; Holding Company.
- (a) Each Acquired Company is duly organized, validly existing and in good standing (to the extent such concept is applicable) under the Laws of jurisdiction of its incorporation, and

has all corporate power and authority to conduct its business as presently conducted and to own and lease the properties and assets used in connection therewith and to perform all of its obligations under each agreement and instrument by which it is bound. The Representative is duly organized, validly existing and in good standing (to the extent such concept is applicable) under the Laws of jurisdiction of its incorporation, and has all corporate power and authority to conduct its business as presently conducted and to own and lease the properties and assets used in connection therewith and to perform all of its obligations under each agreement and instrument by which it is bound. Each Acquired Company is qualified to do business and is in good standing in the jurisdictions set forth on Schedule 3.1(a), which are the only jurisdictions where the nature or character of the property owned, leased or operated by it or the nature of the business transacted by it makes such qualification necessary, except where the failure to be so qualified or be in good standing would not have a Material Adverse Effect. The Company has made available to Parent a true, accurate and complete copy of the organizational documents of each Acquired Company, each as in effect on the date hereof and as of the Closing Date.

(b) Each of the Company, Desser and AOG-Seginus is, and always has been, a holding company with no material operations or assets other than, (i) in the case of the Company, the ownership of the Equity Interests of Desser, (ii) in the case of Desser, the ownership of the Equity Interests of the other Acquired Companies and (iii) in the case of AOG-Seginus, the ownership of the Equity Interests of AOG Aviation, and Seginus Aerospace. No Acquired Company holds, has held since August 12, 2014, or to the Company's Knowledge has ever held prior to August 12, 2014, any Equity Interests in any Person other than another Acquired Company.

# 3.2 No Conflicts.

- (a) The execution, delivery and performance by the Company and the Representative of this Agreement and the Transaction Documents to which the Company or the Representative are party or otherwise to be delivered by the Company or the Representative pursuant hereto do not and will not (with or without the passage of time or the giving of notice): (i) violate or conflict with the certificate of incorporation, bylaws or other organizational documents of any Acquired Company or the Representative; (ii) violate or conflict with any Law binding upon any Acquired Company or the Representative; (iii) require the consent, notice to or other action by any Person under, or violate or conflict with, result in a breach of, or constitute a default or otherwise cause any loss of benefit under, any agreement or other obligation to which any Acquired Company or the Representative is a party or by which it or any of its assets are bound, or give to others any rights (including rights of termination, foreclosure, cancellation or acceleration), in or with respect to any assets of any Acquired Company; or (iv) result in, require or permit the creation or imposition of any Lien upon or with respect to the assets of any Acquired Company; except, in the case of clauses (ii)-(iv) above, for any violations, breaches, defaults, or Liens that would not be material to the Acquired Companies, taken as a whole.
- (b) No permit, authorization, consent or approval of, or filing with or notification to, any Governmental Entity or any other Person is required in connection with the execution and delivery of this Agreement and the applicable Transaction Documents by the Company or the Representative or the consummation of the transactions contemplated herein and therein.

#### 3.3 Capitalization; Investments and Subsidiaries.

(a) Each Person listed on Schedule 3.3(a) is the true and lawful beneficial owner of, and owns all right, title and interest in and to, the Company Equity Interests listed next to its name on Schedule 3.3(a), free and clear of all Liens (other than Liens being terminated in connection with the Closing). The Company Equity Interests constitute all of the issued and outstanding limited and general partnership interests or other Equity Interests of the Company. All outstanding Equity Interests of the Company have been duly authorized and validly issued and are fully paid and non-assessable and were not issued in violation of any preemptive or similar rights created by statute, the Company's organizational documents or any agreement to which the Company is a party or by which it is bound. There are no offers, options, warrants, rights, agreements, subscriptions or commitments of any kind (contingent or otherwise) relating to the issuance, purchase, acquisition, conversion, registration, voting, sale or transfer of any of the Company's Equity Interests or obligating the Company or any other Person to purchase or redeem the Company's Equity Interests.

- (b) The authorized capital of Desser consists of 449,363 units, 405,222 of which are issued and outstanding and owned by the Company, free and clear of all Liens (other than Liens being terminated in connection with the Closing) and 44,140 of which are reserved for issuance upon the exercise of options occurring simultaneously with the Closing. All outstanding Equity Interests of Desser have been duly authorized and validly issued and are fully paid and non-assessable and were not issued in violation of any preemptive or similar rights created by statute, Desser's organizational documents or any agreement to which Desser a party or by which it is bound. Other than those listed on Schedule 3.3(b)(i) and Schedule 3.3(b)(ii), there are no offers, options, warrants, rights, agreements, subscriptions or commitments of any kind (contingent or otherwise) relating to the issuance, purchase, acquisition, conversion, registration, voting, sale or transfer of any Equity Interests of Desser or obligating Desser or any other person to purchase or redeem any such Equity Interests. Schedule 3.3(b)(ii) sets forth a true and complete list of all outstanding Unit Options specifying, with respect to each such Unit Option: (i) the name of each Option Holder, (ii) the number of Units subject to each such Unit Option, (iii) the date of grant, (iv) the exercise price per Unit and (v) whether each such Unit Option is intended to qualify as an incentive stock option. All Unit Options have been issued pursuant to the Unit Option Plan. The exercise price of each Unit Option was equal to or greater than the fair market value of a Unit on the grant date for such Unit Option.
- Interest, including its name, its jurisdiction of incorporation, authorized Equity Interests and issued and outstanding Equity Interests. All of the issued and outstanding Equity Interests of each such Subsidiary are owned directly or indirectly by the Company, free and clear of all Liens, except for Liens arising under applicable securities Laws or such Subsidiaries' organizational documents, are duly authorized and validly issued and are fully paid and non-assessable and were not issued in violation of any preemptive or similar rights created by statute, the Company's organizational documents or any agreement to which any such Subsidiary is a party or by which it is bound. There is no offer, subscription, option, warrant, right, agreement or commitment of any kind (contingent or otherwise) relating to the issuance, purchase, acquisition, conversion, registration, voting, sale, delivery, transfer or redemption of the Equity Interests of any of such Subsidiaries (including any right of conversion or exchange under any outstanding security or other instrument).
- (d) Except as set forth on Schedule 3.3(d), no Company Equity Interests or any Equity Interests of any other Acquired Company are subject to vesting or forfeiture rights or repurchase by the Company or such other Acquired Company. There are no outstanding or authorized equity appreciation, dividend equivalent, phantom stock, profit participation or other similar rights with respect to any Acquired Company.

#### 3.4 Power and Authorization.

(a) The Company has full legal right, power and authority to enter into and perform its obligations under this Agreement and under any Transaction Documents to which it is party or otherwise required to be delivered by it pursuant hereto and to consummate the transactions contemplated hereby and thereby and thereby and thereby and thereby and thereby have been duly authorized by all necessary limited partnership action and no other proceedings on the part of the Company are necessary to authorize the execution, delivery and performance of this Agreement or to consummate the Merger and the other transactions contemplated hereby and thereby. This Agreement has been duly and validly executed and delivered by the Company and constitutes the legal, valid and binding obligation of the Company enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws related to the enforceability of creditors' rights generally, general equitable principles, the discretion of the Company is party or that are otherwise required to be delivered by the Company pursuant hereto shall constitute the legal, valid and binding obligation of the Company, enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws related to the enforceability of creditors' rights generally, general equitable principles, the discretion of the courts in granting equitable remedies, and matters of public policy.

- (b) The Representative has full legal right, power and authority to enter into and perform its obligations under this Agreement and under any Transaction Documents to which it is party or otherwise required to be delivered by it pursuant hereto and to consummate the transactions hereby and thereby. The execution, delivery and performance by the Representative of this Agreement and any Transaction Documents to which it is party or otherwise required to be delivered by it pursuant hereto and the consummation by the Representative of the transactions contemplated hereby and thereby have been duly authorized by all necessary action and no other proceedings on the part of the Representative are necessary to authorize the execution, delivery and performance of this Agreement or to consummate the Merger and the other transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by the Representative and constitutes the legal, valid and binding obligation of the Representative enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws related to the enforceable is party or that are otherwise required to be delivered by the Representative pursuant hereto shall constitute the legal, valid and binding obligation of the Representative, enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws related to the enforceability of creditors' rights generally, general equitable principles, the discretion of the courts in granting equitable remedies, and matters of public policy.
- Desser and its Subsidiaries dated as of December 31, 2021 and 2020 and the related audited consolidated statements of comprehensive loss, members' capital and cash flows for the years then ended, including the notes thereto (the "Audited Financial Statements"), (b) the unaudited consolidated balance sheets of Desser and its Subsidiaries dated as of December 31, 2022 and the related unaudited consolidated statements of comprehensive loss, members' capital and cash flows for the years then ended, including the notes thereto (the "2022 Financial Statements") and (b) unaudited consolidated balance sheet of Desser and its Subsidiaries dated as of March 31, 2023 (the "Balance Sheet" and the date thereof, the "Balance Sheet Date") and the related unaudited consolidated statement of profit and loss for the three (3)-month period then ended (the "Interim Financial Statements") and, collectively with the Audited Financial Statements and the 2022 Financial Statements, the "Financial Statements"). The Financial Statements present fairly, in all material respects, the financial position of Desser and its Subsidiaries, on a consolidated basis, as of the dates thereof and the results of their operations and cash flows for the years then ended in conformity with GAAP, subject, in the case of the Interim Financial Statements, to the absence of footnotes (that, if presented, would not differ materially from those presented in the Audited Financial Statements and 2022 Financial Statements) and to normal year-end reclassifications and adjustments (the effect of which would not be materially adverse). The Financial Statements are based on the books and records of the Acquired Companies. The Acquired Companies maintain a standard system of accounting established and administered in accordance with GAAP. There are no liabilities or obligations of the Acquired Companies other than those that (i) are disclosed or reserved against on the Balance Sheet, (ii) have been incurred in the ordinary course of business since the Balance Sheet Da

## 3.6 Real Property.

- (a) No Acquired Company owns, has owned since August 12, 2014, or to the Company's Knowledge has ever owned prior to August 12, 2014, any real property.
- (b) <u>Schedule 3.6</u> sets forth the address of all real property leased by any Acquired Company (collectively, the "<u>Properties</u>") and identifies each lease, sublease, license or other occupancy agreement, as applicable, under which such Properties are occupied by the Acquired Companies (collectively, the "<u>Company Leases</u>"). The Acquired Companies listed on <u>Schedule 3.6</u> own all right, title and interest in all leasehold estates and other rights purported to be granted to each of them by the Company Leases, in each case free and clear of any Lien except for: (i) Liens for current taxes, assessments and governmental charges and levies that are not yet due and payable or are being contested in good faith by appropriate proceedings and for which reserves have been established on the Balance

Sheet in accordance with GAAP; (ii) any zoning or other governmentally-established restrictions or Liens; (iii) materialmen's, mechanics', carriers', warehousemen's, landlords', workmen's, repairmen's, or other like Liens arising in the ordinary course of business and for which adequate reserves have been established in accordance with GAAP; (iv) Liens incurred or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance, social security and other like Laws, or to secure the performance of construction contracts, leases, statutory obligations, surety, appeal or performance bonds; and (v) such minor Liens, including utility and municipal easements and restrictions, in each case as do not, individually or in the aggregate, detract in any material respect from the value of the Property subject thereto and do not materially interfere with the use of the Property subject thereto in the ordinary conduct of the business of the applicable Acquired Company as presently conducted (collectively, the "Permitted Liens").

- (c) Each Company Lease is the legal, valid and binding obligation of the applicable Acquired Company, and, to the Knowledge of the Company, the other parties thereto, enforceable against the applicable Acquired Company in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws related to the enforceability of creditors' rights generally, general equitable principles, the discretion of the courts in granting equitable remedies, and matters of public policy. No breach or default in any material respect by the applicable Acquired Company under any Company Lease has occurred and is continuing, and no event has occurred which with notice or lapse of time would constitute such a breach or default. To the Knowledge of the Company, no breach or default by any other Person under any Company Lease has occurred and is continuing, and no event has occurred which with notice or lapse of time would constitute such a breach or default. To the Knowledge of the Company, no offsets or defenses are currently available to any party under any Company Lease. The Company has made available to Parent true, accurate and complete copies of each Company Lease and any amendments, modifications or other agreements related thereto and in effect.
- (d) There are no licenses, concessions, occupancy agreements or other Contracts to which any of the Acquired Companies is a party granting to any other Person the right of use or occupancy of the Properties, and there is no Person other than an Acquired Company in possession of the Properties. No condemnation Proceeding is pending or, to the Knowledge of the Company, threatened with respect to any Property.
- (e) No Acquired Company has received any written notice of either (i) violations of building codes and/or zoning ordinances or other Laws affecting any Property or (ii) any existing, pending or threatened zoning, building code or other moratorium proceedings, or similar matters which could reasonably be expected to adversely affect the ability of the Acquired Companies to occupy any Property as currently occupied. To the Knowledge of the Company, each applicable Acquired Company has obtained all certificates of occupancy and other permits or approvals required from any Governmental Entity with respect to the use and occupancy of each Property. All leasehold improvements made by or on behalf of an Acquired Company at any Property are in compliance with all applicable Laws. All of the buildings, fixtures and structures leased by any applicable Acquired Company pursuant to a Company Lease are in good operating condition, subject to ordinary wear and tear, and are adequate and suitable for their present uses and purposes. Neither the whole nor any material portion of any Property has been damaged or destroyed by fire or other casualty during the time any Acquired Company has operated at such Property.
- 3.7 <u>Personal Property</u>. Each Acquired Company has good and valid title to the personal property owned by it, free and clear of all Liens, except for any Permitted Liens. All personal property used but not owned by an Acquired Company is used pursuant to valid, subsisting and enforceable leases, subleases, licenses and other agreements binding upon such Acquired Company (as applicable) and, to the Knowledge of the Company, each other party thereto, in accordance with their terms.

## 3.8 Taxes.

(a) Each Acquired Company has duly and timely (taking into account extensions) filed all income and other material Tax Returns required to be filed by it. All such income and other material Tax Returns were correct and complete in all material respects when filed or when

amended. Each Acquired Company has timely paid all income and other material Taxes of the Acquired Companies (whether or not shown as due and payable on such Tax Returns).

- (b) All material Taxes that any Acquired Company has been required to collect or withhold have been duly collected or withheld and, to the extent required by applicable Law when due, have been duly and timely paid to the proper Governmental Entity.
- (c) No Acquired Company has requested an extension of time to file any Tax Return not yet filed, nor has it been granted any waiver of any statute of limitations with respect to, or any extension of a period for the assessment of, any Tax for any open Tax year.
- (d) All Tax deficiencies that have been assessed or asserted in writing against any Acquired Company have been fully paid or finally settled or otherwise resolved. There is no audit, Proceeding or assessment pending against any Acquired Company in respect of any Tax or Tax Return and no Acquired Company has received any written notice from any Governmental Entity that any such audit, Proceeding or assessment concerning any income or other material Tax Return or Taxes of the Acquired Companies is pending, threatened or proposed.
- (e) There are no Liens for Taxes (other than Liens described in clause (i) of the definition of Permitted Liens) upon the assets of the Company.
- (f) No Acquired Company is party to or bound by any Tax indemnity or Tax sharing agreement, other than (i) an agreement the principal purpose of which is not related to Taxes or (ii) the Tax distribution provision of any operating agreement of an Acquired Company.
- (g) No Acquired Company is liable for the Taxes of any other Person (other than the Acquired Companies) under Treasury Regulations Section 1.1502-6 or any applicable Law, as a transferee or successor or by Contract (other than any Contract the principal purpose of which does not relate to Taxes) or otherwise. Other than a group the common parent of which is Aero-Cee Bailey's Holding Company, no Acquired Company has been a member of an "affiliated group" of corporations within the meaning of Section 1504 of the Code, or any other affiliated, consolidated, combined, unitary or aggregate Tax group for any taxable period under applicable Law.
- (h) No Acquired Company will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date under the Laws in effect as of the Closing Date as a result of any (i) change in method of accounting for a taxable period ending on or prior to the Closing Date, (ii) "closing agreement" as described in Code Section 7121 (or any corresponding or similar provision of state, local or non-U.S. income Tax Law) executed on or prior to the Closing Date, (iii) intercompany transactions entered into prior to the Closing or any excess loss account in existence as of the Closing, in each case, as described in Treasury Regulations under Code Section 1502 (or any corresponding or similar provision of state, local or non-U.S. income Tax Law), (iv) installment sale or open transaction disposition made prior to the Closing, or (v) prepaid amount received or deferred revenue accrued outside the ordinary course of business prior to the Closing. No Acquired Company will be required to pay any Tax after the Closing Date as a result of an election made pursuant to Code Section 965(h).
- (i) No Acquired Company participated in a "reportable transaction" within the meaning of Treasury Regulations Section 1.6011-4(b) (other than a "loss transaction" in respect of which the Acquired Company properly complied with the applicable disclosure requirements under the Code with respect thereto), or any transaction requiring disclosure under a similar or comparable provision of applicable state, local or non-U.S. Law.
- (j) No Acquired Company has been a party to, nor has been otherwise involved in, any transaction, scheme or arrangement designed wholly or mainly or containing stages or steps having no commercial purpose and designed wholly or mainly for the purpose of improperly avoiding, deferring or reducing a liability to Tax.

- (k) No Acquired Company has deferred any Taxes or claimed any Tax credits pursuant to the CARES Act and COVID Relief Programs.
- (l) The Company has always been classified as a partnership for U.S. federal (and applicable state and local) income Tax purposes and the U.S. federal (and applicable state and local) income Tax classification of each other Acquired Company is set forth on Schedule 3.8(l). No IRS Form 8832 or other entity classification election for Tax purposes has been made or filed by or on behalf of any Acquired Company.
- (m) No Acquired Company has constituted either a "distributing corporation" or a "controlled corporation" in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code in the three (3) years prior to the Closing Date.
- (n) No Non-U.S. Acquired Company which was incorporated in the United Kingdom has ever been subject to the loan to participator rules as set out in Part 10 of the U.K. Corporation Tax Act 2010.
- (o) The unpaid Taxes of the Acquired Companies for all periods ending on or before the date of the 2022 Financial Statements, if any, did not, as of the date of the 2022 Financial Statements, exceed the reserve for Tax liabilities set forth on the face of such 2022 Financial Statements (rather than any notes thereto) and the unpaid Taxes of the Acquired Companies for all periods ending on or before the Closing Date will not exceed that reserve as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of the Acquired Companies in filing their Tax Returns.
- (p) No Acquired Company has any deferred intercompany gain or excess loss account described in the Treasury Regulations promulgated under Section 1502 of the Code.
- (q) No Acquired Company is, nor at any time has ever been, subject to the application of the loss deferral rules contained in Section 267(f) of the Code and the Treasury Regulations promulgated thereunder.
- (r) No Acquired Company has ever had a permanent establishment (within the meaning of an applicable Tax treaty or convention) in any jurisdiction other than the jurisdiction in which it is organized. Each Acquired Company is, and has only ever been, Tax resident in the country in which it is organized.
- (s) No Acquired Company owns any equity interests in any entity treated as a partnership for U.S. federal (or applicable state and local) income tax purposes.
  - (t) No Acquired Company is subject to any gain recognition agreement under Section 367 of the Code.
- (u) No Acquired Company has engaged in any transactions that required recapture of dual consolidated losses within the meaning of Section 1503(d) of the Code.
- (v) No Acquired Company is party to any Contract or agreement with a Governmental Entity regarding, or the recipient of, any Tax incentive, holiday, or abatement (in each case, relating to material Taxes) in any jurisdiction that, based on applicable Law, is subject to recapture or will be terminated as a result of the transactions contemplated hereby.
  - (w) No Acquired Company is liable for any material amounts under any escheat or unclaimed property laws.
- (x) Each Acquired Company has registered for the purposes of value added tax (or equivalent material taxes) when required to do so, and such Acquired Companies have complied in all material respects with the laws relating to value added tax (or equivalent material taxes).

3.9 <u>Litigation</u>. Except as set forth on <u>Schedule 3.9(a)</u> there are, and in the past three (3) years there have been, no actions, suits, proceedings, arbitrations, mediations, claims, investigations or audits (each, a "<u>Proceeding</u>") against, brought or threatened by or, to the Knowledge of the Company, otherwise involving or threatened against, any Acquired Company, its businesses or assets, or its directors, officers, or equityholders in their capacities as such. Other than with respect to any Orders regarding Tax matters (which are the subject of <u>Section 3.8</u>), <u>Schedule 3.9(b)</u> sets forth any Order to which any Acquired Company is subject, and no Acquired Company is in default with respect to any such Order. There are no actions, suits or proceedings pending against or, to the Knowledge of the Company, investigations involving, the Representative, any Former Equityholder or any Acquired Company that, in any such case, may have the effect of preventing, delaying, or making illegal the transactions contemplated herein.

## 3.10 Labor Matters.

- (a) <u>Schedule 3.10</u> contains a complete and accurate list of the employees and individual independent contractors of the Acquired Companies and shows for each such employee as of the date hereof: his or her name, position, employing entity, employment location, date of hire, status as active or inactive, notice period (if applicable), status as full-time or part-time, base salary or hourly wage (as applicable) and any variable remuneration, for employees, whether exempt or non-exempt under the Fair Labor Standards Act (if applicable), and for independent contractors, the duties each is retained to provide.
- (b) The Acquired Companies are not party to or bound by any collective bargaining agreements or any other agreement or arrangement with any trade union or any other body representing employees, and there are no collective bargaining agreements applicable to any Persons employed by any Acquired Company. To the Knowledge of the Company, (i) no petition has been filed or Proceeding instituted by an employee or group of employees of any Acquired Company with any labor relations board seeking recognition of a bargaining representative with respect to any Acquired Company and (ii) there is no organizational effort currently being made or threatened by, or on behalf of, any labor union to organize employees of any Acquired Company. During the last three (3) years, there have not been any, and there are not currently any, work slowdowns, lockouts, stoppages, picketing strikes or other form of industrial dispute pending, or to the Knowledge of the Company, threatened between any Acquired Company, on the one hand, and its employees, on the other hand.
- (c) Each Acquired Company is, and for the last three (3) years has been, in compliance in all material respects with all applicable Laws relating to such Acquired Company's employees, including those relating to wages, hours, collective bargaining and labor management relations, employee classification, unemployment insurance, worker's compensation, equal employment opportunity, discrimination, sexual harassment, civil rights, affirmative action, the calculation of holiday pay, work authorization and immigration control, leave, and the payment and withholding of Taxes. There are no Proceedings against the Acquired Companies pending or to the Knowledge of the Company, threatened, and there have been no Proceedings against any Acquired Company during the last three (3) years under any Law relating to employees or employment practices or with respect to breaches of any such Law.
- (d) During the last three (3) years, no Acquired Company has taken any action which would constitute a "plant closing" or "mass layoff" within the meaning of the WARN Act or any similar state or local Law or carried out any collective redundancies in the United Kingdom, issued any notification of a plant closing or mass layoff required by the WARN Act or similar state or local law or given notice of any redundancies to the relevant United Kingdom Secretary of State, or incurred any liability or obligation under WARN, the United Kingdom Trade Union and Labour Relations (Consolidation) Act 1992 or any similar state or local law that remains unsatisfied.
- (e) During the last three (3) years, no Acquired Company (i) has been found in violation of any Laws relating to employees or other labor-related matters; (ii) is or has been a party to, or otherwise bound by, any consent decree with, or citation by, any Governmental Entity relating to current or former employees, officers or directors or employment practices; and (iii) is or has been subject to any audit or investigation by the National Labor Relations Board, the Equal Employment Opportunity

Commission, Occupational Safety and Health Administration, the Department of Labor, U.S. Citizen and Immigration Services, or any comparable Governmental Entity, or subject to fines, penalties, or assessments associated with such audits or investigations.

- (f) In the past three (3) years, there has been no allegation of sexual harassment or employment discrimination brought (internally or otherwise) by any current or former employee of, or any current or former independent contractor or consultant to, any Acquired Company or by any applicant for employment with any Acquired Company.
- (g) All employees of the Acquired Companies employed in the United States are: (i) United States citizens or lawful permanent residents of the United States; (ii) aliens whose right to work in the United States is unrestricted; or (iii) aliens who have valid, unexpired work authorizations issued by the United States government. No Acquired Company has been the subject of an immigration compliance or employment visit from, been assessed any fine or penalty by, or been the subject of any order or directive of, the United States Department of Labor, the United States Department of Justice, the United States Department of Homeland Security, the U.S. Citizen and Immigration Services, or any other similar Governmental Entity. Each Acquired Company has obtained and maintained all required records regarding each employee's eligibility to work in the jurisdiction in which they are currently working in accordance with applicable Laws. Employees of the Acquired Companies have a valid and subsisting permission or authority to remain in the jurisdiction in which they are currently working and work for the relevant Acquired Company and no such permission or authority will expire within six months of Closing.
- (h) No employees based in the United Kingdom are subject to terms and conditions that are materially different to those that have been disclosed in <u>Schedule 3.12(p)</u>.

## 3.11 <u>Intellectual Property Rights and Data Protection</u>.

- (a) <u>Schedule 3.11(a)</u> contains a complete and correct list of (i) all active registrations of, and all pending applications to register, any Company Intellectual Property Rights (the "<u>Company Registered IP</u>"), indicating for each item the owner, registration or application number, the registration or application date, and the applicable filing jurisdiction; and (ii) all material unregistered Company Intellectual Property Rights. The Company Registered IP identified on <u>Schedule 3.11(a)</u> are held and/or recorded in the name of the Acquired Company listed thereon, not subject to any pending cancellation, interference, reissue or reexamination proceeding, are subsisting, valid and, to the Knowledge of the Company, enforceable, and are not subject to any outstanding Order adversely affecting the validity or enforceability of, or any Acquired Company's ownership or use of, or rights in or to, any such Intellectual Property Rights. The Acquired Companies are current in the payment of all registration, maintenance and renewal fees with respect to the Company Registered IP, except in each case as the Acquired Company has elected in its reasonable business judgment to abandon or permit to lapse a registration or application.
- (b) The Acquired Companies own the exclusive right, title and interest to the Company Registered IP, free and clear of all Liens (other than Permitted Liens). Each Acquired Company owns or has sufficient and valid rights to use pursuant to a Contract listed in <u>Section 3.12(c)(i)</u> all Intellectual Property Rights to, and used in, the conduct of its business as currently conducted.
- (c) Each Acquired Company has taken commercially reasonable steps to protect and maintain (as applicable) the Company Intellectual Property Rights (including the Company Registered IP) and the confidentiality of all material Trade Secrets that are owned, used or held by any Acquired Company, and to the Knowledge of the Company, no such material Trade Secrets have been used, disclosed to or otherwise discovered by any Person except pursuant to a valid and enforceable non-disclosure agreement that has not been breached by such Person.
- (d) Neither the conduct of the respective businesses of the Acquired Companies, nor the development, display, manufacture, use, sale or other exploitation of any of their products or services has infringed, misappropriated or otherwise violated any Intellectual Property Rights of any Person. No Acquired Company has received any written communication in the past three (3) years

alleging that an Acquired Company has infringed or misappropriated any Intellectual Property Rights of any Person or contesting or challenging an Acquired Company's use, validity, enforceability or ownership of any Company Intellectual Property Rights or Intellectual Property Rights exclusively licensed to an Acquired Company or any written claim, notice or invitation to license or similar communication. There are no Proceedings pending or, to the Knowledge of the Company, threatened, against any Acquired Company that allege that any Acquired Company is infringing, misappropriating or otherwise violating the rights of any Person with regard to any Intellectual Property Rights and, to the Knowledge of the Company, there is no existing fact or circumstance that would be reasonably expected to give rise to any such Proceeding. To the Knowledge of the Company, no third party is infringing upon or violating any of the Company Intellectual Property Rights or has been infringing upon or violating any of the Company Intellectual Property Rights. Except as set forth in Schedule 3.11(d), in the past three (3) years, no Acquired Company has sent any notice to or asserted or threatened any action or claim against any Person involving or relating to any infringement of any Company Intellectual Property Rights.

- (e) Each Person (including any employees, officers, contractors or service providers) whose work for or on behalf of any Acquired Company has involved, or is reasonably expected to involve, the development, discovery, conception or reduction to practice of any material Intellectual Property Rights, has executed a written agreement containing a valid and enforceable present assignment to an Acquired Company of all Intellectual Property Rights developed during the course of such Person's work for or on behalf of an Acquired Company. No such Person retains or purports to retain any right, title or interest in or to any such Intellectual Property Rights.
- (f) No Acquired Company owns or purports to own any proprietary software, and there is no software included in the Company Intellectual Property Rights.
- (g) All IT Assets owned, used or held for use by any Acquired Company (i) are free from any material defect, to the extent applicable, and (ii) do not contain any virus, software, routine or hardware component designed to permit unauthorized access or control, or to disable or otherwise harm any computer, systems or Software, including with the passage of time. The IT Assets owned, used or held for use by any Acquired Company are sufficient in all material respects for the operation of the businesses of the Acquired Companies as currently conducted. Except as set forth on Schedule 3.11(g), in the past three (3) years, there has been no unauthorized access to or unauthorized use of any such IT Assets.
- (h) Each Acquired Company is in compliance with, and has during the last three (3) years complied with, all applicable Privacy Laws in all material respects. Each Acquired Company materially follows its posted Privacy Policies regarding the Processing of Personal Information. In the last three (3) years, each Acquired Company has at all times made all required disclosures to, and obtained any required consents from, users, customers, employees, contractors, Governmental Entities and other applicable Persons as required by applicable Privacy Laws. Neither the execution, delivery, or performance of this Agreement nor the consummation of any of the transactions contemplated by this Agreement will violate any applicable Privacy Law or Privacy Policy. Each Acquired Company does not "sell" Personal Information, as such term and equivalent terms are defined in applicable Privacy Law.
- (i) Except as set forth in Schedule 3.11(g), during the last five (5) years, no Person has gained unauthorized access to, acquired, or engaged in unauthorized Processing of (i) any Personal Information or Company Data held by or on behalf of any Acquired Company, or (ii) any databases, computers, servers, storage media (e.g., backup tapes), network devices, or other devices or systems that processes Personal Information or Company Data and owned or maintained by each, their customers, subcontractors or vendors, or any other Persons on their behalf (a "Security Breach"), and no Acquired Company has any reason to reasonably suspect a Security Breach has occurred.
- (j) Each Acquired Company has at all times in the last three (3) years used commercially reasonable controls, technologies, processes, and practices to detect, identify and remediate Security Breaches. Each Acquired Company has at all times in the last three (3) years maintained commercially reasonable security measures, controls, technologies, polices and safeguards sufficient to comply with applicable Privacy Laws and Contracts and to the extent required by Privacy Laws, requires the same of all vendors that Process Personal Information and Company Data on its behalf. Each

Acquired Company maintains disaster recovery and business continuity plans, procedures and facilities that are commercially reasonable and that satisfy each Acquired Company's contractual and legal obligations.

- (k) There is no Proceeding initiated by any other Person pending or, threatened in writing against any Acquired Company alleging a violation of any Person's rights under any Privacy Law, nor has there been any court order adversely affecting any Acquired Company's use, disclosure or other processing of any Personal Information. There are no facts or circumstances that could constitute a reasonable basis for any such Proceeding relating to data privacy, data protection or data security. No Acquired Company has received any communications from or, to the Knowledge of the Company, been the subject of any investigation by, the Federal Trade Commission, a data protection authority, or any other Governmental Entity regarding its acquisition, use, disclosure or other processing of any Personal Information.
  - 3.12 <u>Contracts and Commitments.</u> <u>Schedule 3.12</u> sets forth a complete and accurate list of the following:
- (a) each contract, agreement, license, commitment, obligation or understanding, whether written or oral, to which any Acquired Company is party or by which any of their assets are bound, in each that is legally binding (a "Contract"), with any Top Customer or that involves the performance of services or the delivery of goods or materials by any Acquired Company of an amount or value in excess of \$150,000; provided, however, open sales orders that constitute Contracts shall be considered Contracts for purposes of this Agreement but shall not be required to be listed on Schedule 3.12 solely because such orders are responsive to this Section 3.12(a);
- (b) each Contract with any Top Supplier or that involves the performance of services for or the delivery of goods or materials to any Acquired Company of amount or value in excess of \$150,000; provided, however, open purchase orders that constitute Contracts shall be considered Contracts for purposes of this Agreement but shall not be required to be listed on Schedule 3.12 solely because such orders are responsive to this Section 3.12(b);
- (c) each Contract pursuant to which an Acquired Company: (i) receives a license, assignment, covenant not to sue, release, option or other right to any Intellectual Property Rights that are material to the business of any Acquired Company or are currently used or embodied in any product or service sold or made available by any Acquired Company, other than with respect to non-exclusive licenses to non-customized commercially available software granted on generally available terms; (ii) grants or agrees to grant to a third person a license, assignment, covenant not to sue, release, option or other right under any Company Intellectual Property Rights, excluding licenses granted by an Acquired Company to its customer for the purpose of utilizing the products or services of an Acquired Company, or to employees, consultants, agents, distributors or sales representatives for the purpose of permitting such Persons to provide services to or on behalf of, an Acquired Company; or (iii) otherwise agrees to limit its use of any trademark owned by an Acquired Company (such as any co-existence or consent agreement);
  - (d) each Contract for capital expenditures in excess of \$150,000;
  - (e) each Contract relating to Funded Indebtedness;
  - (f) each representative, distribution, marketing or sales agency Contract;
- (g) each Contract containing covenants limiting the freedom of any Acquired Company or any of its Affiliates to engage in any line of business, to compete with any Person or covenants of another Person not to compete with any Acquired Company or to solicit for employment, hiring or employing any Person;
- (h) each Contract entered into by any Acquired Company since January 1, 2017 or under which any Acquired Company has outstanding obligations or liabilities, in each case with

respect to the acquisition or disposition of any other entity, business, line of business or material amount of assets;

- (i) all Contracts that contain an "exclusivity" or similar provision that requires the Company or its Subsidiaries to purchase its total or a material portion of its requirements of any product or service from a third party;
- (j) each employment, severance, change of control, deferred compensation, or consulting agreement with executive officers of any Acquired Company and each commission, agency and representative agreement or arrangement to which any Acquired Company is a party or is otherwise bound and agreements or arrangements that have expired by their terms;
- (k) all Contracts that contain any fixed or indexed pricing, "most-favored nation" pricing or similar pricing terms or provisions regarding minimum volumes, volume discounts, or rebates or that grant any Person a right of first refusal, first offer or similar preferential right to purchase or acquire any right, asset, property or service of any Acquired Company;
  - (1) all collective bargaining agreements;
  - (m) all Government Contracts in effect as of the date hereof;
- (n) each Contract with respect to a joint venture, partnership or the sharing of profits, revenues, losses, costs or liabilities of any Person other than a Acquired Company;
- (o) each Contract with a Former Equityholder or any Affiliate thereof that will remain in effect or under which there will be any further obligations or restrictions following the Closing; and
- (p) current pro forma employment contracts applicable to each category of employee based in the United Kingdom who is employed by an Acquired Company.

Each Contract listed or required to be listed on Schedule 3.12 (including any open sales orders) is in full force and effect and is the legal, valid and binding obligation of the applicable Acquired Company and, to the Knowledge of the Company, the other parties thereto, enforceable against the applicable Acquired Company in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws related to the enforceability of creditors' rights generally, general equitable principles, the discretion of the courts in granting equitable remedies, and matters of public policy. No breach or default in any material respect by any of the Acquired Companies under any of the Contracts listed or required to be listed on Schedule 3.12 (including any open sales or purchase orders) has occurred and is continuing, and no event has occurred which with notice or lapse of time would constitute such a breach or default or which would give rise to any right of notice, modification, acceleration, payment, cancellation or termination of or by another party under, or in any manner release any party thereto from any obligation under, any such Contract. To the Knowledge of the Company, no breach or default by any other Person under any of the Contracts listed or required to be listed on Schedule 3.12 (including any open sales or purchase orders) has occurred and is continuing, and no event has occurred which with notice or lapse of time would constitute such a breach or default or which would give rise to any right of notice, modification, acceleration, payment, cancellation or termination of or by another party under, or in any manner release any party thereto from any obligation under, any such Contract. The Company has made available to Parent true, accurate and complete copies of each of the Contracts listed or required to be listed on Schedule 3.12 (including any open sales or purchase orders), in each case, as amended or otherwise modified and in effect. Neither the Representative nor any Acquired Company has

3.13 <u>Recent Events</u>. Except as set forth on <u>Schedule 3.13</u>, since the Balance Sheet Date, (x) the Acquired Companies have conducted their business in the ordinary course of business consistent with past practice and (y) there has been no Material Adverse Effect. Without limiting the

generality of the foregoing, except as disclosed in the applicable subsection of Schedule 3.13, since the Balance Sheet Date:

- (a) no Acquired Company has sold, leased, transferred, or assigned any of its assets, other than sales of inventory in the ordinary course of business;
- (b) no party (including any Acquired Company) has entered into, terminated, cancelled, amended, modified or accelerated any Contract that is listed or required to be listed on <a href="Schedule 3.12">Schedule 3.12</a>;
  - (c) no Acquired Company has imposed any Liens (other than Permitted Liens) upon any of its assets;
- (d) no Acquired Company has made any capital expenditures, or committed to make any capital expenditures, in excess of \$150,000;
- (e) no Acquired Company has made any capital investment in, any loan to, or any acquisition of the securities or assets (other than purchases of inventory, supplies and equipment in the ordinary course of business) of, any Person or business;
- (f) no Acquired Company has created, incurred, assumed or guaranteed any Funded Indebtedness (other than draws under a revolving line of credit in the ordinary course of business consistent with past practice);
- (g) no Acquired Company has delayed the payment of accounts payable past the date when such obligation would have been paid in the ordinary course consistent with past practice, or accelerated the collection of account receivable in advance of when such receivable would have been collected in the ordinary course of business consistent with past practice;
- (h) no Acquired Company has cancelled, compromised, waived, or released any right or claim (or series of related rights and claims) involving more than \$75,000;
- (i) no Acquired Company has transferred, assigned or granted any license or sublicense of any rights under or with respect to any Intellectual Property Rights other than non-exclusive licenses to customers in the ordinary course of business consistent with past practice;
- (j) no Acquired Company has issued, sold, or otherwise disposed of any of its Equity Interests, or granted any options, warrants, or other rights to purchase or obtain (including upon conversion, exchange, or exercise) any of its Equity Interests;
- (k) no Acquired Company has declared, set aside or paid any dividend or made any distribution with respect to its Equity Interests (whether in cash or in kind) or redeemed, purchased or otherwise acquired any of its Equity Interests;
- (l) no Acquired Company has experienced any material damage, destruction or loss (whether or not covered by insurance) to any of its property;
  - (m) no Acquired Company has entered into any transaction with any of its directors, managers, officers or employees;
- (n) no Acquired Company has (i) made or promised any increase in excess of 6% the base compensation of any of its directors, managers, officers or employees, (ii) adopted, amended, modified, or terminated any Employee Plan, (iii) hired any individual with salary or overall compensation greater than \$100,000, (iv) taken action or used discretion to adjust the payment of bonuses under the Desser Holding Company, LLC Annual Bonus Plan, including, but not limited to, providing for the payment of pro-rata bonuses or providing for payment of bonuses at target levels without regard to the

achievement of applicable performance metrics, or (v) made any other change in employment terms for any of its directors, managers, officers or employees;

- (o) no Acquired Company has made any loans or advances of money;
- (p) no Acquired Company has made any material changes in its accounting policies, methods, principles or practices or any of its cash management practices;
- (q) no Acquired Company has made any material Tax election, changed its method of Tax accounting (except as the result of any change in Law), prepared any Tax Returns in a manner which is inconsistent in any material respect with the past practices of such Acquired Company with respect to the treatment of items on such Tax Returns, incurred any material liability for Taxes other than in the ordinary course of business consistent with past practice, filed an amended Tax Return or a claim for refund of Taxes with respect to the income, operations or property of such Acquired Company, or settled any claim relating to Taxes; and
- (r) no Acquired Company has entered into any agreement, understanding or commitment by any Acquired Company to do any of the foregoing.

## 3.14 Employee Benefit Plans.

- (a) Schedule 3.14(a) contains a complete list of (i) all "employee benefit plans," as defined in Section 3(3) of ERISA (whether or not subject to ERISA), (ii) all other employment, severance pay, salary continuation, bonus, incentive, stock option, equity-based, welfare benefit, retirement, pension, profit sharing or deferred compensation plans, contracts, programs or arrangements of any kind including, in the United Kingdom, any custom and practice of making enhanced redundancy payments or any agreed procedure for redundancy selection, and (iii) all other employee benefit plans, contracts, programs, funds or arrangements, whether or not funded, in each such case (A) that are sponsored or maintained by any Acquired Company, (B) with respect to which any Acquired Company has or could reasonably be expected to have any current or contingent Liability (including on account of any member of the Controlled Group) (all of the above being hereinafter individually or collectively referred to as an "Employee Plan" or "Employee Plans," respectively). Schedule 3.14(a) separately identifies each Employee Plan that is maintained outside the United States and, in respect of any schemes, plans or practices (whether discretionary or otherwise) under which any employee of an Acquired Company that is based in the United Kingdom may be entitled to receive a bonus, commission or other incentive award, includes details of (y) all payments that are proposed and yet to be paid; and (z) payments made for the two financial years preceding the year in which Closing takes place.
- (b) All Employee Plans have been administered in compliance in all material respects with their terms, any related documents or agreements, and applicable Law (including the applicable provisions of ERISA and the Code). With respect to each Employee Plan, (i) each Employee Plan that is intended to be qualified within the meaning of Section 401(a) of the Code is so qualified and is, as most recently amended, the subject of a favorable determination and/or opinion or advisory letter as to its qualification and, to the Knowledge of the Company, nothing has occurred, whether by action or failure to act, that would reasonably be expected to cause the loss of such qualification; (ii) none of the Acquired Companies, any employee or director of the Acquired Companies, or, to the Knowledge of the Company, any other "disqualified person" (as defined in Section 4975 of the Code) or any other "party in interest" (as defined in Section 3(14) of ERISA) has engaged in, or been a party to, any "prohibited transaction" (as described in Section 4975 of the Code or in Part 4 of Subtitle B of Title I of ERISA) with respect to any Employee Plan (other than any such transactions for which a statutory, regulatory or administrative exemption is available) that could reasonably be expected to result in liability for any Acquired Company; and (iii) there are no Proceedings pending or, to the Knowledge of the Company, threatened by any Governmental Entity or by any participant or beneficiary against any of the Employee Plans, other than routine claims for benefits. Each Employee Plan that is maintained outside the United States (A) that is required to be registered or meet requirements to receive tax favored treatment, has been registered or met such requirements, and (B) has been maintained in good standing with the applicable regulatory authorities.

- (c) Copies of the following materials have been delivered or made available to Parent, if applicable: (i) true and complete copies of the current plan documents for each Employee Plan or, in the case of an unwritten Employee Plan, a written description thereof, (ii) the most recent determination, advisory or opinion letters from the IRS with respect to any Employee Plan, (iii) the currently effective summary plan descriptions, together with any summaries of material modifications thereto, with respect to any Employee Plan, (iv) the three most recent Form 5500 annual reports with respect to any Employee Plan, (v) all trust agreements and insurance contracts relating to the funding or payment of benefits under any Employee Plan, and (vi) all material correspondence relating to any Employee Plan between an Acquired Company or its representatives and any Governmental Entity within three (3) years prior to the date hereof and as of the Closing Date.
- (d) No Employee Plan is a "multiemployer plan" as such term is defined in Section 3(37) of ERISA or a plan that is subject to the minimum funding requirements of Part 3 of Subtitle B of Title I of ERISA or Section 412 of the Code. None of the Acquired Companies nor any other member of the Controlled Group has had an obligation to contribute to a "defined benefit plan" (as defined in Section 3(35) of ERISA), a pension plan subject to the funding standards of Section 302 of ERISA or Section 412 of the Code, a "multiemployer plan" (as defined in Section 3(37) of ERISA or Section 414(f) of the Code), a "multiple employer plan" (within the meaning of Section 210(a) of ERISA or Section 413(c) of the Code) or a "multiple employer welfare arrangement" (as defined in Section 3(40) of ERISA), in each case, that would result in liability of any Acquired Company on or after the Closing.
- (e) Except as disclosed on <u>Schedule 3.14(e)</u>, the consummation of the transactions contemplated herein will not, either alone or in combination with another event, (i) entitle any current or former employee, officer, director, or consultant of any Acquired Company to severance pay or any other payment under any Employee Plan, (ii) accelerate the time of payment or vesting of, or increase the amount of, compensation or benefits due to any such individual, (iii) result in any forgiveness of indebtedness with respect to any such individual, (iv) trigger any funding obligation under any Employee Plan or impose any restrictions or limitations on the rights of any Acquired Company to administer, amend or terminate any Employee Plan other than as imposed by Law.
- (f) Other than such continuation of benefit coverage under group health plans as is required by applicable Law, no Employee Plan provides benefits or coverage in the nature of health or life insurance for any employee or any beneficiary of an employee after the employee's termination of employment.
- (g) With respect to each group health plan benefiting any current or former employee of an Acquired Company or any other member of the Controlled Group that is subject to Section 4980B of the Code, the Acquired Companies and each member of the Controlled Group have complied with the continuation coverage requirements of Section 4980B of the Code and Part 6 of Subtitle B of Title I of ERISA in all material respects.
- (h) Each Employee Plan, and each Acquired Company, is in compliance with, and has been operated in compliance with, the applicable requirements of the Patient Protection and Affordable Care Act and the Health Care Education and Reconciliation Act of 2010, and no Acquired Company has incurred (whether or not assessed) or is reasonably expected to incur or to be subject to, any tax, penalty, or other payment under Section 4980D or 4980H of the Code.
- (i) All (i) insurance premiums required to be paid with respect to, (ii) benefits, expenses and other amounts due and payable under, and (iii) contributions, transfers or payments required to be made to, any Employee Plan on or prior to the Closing Date will have been paid, made or accrued on or prior to the Closing Date.
- (j) No Acquired Company has agreed or committed to institute any additional plan, program, arrangement or agreement for the benefit of any current or former employees, officers, directors, managers, or independent contractors (who are natural Persons) of the Acquired Company that would be considered an Employee Plan, or to make any additional amendments to any of the Employee Plans.

- (k) No amount, economic benefit or other entitlement that could be received (whether in cash or property or the vesting of property) as a result of any of the transactions contemplated by this Agreement (alone or in combination with any other event) by any "disqualified individual" (as such term is defined in Treasury Regulation Section 1.280G-1) with respect to an Acquired Company under any employment, severance or termination agreement, other compensation arrangement or Employee Plan in effect as of the Closing Date would be characterized as an "excess parachute payment" (as such term is defined in Section 280G(b)(1) of the Code), and no such disqualified individual is entitled to receive any "gross-up", indemnification or other payment from an Acquired Company in the event that the excise tax required by Section 4999(a) of the Code is imposed on such disqualified individual.
- (l) Each Employee Plan and any other payment or arrangement for which the Acquired Company has liability that is subject to Section 409A of the Code is in documentary compliance with, and has been operated in compliance with, Section 409A of the Code, and no person has a right to any gross up or indemnification from an Acquired Company with respect to any such Employee Plan, payment or arrangement subject to Section 409A of the Code.
- (m) No Acquired Company is or has at any time been (i) an "employer" for the purposes of sections 38 to 51 of the UK Pensions Act 2004 and section 75 of the UK Pensions Act 1995 in relation to an occupational pension scheme (as that term is defined in those Acts), (ii) either "connected" with or an "associate" of (as those terms are used in sections 38 to 51 of the UK Pensions Act 2004) such an employer, or (iii) issued with a "financial support direction", "contribution notice" or "restoration order" in respect of any occupational pension scheme (as those terms are used in sections 38 to 56 of the UK Pensions Act 2004) or a warning notice or determination notice in relation to a contribution notice.
- (n) No employee of an Acquired Company that is based in the United Kingdom or is subject to an employment agreement governed by English law is entitled: (i) to notice of termination in excess of three months or (ii) to a payment on termination of employment other than in accordance with their contract of employment or as set out in <u>Schedule 3.14(a)</u>.

## 3.15 Compliance with Laws.

- (a) Each Acquired Company is, and has been for the past three (3) years, in compliance in all material respects with all applicable Laws, and no Acquired Company has received any unresolved written notice of any violation or delinquency, or potential violation or delinquency with respect to any applicable Laws.
- (b) Each Acquired Company has applied for or filed, or owns, possesses, has or otherwise lawfully uses (and has for the past three (3) years owned, possessed or otherwise lawfully used) all material federal, foreign, state, local and other governmental consents, licenses, permits, franchises and approvals (collectively "Authorizations") that are necessary or required for such Acquired Company to conduct its business as presently conducted and all such Authorizations are valid and in full force and effect, and (i) no Acquired Company is in default in any material respect under any Authorization, (ii) each Acquired Company is in compliance in all material respects, and has been in compliance in all material respects for the last three (3) years, with the terms and requirements of such Authorization, as applicable, (iii) all fees and charges with respect to such Authorization which are due and payable have been paid in full and (iv) no event has occurred that, with or without notice or lapse of time or both, would reasonably be expected to result in the revocation, suspension, lapse or limitation of any Authorization
- (c) Each Acquired Company has retained all written records as required by all applicable Laws, including, without limitation, regulations promulgated by any Governmental Entity with respect to the Authorizations held by each of them. Each Acquired Company possesses all necessary Intellectual Property Rights required to retain, maintain, and utilize the Authorizations.

## 3.16 Environmental Compliance. Except as set forth on Schedule 3.16:

- (a) Each Acquired Company is, and for the past five (5) years has been, in compliance in all material respects with and does not presently have any unresolved written notice of any material violation or delinquency with respect to, any Environmental Laws. Neither the Acquired Companies nor, to the Knowledge of the Company, any of the Properties, has been made subject to any unresolved Proceeding or Order pursuant to applicable Environmental Laws.
- (b) Each Acquired Company over the last five (5) years has obtained, or is in the process or renewing in the ordinary course of business, all material Authorizations required under applicable Environmental Laws for the conduct of such Acquired Company's business as conducted during the last five (5) years.
- (c) There is no Environmental Claim pending, to the Knowledge of the Company, or threatened in writing against any Acquired Company or the Properties.
- (d) No Acquired Company has released any Hazardous Material that requires any material investigation or material remediation by the Acquired Company pursuant to Environmental Law.
- (e) To the Knowledge of the Company, there are no pending Environmental Claims related to per- or polyfluoroalkyl substances (commonly referred to as "PFAS") that are presently regulated as Hazardous Materials as defined herein arising from the properties listed on Schedule 3.16(e).
- (f) The Company has made available to Parent true and correct copies of all material, final, non-privileged reports regarding the current environmental condition of any Property used by the Acquired Companies in the possession or control of the Company or any Acquired Company.
- (g) Excepting leases or loans or similar financing facilities including customary commercial terms indemnifying lenders, and further excepting product warranties of merchantability, fitness and compliance with Law made by any Acquired Company in the ordinary course of business for the products or services sold by the Company or any Acquired Company, no Acquired Company has entered into, assumed, undertaken, become subject to, or provided an indemnity with respect to, by Contract or operation of Law or otherwise, any material liability arising under or related to Environmental Laws or Order relating to or arising under Environmental Laws.
- (h) "Environmental Claim" shall mean any Proceeding by any Person alleging potential liability arising out of, based on, or resulting from (i) the release or disposal into the environment of any Hazardous Materials by any Acquired Company at the Properties (or at or from any other location) on or prior to the Closing Date, or (ii) circumstances forming the basis of any violation, or alleged violation, of any Environmental Laws by any Acquired Company. "Environmental Laws" shall mean all Laws (A) related to releases or threatened releases of any Hazardous Materials in soil, surface water, groundwater or air; or (B) protection of human health or employee safety with respect to Hazardous Materials exposures, natural resources or the environment; or (C) governing the use, treatment, storage, disposal, transport, or handling of Hazardous Materials. Such Environmental Laws shall include, but are not limited to, the Resource Conservation and Recovery Act, the Comprehensive Environmental Response, Compensation and Liability Act, the Safe Drinking Water Act, the Clean Water Act, the Toxic Substances Control Act, the Clean Air Act, the Emergency Planning and Community Right-to-Know Act, the Hazardous Materials Transportation Act and all analogous or related Laws of any state, local or international authority. "Hazardous Materials" shall mean any pollutant, toxic substance, hazardous waste, hazardous substance or extremely hazardous material regulated under any Environmental Laws, including any petroleum product or by-product, petroleum-derived substances wastes or breakdown products, friable asbestos, lead-based paint, polychlorinated biphenyls.

## 3.17 <u>Customers and Suppliers</u>.

- (a) <u>Schedule 3.17(a)</u> sets forth, for the fiscal years ended December 31, 2021 and December 31, 2022, the top twenty (20) customers of each of the Acquired Companies (collectively, the "<u>Top Customers</u>"), and with respect to each Top Customer, the aggregate dollar value of sales
- (b) No Top Customer has terminated or adversely modified the amount, pricing, frequency or terms of the business such Top Customer conducts with any Acquired Company. No Acquired Company has received written notice or, to the Knowledge of the Company, oral notice, that any of the Top Customers plans to discontinue its relationship with any of the Acquired Companies or to adversely modify the amount, pricing, frequency or terms of the business such Top Customer conducts with any Acquired Company. No Acquired Company is currently engaged in, and no conditions, facts or circumstances exist that would reasonably be expected to result in, a material dispute with any Top Customer.
- (c) <u>Schedule 3.17(c)</u> sets forth, for the fiscal years ended December 31, 2021 and December 31, 2022, the top twenty (20) suppliers of each of the Acquired Companies (collectively, the "<u>Top Suppliers</u>"), and with respect to each Top Supplier, the aggregate dollar value of purchases.
- (d) No Top Supplier has terminated or adversely modified the amount, pricing, frequency or terms of the business such Top Supplier conducts with any Acquired Company. No Acquired Company has received written notice, or to the Knowledge of the Company, oral notice, that any of the Top Suppliers plans to discontinue its relationship with any of the Acquired Companies or to adversely modify the amount, pricing, frequency or terms of the business such Top Supplier conducts with any Acquired Company. No Acquired Company is currently engaged, and no conditions, facts or circumstances exist that would reasonably be expected to result in, in a material dispute with any such Top Supplier. No supplier to any Acquired Company represents a sole source of supply for goods or services used in the conduct of the business of any Acquired Company.
- 3.18 Transactions with Affiliates. Except as set forth on Schedule 3.18(a), and other than any intercompany transactions among the Acquired Companies, none of the Acquired Companies, the Former Equityholders or any of their respective Affiliates, nor any current or former equityholder, member, partner, director, manager, officer, employee or consultant of any Acquired Company: (a) has, or during the last three (3) years has had, any direct or indirect interest (i) in, or is, or during the last three (3) years was, an equityholder, member, partner, director, manager, officer, employee or consultant of, any Person that is a client, customer, supplier, lessor, lessee, debtor, creditor or competitor of any Acquired Company or (ii) in any material property, asset or right that is owned or used by any Acquired Company in the conduct of its business; or (b) is, or during the last three (3) years has been, a party to any agreement or transaction with any Acquired Company. Except as set forth in Schedule 3.18(b), there is no outstanding Funded Indebtedness owed to any Acquired Company from any current or former equityholder, member, partner, director, manager, officer, employee or consultant of any Acquired Company or any Former Equityholder or any of its, his or her respective Affiliates.
- 3.19 Insurance. Each Acquired Company maintains insurance under various insurance policies, a true and complete list of which are set forth on Schedule 3.19, which policies are in full force and effect and are valid and enforceable in accordance with their terms and shall remain in full force and effect and valid and enforceable in accordance with their respective terms immediately following the Closing. The Acquired Companies are current in all premiums or other payments due under such insurance policies and have otherwise complied in all material respects with all of their obligations under each such insurance policy. The Acquired Companies have given timely notice to the insurer of all material claims that may be insured thereby under any such insurance policy. During the past three (3) years, no Acquired Company has been refused any insurance by, not has coverage been limited by, any insurance carrier with which any Acquired Company has carried insurance or any other insurance carrier to which any Acquired Company has applied for insurance. No Acquired Company has received any written notice, or to the Knowledge of the Company, oral notice (a) of cancellation of any such policy or binder of insurance coverage thereunder; (b) that any issuer of such policy or binder has filed for protection under applicable bankruptcy or insolvency Laws or is otherwise in the process of liquidating or

has been liquidated; or (c) that any such policy or binder may no longer be in full force or effect or that the issuer of any such policy or binder may be unwilling or unable to perform its obligations thereunder.

- 3.20 <u>Brokers.</u> No Person acting on behalf of any Acquired Company, or any of their respective Affiliates, is or will be entitled to any brokers' or finders' fee or any other commission or similar fee, directly or indirectly, from any of such parties in connection with any of the transactions contemplated by this Agreement, other than Jefferies LLC and Lincoln International LLC, the fees of which shall be paid by the Representative and shall constitute Seller Expenses.
- 3.21 <u>Condition and Sufficiency of Assets</u>. The tangible assets and properties, including buildings, machinery and equipment, used by each Acquired Company in the operation of its business are in good operating condition and repair, ordinary wear and tear excepted, and are suitable for the purposes for which they are used. Each Acquired Company's assets, rights and properties are sufficient to permit the continued conduct of such Acquired Company's business in substantially the same manner as conducted immediately prior to Closing.
- 3.22 Records. The minute books and corporate record books of the Acquired Companies provided to Parent contain minutes of all meetings of, and accurate consents to all actions taken without meetings by, the boards of directors or any committees thereof and the holders of any securities of each Acquired Company. The books and records of the Acquired Companies are otherwise correct and complete in all material respects and have been maintained in accordance with sound business practices. No Acquired Company nor any class of its holders of Equity Interests has passed any resolution required to be filed with any Governmental Entity that has not yet been so filed.
- 3.23 <u>Bank Accounts.</u> Schedule 3.23 sets forth the names of all financial institutions or Persons which are depositories of the funds of any Acquired Company, the accounts established at such depositories, the names of all Persons authorized to draw or sign checks or drafts upon the accounts established in such depositories, the name of any depositories or other organization in which any Acquired Company has a safe deposit box, and the names of the Persons having access thereto.
  - 3.24 Government Contracts. Except as set forth at Schedule 3.24:
- (a) All of the Government Contracts were legally awarded, are binding on the parties thereto, and are in full force and effect in accordance with their terms, and the Acquired Companies have complied with all terms and conditions of each Government Contract and Government Bid, including all clauses, provisions and requirements incorporated expressly, by reference, or operation of Law therein.
- (b) All representations, certifications and disclosures made by each of the Acquired Companies with respect to any Government Contract or Government Bid were current, accurate and complete at the time of submission, and in the past three (3) years, no Governmental Entity nor any prime contractor, subcontractor or other Person has notified any of the Acquired Companies that it has, or is alleged to have, breached or violated any Law, representation, certification, disclosure, clause, provision or requirement pertaining to any Government Contract or Government Bid. None of the Acquired Companies is subject to any outstanding or anticipated request for price adjustment or repayment on any Government Contract.
- (c) None of the Acquired Companies is subject to any notice of termination for convenience, notice of termination for default, stop work order, cure notice or show cause notice (or the functional equivalents thereof) pertaining to any Government Contract, and no Governmental Entity has threatened to issue any such notice. In the past three (3) years, none of the Acquired Companies has received any notice from any Governmental Entity pertaining to any Government Contract indicating that it intends to cancel or otherwise adversely modify its relationship with the Company, other than in a manner that would not materially adversely impact such Government Contract.
- (d) In the past three (3) years, none of the Acquired Companies, nor any director, manager or officer of the Acquired Companies, has been debarred, suspended, proposed for

debarment or suspension, or received a show cause notice related to contemplated suspension or debarment, from participation in Government Contracts.

- (e) None of the Acquired Companies has represented or certified itself as "small" under the regulations at 13 C.F.R. Part 121 in connection with a Government Contract, and each of the Acquired Companies has at all times been in compliance with all Laws regarding business classifications (including woman-owned, minority-owned, veteran-owned, and HUBZone) that confer or potentially confer preferential status in connection with the award of Government Contracts.
- (f) None of the Acquired Companies is currently subject to any Proceeding related to any Government Contract or Government Bid, and none of the Acquired Companies has been subject to any Proceeding related to any Government Contract in the past three (3) years.
- (g) None of the Acquired Companies has received a past performance evaluation (or equivalent) that would be reasonably expected to have a materially adverse effect on the ability of any of the Acquired Companies to compete for a Government Contract.
- 3.25 Inventory; Accounts Receivable. The inventory of the Acquired Companies is in all material respects merchantable and fit for the purpose for which it was procured or manufactured, and is not slow-moving, obsolete, damaged, or defective, subject to the reserve for inventory writedown set forth on the Balance Sheet as adjusted for the passage of time through the Closing Date in the ordinary course of business. All notes and accounts receivable of the Acquired Companies are reflected properly on their books and records, are valid receivables subject to no setoffs or counterclaims, and are current and collectible subject to the reserve for bad debts set forth on the Balance Sheet as adjusted for the passage of time through the Closing Date in accordance with the ordinary course of business of the Acquired Companies. The accounts payable and accruals of the Acquired Companies have arisen in bona fide arm's-length transactions in the ordinary course of business, and each Acquired Company has been paying its accounts payable as and when due.
- 3.26 Products. All products manufactured, sold or delivered by any Acquired Company have been in conformity with all applicable warranties and product specifications in all material respects, and no Acquired Company has any material liability for replacement thereof or other damages in connection therewith in excess of any warranty reserve established with respect thereto on the Balance Sheet as adjusted for the passage of time through the Closing Date in accordance with the ordinary course of business of the Acquired Companies. No products manufactured, sold or delivered by any Acquired Company are subject to any contractual or implied guaranty, warranty or other indemnity beyond the applicable standard terms and conditions of sale with respect thereto which, in each case, have been made available to Parent. No Acquired Company has received any written or, to the Company's Knowledge, oral, notice of any claims for, and to the Knowledge of the Company, there is no reasonable basis for, any product recalls, returns, warranty obligations or service calls relating to any of its products or services. In the past three (3) years, no Acquired Company has had any material liability arising out of any injury to individuals or property as a result of the ownership, possession or use of any products manufactured, sold or delivered by any Acquired Company or with respect to any services rendered by any Acquired Company.

#### 3.27 Anti-Corruption.

- (a) During the three (3)-year period prior to the Closing Date, each Acquired Company has been in compliance with Anti-Corruption Laws, and no Acquired Company nor any of its directors, managers, officers, employees nor any other Person acting on its behalf has violated any Anti-Corruption Law.
- (b) During the three (3)-year period prior to the Closing Date, no Acquired Company nor any of its directors, managers, officers, employees, nor, to the Company's Knowledge, any other Person acting on its behalf has paid, offered, promised, or authorized the payment of money or anything of value, directly or indirectly, to any official of a Governmental Entity, any political party or any other Person for the purpose of influencing any act or decision or to secure any improper advantage.

(c) During the three (3)-year period prior to the Closing Date, no Acquired Company nor any of its directors, managers, officers, employees, nor, to the Company's Knowledge, any other Person acting on its behalf has received from any Governmental Entity or any other Person any notice of any violation, alleged violation, or any suspected violation of any Anti-Corruption Law, or conducted any internal investigation with respect to, or made any voluntary or involuntary disclosure to a Governmental Entity concerning, any actual, suspected, or alleged violation of any Anti-Corruption Law.

#### 3.28 International Trade.

- (a) Within the prior five (5) years, (i) no Acquired Company has made voluntary disclosures to any Governmental Entity under the Trade Compliance Laws that are applicable to such Acquired Company, and (ii) there have been no actual or threatened Proceedings by any Governmental Entity regarding compliance by any Acquired Company with such applicable Trade Compliance Laws, and no Governmental Entity has assessed any fine or penalty against, or issued any warning letter to, any Acquired Company with regard to compliance with such applicable Trade Compliance Laws.
- (b) Each Acquired Company is, and at all times during the past five (5) years has been, in compliance in all material respects with the Trade Compliance Laws. Without limiting the foregoing, during the five (5)-year period prior to the Closing Date, no Acquired Company has made or provided any material false statement or omission to any Governmental Entity or to any purchaser of products, in connection with the importation of items, the valuation or classification of imported items, the duty treatment of imported items, the eligibility of imported items for favorable duty rates or other special treatment, country-of-origin marking, NAFTA or USMCA Certificates, marking and labeling requirements, other statements or certificates concerning origin, quota or visa rights, export licenses or other export authorizations, U.S.-content requirements, licenses or other approvals required by a Governmental Entity.
- (c) For the past five (5) years, each Acquired Company has conducted its export activities in compliance with Trade Compliance Laws. Without limiting the foregoing, in the past five (5) years, no Acquired Company has engaged in any transaction or other dealing involving any entity or person with respect to which such Acquired Company was at the time of such transaction prohibited from dealing under Trade Compliance Laws; nor, for the past five (5) years, has any Acquired Company engaged in any transactions or other dealings that at the time of such transaction(s) involved any comprehensively sanctioned country or region under the Trade Compliance Laws.
- (d) In the past five (5) years, none of the products or materials imported by, for or on behalf of each Acquired Company for which final liquidation has not yet occurred is subject to or otherwise covered by an antidumping duty order or countervailing duty order that remains in effect or is subject to or otherwise covered by any pending antidumping or countervailing duty investigation by any Governmental Entity. In the past five (5) years, none of the products or materials imported by, for or on behalf of any Acquired Company are subject to, or meet the criteria to be subject to, any Withhold Release Order administered by U.S. Customs and Border Protection or any equivalent Order outside the United States.
- 3.29 <u>Funded Indebtedness; Seller Expenses</u>. The Funded Indebtedness and Seller Expenses set forth in the Estimated Closing Merger Consideration are the only outstanding Funded Indebtedness and unpaid Seller Expenses, respectively.

# 3.30 CARES Act and COVID Relief Programs.

(a) Except as set forth on Schedule 3.30(a), no Acquired Company has (i) received any Funded Indebtedness (other than with respect to the deferral of Taxes) or grant under the CARES Act and COVID Relief Programs, or (ii) taken any action related to the COVID-19 pandemic with respect to their employees or other service providers, including implementing workforce reductions, terminations, furloughs or changes to compensation, benefits or working schedules, or changes to Employee Plans. Each Acquired Company has complied in all respects with the CARES Act and COVID Relief Programs and any documentation related thereto. All applications and certifications made by any

Acquired Company pursuant to the CARES Act and COVID Relief Programs are true and accurate. True and correct copies of all documentation related to the eligibility of the Acquired Companies for relief pursuant to the CARES Act and COVID Relief Programs have been made available to Parent.

- (b) Except as set forth on Schedule 3.30(b), no officer or executive of any Acquired Company nor any Former Equityholder is currently subject to the compensation restrictions set forth in Section 4004 of the CARES Act and COVID Relief Programs, and each officer and executive set forth on Schedule 3.30(b) has consented to any reductions to his or her compensation and/or severance or termination benefits that may be implemented as a result of the Acquired Companies' covenants to be bound by such compensation restrictions.
- 3.31 <u>Disclaimer of Warranties</u>. EXCEPT AS EXPRESSLY SET FORTH IN THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS <u>ARTICLE 3</u> (INCLUDING THE RELATED PORTIONS OF THE DISCLOSURE SCHEDULE) AND THE TRANSACTION DOCUMENTS, THE COMPANY DISCLAIMS ALL OTHER REPRESENTATIONS, WARRANTIES AND GUARANTEES, WHETHER EXPRESS OR IMPLIED. NOTWITHSTANDING THE FOREGOING, NOTHING IN THIS AGREEMENT SHALL LIMIT OR RESTRICT ANY CLAIM BASED ON OR ARISING OUT OF FRAUD.

## ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Parent and Merger Sub jointly and severally represent and warrant to the Representative that, as of the date hereof and as of the Closing Date:

## 4.1 Organization and Good Standing.

- (a) Parent is a corporation duly organized, validly existing and in good standing under the Laws of the State of Florida, and has all corporate power and authority to conduct its business as presently conducted and to own and lease the properties and assets used in connection therewith and to perform all of its obligations under each agreement and instrument by which it is bound.
- (b) Merger Sub is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware, and has all corporate power and authority to conduct its business as presently conducted and to own and lease the properties and assets used in connection therewith and to perform all of its obligations under each agreement and instrument by which it is bound.
- 4.2 <u>Power and Authorization</u>. Both Parent and Merger Sub have full legal right, power and authority to enter into and perform their respective obligations under this Agreement and under any Transaction Documents to which they are party or otherwise required to be delivered by them pursuant hereto and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by Parent and Merger Sub of this Agreement and the Transaction Documents to which they are party or otherwise required to be delivered by them pursuant hereto have been duly authorized by all necessary action. This Agreement has been duly and validly executed and delivered by Parent and Merger Sub and constitutes the legal, valid and binding obligation of Parent and Merger Sub enforceable against Parent and Merger Sub in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws related to the enforceability of creditors' right generally, general equitable principles, the discretion of the courts in granting equitable remedies, and matters of public policy. When executed and delivered as contemplated herein, the Transaction Documents to which the Parent and Merger Sub are party or that are otherwise required to be delivered by them pursuant hereto shall constitute the legal, valid and binding obligation of Parent and Merger Sub, enforceable against Parent and Merger Sub in accordance with its terms, subject to applicable bankruptcy, insolvency and similar

Laws attending the enforceability of creditors' right generally, general equitable principles, the discretion of the courts in granting equitable remedies, and matters of public policy.

## 4.3 No Conflicts.

- (a) The execution, delivery and performance by Parent and Merger Sub of this Agreement and the Transaction Documents to which they are party or that are otherwise required to be delivered by Parent and Merger Sub pursuant hereto, as applicable, do not and will not (with or without the passage of time or the giving of notice) violate or conflict with the certificate or articles of incorporation or bylaws (or other organizational or governing documents) of Parent or Merger Sub, or any Law binding upon Parent or Merger Sub.
- (b) No permit, authorization, consent or approval of, or filing with or notification to, any Governmental Entity or any other Person is required in connection with the execution and delivery of this Agreement and the applicable Transaction Documents by Parent or Merger Sub or the consummation of the transactions contemplated herein or therein.
- 4.4 <u>Litigation</u>. There are no actions, suits or proceedings pending against or, to the Knowledge of Parent, investigations involving, Parent or Merger Sub that, in any such case, may have the effect of preventing, delaying, or making illegal the transactions contemplated herein.
- 4.5 <u>Solvency</u>. Neither Parent nor Merger Sub is entering into this Agreement or the transactions contemplated hereby with the actual intent to hinder, delay or defraud either present or future creditors of any Acquired Company. Assuming that the representations and warranties of the Company contained in this Agreement are true and correct in all material respects, and after giving effect to the transactions contemplated by this Agreement, the Surviving Company, as of immediately following the Effective Time, (a) will be solvent (in that both the fair value of its assets will not be less than the sum of its liabilities and that the present saleable value of its assets will not be less than the amount required to pay its probable liabilities as they become absolute and matured), (b) will have adequate capital with which to engage in its business, and (c) will not have incurred and will not plan to incur liabilities beyond its ability to pay as they become absolute and matured.
- 4.6 <u>Brokers</u>. No Person acting on behalf of Parent, Merger Sub, or any of their respective Affiliates or under the authority of any of the foregoing is or will be entitled to any brokers' or finders' fee or any other commission or similar fee, directly or indirectly, from any of such parties in connection with any of the transactions contemplated by this Agreement.

# 4.7 <u>Investigation and Evaluation; Disclaimer Regarding Projections.</u>

- (a) Each of Parent, Merger Sub, and their respective directors, officers, employees, lawyers, accountants and advisors have been given the opportunity to examine to the full extent deemed necessary and desirable by Parent and Merger Sub all books, records and other information with respect to the Acquired Companies. Each of Parent and Merger Sub has taken full responsibility for determining the scope of its investigations of the Acquired Companies and for the manner in which such investigations have been conducted, and has examined the Acquired Companies to Parent's and Merger Sub's full satisfaction. Parent and Merger Sub are fully capable of evaluating the adequacy and accuracy of the information and material obtained by Parent or Merger Sub in the course of such investigations. Neither Parent nor Merger Sub has relied on any Acquired Company, or any information delivered or made available to Parent or Merger Sub with respect to any matter in connection with Parent's and Merger Sub's evaluation of the Acquired Companies, other than the representations and warranties of the Company expressly set forth in Article 3 of this Agreement (including the related portions of the Disclosure Schedule) and in each Transaction Document.
- (b) In connection with their investigation of the Acquired Companies and their business, Parent and Merger Sub have received from the Acquired Companies and their agents certain projections and other forecasts, including projected financial statements, cash flow items, certain business plan information and other data of the business of the Acquired Companies. Each of Parent and Merger Sub acknowledges that (a) there are uncertainties inherent in attempting to make such projections,

forecasts and plans and, accordingly, is not relying on them, (b) such Person is familiar with such uncertainties and is taking full responsibility for making its own evaluation of the adequacy and accuracy of all projections, forecasts and plans so furnished to it and (c) Parent and Merger Sub shall have no claim against anyone with respect to any of the foregoing. Accordingly, each of Parent and Merger Sub acknowledges that neither Acquired Company nor any Company Affiliate has made any representation or warranty with respect to such projections and other forecasts and plans other than as expressly set forth in Article 3 of this Agreement (including the related portions of the Disclosure Schedule) and in each Transaction Document.

(c) Notwithstanding the foregoing, nothing in this Agreement shall limit or restrict any claim based on or arising out of Fraud.

#### 4.8 Sufficiency of Funds.

- (a) Parent has or will have, and will cause Merger Sub to have, prior to or at the Effective Time, sufficient funds to pay the Closing Merger Consideration contemplated by this Agreement. As of the date hereof: (i) no Default or Event of Default has occurred and is continuing under and as defined in the Credit Agreement, and (ii) all representations and warranties in the Credit Agreement and the other Loan Documents (as defined therein) are true and correct in all material respects (except for any representation and warranty that is qualified by materiality or reference to Material Adverse Effect (as defined therein), which such representation and warranty shall be true and correct in all respects.
- (b) Concurrently with the execution and delivery of this Agreement, Parent has entered into that certain Signing Date Agreement, dated as of the date hereof, by and between Parent and Loar, pursuant to which the parties thereto will enter into that certain Membership Interest Purchase Agreement attached thereto (collectively, the "Loar Transaction Documents") in order to effect the subsequent sale of certain of the Acquired Companies to Loar (the "Loar Transaction"). The Loar Transaction Documents are in substantially the form provided to the Company, and there are no other agreements or understandings among the parties thereto that would adversely affect the ability of the parties thereto to close the transactions contemplated therein immediately following the Closing. There are no conditions precedent or other contingencies relating to the closing of the Loar Transaction except as expressly set forth in the Loar Transaction Documents.

#### 4.9 NO ADDITIONAL REPRESENTATIONS.

EACH OF PARENT AND MERGER SUB HEREBY ACKNOWLEDGES, AGREES, REPRESENTS AND WARRANTS THAT, EXCEPT AS SPECIFICALLY AND EXPRESSLY SET FORTH IN <u>ARTICLE 3</u> HEREOF (IN EACH CASE, AS QUALIFIED BY THE DISCLOSURE SCHEDULE) AND IN EACH TRANSACTION DOCUMENT: (A) NEITHER THE ACQUIRED COMPANIES NOR ANY COMPANY AFFILIATE MAKES, OR SHALL BE DEEMED TO MAKE, NOR HAS PARENT OR MERGER SUB RELIED UPON, ANY REPRESENTATION OR WARRANTIES OF ANY KIND OR NATURE (WHETHER WRITTEN OR ORAL, EXPRESS OR IMPLIED) WITH RESPECT TO THE MATTERS CONTEMPLATED HEREBY; (B) ALL WARRANTIES (WHETHER WRITTEN OR ORAL, EXPRESS OR IMPLIED) IN REGARD TO MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, CONDITION OR DESIGN OR ARISING FROM A COURSE OF DEALING OR USAGE OF TRADE OR OTHERWISE ARE EXPRESSLY EXCLUDED; (C) IT IS NOT RELYING ON ANY STATEMENT OR REPRESENTATION MADE BY OR ON BEHALF OF ANY ACQUIRED COMPANY OR ANY COMPANY AFFILIATE EXCEPT AS SPECIFICALLY SET FORTH IN <u>ARTICLE 3</u> HEREOF (IN EACH CASE, AS QUALIFIED BY THE DISCLOSURE SCHEDULE) AND IN EACH TRANSACTION DOCUMENT; (D) NO PERSON HAS BEEN AUTHORIZED, EXPRESSLY OR IMPLICITLY, BY ANY ACQUIRED COMPANY OR ANY COMPANY AFFILIATE TO MAKE ANY REPRESENTATION OR WARRANTY RELATING TO ANY ACQUIRED COMPANY OR OTHERWISE IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY EXCEPT AS SET FORTH IN <u>ARTICLE 3</u> HEREOF (IN EACH CASE, AS QUALIFIED BY THE DISCLOSURE SCHEDULE) AND IN EACH TRANSACTION DOCUMENT AND THAT, IF MADE, ANY SUCH REPRESENTATION

OR WARRANTY HAS NOT BEEN RELIED UPON; AND (E) HEREBY EXPRESSLY WAIVES AND RELINQUISHES ANY AND ALL RIGHTS, CLAIMS AND CAUSES OF ACTION AGAINST EACH ACQUIRED COMPANY AND EACH COMPANY AFFILIATE IN CONNECTION WITH THE ACCURACY, COMPLETENESS OR MATERIALITY OF ANY STATEMENTS, INFORMATION, DATA OR OTHER MATERIALS, WRITTEN OR ORAL, OR DOCUMENTS HERETOFORE PROVIDED OR MADE AVAILABLE TO PARENT OR MERGER SUB IN CONNECTION WITH THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY (EXCEPT AS SPECIFICALLY SET FORTH IN <u>ARTICLE 3</u> HEREOF, AS QUALIFIED BY THE DISCLOSURE SCHEDULE, AND IN EACH TRANSACTION DOCUMENT). NOTWITHSTANDING THE FOREGOING, NOTHING IN THIS AGREEMENT SHALL LIMIT OR RESTRICT ANY CLAIM BASED ON OR ARISING OUT OF "FRAUD" AS DEFINED IN THIS AGREEMENT.

## ARTICLE 5. CERTAIN OBLIGATIONS OF THE PARTIES

5.1 Public Announcements. Except as may be required by Law or by obligations pursuant to any listing agreement with any securities exchange or any stock exchange regulations or previously contained in the contents of any press release, public announcement or other disclosure approved in writing by Parent and the Representative, neither the Representative or any of its Affiliates, on the one hand, or Parent, any of the Acquired Companies or any of their respective Affiliates, on the other hand, shall make any public statements, including any press releases, or make any disclosure to any Person, other than to their respective officers, directors, employees, representatives, counsel and other professional advisors and Loar and its Affiliates and their respective officers, directors, employees, representatives, counsel and other professional advisors, with respect to the transactions contemplated hereby that contain any terms of this Agreement, including the Base Merger Consideration, the Estimated Closing Merger Consideration or the final Closing Merger Consideration, without the prior written consent of Parent and the Representative, and no such public statement or other disclosure may contain the identity of Parent, the Acquired Companies, the Representative or any of their respective Affiliates (or any information that would reasonably allow an unaffiliated third party to determine such identity), without the prior written consent of Parent and the Representative. Notwithstanding anything to the contrary in this Section 5.1, (a) the Representative may (without the consent of Parent) disclose information (on a confidential basis) to (i) its Affiliates and (ii) its Affiliates' investors, potential investors, annual meeting invitees and lenders announcing the consummation of the transactions contemplated hereby if the information included in such announcement is of a nature customarily conveyed by a private equity fund to its investors, potential investors, annual meeting invitees or lenders with respect to the announcement of the closing of a sale of one of its portfolio companies in a private negotiated transaction, and (b) if the transactions contemplated hereby or the Base Merger Consideration, the Estimated Closing Merger Consideration, the final Closing Merger Consideration or the final Merger Consideration are required by Law to be disclosed, the Representative and Parent shall consult with each other in advance as to the contents thereof.

## 5.2 <u>Indemnification and Insurance</u>.

- (a) Parent shall not, nor shall it cause or permit any Acquired Company to, as of the Closing Date or for a period of six (6) years following the Closing Date, amend, repeal or otherwise modify any of the provisions relating to the indemnification of directors and officers in the certificate of incorporation, bylaws, limited liability company agreement or other organizational documents (as applicable) of any Acquired Company as in effect immediately prior to the date hereof in any manner that would adversely affect in any material respect the rights thereunder of individuals who at any time prior to the Closing were directors or officers of any Acquired Company in respect of actions or omissions occurring at or prior to the Closing (including, without limitation, the transactions contemplated by this Agreement), unless such modification is required by Law.
- (b) The Company shall have arranged for an irrevocable D&O tail insurance policy that provides coverage for at least six (6) years following the Closing Date on terms substantially the same as or better than the existing coverage maintained as of the date hereof by the Acquired Companies. Parent (i) shall pay the entire premium for such insurance policy at Closing and (ii) shall not cancel, or permit any Acquired Company to cancel, such insurance policy during such six (6)-year period.

- (c) Notwithstanding anything contained in this Agreement to the contrary, this Section 5.2 shall survive the Closing until the sixth (6<sup>th</sup>) anniversary thereof and shall be binding, jointly and severally, on all successors and assigns of Parent. In the event that Parent or any Acquired Company, or any of their respective successors or assigns, consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or transfers or conveys all or substantially all of its properties and assets to any Person, then, and in each case, proper provision shall be made so that the successors and assigns of Parent or the applicable Acquired Company, as the case may be, shall succeed to the obligations set forth in this Section 5.2.
- (d) The provisions of this Section 5.2 are intended to be for the benefit of, and shall be enforceable by, each of the individuals who at any time on or prior to the Closing were directors or officers of any Acquired Company and his or her heirs and legal representatives, and shall be in addition to any other rights any such person may have.

#### 5.3 Certain Tax Matters.

- (a) <u>Transfer Taxes</u>. Any Transfer Taxes incurred in connection with this Agreement shall be paid one-half by Parent and one-half by Representative on behalf of the Represented Parties when due, and the party with the obligation under applicable Law shall, at its sole cost and expense, prepare and timely file, or cause to be prepared and timely filed, all Tax Returns and other documentation with respect to all such Transfer Taxes. Parent and the Representative agree to reasonably cooperate with the other Parties in the preparation and filing of any Tax Returns with respect to Transfer Taxes, including promptly supplying any information in its possession that is reasonably necessary to complete such Tax Returns.
- (b) Additional Cooperation on Tax Matters. From and after the Closing, Parent, the Acquired Companies, and the Representative shall cooperate fully and in good faith, as and to the extent reasonably requested by any other Party, in connection with the preparation and filing of any Tax Return and any audit, litigation or other proceeding with respect to Taxes, including, without limitation, to the extent such cooperation is reasonably related to the preparation of Tax Returns pursuant to Section 5.3(d). Such cooperation shall include the retention and (upon the other Party's request) the provision of records and information which are reasonably relevant to any such Tax Return, audit, litigation or other proceeding or any tax planning, as well as making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Parent, the Acquired Companies, and the Representative further agree, upon request, to use their commercially reasonable efforts to obtain any certificate or other document from any Governmental Entity or any other Person as may be necessary to mitigate, reduce or eliminate any Tax (including withholding Taxes) that could be imposed in connection with the transactions contemplated by this Agreement.

# (c) <u>U.S. Federal Income Tax Treatment; Allocation</u>.

(i) The Parties intend that, for U.S. federal (and applicable state and local) income tax purposes: (A) the purchase and sale of the Equity Interests of the Company pursuant to this Agreement be treated as a transaction governed by Rev. Rul. 99-6, 1999-1 C.B. 434 (Situation 2), with the Former Equityholders treated as having sold all of their partnership interests in the Company to Parent, and Parent treated as acquiring (w) all of the stock of the Aero-Cee Bailey's Holding Company, (x) the stock of Watts Desser Limited, (y) all of the stock of Desser Tire & Rubber Co., Pty Ltd (together with Watts Desser Limited and its subsidiaries, the "Non-U.S. Acquired Companies") and (z) all of the remaining assets of the Company and each other Acquired Company disregarded as separate from the Company, in each case, in exchange for the Merger Consideration (and any other amounts treated as consideration for such purposes) pursuant to this Agreement; (B) the existence of the Company as a partnership be treated as terminating as of the end of the Closing Date, (C) the taxable year of Aero-Cee Bailey's Holding Company consolidated group be treated as ending as of the end of day on the Closing Date under Treasury Regulation Section 1.1502-76(b)(1)(ii)(A)(1), and (D) to the extent that Parent makes (or causes to be made) any election under Section 338(g) of the Code (or any similar provision of state, local or non-U.S. Law) with respect to the direct or indirect acquisition of any Non-U.S. Acquired

Company, the taxable year of such Non-U.S. Acquired Company shall be treated as ending as of the end of day on the Closing Date (collectively, the "Intended Tax Treatment").

(ii) As soon as reasonably practicable after the date hereof, the Parties shall use their reasonable best efforts to agree on an allocation of the Base Merger Consideration among the Equity Interests of each of Desser Tire & Rubber Co., LLC, Aero-Cee Bailey's Holding Company, Watts Desser Limited, and Desser Tire & Rubber Co., Pty Ltd, with such agreed allocation reflected in the "First-Tier Allocation" section of Schedule 5.3(c) (the "Entity Allocation"); provided that the Parties agree that the allocation of the Base Merger Consideration to AOG-Seginus and DAC together shall equal \$30,000,000. The Parties agree that all amounts treated as consideration for U.S. federal (and applicable state and local) income tax purposes other than the Base Merger Consideration, and any adjustments to the Closing Merger Consideration, shall, in each case, be allocated among the Equity Interests of the Acquired Companies in accordance with the principles set forth in the "First-Tier Allocation" section of Schedule 5.3(c). As soon as reasonably practicable following the Closing (and not later than 120 days thereafter), Parent shall deliver to the Representative a proposed allocation (the "Allocation") of the Closing Merger Consideration (and any other amounts treated as consideration for U.S. federal (and applicable state and local) income tax purposes) among the assets of the Company (and, as applicable, any other Acquired Company) in accordance with Section 1060 of the Code, the Entity Allocation and the principles set forth on Schedule 5.3(c) (the "Allocation Schedule"). The proposed Allocation shall be considered final and binding upon Parent, the Acquired Companies, the Representative and the Former Equityholders, unless the Representative shall, within thirty (30) days of receipt of the proposed Allocation, provide to Parent written notice of its comments thereto, in which event Parent and the Representative shall endeavor in good faith to agree upon an Allocation. If Parent and the Representative cannot agree on the Allo

(iii) Each Party agrees to file, and to cause their Affiliates to file, all U.S. federal, state and local Tax Returns consistent with the Intended Tax Treatment, Entity Allocation and any agreed-upon Allocation and shall not take any Tax position contrary thereto; provided that nothing herein shall be construed so as to prevent any Party from settling, or require any Party to commence or participate in any litigation or administrative process challenging, any determination made by any Governmental Entity based upon or arising out of the Entity Allocation or the agreed-upon Allocation. Any adjustments to the Closing Merger Consideration pursuant to this Agreement shall be allocated consistent with the principles set forth on the Allocation Schedule and any agreed-upon Allocation.

## (d) Preparation of Tax Returns.

(i) The Representative shall prepare, or cause to be prepared, all Flow-Through Tax Returns for any Tax period that ends on or before the Closing Date (but for which the relevant Tax Return is not due until after the Closing Date) (the "Seller Tax Returns"). Each Seller Tax Return shall be prepared in accordance with the past practice of the Acquired Companies, unless otherwise required by Law or provided by this Agreement, and such Tax Returns shall report the Transaction Tax Deductions in a Pre-Closing Tax Period to the maximum extent permitted by applicable Law. No later than thirty (30) days prior to the due date (subject to any extension thereof) of any Seller Tax Return (or such shorter time as is commercially reasonable), the Representative shall provide to Parent a draft of such Seller Tax Return for Parent's review and comment. The Representative shall consider in good faith all reasonable comments requested by Parent in writing no later than fifteen (15) days after receiving such draft Seller Tax Return (or such shorter time as is commercially reasonable). Parent shall timely file or cause to be timely filed all Seller Tax Returns so prepared by the Representative (as updated to reflect any reasonable comments made by Parent that are accepted by the Representative).

(ii) Parent shall prepare and file or cause to be prepared and timely filed all Tax Returns (other than the Seller Tax Returns) of any Acquired Company for Pre-Closing Tax Periods, in each case, that are first due after the Closing Date (the "Parent Tax Returns"). Until the date

on which the Final Statement of Merger Consideration has been agreed or otherwise determined, the Parent shall deliver a draft of each Parent Tax Return reflecting any income or material Taxes that will be taken into account for purposes of calculating the Merger Consideration no later than thirty (30) days prior to the due date (subject to any extension thereof) of such Parent Tax Return (or such shorter time as is commercially reasonable). The Parent shall consider in good faith all reasonable comments requested by the Representative in writing no later than fifteen (15) days after receiving such draft Parent Tax Return (or such shorter time as is commercially reasonable). For the avoidance of doubt, if and to the extent there is any inconsistency between the amount of Taxes shown on a Parent Tax Return as filed and the amount of such Taxes included in Funded Indebtedness or Net Working Capital reflected in the Final Statement of Closing Merger Consideration shall control for purposes of Section 1.5(d)-(e).

- (e) Post-Closing Restrictions. Except as related to the Employee Matters Escrow Issue, Parent covenants that, on or after the Closing Date, it shall not, and shall not cause or permit the Acquired Companies or any of their or Parent's Affiliates to, without prior written consent of the Representative (such consent not to be unreasonably withheld, conditioned or delayed), (i) make, change, revoke or file any Tax election that has any retroactive effect to any taxable period ending on or before the Closing Date (other than any elections under Section 338(g) (or any similar provision of state, local or non-U.S. Law) for any Non-U.S. Acquired Company), (ii) refile, revoke, amend or otherwise modify (or grant an extension or waiver of any applicable statute of limitations period with respect to) any Tax Return filed by or with respect to an Acquired Company, or initialize or otherwise approach any Governmental Entity regarding any voluntary disclosure (or similar) agreement or procedure related to any Pre-Closing Tax Period, (iii) take any action after the Closing on the Closing Date outside the ordinary course of business, or (iv) make or file any election under Section 338(g) of the Code (or any similar provision of state, local or non-U.S. Law) with respect to any Acquired Company other than a Non-U.S. Acquired Company, in each case, if such action could reasonably be expected to (1) impact any amount included in Funded Indebtedness or Net Working Capital reflected in the Final Statement of Closing Merger Consideration, or (2) increase the liability of any Former Equityholder (or the direct or indirect owner(s) of any Former Equityholder) of the Company with respect to Taxes, including in respect of any obligation to indemnify for UK Exchange Taxes or UK Withholding Taxes, or in respect of any Flow-Through Tax Return.
- (f) Straddle Period. If applicable, in preparing any Tax Return of the Acquired Companies for a Straddle Period, the distributive share of such Acquired Company's items allocable to any Former Equityholder shall be determined using an interim closing of the Acquired Company's books as of the end of the day on the Closing Date. If, notwithstanding the preceding sentence, it is necessary to determine the liability for Taxes for a Straddle Period, the portion of any Taxes that are payable for such period that relate to a Pre-Closing Tax Period shall (i) in the case of Taxes charged on a periodic basis (such as ad valorem or property Taxes), be deemed to be the amount of such Taxes for the entire Straddle Period multiplied by a fraction the numerator of which is the number of days of the portion of such Straddle Period in the Pre-Closing Tax Period and the denominator of which is the number of days in the entire Straddle Period, and (ii) in the case of any other Taxes, be deemed equal to the amount which would be payable computed on a closing of the books basis as if the relevant Tax period ended as of the end of the Closing Date. Notwithstanding the foregoing or anything to the contrary in this Agreement, whether in respect of a Straddle Period or a taxable period ending on the Closing Date, to the maximum extent permitted by applicable Law, all Transaction Tax Deductions shall be treated as accruing on or before the Closing Date.
- (g) <u>Tax Contests</u>. Notwithstanding anything in this Agreement to the contrary, the Representative shall control the appointment and identity (including any changes) of the "partnership representative" (as defined for purposes of Section 6223(a) of the Code and the Treasury Regulations promulgated thereunder) and of any designated individual (as defined under Treasury Regulations Section 301.6223-1(b)(3)(ii)) of the Company for any taxable period ending on or prior to the Closing Date. Parent agrees to cooperate and to cause the Acquired Companies to cooperate with the Representative to the extent reasonably required after the Closing Date in connection with any audits, examinations, and administrative or judicial proceedings (including Tax contests, assessments, notices of deficiency or other adjustments or proposed adjustments) relating to a Flow-Through Tax Return or Taxes that may be reflected thereon conducted by Tax authorities with respect to any Pre-Closing Tax Period

(including the applicable portion of any Straddle Period) ("<u>Tax Contests</u>")). Within a reasonable time (but not more than five (5) days) after the Parent, any of the Acquired Companies or any of their respective Affiliates receives written notice of any such Tax Contest, Parent shall notify the Representative in writing of such Tax Contest. Parent shall, and shall cause the relevant Acquired Company to, provide the Representative with all necessary powers of attorney and other necessary documents and assistance to allow the Representative, at its sole cost and expense, to effectively conduct and control the defense of any Tax Contest. Parent shall (i) have the right, directly or through its designated representatives and at Parent's sole cost and expense, to review in advance and comment upon all submissions made in the course of any Tax Contest (including any administrative appeals thereof), the Representative shall keep Parent reasonably informed regarding the progress of such Tax Contest, and the Representative shall not settle any Tax Contest without the prior written consent of Parent (not to be unreasonably withheld, conditioned or delayed), and (ii) at Parent's sole election, have the right to require the Company, partnership representative and designated individual to make a "push out" election under Section 6226 of the Code with respect to any Liability for Taxes arising from any such Tax Contest involving an applicable Flow-Through Tax Return, in which case the former partners in the Company for U.S. federal income tax purposes shall comply with all requirements in respect of making such election.

## (h) <u>UK Exchange Tax and UK Withholding Tax Matters</u>

- (i) The Former Equityholders shall indemnify, defend and hold the Parent Indemnitees harmless from all Losses resulting from, or otherwise attributable to, any UK Exchange Taxes or UK Withholding Taxes; provided that the sole recourse of the Parent Indemnitees in respect of such indemnity, absent Fraud, shall be the release of amounts from the applicable UK Tax Escrow Account. Amounts from the UK Exchange Tax Escrow Account and UK Withholding Tax Escrow Account shall be released to the Representative (for the benefit of the Former Equityholders) or the Parent Indemnitees, or each of them, in such amounts and at such times as set forth in accordance with the provisions of the Escrow Agreement.
- (ii) The Representative shall, at its sole cost and expense, control the conduct of any audits, examinations, processes (including with respect to any voluntary disclosures) and administrative or judicial proceedings with respect to any UK Exchange Taxes and any UK Withholding Taxes (the "<u>UK Tax Contests</u>" and each a "<u>UK Tax Contests</u>" for so long as any amount remains in the applicable UK Tax Escrow Account; <u>provided</u> that (W) the Representative shall use reasonable best efforts to fully resolve such UK Tax Contests as soon as reasonably practicable after the Closing Date, but in no event later than the date that is five (5) Business Days prior to the date that is 24 months after the Closing Date, (X) Parent shall provide prompt written notice to the Representative of any UK Tax Contests it receives, (Y) Parent shall, and shall cause the relevant Acquired Companies to, provide the Representative with all necessary powers of attorney and other necessary documents and assistance in respect thereto, and (Z) Parent shall have the right, directly or through its designated representatives and at Parent's sole cost and expense, to review in advance and comment upon all submissions made in the course of any UK Tax Contest, the Representative shall keep Parent reasonably informed regarding the progress of such UK Tax Contest, and the Representative shall not settle any UK Tax Contest without the prior written consent of Parent (not to be unreasonably withheld, conditioned or delayed).
- 5.4 <u>R&W Insurance Policy</u>. Parent has delivered a copy of the form of R&W Insurance Policy (and/or the binder agreement for the R&W Insurance Policy) to the Representative. Parent shall not (and shall cause its Affiliates not to) amend, modify, terminate, or waive any provision of the R&W Insurance Policy that waives or limits subrogation or similar rights the insurer may have to seek recoveries from the Former Equityholders or any of their respective Affiliates, any of their respective directors, officers, employees, agents or representatives, or the Representative, in each case in a manner that would adversely affect any of the foregoing without the prior written consent of the Representative. In connection with Loar's acquisition from Parent of certain Acquired Companies, Parent shall require

Loar to agree to terms substantially similar to those contained in the second sentence of this Section 5.4 with respect to any representation and warranty insurance policy it obtains.

## 5.5 Employee Matters.

- (a) For ninety (90) days following the Closing Date, Parent shall: (i) cause the Acquired Companies to provide to the employees of the Acquired Companies (each, a "Continuing Employee", and collectively, the "Continuing Employees") compensation and benefits (including retirement, health and welfare (including medical, dental, vision and prescription benefits covering each Continuing Employee and their eligible spouses and dependents), paid time off and fringe benefits, severance policies, and bonus plans, as applicable, but excluding any equity-based compensation) that in the aggregate are substantially similar to the Continuing Employee as the compensation and benefits made available to such persons immediately prior to the Closing Date. Nothing in this Section 5.5 (i) shall confer upon any Continuing Employee or other person any right to continued employment with Parent or an Acquired Company, (ii) shall be deemed to establish, amend, or modify, or limit the ability to amend, modify or terminate any Employee Plan or any other benefit or employment plan or arrangement, or (iii) confer upon any current or former employee, officer, director or consultant of an Acquired Company or any other person (other than the Parties) any third party beneficiary rights or any other right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- (b) The Former Equityholders shall indemnify, defend and hold the Parent Indemnitees harmless from all Losses resulting from any Acquired Company's failure to file or furnish any return or statement required under Code Section 6055 or Code Section 6056 (the "Employee Matters Escrow Issue"); provided that the sole recourse of the Parent Indemnitees in respect of such indemnity, absent Fraud, shall be the release of amounts from the Employee Matters Escrow Account within 24 months following the Closing Date. Amounts from the Employee Matters Escrow Account shall be released to the Representative (for the benefit of the Former Equityholders) or the Parent Indemnitees, or each of them, in such amounts and at such times as set forth in accordance with the provisions of the Escrow Agreement. For the avoidance of doubt, notwithstanding anything to the contrary in this Agreement, after the Closing Date, Parent and the Acquired Companies may take all actions necessary to correct the Employee Matters Escrow Issue, including filing or furnishing any return or statement required under Code Section 6055 or Code Section 6056.
- 5.6 <u>Data Room</u>. As promptly as practicable, but no later than seven Business Days following the Closing Date, the Representative will deliver, or cause to be delivered, to Parent a copy of the data room maintained by or on behalf of the Company for purposes of the transactions contemplated by this Agreement, as such data room existed as of the Closing, in USB format or another format agreed upon by Parent.
- 5.7 <u>Release of Claims</u>. Effective as of the Closing, each Former Equityholder, on behalf of itself, himself or herself and its, his or her Affiliates and their respective heirs, executors, administrators, successors, assigns, and personal and legal representatives, in each case as applicable (collectively, "<u>Releasors</u>"), shall irrevocably, knowingly and voluntarily (a) fully releases, discharges, and covenants not to sue each of the Company, the Surviving Company and each of the other Acquired Companies, and their respective Affiliates and each of their respective past, present and future directors, managers, officers, employees, agents, predecessors, successors, assigns, equity holders, subsidiaries, partners, joint ventures, attorneys, representatives and insurers, and all others connected with any of them, both individually and in their official capacities (collectively, the "<u>Released Parties</u>"), from any and all disputes, claims, controversies, demands, damages, rights, obligations, judgments, causes of action, and liabilities of any nature whatsoever existing at or prior to Closing (the "<u>Released Claims</u>") and (b) agrees that no Releasor will bring or voluntarily participate in or assist in any Proceeding that relates to any matter released pursuant to this <u>Section 5.7</u>. Notwithstanding the foregoing, the Released Claims do not include, and the provisions of this <u>Section 5.7</u>, shall not release or otherwise diminish the obligations of Purchaser or its Affiliates (other than the Acquired Companies) set forth in or arising under any provisions of this Agreement or the Transaction Documents.

Each Former Equityholder hereby acknowledges that the Releasors or the Released Parties may hereafter discover facts different from or in addition to those now known, or believed to be true, regarding

the subject matter of the release set forth in this <u>Section 5.7</u>, and such Former Equityholder further acknowledges that such release will remain in full force and effect, notwithstanding the existence of any different or additional facts that might have materially affected such Former Equityholder's willingness to enter into such release on its, his or her own behalf and on behalf of the Releasors. Further, each Former Equityholder, on its, his or her own behalf and on behalf of the Releasors, knowingly waives the provisions of Section 1542 of the Civil Code of the State of California, which read as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR."

# 5.8 Regulatory Matters.

- (a) Subject to the terms and conditions of this Agreement, each Party will use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Laws to consummate the Merger as soon as reasonably practicable after the date hereof, including (i) preparing and filing, in consultation with the other Party and as promptly as practicable and advisable after the date hereof and in any event within two Business Days after the date hereof, all documentation to effect notice of the Merger pursuant to section 14 of the FDI Law to the Secretary of State and, thereafter, (ii) using their reasonable best efforts to obtain as promptly as practicable all consents, clearances, waivers, licenses, orders, registrations, approvals, permits, and authorizations necessary to be obtained by such Party from any Governmental Entity under the FDI Law in order to consummate the Merger (unless the foregoing are no longer required or have been validly withdrawn). For the avoidance of doubt, it is the responsibility and obligation of the Parent to file and give notice of the Merger pursuant to section 14 of the FDI Law to the Secretary of State.
- (b) Each of Parent, the Company and Representative shall, in connection with the efforts referenced in Section 5.8(a) to use reasonable best efforts to obtain all consents, clearances, notifications, orders, approvals, and authorizations for the Merger under the FDI Law, (i) cooperate in all respects and consult with each other in connection with any notification, filing or submission and in connection with any investigation or other inquiry, including any proceeding initiated by a private party under the FDI Law, including by allowing the other Parties to have a reasonable opportunity to review in advance and comment on drafts of filings and submissions; (ii) promptly inform the other Parties of any communication received by such Party from, or given by such Party to, any Governmental Entity with respect to the FDI Law, including by promptly providing copies to the other Parties of any such written communications, and of any material communication received or given in connection with any proceeding by a private party under the FDI Law, in each case regarding the Merger; and (iii) to the extent practicable, permit the other Party to review in advance any communication that it gives to, and consult with each other in advance of any meeting, telephone call or conference with, any Governmental Entity with respect to the subject matter of this Section 5.8(b), or, in connection with any proceeding by a private party under the FDI Law, with any other Person, and to the extent not prohibited by the applicable Governmental Entity or other Person with respect to the subject matter of this Section 5.8(b), give the other Parties the opportunity to attend and participate in any in-person meetings with the applicable Governmental Entity or other Person with respect to the subject matter of this Section 5.8(b). Parent shall have the principal responsibility, with reasonable consultation with the Company and Representative, for devising and implementing the strategy under or relating to the FDI Law or approval of any Governme
- (c) In furtherance and not in limitation of the foregoing, if any administrative or judicial action or proceeding, including any proceeding by a private party, is instituted (or threatened to be instituted) challenging the Merger as violative of the FDI Law, each of the Parties hereto shall use its commercially reasonable efforts to contest and resist any such action or proceeding; provided, however, nothing herein shall require any Party or any of its Affiliates to propose, negotiate and agree to any sale, divestiture, license, holding separate or other similar arrangement with respect to, or other disposition of or restriction on, any of their, or any of its businesses, product lines, divisions or

assets or interests therein, or to take such action or actions that would have a similar effect, including agreeing to change or modify any course of conduct regarding its future operations or otherwise take actions that would limit its freedom of action with respect to, or ability to retain, one or more of its businesses, product lines, divisions or assets or interests therein; provided further, that Parent and its Affiliates shall use reasonable best efforts to comply with any requirement relating solely to the Contracts or other activities triggering the notice requirement under the FDI Law, and not to any other business of any of the Acquired Companies.

- 5.9 <u>Interim Operations of the Business</u>. From the date hereof until the Closing or the earlier termination of this Agreement, except as set forth on <u>Schedule 5.9</u>, or Parent has previously consented thereto in writing (such consent not to be unreasonably withheld, conditioned or delayed):
- (a) the Company shall, and the Company and Representative shall cause the other Acquired Companies to: (i) conduct their respective business and operations in the ordinary course of business, (ii) preserve intact and maintain, as applicable, their respective existence, business organization, properties and assets, (iii) preserve and maintain all of their respective permits and licenses, (iv) pay their respective debts, Taxes and other obligations when due (taking into account extensions), (v) continue in full force and effect without modification all insurance policies, except as required by applicable Law and (vi) use commercially reasonable efforts to keep available the services of its employees and to preserve the goodwill and present relationships (contractual or otherwise) with all customers, suppliers, employees, licensors and other Persons doing business with the applicable Acquired Company, and
- (b) without limiting the foregoing clause (a), the Company shall not, and the Company and the Representative shall not permit any Acquired Company to (i) take any action, or agree, authorize or commit (whether in writing or otherwise) to take any action that, if taken prior to the date hereof, would be required to be disclosed on Schedule 3.13 (other than any such action taken with respect to a UK Tax Contest), (ii) make or promise any increase in compensation of any Acquired Company's directors, managers, officers or employees, (iii) hire any individual with salary or overall compensation greater than \$50,000, (iv) enter into any Contract with a customer or supplier involving more than \$100,000, other than purchase and sale orders made in the ordinary course of business, (v) enter into any lease, sublease, license or other occupancy agreement, (vi) enter into any Contract with another Acquired Company or any of their Affiliates; (vii) make or commit to make any capital expenditures in excess of \$50,000 per Acquired Company; or (viii) make any NPI & EPI Development expenditures in excess of the currently planned amount of \$1,463,000.
  - 5.10 Reasonable Access; Information Technology. From the date hereof until the Closing or the earlier termination of this Agreement:
- (a) the Company shall, and the Company and Representative shall cause each Acquired Company to (i) provide Parent and Loar and their respective representatives, upon reasonable advance notice to Representative, reasonable access, during normal business hours, to inspect the assets, properties, facilities, offices, books, records, Contracts, personnel and representatives of each Acquired Company and (ii) furnish Parent and Loar and their representatives with all such information relating to the Acquired Companies and their respective businesses as Parent or Loar may from time to time reasonably request. Loar is intended to be an express third-party beneficiary of this Section 5.10(a).
- (b) subject to applicable Laws and the terms of this Agreement, the Parties shall use commercially reasonable efforts, at Parent's sole cost and expense, to cooperate to adapt the Acquired Companies' information systems, including cybersecurity evaluations and preparation for enterprise resource planning migration work, with a view to enable the Acquired Companies to operate without disruption at Closing; provided, that this Section 5.10(b) shall be disregarded for purposes of determining whether the condition set forth in Section 6.2(b) has been met.
- 5.11 Exclusivity. From the date hereof until the Closing or the earlier termination of this Agreement, neither the Company nor Representative will, and will not permit their respective Affiliates (including the other Acquired Companies) or any representative of any of the foregoing to, (a) solicit, initiate or encourage the submission of any proposal or offer from any Person relating to the

acquisition of any Equity Interests or any assets of any Acquired Company (whether by merger, purchase or sale of Equity Interests or assets or otherwise) (an "Alternative Proposal"), (b) enter into, maintain or continue discussions or negotiations regarding, or furnish or disclose to any Person any information in connection with, any Alternative Proposal, (c) enter into any binding or non-binding letter of intent, term sheet, memorandum of understanding, purchase agreement, merger agreement or other similar agreement with any Person with respect to any Alternative Proposal or (d) otherwise cooperate in any way with, or assist or participate in, facilitate or encourage, any effort or attempt by any other Person, including Jefferies LLC and Lincoln International, to do or seek any of the foregoing, in each case, other than Parent and its Affiliates and representatives. Representative will notify Parent promptly after receipt by any Acquired Company, Representative or a Former Equityholder of any proposal or offer from any Person other than Parent to effect an Alternative Proposal.

- 5.12 Parachute Payments. If required to avoid the imposition of Taxes under Section 4999 of the Code or the loss of deduction under Section 280G of the Code and the Treasury Regulations promulgated thereunder (collectively, "Section 280G") with respect to any payment or benefit in connection with the Merger and to the extent that any individual who could be considered to be a "disqualified individual" (within the meaning of Section 280G(c) of the Code and the Treasury Regulations promulgated thereunder) has the right to receive or retain any payments or benefits that could result in an "excess parachute payment" (within the meaning of Section 280G(b)(1) of the Code and the Treasury Regulations promulgated thereunder) (any such payments or benefits, together, the "Section 280G Payments"), the Company will cause Desser to (a) solicit from each such "disqualified individual" a waiver of such disqualified individual's right to receive or retain some or all of such Section 280G Payments such that any remaining Section 280G Payments would not result in "excess parachute payments" (within the meaning of Section 280G(b)(1) of the Code and the Treasury Regulations promulgated thereunder) (the "Waived 280G Benefits") and (b) prior to the Closing Date, with respect to each such individual, solicit the vote of the holders of voting power with respect to Desser in the manner required under Section 280G(b)(5)(ii) of the Code and the Treasury Regulations promulgated thereunder, including Q-7 of Section 1.280G-1 of such Treasury Regulations, the right of such "disqualified individual" to receive or retain the Waived 280G Benefits. In connection with the foregoing, Parent will provide Desser with descriptions of any Section 280G Payments made or to be made, or benefits granted or to be granted, pursuant to any agreement, arrangement or contract entered into, modified or negotiated by Parent or any of its Affiliates to the extent such amounts (when taken into account along with any other Section 280G Payments that could be received by the applicable individual in connection with any of the Transactions) would result in an "excess parachute payment" (within the meaning of Section 280G(b)(1) of the Code and the Treasury Regulations promulgated thereunder) prior to the Closing Date. The Company will cause Desser to provide Parent and its counsel for review drafts of the waivers and disclosure statement and any mathematical analysis of the Section 280G Payments contemplated by this Section 5.12 prior to delivery to each disqualified individual and the holders of voting power with respect to Desser entitled to vote on such matters of such materials, and Desser will consider in good faith any changes reasonably requested by Parent or its counsel.
- 5.13 Obligations Related to Employee Plan Matters. Prior to the Closing Date, the Company shall use commercially reasonable efforts to deliver to Parent a fully executed copy of the Section 125 cafeteria plan document that is sponsored by an Acquired Company and covers the employees of all the Acquired Companies. Prior to the Closing Date (and in all events prior to August 1, 2023), the Company shall use commercially reasonable efforts to (i) submit a Voluntary Correction Program application (in a manner and form approved by Parent, such approval not to be unreasonably withheld, conditioned or delayed) pursuant to the IRS Employee Plans Complaince Resolution System (EPCRS) in connection with any failures related to an Acquired Company or Controlled Group member sponsoring or maintaining a SIMPLE IRA and (ii) correct (in a manner and form reasonably acceptable to Parent) the failure of the Acquired Companies to properly file and furnish any form under Code Sections 6055 or 6056 with respect to all years prior to 2023.
- 5.14 <u>Notice of Certain Events</u>. If any of the representations in <u>Section 4.8</u> cease to be true and correct prior to the Closing Date, Parent shall promptly, and in any event within two (2) Business

Days after it becomes aware of such fact, notify the Representative in writing of the circumstances causing the representation to not be true and correct.

#### ARTICLE 6. CONDITIONS PRECEDENT

- 6.1 <u>Conditions to Obligations of the Company and Representative</u>. The obligations of the Company and Representative to consummate the transactions contemplated hereby are subject to the satisfaction, or waiver by Representative (if permitted by applicable Law), at or prior to the Closing of each of the following conditions:
- (a) The representations and warranties of Parent and Merger Sub set forth in this Agreement must be true and correct in all respects as of the date hereof and as of the Closing Date as though made on and as of the date hereof and as of the Closing Date (except to the extent expressly made as of an earlier date, in which case, as of such date), except where the failure of such representations and warranties to be so true and correct has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on Parent's and Merger Sub's ability to timely consummate the transactions contemplated hereby.
- (b) Parent and Merger Sub must have performed, satisfied and complied with in all material respects all obligations, covenants and agreements required to be performed, satisfied and complied with by them under this Agreement at or prior to the Closing.
  - (c) Parent must have delivered or caused to be delivered to Representative the items required by Section 2.3(b) through (e).
- 6.2 <u>Conditions to Obligations of Parent and Merger Sub</u>. The obligations of Parent and Merger Sub to consummate the transactions contemplated hereby are subject to the satisfaction or waiver (if permitted by applicable Law) at or prior to the Closing of each of the following conditions:
- (a) The (i) Fundamental Representations must be true and correct in all respects as of the date hereof and as of the Closing Date as though made on and as of the date hereof and as of the Closing Date (except to the extent expressly made as of an earlier date, in which case, as of such date), and (ii) the remaining representations and warranties of the Company must be true and correct in all respects (without giving effect to any materiality or Material Adverse Effect or similar materiality qualifiers contained therein) as of the date hereof and as of the Closing Date as though made on and as of the date hereof and as of the Closing Date (except to the extent expressly made as of an earlier date, in which case, as of such date), except where the failure of such representations and warranties to be true and correct would not have a Material Adverse Effect.
- (b) The Company and Representative must have performed, satisfied and complied with in all material respects all obligations, covenants and agreements required to be performed, satisfied and complied with by it under this Agreement at or prior to the Closing.
  - (c) The Company and Representative must have delivered or caused to be delivered to Buyer the items required by Section 2.2.
  - (d) There has been no Material Adverse Effect.
- (e) Loar stands ready, willing and able to proceed with the closing of the Loar Transaction immediately after the Closing; provided, that this condition shall be deemed to have been satisfied or waived unless, within one Business Day of Parent receiving written notice from the Company and Representative that (i) each of the conditions set forth in Sections 6.1, 6.2 and 6.3 have been satisfied or waived (other than those conditions intended to be fulfilled at the Closing) and (ii) that the Company and Representative stand ready, willing and able to proceed to Closing in accordance with Section 2.1, Parent certifies to the Representative in writing, that (x) all conditions to Loar's obligation to close the Loar Transaction have been satisfied or waived (other than those conditions intended to be fulfilled at the closing of the Loar Transaction), and (y) Loar has either confirmed to Parent in writing that Loar does not stand ready, willing and able to proceed with the Loar Transaction or has not confirmed to

Parent in writing that Loar stands ready, willing and able to proceed with the Loar Transaction (a "Non-Closing Notice").

In the event the Representative receives a Non-Closing Notice from Parent, Representative may, at its option, elect within five Business Days following its receipt of a Non-Closing Notice to take any one or more of the following actions: (A) require Parent to join Representative and the Company in seeking any applicable legal or equitable remedies from Loar, including specific performance or similar remedies under the Loar Transaction Documents; (B) terminate this Agreement in accordance with Section 7.1(f); or (C) require Parent to negotiate an alternative transaction with the Company and the Representative pursuant to which Parent would acquire those Acquired Companies not subject to the Loar Transaction at the price (net of the portion of the purchase price payable by Loar pursuant to the Loar Transaction (including, for the avoidance of doubt, the Earn Out Payments, which would not be payable by Parent)) and on the terms set forth herein, with such changes as are reasonably necessary to facilitate an appropriate alternative transaction structure (including any necessary changes to post-closing commercial arrangements among the parties to the alternative transaction). The parties agree that, in the event that an alternative transaction is not entered into within thirty (30) days after a Non-Closing Notice, the restrictions in Section 5.11 shall terminate and be of no further force or effect.

- 6.3 <u>Conditions to Obligations of All Parties</u>. The obligations of the Parties to consummate the transactions contemplated hereby are subject to the satisfaction or waiver (if permitted by applicable Law) at or prior to the Closing of each of the following conditions:
- (a) None of the Parties will be subject to any Order or Law that prohibits or makes illegal the consummation of the transactions contemplated hereby.
  - (b) No Proceeding will have been instituted and remain pending to threaten or restrain, prohibit or delay the Transactions.
- (c) Either: (a) following the notification of the Merger in accordance with the requirements of the FDI Law and this Agreement under Section 5.8, the Secretary of State will have notified the Parent (and/or other "relevant person" (as such term is defined under the FDI Law)) pursuant to section 14(8)(b)(ii) of the FDI Law that no further action will be taken in relation to the Merger; (b) in the event that a call-in notice under section 1(1) of the FDI Law is given by the Secretary of State in relation to the Merger, the Secretary of State will have either: (i) given a final notification confirming that no further action will be taken in relation to the Merger under the FDI Law, or (ii) made a final order under section 26(3) of the FDI Law permitting the Merger to proceed, subject only to such conditions or obligations that are limited solely to the Contracts or other activities triggering the notice requirement under FDI Law (and not to any other business of any of the Acquired Companies), taken as a whole (and to the extent relevant, any such conditions or obligations contained in such an order necessary for the Closing of the Merger having been satisfied or complied with); or (c) such approvals or clearances are no longer required or have been validly withdrawn.

#### ARTICLE 7. TERMINATION

- 7.1 <u>Termination</u>. Notwithstanding any other provision of this Agreement, this Agreement may be terminated at any time prior to the Closing only as follows:
  - (a) by the mutual written consent of Parent and Representative;
- (b) by Parent, on the one hand, or Representative, on the other hand, upon written notice to the other, if the transactions contemplated hereby have not been consummated on or prior to October 31, 2023 (the "Termination Date"); provided, however, that the right to terminate this Agreement pursuant to this Section 7.1(b) is not available to any Party whose material breach of any provision of this Agreement is a principal cause of the failure of the transactions to be consummated by such time;

- (c) by Parent, on the one hand, or Representative, on the other hand, upon written notice to the other, if a Governmental Entity has issued an Order or any other action permanently enjoining or otherwise prohibiting the consummation of the transactions contemplated hereby, and such Order has become final and non-appealable; provided, however, that the right to terminate this Agreement pursuant to this Section 7.1(c) is not available to any Party whose material breach of any provision of this Agreement is a principal cause of such Order;
- (d) by Representative if: (i) Parent or Merger Sub has breached or failed to perform any of its obligations, covenants or agreements contained in this Agreement to be complied with by Parent or Merger Sub, as applicable, such that the closing condition set forth in <u>Section 6.1(b)</u> would not be satisfied; or (ii) there exists a breach of any representation or warranty of Parent or Merger Sub contained in this Agreement such that the closing condition set forth in <u>Section 6.1(a)</u> would not be satisfied, and, in the case of clauses (i) and (ii) of this <u>Section 7.1(d)</u>, such breach or failure to perform is not cured within five days after receipt of written notice thereof from Representative or is incapable of being cured by Parent or Merger Sub by the Termination Date;
- (e) by Parent if: (i) the Company or Representative has breached or failed to perform any of its obligations, covenants or agreements contained in this Agreement to be complied with by them such that the closing condition set forth in Section 6.2(b) would not be satisfied; or (ii) there exists a breach of any representation or warranty of the Company contained in this Agreement such that the closing condition set forth in Section 6.2(a) would not be satisfied, and, in the case of clauses (i) and (ii) of this Section 7.1(e), such breach or failure to perform is not cured within five days after receipt by Representative of written notice thereof from Parent or is incapable of being cured by the Company or Representative, as applicable, by the Termination Date;
  - (f) by Representative following the receipt of a Non-Closing Notice; or
  - (g) by Parent if there has been a Material Adverse Effect.
- 7.2 <u>Effect of Termination</u>. In the event of termination of this Agreement pursuant to <u>Section 7.1</u> by either Parent or Representative, this Agreement will become void and have no effect, without any liability or obligation on the part of any party hereto, other than the provisions of this <u>Section 7.2</u> and <u>Article 10</u>, which will survive any termination of this Agreement; provided, however, that nothing herein will relieve any party hereto from any liability for any Fraud or willful breach of this Agreement occurring prior to such termination.

#### ARTICLE 8. NO SURVIVAL; EXCLUSIVE REMEDY; NONRECOURSE

8.1 No Survival. None of the representations and warranties of the Company, the Representative or Parent contained in this Agreement shall survive the Closing (except in the case of a claim based upon or arising out of Fraud). Except (a) pursuant to and in accordance with the R&W Insurance Policy and (b) with respect to claims based upon or arising out of Fraud, no Person shall be entitled to any recovery in respect of, or to make any claim whatsoever (including any claim of detrimental reliance or breach of contract or otherwise) with respect to, or to assert any other right or remedy (whether in contract, in tort, in equity or at law or otherwise) regarding, any inaccuracy or breach of any representations or warranties contained in this Agreement. Except for claims based upon or arising out of Fraud, (a) the sole recourse for any Losses of Parent, the Surviving Company and/or their respective Affiliates and/or agents (each, a "Parent Indemnitee" and together, the "Parent Indemnitees") resulting from, or that exists or arises due to any inaccuracy in or breach of any of the representations and warranties contained in this Agreement, shall be from the proceeds (if any) that may be available under the R&W Insurance Policy, and (b) no Parent Indemnitee shall have any recourse against Representative, any Affiliate thereof, or any Company Affiliate with respect to any such Losses or for any claim based on, in respect of, or by reason of any inaccuracy, omission or breach of any representations or warranties in this Agreement, regardless as to whether or not all or some of any such Loss or claim is covered by the R&W Insurance Policy or denied by the insurer thereunder or whether or not the R&W Insurance Policy has expired, been terminated or lapsed. For the avoidance of doubt, this

Section 8.1 does not apply to any recovery of the Parent Indemnitees under the indemnities provided for in Section 5.3(h) and Section 5.5(b).

#### 8.2 EXCLUSIVE REMEDY; NONRECOURSE.

- (a) Notwithstanding anything else contained in this Agreement to the contrary, except with respect to claims based upon or arising out of Fraud, the provisions and limitations of the R&W Insurance Policy shall be the sole and exclusive remedy with respect to any and all claims by Parent Indemnitees relating to any inaccuracy, omission or breach of any representations or warranties in this Agreement. Without limiting the generality or effect of the foregoing, Parent (for itself and on behalf of each other Parent Indemnitee) hereby (i) acknowledges and agrees that the limitations on liability regarding any inaccuracy, omission or breach of any representations or warranties in this Agreement set forth herein were expressly bargained for and relied upon by, and are a material inducement for, the Company and Representative to enter into this Agreement and consummate the transactions contemplated hereby, and (ii) except for claims based on or arising out of Fraud, knowingly, voluntarily and irrevocably waives and releases any and all rights and remedies to which any such Parent Indemnitee would otherwise be entitled (whether known or unknown, foreseen or unforeseen, in contract, in tort, in equity or at law or otherwise) against Representative, any Affiliate thereof and any Company Affiliate with respect to any inaccuracy, omission or breach of any representations or warranties in this Agreement. No Parent Indemnitee may avoid the exclusion from liability of the Company Affiliates set forth in this Agricle 6 by seeking damages for breach of contract, tort or pursuant to any other theory of liability.
- (b) The provisions of this <u>Article 6</u>, including the limitations on remedies provided herein, were specifically bargained for by the Parties at arms' length and were taken into account by the Parties in agreeing on the amount of the Merger Consideration and the other terms and conditions of this Agreement, and without such limitations the Parties would not have entered into this Agreement and consummated the transactions contemplated hereby.
- (c) The Parties hereto expressly intend and agree that the provisions of this <u>Article 6</u> alter any applicable Laws and any statutes of limitation thereunder.
- 8.3 <u>Materiality Scrape</u>. For purposes of determining whether there has been any inaccuracy, omission or breach of a representation or warranty, and for purposes of determining the amount of Losses resulting therefrom, all qualifications or exceptions in any representation or warranty relating to or referring to the terms "Material Adverse Effect", "material", "materiality", "in all material respects" or any similar term or phrase shall be disregarded and shall be read as if such terms and phrases were not included in them; <u>provided</u>, <u>however</u>, that the foregoing provisions of this <u>Section 8.3</u> shall be deemed not to apply to the representations and warranties in the second sentence of <u>Section 3.5</u> and the first sentence of <u>Section 3.13</u>.

#### ARTICLE 9. DEFINITIONS

Whenever used in this Agreement, the following terms and phrases shall have the following respective meanings:

"2022 Financial Statements" has the meaning set forth in Section 3.5.

"Accounting Firm" has the meaning set forth in Section 1.5(d)(ii).

"Accounting Principles" means (a) the accounting policies, processes, methodologies, practices, estimation techniques, assumptions and principles set forth on Exhibit I, and (b) to the extent not addressed in Exhibit I, GAAP; provided, however, that if and to the extent there is any inconsistency between Exhibit I, on the one hand, and GAAP, on the other hand, then Exhibit I shall control.

"Acquired Companies" has the meaning set forth in the recitals.

- "Actual Adjustment" means (x) the Closing Merger Consideration as set forth on the Final Statement of Closing Merger Consideration minus (y) the Estimated Closing Merger Consideration.
- "Affiliate" means, with respect to any Person, (i) each Person that controls, is controlled by or is under common control with such Person, (ii) each of such Person's officers, directors, stockholders, joint venturers and partners and (iii) each member of the immediate family of any Person that is a natural person.
- "Aggregate Option Exercise Price" means the aggregate exercise price that would be payable upon the exercise of the vested portion of all Unit Options immediately prior to the Effective Time.
  - "Agreement" has the meaning set forth in the preamble.
  - "Allocation" has the meaning set forth in Section 5.3(c)(ii).
  - "Allocation Schedule" has the meaning set forth in Section 5.3(c)(ii).
  - "Alternative Proposal" has the meaning set forth in Section 5.11.
- "Amended and Restated LP Agreement" means that certain Amended and Restated Limited Partnership Agreement to be entered into by the Company and the other parties thereto on the date hereof.
- "Anti-Corruption Laws" means the United States Foreign Corrupt Practices Act of 1977, the UK Bribery Act 2010, any national and international Law enacted to implement the OECD Convention on Combating Bribery of Foreign Officials in International Business Transactions, or any other applicable anti-corruption, anti-bribery or anti-money laundering Law.
  - "AOG Aviation" has the meaning set forth in Section 2.2(k).
  - "AOG-Seginus" has the meaning set forth in Section 2.2(k).
  - "Audited Financial Statements" has the meaning set forth in Section 3.5.
  - "Authorizations" has the meaning set forth in Section 3.15(b).
  - "Balance Sheet" has the meaning set forth in Section 3.5.
  - "Balance Sheet Date" has the meaning set forth in Section 3.5.
  - "Base Merger Consideration" has the meaning set forth in Section 1.5(a).
- "Business Day" means any day other than Saturday, Sunday or any other day on which banks in Philadelphia, Pennsylvania are authorized or required by applicable Law to close.
- "Cash and Cash Equivalents" means the sum (expressed in United States dollars) of all cash and cash equivalents (including bank account balances, marketable securities and short term investments) of the Company and its Subsidiaries as determined in accordance with GAAP. "Cash and Cash Equivalents" shall be calculated (i) net of issued but uncleared checks, wires and drafts issued by the Company or any of its Subsidiaries, provided that the accounts payable to which such checks, wires and drafts relate are correspondingly reduced, (ii) including received and uncleared checks, wires or drafts of the Company and its Subsidiaries and (iii) excluding security deposits and any other cash or cash equivalents that are subject to any contractual restriction on the ability to freely transfer or use such cash or cash equivalents.

"CARES Act and COVID Relief Programs" means, collectively, The Coronavirus Aid, Relief, and Economic Security Act, the Families First Coronavirus Response Act, the Paycheck Protection Program Flexibility Act, any rules and regulations of the U.S. Small Business Association, the U.S. Department of the Treasury or any public health agency or other Governmental Entity (including outside of the United States), the UK Coronavirus Job Retention Scheme and any applicable Law in connection with the COVID-19 pandemic, and all FAQs or interim final rules issued by any Governmental Entity related thereto.

- "Certificate of Merger" has the meaning set forth in Section 1.2.
- "Closing" has the meaning set forth in Section 3.1.
- "Closing Date" has the meaning set forth in Section 3.1.
- "Closing Date Cash and Cash Equivalents" means Cash and Cash Equivalents as of the Measurement Time.
- "Closing Date Funded Indebtedness" means the Funded Indebtedness as of immediately prior to Closing.
- "Closing Merger Consideration" has the meaning set forth in Section 1.5(a).
- "Closing Proceeds" means, for each Former Equityholder, an amount equal to (i) such Former Equityholder's Ownership Percentage multiplied by (ii) the Aggregate Closing Proceeds; provided that, in the case of each Option Holder, such product shall be reduced by the aggregate exercise price of the Unit Options held by such Option Holder. The total Closing Proceeds to be paid to all Former Equityholders shall equal the Aggregate Closing Proceeds
  - "Code" means the Internal Revenue Code of 1986, as amended.
  - "Company" has the meaning set forth in the preamble.
- "Company Affiliate" means any director, officer, employee, incorporator, member, partner, stockholder, lender, Affiliate, agent, attorney or representative of any Acquired Company at any time during the period prior to Closing.
  - "Company Certificate" means the Company's Certificate of Limited Partnership, in effect on the date hereof.
  - "Company Equity Interests" means, collectively, the Preferred Interests and the Original Partner Interests.
- "Company Data" means all customer, employee, vendor or business partner data used, Processed, and/or hosted by or on behalf of any Acquired Company, whether provided by the Acquired Company or any other Person.
- "Company Intellectual Property Rights" means all Intellectual Property Rights owned or purported to be owned in whole or in part by any Acquired Company.
  - "Company Leases" has the meaning set forth in Section 3.6(b).
  - "Company Registered IP" has the meaning set forth in Section 3.11(a).
  - "Continuing Employees" has the meaning set forth in Section 5.5.
  - "Contract" has the meaning set forth in Section 3.12(a).

"Controlled Group" means any trade or business (whether or not incorporated) (i) under common control within the meaning of Section 4001(b)(1) of ERISA with any Acquired Company or (ii) which together with any Acquired Company is treated as a single employer under Section 414(t) of the Code.

"Counsel" has the meaning set forth in Section 10.13.

"Credit Agreement" means that certain Fourth Amended and Restated Business Loan and Security Agreement, dated as of January 5, 2018, by and among VSE Corporation, each of its subsidiaries party thereto as borrowers, the banks and other financial institutions party thereto as Lenders, and Citizens Bank, National Association (as successor by merger to Citizens Bank of Pennsylvania), as Administrative Agent (as amended by that certain First Amendment to Fourth Amended and Restated Business Loan and Security Agreement, dated as of November 26, 2019, that certain Second Amendment to Fourth Amended and Restated Business Loan and Security Agreement, dated as of June 29, 2020, that certain Third Amendment to Fourth Amended and Restated Business Loan and Security Agreement, dated as of July 23, 2021, that certain Fourth Amendment to Fourth Amended and Restated Business Loan and Security Agreement, dated as of October 7, 2022).

"DAC" has the meaning set forth in Section 2.2(k).

"Desser" has the meaning set forth in the recitals.

"Desser 401(k) Plan" has the meaning set forth in Section 5.13.

"<u>Disclosure Schedule</u>" means the disclosure schedule, dated the date hereof, delivered to Parent by the Company in connection with this Agreement.

"Dispute" has the meaning set forth in Section 10.7.

"<u>Distribution</u>" means (i) any payment made to the Former Equityholders as a result of the determination of the Final Statement of Closing Merger Consideration, (ii) any payment made to the Former Equityholders of all or any amounts released from the Merger Consideration Adjustment Escrow Account, (iii) any Earn Out Payment made to the Former Equityholders, and (iv) any other payment made to the Former Equityholders (in their capacity as such) under this Agreement after the Closing Date.

"DLPA" has the meaning set forth in Section 1.1.

"Earn Out Agreement" that certain agreement dated as of the date hereof by and among Representative, Parent and Loar as set forth in Exhibit J.

"Earn Out Payments" has the meaning set forth in Section 1.5(a).

"Effective Time" has the meaning set forth in Section 1.2.

"Electronic Delivery" has the meaning set forth in Section 10.11.

"Employee Plan" has the meaning set forth in Section 3.14(a).

"Employee Matters Escrow Account" has the meaning set forth in Section 1.5(c)(i).

"Employee Matters Escrow Issue" has the meaning set forth in Section 5.5(b).

"Environmental Claim" has the meaning set forth in Section 3.16(h).

"Environmental Law" has the meaning set forth in Section 3.16(h).

"Equity Interests" means, with respect to any Person, all of the shares of capital stock of (or other ownership or profit interests in) such Person, all of the warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, all of the securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or such other interests), and all of the other ownership or profit interests in such Person (including partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such shares, warrants, options, rights or other interests are outstanding on any date of determination.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, or any successor Law, and the regulations and rules issued pursuant to that act or to any successor Law.

"Escrow Agent" means U.S. Bank National Association, a national banking association.

"Escrow Agreement" means the escrow agreement among Parent, the Representative and the Escrow Agent, in the form attached hereto as Exhibit K.

"Estimated Closing Date Funded Indebtedness" has the meaning set forth in Section 1.5(b).

"Estimated Closing Merger Consideration" has the meaning set forth in Section 1.5(b).

"Expense Funds" has the meaning set forth in Section 1.6(d).

"FDI Law" means the National Security and Investment Act 2021 (United Kingdom) (and any regulations made or issued thereunder).

"Final Statement of Closing Merger Consideration" has the meaning set forth in Section 1.5(d)(ii).

"Financial Statements" has the meaning set forth in Section 3.5.

"Flow-Through Tax Return" means any Tax Return of an Acquired Company filed or required to be filed for a Pre-Closing Tax Period with respect to any U.S. federal (and applicable state and local) Tax that is based on or measured by income or gains (however denominated, and including any franchise, business profits or similar Tax incurred in lieu of a Tax on income) for which a Former Equityholder (or any direct or indirect owner(s) of any Former Equityholder) will be liable as a matter of Law, including, for the avoidance of doubt, IRS Form 1065 and Schedule K-1. For the avoidance of doubt, no Tax Return of a Non-U.S. Acquired Company shall constitute a "Flow-Through Tax Return."

"Former Equityholders" means the holders of Company Equity Interests or Unit Options on the Closing Date as of immediately prior to the Effective Time.

"Fraud" means actual common law fraud under the Laws of the State of Delaware in the making of the representations and warranties in this Agreement or any other Transaction Document (which, for the avoidance of doubt, shall not include any equitable fraud, constructive fraud or fraud arising out of negligence or recklessness).

"Fully-Diluted Units" means the number of Units outstanding immediately prior to the Effective Time, assuming the exercise of the vested portion of all outstanding Unit Options immediately prior to the Effective Time.

"Fundamental Representations" means those representations and warranties set forth in Sections 3.1 (Organization and Good Standing; Holding Company), 3.3 (Capitalization; Investments and Subsidiaries), 3.4 (Power and Authorization) and 3.20 (Brokers).

"Funded Indebtedness" means, as of any date, without duplication, the outstanding principal amount of, accrued and unpaid interest on and other obligations (including any prepayment premiums payable as a result of the consummation of the Merger) arising under any obligations of any Acquired Company for (i) borrowed money, (ii) any note, bond, debenture or other similar instrument or debt security, (iii) under leases which in accordance with Accounting Principles have been or should be recorded as a capital lease, (iv) letters of credit and bankers' acceptances only to the extent drawn, (v) the deferred purchase price of any property or services (other than trade accounts payable and accrued expenses incurred in the ordinary course of business and reflected as accounts payable or accrued expenses in Net Working Capital, as finally determined pursuant to Section 1.5), including earn-outs, payments under non-compete agreements and seller notes, (vi) conditional sale or other title retention agreements, (vii) without duplication of any amounts treated as Seller Expenses, all unpaid income and other similar Tax liabilities of the Acquired Companies for Pre-Closing Tax Periods (calculated consistently with the past practices of the Acquired Companies and on a jurisdiction-by-jurisdiction basis, taking into account (A) only those jurisdictions in which the Acquired Companies are currently filing income and similar Tax Returns, and (B) any prepayments, estimated or similar Tax payments, and any Tax overpayments and refunds for such Pre-Closing Tax Periods (calculated by taking into account any applicable Transaction Tax Deductions) in each such jurisdiction), treating for this purpose the taxable year of or with respect to each of the Non-U.S. Acquired Companies, and any entity treated as a partnership for U.S. federal income tax purposes in which any Acquired Company holds an interest, as ending as of the end of the Closing Date (including Taxes the due date for which has been deferred until after the Closing Date pursuant to the CARES Act and COVID Relief Programs and any U.S. state or local "pass through entity" or similar Taxes imposed on the Company), excluding any and all UK Exchange Taxes and UK Withholding Taxes, and provided that, for the avoidance of doubt, if and to the extent that such amount in a jurisdiction is less than zero due to the items identified in subclause (B) hereof, such negative amount shall be netted against any and all positive amounts in other jurisdictions, or (viii) liabilities of any Person (other than the Company or one of its Subsidiaries) that are directly or indirectly guaranteed by the Company or one of its Subsidiaries or in respect of which the Company or such Subsidiary has otherwise assured an obligee against loss. Notwithstanding the foregoing, Funded Indebtedness does not include (A) any operating lease obligations, (B) any performance bonds, bankers' acceptances or similar obligations to the extent undrawn, (C) any intercompany obligations between or among the Company and any of its Subsidiaries and (D) any obligations in respect of any financing obtained by Parent in connection with the transactions contemplated hereby.

"GAAP" means generally accepted accounting principles in the United States as in effect on the dates of the respective Financial Statements consistently applied in accordance with past practice; provided, however, to the extent more than one practice is permitted by GAAP, the practice historically employed by the Acquired Companies shall apply.

"General Partner" has the meaning set forth in the recitals.

"General Partner Interest" has the meaning set forth in the LP Agreement.

"Governmental Entity" means any United States or foreign national, state or local government or any court of competent jurisdiction, administrative agency, board, commission, bureau, tribunal or other governmental authority or instrumentality or political subdivision thereof, or any quasi-governmental or private body exercising or entitled to exercise any regulatory, administrative, executive, judicial, legislative or taxing authority thereunder (and, in respect of the FDI Law, means the relevant Secretary of State).

"Government Bid" means any offer, quotation, bid or proposal which, if accepted or awarded, would result in a Government Contract.

"Government Contract" means any prime contract, subcontract, basic ordering agreement, purchase order, task order, delivery order, change order, or any other agreement between any Acquired Company and (i) any Governmental Entity, (ii) any prime contractor of a Governmental Entity in its capacity as a prime contractor, or (iii) any subcontractor with respect to any Contract described in clause (i) or (ii) hereto.

- "Hazardous Materials" has the meaning set forth in Section 3.16(h).
- "Intellectual Property Rights" means any and all intellectual property rights in any jurisdiction, whether registered or unregistered, pertaining to or deriving from: (i) patents and patent applications, reexaminations, reissues, continuations, continuations in part, renewals, substitutes and extensions, and invention disclosures, inventions and discoveries; (ii) computer software; (iii) copyrights, data, databases, and works of authorship; (iv) trade secrets, know-how and confidential information ("Trade Secrets"); (v) trademarks, trade names, service marks, certification marks, service names, brands, trade dress and logos and the goodwill associated therewith ("Trademarks"); (vi) domain names, URLs and social media accounts; and (vii) any other intellectual property, industrial and proprietary rights.
  - "Intended Tax Treatment" has the meaning set forth in Section 5.3(c)(i).
  - "Interim Financial Statements" has the meaning set forth in Section 3.5.
- "IT Asset" means any technology device, computer, software, server, network or other information technology asset or equipment, and all data stored therein or processed thereby, and all associated documentation.
- "Knowledge" means, with respect to a particular matter, the actual knowledge, after reasonable inquiry, of, with respect to the Company or the Representative, Jeffrey Johnston and Adam Gallo (each, a "Company Knowledge Party").
- "Law" means any law, statute, regulation, permit, license, certificate, Order, award, code, ordinance, rule, judicial or similar other decision or requirement of any Governmental Entity.
- "Lien" means any mortgage, pledge, security interest, encumbrance, lien or charge of any kind, including, without limitation, any conditional sale or other title retention agreement, any lease in the nature thereof and the filing of or agreement to give any financing statement under the Uniform Commercial Code of any jurisdiction and including any lien or charge arising by statute or other Laws, including those that secure the payment of a debt (including, without limitation, any Tax) or the performance of an obligation.
  - "Limited Partner" has the meaning set forth in the recitals.
  - "Loar" has the meaning set forth in the recitals.
  - "Loar Transaction" has the meaning set forth in Section 4.8(b).
  - "Loar Transaction Documents" has the meaning set forth in Section 4.8(b).
- "Losses" means all losses, damages, liabilities, fines, penalties, claims, judgments, awards, settlements, Taxes and costs and expenses and disbursements (including settlement costs, court costs and any reasonable legal expenses); provided, however, that Losses shall not include punitive or exemplary damages, except to the extent actually paid in connection with a third-party claim.
  - "LP Agreement" has the meaning set forth in the recitals.
- "Material Adverse Effect" means an event, occurrence, fact, condition or change that has, or would reasonably be expected to have, a material adverse effect on (a) the operations, properties or financial condition of (x) the Acquired Companies, taken as a whole or (y) AOG-Seginus, AOG Aviation, Seginus Aerospace and DAC, taken as a whole or (b) the ability of Representative or the Company to consummate the transactions contemplated hereby on a timely basis; provided, however, that with respect to the foregoing clause (a), none of the following will be deemed, either alone or in combination, to constitute, and none of the following will be taken into account in determining whether there has been or will be, a "Material Adverse Effect": (i) events affecting the economy of the United States, the other

geographic areas in which the Acquired Companies operate or the global economy generally (including changes in the securities, financial or banking markets, interest rates or the prices of securities generally); (ii) changes in conditions in the industries in which the Acquired Companies conduct business; (iii) changes in Laws or in (A) the authoritative interpretations thereof or (B) regulatory guidance related thereto; (iv) changes in GAAP, other accounting requirements or any Laws or the enforcement, implementation or interpretation of any of the foregoing; (v) natural disasters, man-made disasters, acts of war, sabotage, terrorism, hostilities, military action or any escalation or worsening thereof; (vi) any disease, public health emergency, pandemic (e.g., COVID-19), or endemic or acts of any Governmental Entity in response to any of the foregoing matters identified in this clause (vi), including domestic or international travel bans or restrictions on travel operations; (vii) any failure by the Acquired Companies to meet any internal projections, forecasts or revenue or earnings predictions (provided that the underlying facts or basis for any failure to meet such projections, forecasts or revenue or earnings predictions may be taken into account in determining whether there is a Material Adverse Effect); (viii) the announcement, pendency or completion of the transactions contemplated by this Agreement; (ix) any matter referred to on the Disclosure Schedules (but not any worsening of or change in circumstances with respect thereto); and (x) any action expressly required by this Agreement or any action taken (or omitted to be taken) with the written consent of or at the written request of Parent; provided, further, that any event, occurrence, fact, condition or change referred to in clauses (i) through (viii) immediately above shall be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur to the extent that such event, occurrence, fact, con

- "Measurement Time" means 12:01 am Pacific Time on the Closing Date.
- "Merger" has the meaning set forth in the recitals.
- "Merger Consideration" has the meaning set forth in Section 1.5(a).
- "Merger Consideration Adjustment Escrow Account" has the meaning set forth in Section 1.5(c)(i).
- "Merger Consideration Dispute Notice" has the meaning set forth in Section 1.5(d)(ii).
- "Merger Consideration Adjustment Escrow Funds" has the meaning set forth in Section 1.5(c)(i).
- "Merger Sub" has the meaning set forth in the preamble.
- "Merger Sub Capital Stock" has the meaning set forth in Section 1.3(b).

"Net Working Capital" means all current assets <u>less</u> all current liabilities of the Company and its Subsidiaries on a consolidated basis, in each case in accordance with the Accounting Principles. For the avoidance of doubt, the calculation of Merger Consideration will be determined in all instances in accordance with the Accounting Principles, and without duplication to any items counted in such calculation. The parties agree that the purpose of preparing and calculating the Net Working Capital hereunder is to measure changes in Net Working Capital in a manner consistent with the Accounting Principles, without including any changes in assets or liabilities as a result of purchase price accounting adjustments or other changes arising from or resulting as a consequence of the transactions contemplated hereby. Notwithstanding anything to the contrary contained herein, "Net Working Capital" shall not include any amounts: (i) with respect to UK Exchange Taxes, UK Withholding Taxes, income Tax assets, income Tax liabilities, deferred Tax assets or deferred Tax liabilities or (ii) included in Cash and Cash Equivalents, Funded Indebtedness or Seller Expenses, as finally determined pursuant to Section 1.5. A sample calculation of Net Working Capital is attached hereto as Exhibit E.

"Net Working Capital Adjustment" means (i) the amount by which Net Working Capital as of the Measurement Time exceeds the Reference Amount or (ii) the amount by which Net

Working Capital as of the Measurement Time is less than the Reference Amount; <u>provided</u> that any amount which is calculated pursuant to clause (ii) above shall be deemed to be a negative number.

- "Non-Closing Notice" has the meaning set forth in Section 6.2(e).
- "Non-U.S. Acquired Companies" has the meaning set forth in Section 5.3(c)(i).
- "<u>Open Source Software</u>" means any software that is subject to an open source or copyleft license, including any license approved as an open source license by the Open Source Initiative (opensource.org) or any variant or derivative thereof.
  - "Option Holders" means holders of Unit Options as of immediately prior to the Effective Time.
  - "Option Payments" has the meaning set forth in Section 1.4.
  - "Order" means any judgment, award, determination, order, writ, injunction or decree of any Governmental Entity.
- " $\underline{Ownership\ Percentage}$ " means, for each Former Equityholder, the "Ownership Percentage" set forth across from such Person's name on  $\underline{Appendix\ I}$ .
  - "Original Partner Interests" has the meaning set forth in the LP Agreement.
  - "Parent" has the meaning set forth in the preamble.
  - "Parent Indemnitee" has the meaning set forth in Section 6.1.
  - "Parties" has the meaning set forth in the preamble.
  - "Partner" has the meaning set forth in the LP Agreement.
- "Payoff Letters" means the letters provided to an Acquired Company by the lenders or other holders of Funded Indebtedness in connection with the repayment and/or forgiveness of the outstanding Funded Indebtedness due thereunder as contemplated by this Agreement.
  - "Permitted Liens" has the meaning set forth in Section 3.6(b).
- "Person" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a business or other trust, a joint venture, any other business entity or an unincorporated organization.
- "Personal Information" means (a) any information about an individual that alone or in combination with other information could be used to identify an individual or otherwise facilitate decisions regarding the individual; or (b) any information regarding a person or household that is defined as "personal information", "sensitive personal information", "personally identifiable information", "personal data", or any similar terms under Privacy Laws.
- "Pre-Closing Tax Period" means any taxable period that ends on or before the Closing Date and, with respect to any Straddle Period, the portion thereof ending at the end of the day on the Closing Date.
- "Privacy Law" means all applicable Laws, whether foreign, federal, state or local related to data privacy, data protection, data security, payment card processing standards or consumer marketing.
- "Privacy Policy" means any written notices, policies, disclosures or representations by any Acquired Company applicable to Personal Information used, Processed, and/or hosted in connection

with the business of any Acquired Company that is externally-facing to customers, website visitors, employees, or contractors, or any other Person.

- "Process" means any operation or set of operations which is performed upon information, whether or not by automatic means, such as collection, recording, organization, storage, adaptation or alteration, retrieval, consultation, use, disclosure by transmission, dissemination or otherwise making available, alignment or combination, blocking, erasure or destruction.
  - "Preferred Interests" has the meaning set forth in the LP Agreement.
  - "Proceeding" has the meaning set forth in Section 3.9.
  - "Properties" has the meaning set forth in Section 3.6(b).
  - "Proposed Aggregate Option Exercise Price" has the meaning set forth in Section 1.5(d)(i).
  - "Proposed Closing Date Calculation" has the meaning set forth in Section 1.5(d)(i).
  - "Proposed Closing Date Cash and Cash Equivalents" has the meaning set forth in Section 1.5(d)(i).
  - "Proposed Closing Date Indebtedness" has the meaning set forth in Section 1.5(d)(i).
  - "Proposed Closing Date Net Working Capital" has the meaning set forth in Section 1.5(d)(i).
  - "Proposed Inventory" has the meaning set forth in Section 1.5(d)(i).
  - "Proposed Closing Merger Consideration Calculation" has the meaning set forth in Section 1.5(d)(i).
  - "Proposed Seller Expenses" has the meaning set forth in Section 1.5(d)(i).
  - "Reference Amount" means \$34,877,282.
  - "Released Claims" has the meaning set forth in Section 5.7.
  - "Released Parties" has the meaning set forth in Section 5.7.
  - "Releasors" has the meaning set forth in Section 5.7.
  - "Representative" has the meaning set forth in the preamble.
  - "Represented Parties" has the meaning set forth in Section 1.6(a).
  - "Restrictive Covenant Agreements" has the meaning set forth in Section 2.2(j).
- "R&W Insurance Policy" means the "buyer-side" representations and warranties insurance policy being obtained by Parent in connection with the transactions contemplated herein.
  - "Sale Bonus Plan" means the Desser Holding Company, LLC Sale Bonus Plan.
  - "Security Breach" has the meaning set forth in Section 3.11(i).
  - "Section 280G" has the meaning set forth in Section 5.12.

"Section 280G Payments" has the meaning set forth in Section 5.12.

"Seginus Aerospace" has the meaning set forth in Section 2.2(k).

"Seller Expenses" means the sum of (i) the amount of sale bonuses, change in control bonuses, retention bonuses, bonuses paid pursuant to the Sale Bonus Plan or any similar bonuses payable to any officer, director, manager, employee, agent or consultant of any Acquired Company or payable by an Acquired Company to any Person, in each case that become payable upon or by reason of, the consummation of the transactions contemplated by this Agreement, and the employer portion of payroll or similar taxes with respect to such amounts; (ii) the amount of any investment banking, accounting, attorney or other professional fees or other obligations owed to any of the foregoing incurred by any Acquired Company or the Former Equityholders with respect to the transactions contemplated by this Agreement; (iii) the amount of any management or transaction fees (including any accelerated management fees) incurred by any Acquired Company or the Former Equityholders in connection with any of the transactions contemplated by this Agreement; (iv) the Expense Funds payable to the Representative pursuant to Section 1.6(e); (v) the employer portion of payroll or similar taxes owed with respect to the Option Payments (including any UK National Insurance contribution taxes and the UK apprenticeship levy); (vi) any severance, termination, end-of-service or similar payments owed by an Acquired Company relating to a termination of employment that occurred prior to the Closing and/or with respect to [\*\*\*] (regardless of when such termination of employment occurs) and the employer portion of payroll or similar taxes with respect to such payments; (vii) to the extent not included in Net Working Capital, the amounts of any bonuses, commissions or other incentive compensation that have been or should have been accrued or are earned and unpaid for, or are payable to, any office, director, manager, employee, agent or consultant of any Acquired Company as of the Closing Date, and the employer portion of payroll or similar taxes with respect to such amounts; and (viii) one-half of (x) any fees and expenses payable to the Escrow Agent in connection with the Merger Consideration Adjustment Escrow Account, UK Tax Escrow Accounts and Employee Matters Escrow Account and (y) any filing fees required under the FDI Law.

"Seller Tax Returns" has the meaning set forth in Section 5.3(d).

"Straddle Period" means any taxable period that begins before and ends after the Closing Date.

"Subsidiary" means, with respect to any Person, an entity as to which such Person owns directly or indirectly 20% or more of such entity's voting power or similar interest.

"Surviving Company" has the meaning set forth in Section 1.1.

"Tax" or "Taxes" means all federal, state, local or foreign income, gross receipts, license, duties, value added, franchise, profits, sales, use, withholding, contributions, rates, social security (or similar), unemployment, disability, payroll, employment, premium, excise, severance, windfall profits, environmental, customs duties, capital stock, occupation, stamp, registration, property, alternative or add-on minimum, or estimated taxes, together with any interest, penalties or addition thereto, whether disputed or not.

"Tax Contests" has the meaning set forth in Section 5.3(g).

"Tax Refund" has the meaning set forth in Section 5.3(e).

"<u>Tax Return</u>" means any return (including any information return), report, statement, claim for refund, election, request for extension, estimated Tax filing or other documents filed or required to be filed with or submitted to, any Governmental Entity in connection with or with respect to Taxes, including any attachment thereto and any amendment thereof.

"Termination Date" has the meaning set forth in Section 7.1(b).

- "Top Customers" has the meaning set forth in Section 3.17(a).
- "Top Suppliers" has the meaning set forth in Section 3.17(c).
- "Trade Compliance Laws" means any requirement of applicable Law relating to the regulation of imports, exports, re-exports, transfers, releases, shipments, transmissions or any other provision of goods, technology, software or services, including, without limitation: (a) International Traffic in Arms Regulations (ITAR), 22 C.F.R. Parts 120 et. seq.; (b) the Export Administration Regulations (EAR), 15 C.F.R. Parts 730 et seq., (c) the regulations administered by the U.S. Department of the Treasury's Office of Foreign Assets Control, 31 C.F.R. Parts 500 et seq. and related executive orders authorizing or implementing economic sanctions, (cd) Laws and regulations relating to economic sanctions administered by His Majesty's Treasury and, as applicable, the European Union and any European Union Member State, (e) the customs regulations set forth in Title 19 of the Code of Federal Regulations and/or the UK Control Act 2002 and the Export Control Order 2008, (f) the antiboycott Laws administered by the U.S. Department of Commerce and the U.S. Department of the Treasury's Internal Revenue Service and (g) similar trade compliance Laws in which any Acquired Company is operating or has operated.
- "Transaction Document" means any agreement, document, certificate or instrument delivered pursuant to or in connection with this Agreement or the transactions contemplated hereby.
- "Transaction Tax Deductions" means, without duplication, to the extent deductible at a "more likely than not" level of comfort, all income tax deductions attributable to (a) Funded Indebtedness, including the fees and expenses (including any breakage fees or accelerated deferred financing fees) incurred by the Acquired Companies with respect to the payment of Funded Indebtedness in connection with the consummation of the transactions contemplated hereby, (b) the Seller Expenses, (c) the vesting, cancellation or exercise of the Options in connection with the transactions contemplated hereby, and (d) the amount of any employment or similar Taxes with respect to the amounts set forth in clause (b), (c) or (d) of this definition; provided, that 70% of any of the foregoing that constitutes a success based fee within the scope of Internal Revenue Service Revenue Procedure 2011-29 shall be considered to be a Transaction Tax Deduction.
- "Transfer Taxes" means any transfer, excise, sales, use, goods and services, value added, registration, stamp, recording, documentary, conveyancing, transfer, transaction privilege, land appreciation, land transfer, business, personal property transfer, filing, permit, stock, license and similar Taxes (including penalties and interest) incurred in connection with the transactions contemplated by this Agreement.
- "<u>UK Exchange Taxes</u>" means any Taxes imposed on Watts Desser Limited under the U.K. Corporation Tax Act 2009 with respect to those certain exchange gains and losses generated, incurred or accrued on or before the Closing Date that are described in the letter from Crowe U.K. to the Corporation Tax Services, HM Revenue and Customs, dated November 29, 2022.
  - "UK Exchange Tax Escrow Account" has the meaning set forth in Section 1.5(c)(i).
  - "UK Tax Contests" has the meaning set forth in Section 5.3(h)(ii).
  - "UK Tax Escrow Account" has the meaning set forth in Section 1.5(c)(i).
- "UK Withholding Taxes" means any Taxes imposed on Watts Desser Limited on account of any shortfall in the amount of UK Tax withheld or failure to withhold an amount of UK Tax by Watts Desser Limited under the U.K. Income Tax Act 2007 with respect to interest payments made by it to Desser Holding Company, LLC, prior to the Closing Date pursuant to the two tranches of promissory notes issued by Watts Desser Limited to Desser Holding Company, LLC in 2015 and 2017, respectively.
  - "UK Withholding Tax Escrow Account" has the meaning set forth in Section 1.5(c)(i).

- "Unit Option Plan," means the Desser Holding Company, LLC 2014 Unit Option Plan, as amended from time to time.
- "Unit Options" means the options that are outstanding to purchase Units.
- "Units" means the Units of Desser.
- "Waived 280G Benefits" has the meaning set forth in Section 5.12.

#### ARTICLE 10. MISCELLANEOUS

#### 10.1 Construction.

- (a) Within this Agreement, all Transaction Documents and all other documents required to consummate the transactions contemplated herein, the singular shall include the plural and the plural shall include the singular, and any gender shall include all other genders, all as the meaning and the context of this Agreement shall require. Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. If any action is to be taken or given on or by a particular calendar day, and such calendar say is not a Business Day, then such action may be deferred until the next Business Day. Any reference to any federal, state, local, or foreign statute or Law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. To the extent that any documents or other materials were present in the online data room maintained by or on behalf of the Company for purposes of the transactions contemplated by this Agreement at least one Business Day prior to the execution of this Agreement, such documents or other materials shall be deemed "provided" and "made available" (and all similar phrases used herein with a correlative meaning) to Parent or Merger Sub for all purposes of this Agreement. The terms "including" shall mean "including without limitation" (if such words do not already follow). References to "ordinary course of business" shall mean the ordinary and usual course of business of the referenced Person or Persons, as applicable, consistent with past practice (if such words do not already follow). The terms "true and correct," "complete and accurate" and similar phrases used in representations or warranties shall mean "true, correct and complete," regardless of the formulation used.
- (b) Each of the Parties hereto is a sophisticated legal entity or Person that was advised by experienced counsel and, to the extent it deemed necessary, other advisors in connection with this Agreement. The Parties hereto have been represented by counsel who have carefully negotiated the provisions hereof. As a consequence, the Parties hereto do not intend that the presumptions of any Laws or rules relating to the interpretation of contracts against the drafter of any particular clause should be applied to this Agreement and therefore waive their effects. The language used in this Agreement will be deemed to be the language chosen by the Parties hereto to express their mutual intent, and no rule of strict construction will be applied against any Person. The Parties hereto agree that prior drafts of this Agreement or the Transaction Documents and the documents referred to herein and therein will be deemed not to provide any evidence as to the meaning of any provision hereof or thereof or the intent of the Parties hereto or thereto with respect hereto or thereto and that such drafts will be deemed joint work product of the Parties hereto. Furthermore, the Parties hereto each hereby acknowledge that this Agreement embodies the justifiable expectations of sophisticated parties derived from arm's-length negotiations and specifically acknowledge that no Party has any special relationship with another Party that would justify any expectation beyond that of an ordinary purchaser and an ordinary seller in an arm's-length transaction.
- 10.2 Exhibits and Schedules. For all purposes of this Agreement, any matter that is disclosed in a particular Disclosure Schedule to this Agreement shall be deemed to have been included in the other Disclosure Schedules, notwithstanding the omission of a cross reference thereto, so long as the relevance of such matter to such other Disclosure Schedules is reasonably apparent on such disclosure's face. Matters reflected in the Disclosure Schedules are not necessarily limited to matters required by this Agreement to be set forth on the Disclosure Schedules, and any such additional matters are for informational purposes only and do not necessarily include other matters of a similar nature. Disclosure of any fact or item in any Disclosure Schedule shall not necessarily mean that such fact or item is material to the Acquired Companies. The information contained in the Disclosure Schedules is disclosed solely

for purposes of this Agreement, and no information contained therein shall be deemed to be an admission by any party hereto to any third party of any matter whatsoever, including any violation of Law or breach of any agreement or obligation.

- 10.3 <u>Further Assurances</u>. Each Party shall, without further consideration, execute such further instruments and take such other actions as any other Party shall reasonably request in order to fulfill its obligations under this Agreement and to effectuate the purposes of this Agreement.
- 10.4 <u>Costs and Expenses</u>. Except as otherwise expressly provided herein, each Party shall bear its own expenses in connection herewith. In furtherance of the foregoing, any amounts owed or payable to Jefferies LLC or Lincoln International LLC with respect to the transactions contemplated by this Agreement shall be borne by the Former Equityholders, and Representative shall pay any amounts that remain unpaid or become payable to Jefferies LLC and Lincoln International LLC following the Closing.
- 10.5 Notices. All notices or other communications permitted or required under this Agreement shall be in writing and shall be sufficiently given if and when sent by electronic transmission or hand delivered to the Persons set forth below or if sent by documented overnight delivery service or registered or certified mail, postage prepaid, return receipt requested, or by electronic mail addressed as set forth below or to such other Person or Persons and/or at such other address or addresses as shall be furnished in writing by any Party to the others. Any such notice or communication shall be deemed to have been given as of the date received, in the case of personal delivery, or on the date shown on the receipt or confirmation therefor in all other cases.

### To Parent:

VSE Aviation, Inc. c/o VSE Corporation 6348 Walker Lane Alexandria, VA 22310 Attention: John A. Cuomo Farinaz Tehrani

Email: JACuomo@VSECorp.com FTehrani@VSECorp.com

With copies (which shall not constitute notice) to:

Jones Day 600 Brickell Avenue, Suite 3300 Miami, Florida 33131 Attention: Lorne S. Cantor Ashley L. Gullett

Email: lcantor@jonesday.com agullett@jonesday.com

#### To the Representative:

Desser Holdings Partnership GP, LLC c/o Graham Partners, Inc.
3811 West Chester Pike
Building 2, Suite 200
Newtown Square, Pennsylvania 19073
Attention: William P. McKee, Jr., Chief Operating Officer
Email: BMcKee@grahampartners.net

With copies (which shall not constitute notice) to:

Faegre Drinker Biddle & Reath LLP

105 College Road East Princeton, NJ 08542

Attention: John E. Stoddard III Adam Weinstock

Email: John.Stoddard@faegredrinker.com Adam.Weinstock@faegredrinker.com

#### 10.6 Assignment and Benefit.

- (a) This Agreement shall be binding upon and inure to the benefit of the Parties and their respective permitted successors and permitted assigns. Neither this Agreement, nor any of the rights hereunder, may be assigned by any Party, nor may any Party delegate any obligations hereunder, without the written consent of the other Parties. Any non-permitted assignment or attempted assignment shall be void. Notwithstanding the foregoing, Parent or the Surviving Company (i) may assign and grant a security interest in its rights, title and interest under this Agreement and the other agreements contemplated hereby for collateral security purposes to any lender(s) providing financing to Parent or the Surviving Company or for any purpose to its Affiliates and (ii) may assign this Agreement to a purchaser of substantially all of the assets or business of Parent or the Surviving Company.
- (b) Except as otherwise expressly provided herein, including pursuant to Section 5.2, Section 5.3(h) and Section 5.7, this Agreement shall not be construed as giving any Person, other than the Parties and their permitted successors, heirs and assigns, any legal or equitable right, remedy or claim under or in respect of this Agreement or any of the provisions herein contained, this Agreement and all provisions and conditions hereof being intended to be, and being, for the sole and exclusive benefit of such Parties, and permitted successors, heirs and assigns and for the benefit of no other Person.
- 10.7 Governing Law; Jurisdiction. This Agreement, and all claims, disputes or controversies, whether in contract, equity, statute, tort or otherwise ("Disputes"), that may be based upon, arise out of or relate to this Agreement or the negotiation, execution or performance of this Agreement (including any Dispute based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement, but excluding any Merger Consideration Dispute pursuant to Section 1.5(d)(ii)), shall be solely and exclusively governed by, and enforced in accordance with, the internal Laws of the State of Delaware, including its statutes of limitations, but without reference to the choice-of-law or conflicts-of-law principles thereof. The Parties hereto hereby agree and consent to be subject to the sole and exclusive jurisdiction of the federal or state courts located in the State of Delaware in connection with any such Dispute and hereby irrevocably waive the right to assert the lack of personal or subject-matter jurisdiction or improper venue in connection with any such Dispute. In furtherance of the foregoing, each of the Parties (i) hereby irrevocably waives the defense of inconvenient forum, (ii) shall not commence any such Dispute other than in any such court, (iii) agrees that a final judgment in any such Dispute shall be conclusive and may be enforced in other jurisdictions by suit or judgment or in any other manner provided by applicable Law and (iv) waives any right it may have to trial by jury in respect of any litigation based on, arising out of, under or in connection with this Agreement or any course of conduct, course of dealing, verbal or written statement or action of any Party hereto.
- 10.8 Amendment, Modification and Waiver. The Parties may amend or modify this Agreement in any respect. Any such amendment or modification shall be in writing and signed by the Representative and Parent. The waiver by a Party of any breach of any provision of this Agreement shall not constitute or operate as a waiver of any other breach of such provision or of any other provision hereof, nor shall any failure to enforce any provision hereof operate as a waiver of such provision or of any other provision hereof.
- 10.9 <u>Section Headings and Defined Terms</u>. The section headings contained herein are for reference purposes only and shall not in any way affect the meaning and interpretation of this Agreement. Except as otherwise indicated, all agreements defined herein refer to the same as from time

to time amended or supplemented or the terms thereof waived or modified in accordance herewith and therewith.

- 10.10 <u>Severability</u>. The invalidity or unenforceability of any particular provision, or part of any provision, of this Agreement shall not affect the other provisions or parts hereof, and this Agreement shall be construed in all respects as if such invalid or unenforceable provisions or parts were omitted.
- 10.11 Counterparts. This Agreement and the other documents required to consummate the transactions contemplated herein may be executed in one or more counterparts, each of which shall be deemed an original (including facsimile signatures), but all of such counterparts together shall be deemed to be one and the same instrument. It shall not be necessary in making proof of this Agreement or any counterpart hereof to produce or account for any of the other counterparts. Delivery of an executed counterpart of a signature page to this Agreement by facsimile, portable document format (.pdf), or any electronic signature complying with the U.S. Federal ESIGN Act of 2000 (including Docusign) shall be effective as delivery of an executed counterpart to this Agreement (any such delivery, an "Electronic Delivery"), shall be treated in all manner and respects as an original agreement or instrument and shall be deemed to have the same binding legal effect as if it were the original signed version thereof delivered in person by a Party. No party shall raise the use of Electronic Delivery to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of Electronic Delivery as a defense to the validity and enforceability of this Agreement, and each Party hereby irrevocably waives any such defense.
- 10.12 Entire Agreement. This Agreement, together with the Disclosure Schedule, the Confidentiality Agreement dated October 26, 2022 entered into by VSE Corporation and Desser Holdings LLC, the Transaction Documents and the agreements, exhibits, schedules and certificates referred to herein or therein or delivered pursuant hereto or thereto, constitute the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings. The submission of a draft of this Agreement or portions or summaries thereof does not constitute an offer to consummate the transactions contemplated hereby, it being understood and agreed that no Party shall be legally obligated with respect to such transactions or to any other terms or conditions set forth in such draft or portion or summary unless and until this Agreement has been duly executed and delivered by all Parties.
- 10.13 Retention of Counsel. In any Dispute arising under or in connection with this Agreement, the Representative shall have the right, at its election, to retain the firms of Faegre Drinker Biddle & Reath LLP and Garofalo Goerlich Hainbach PC (collectively, "Counsel") to represent it and any of its Affiliates in such matter, and Parent, for itself and each Acquired Company and for its and such Acquired Company's successors and assigns, hereby irrevocably waives and consents to any such representation in any such matter and the communication by such Counsel to the Representative or any of its Affiliates in connection with any such representation of any fact known to such Counsel arising by reason of such Counsel's prior representation of any Acquired Company or any of their respective Affiliates. Parent, for itself, the Acquired Companies and its and their respective successors and assigns, hereby irrevocably acknowledges and agrees that all communications between any Acquired Company and their Counsel, made in connection with the negotiation, preparation, execution, delivery and closing under, or any Dispute or proceeding arising under or in connection with, this Agreement which, immediately prior to the Closing, would be deemed to be privileged communications of such Acquired Company and/or their Counsel and would not be subject to disclosure to Parent in connection with any process relating to a Dispute arising under or in connection with his Agreement or otherwise, shall continue after the Closing to be privileged communications with such Counsel and neither Parent nor any Person purporting to act on behalf of or through Parent shall seek to obtain the same by any process on the grounds that the privilege attaching to such communications belongs to the Acquired Company and not the Representative. Other than as explicitly set forth in this Section 10.13, the Parties acknowledge that any attorney-client privilege attaching as a result of any Counsel representing any Acquired Company

prior to the Closing shall survive the Closing and continue to be a privilege of such Acquired Company and not the Representative after the Closing.

10.14 Specific Performance. The Parties agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with its specific terms or is otherwise breached, and that money damages or other legal remedies would not be an adequate remedy for any such damages. Accordingly, the Parties acknowledge and agree that in the event of any breach or threatened breach by the Representative, on the one hand, or Parent, on the other hand, of any of their respective covenants or obligations set forth in this Agreement, the Representative, on the one hand, and Parent, on the other hand, shall be entitled to seek an injunction or injunctions to prevent or restrain breaches or threatened breaches of this Agreement by the other (as applicable), without the necessity of proving the inadequacy of monetary damages and without the necessity of posting a bond or other security, and to specifically enforce the terms and provisions of this Agreement to prevent breaches or threatened breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of the other under this Agreement. The Parties agree not to raise any objections to the availability of the equitable remedy of specific performance to prevent or restrain breaches or threatened breaches of this Agreement by the other Parties, and to specifically enforce the terms and provisions of this Agreement to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of the other Parties under this Agreement to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of the other Parties under this Agreement. The Parties further agree that: (x) by seeking the remedies provided for in this Section 10.14, a Party shall not in any respect waive its right to seek any other form of relief that may be available under this Agreement (including monetary damages) if the remedies provided for in this Section 10.14 are not available

[Signature Pages Follow]

IN WITNESS WHEREOF, each of the Parties has duly executed this Agreement as of the date first above written.

#### **COMPANY:**

### DESSER-GRAHAM PARTNERSHIP, L.P.

By: Desser Holdings Partnership GP, LLC, its General Partner

By: Graham Partners III, L.P., its Member

By: Graham Partners Management III, L.P., its Manager

By: Graham Partners Management GP III, L.L.C., its General Partner

By: Graham Partners, Inc., its Managing Member

By: /s/ Joshua Wilson
Name: Joshua Wilson
Title: Managing Principal

### **REPRESENTATIVE:**

DESSER HOLDINGS PARTNERSHIP GP, LLC

By: Graham Partners III, L.P., its Member

By: Graham Partners Management III, L.P., its Manager

By: Graham Partners Management GP III, L.L.C., its General Partner

By: Graham Partners, Inc., its Managing Member

By: /s/ Joshua Wilson
Name: Joshua Wilson
Title: Managing Principal

[Signature Page to Agreement and Plan of Merger]

## **PARENT:**

# VSE AVIATION, INC.

By:/s/ Benjamin E. Thomas
Name: Benjamin E. Thomas
Title: President

## **MERGER SUB:**

# **DIAMOND MERGER SUB 1, LLC**

By:/s/ Benjamin E. Thomas Name: Benjamin E. Thomas Title: President

[Signature Page to Agreement and Plan of Merger]